Washington, D.C. 20506, or call area code 202-724-0256.

JOHN W. JORDAN. Advisory Committee Management Officer.

[FR Doc.77-22537 Filed 8-4-77;8:45 am]

#### ADVISORY COMMITTEE FELLOWSHIPS PANEL

Notice of Meeting

AUGUST 1, 1977.

Pursuant to the provisions of the Federal Advisory Committee Act (Pub. L. 92-463, as amended), notice is hereby given that a meeting of the Fellowships Panel will be held at 806 15th Street, N.W., Washington, D.C. 20506, in room 314 from 9:30 a.m. to 5:30 p.m. on August

The purpose of the meeting is to review 1978 Summer Seminar applications from prospective seminar directors in the field of Philosophy submitted to the National Endowment for the Humanities for projects beginning in the summer of 1978.

Because the proposed meeting will consider financial information and disclose information of a personal nature the disclosure of which would constitute a clearly unwarranted invasion of personal privacy, pursuant to authority granted me by the Acting Chairman's Delegation of Authority to Close Advisory Committee Meetings, dated April 28, 1977, I have determined that the meeting would fall within exemptions (4) and (6) of 5 U.S.C. 552b(c) and that it is essential to close the meeting to protect the free exchange of internal views and to avoid interference with operation of the Committee.

It is suggested that those desiring more specific information contact the Advisory Committee Management Officer. Mr. John W. Jordan, 806 15th Street, N.W. Washington, D.C. 20506, or call area code 202-724-0256.

JOHN W. JORDAN, Advisory Committee Management Officer.

[FR Doc.77-22538 Filed 8-4-77;8:45 am]

#### ADVISORY COMMITTEE FELLOWSHIPS PANEL

Notice of Meeting

AUGUST 1, 1977.

Pursuant to the provisions of the Federal Advisory Committee Act (Pub. L. 92-463, as amended), notice is hereby given that a meeting of the Fellowships Panel will be held at 806 15th Street, N.W., Washington, D.C. 20506, in room 1130 from 9:30 a.m .to 5:30 p.m. on August 31

The purpose of the meeting is to review 1978 Summer Seminar applications from prospective seminar directors in the field of Political Science submitted to the National Endowment for the Humanities for projects beginning in the summer of 1978.

Because the proposed meeting will consider financial information and disclose information of a personal nature the disclosure of which would constitute a clearly unwarranted invasion of personal privacy, pursuant to authority granted me by the Acting Chairman's Delegation of Authority to Close Advisory Committee Meetings, dated April 28. 1977. I have determined that the meeting would fall within exemptions (4) and (6) of 5 U.S.C. 552b(c) and that it is essential to close the meeting to protect the free exchange of internal views and to avoid interference with operation of the Committee.

It is suggested that those desiring more specific information contact the Advisory Committee Management Officer, Mr. John W. Jordan, 806 15th Street, N.W., Washington, D.C. 20506, or call area code 202-724-0256.

> JOHN W. JORDAN, Advisory Committee Management Officer.

(FR Doc.77-22539 Filed 8-4-77;8:45 am)

#### ADVISORY COMMITTEE FELLOWSHIPS PANEL

Notice of Meeting

AUGUST 1, 1977.

Pursuant to the provisions of the Federal Advisory Committee Act (Pub. L. 92-463, as amended) notice is hereby given that a meeting of the Fellowships Panel will be held at 806 15th Street, N.W., Washington, D.C. 20506, in room 314 from 9:30 a.m. to 5:30 p.m. on August 25

The purpose of the meeting is to review 1978 Summer Seminar applications from prospective seminar directors in the field of Drama, Film, American Studies, Afro-American Literature Composition and Rhetoric submitted to the National Endowment for the Humanities for projects beginning in the summer of 1978.

Because the proposed meeting will consider financial information and disclose information of a personal nature the disclosure of which would constitute a clearly unwarranted invasion of personal privacy, pursuant to authority granted me by the Acting Chairman's Delegation of Authority to Close Advisory Committee Meetings, dated April 28, 1977, I have determined that the meeting would fall within exemptions (4) and (6) of 5 U.S.C. 552b(c) and that it is essential to close the meeting to protect the free exchange of internal views and to avoid interference with operation of the Committee.

It is suggested that those desiring more specific information contact the Advisory Committee Management Officer, Mr. John W. Jordan, 806 15th Street, N.W., Washington, D.C. 20506, or call area code 202-724-0256.

> John W. Jordan, Advisory Committee Management Officer.

[FR Doc.77-22540 Filed 8-4-77;8:45 am]

#### ADVISORY COMMITTEE FELLOWSHIPS PANEL

Notice of Meeting

AUGUST 1, 1977.

Pursuant to the provisions of the Federal Advisory Committee Act (Pub. L. 92-463, as amended.) notice is hereby given that a meeting of the Fellowships Panel will be held at 806 15th Street, N.W., Washington, D.C. 20506, in room 1023 from 9:30 am. to 5:30 pm. on August 26.

The purpose of the meeting is to review 1978 Summer Seminar applications from prospective seminar directors in the field of Linguistics submitted to the National Endowment for the Humanities for projects beginning in the summer of 1978.

Because the proposed meeting will consider financial information and disclose information of a personal nature the disclosure of which would constitute a clearly unwarranted invasion of personal privacy, pursuant to authority granted me by the Acting Chairman's Delegation of Authority to Close Advisory Committee Meetings, dated April 28. 1977. I have determined that the meeting would fall within exemptions (4) and (6) of 5 U.S.C. 552b(c) and that it is essential to close the meeting to protect the free exchange of internal views and to avoid interference with operation of the Committee.

It is suggested that those desiring more specific information contact the Advisory Committee Management Officer, Mr. John W. Jordan, 806 15th Street, N.W., Washington, D.C. 20506, or call area code 202-724-0256.

JOHN W. JORDAN. Advisory Committee Management Officer.

[FR 1)oc.77-22541 Filed 8-4-77;8:45 am]

#### ADVISORY COMMITTEE FELLOWSHIPS PANEL

Notice of Meeting

AUGUST 1, 1977.

Pursuant to the provisions of the Federal Advisory Committee Act (Pub. L. 92-463, as amended), notice is hereby given that a meeting of the Fellowships Panel will be held at 806 15th Street, N.W., Washington, D.C. 20506, in room 314 from 9:30 am. to 5:30 pm. on August 27.

The purpose of the meeting is to review 1978 Summer Seminar applications from prospective seminar directors in the field of English and American Literature submitted to the National Endowment for the Humanities for projects beginning in the summer of 1978.

Because the proposed meeting will consider financial information and disclose information of a personal nature the disclosure of which would constitute a clearly unwarranted invasion of personal privacy, pursuant to authority granted me by the Acting Chairman's Delegation of Authority To Close Advisory Committee Meetings, dated April 28,

1977. I have determined that the meeting would fall within exemptions (4) and (6) of 5 U.S.C. 552b(c) and that it is essential to close the meeting to protect the free exchange of internal views and to avoid interference with operation of the Committee.

It is suggested that those desiring more specific information contact the Advisory Committee Management Officer, Mr. John W. Jordan, 806 15th Street, N.W., Washington, D.C. 20506, or call area code 202-724-0256.

> JOHN W. JORDAN. Advisory Committee Management Officer.

[FR Doc.77-22542 Filed 8-4-77;8:45 am]

# OFFICE OF SCIENCE AND TECHNOLOGY POLICY

# ADVISORY GROUP ON WHITE HOUSE INFORMATION SYSTEMS

#### Establishment

Pursuant to the Federal Advisory Committee Act (Pub. L. 92-463), it is hereby determined that the establishment of the Adivsory Group on White House Information Systems is necessary, appropriate, and in the public interest in connection with the performance of the duties imposed upon the Director. Office of Sciece and Technology Policy (OSTP) by the National Science and Technology Policy, Organization, and Priorties Act of 1976. This determination follows consultation with the Office of Management and Budget (OMB), pursuant to Section 9(a) (2) of the Federal Advisory Committee Act and OMB Circular No. A-63, Revised.

- 1. NAME OF GROUP: Advisory Group on White House Information Systems.
- PURPOSE AND FUNCTION: The Office of Science and Technology Policy, in accordance with the statutory mandate to advise the President and to analyze and interpret significant developments and trends in science and technology, will be identifying the information systems needs and the impact of technological advances in information and data handling as these might support the decision processes of the White House and the Executive Office of the President. The work of the Advisory Group will be based upon inputs from the relevant departments and earlier work carried out by other organizations in the Executive Branch including the Reorganization Team. The Advisory Group will consider the implications for policy initiatives that may be appropriate to exploit advances that may be identified, but it will not consider or propose specific computer and supporting systems or the procurement of same. The Group will submit a report and briefing for appropriate officials in the Executive Branch upon completion of its

visory Group is established to provide advice to the Director of the Office of Science and Technology Policy and to the Special Assistant to the President for Budget and Organization, and is established until December 1, 1977. Due to the need to provide a prompt followup to related efforts by the President's Re-organization Project, the Office of Management and Budget has agreed to waive the requirement for a 15-day delay between the publication of this notice and the filing of the charter for this Advisory

4. MEMBERSHIP: Membership of the Advisory Group on White House Information Systems will consist of approximately six members who are particularly knowledgeable in the areas of information and automation.

ADVISORY GROUP OPERA-TION: The Advisory Group on White House Information Systems will operate in accordance with provisions of the Federal Advisory Committee Act (Pub. L. 92-463), OSTP policy and procedures, OMB Circular No. A-63. Revised, and other directives and instructions issued in implementation of the Act.

> FRANK PRESS, Director.

IFR Doc.77-22564 Filed 8-4-77;8:45 aml

## DEPARTMENT OF STATE

Agency for International Development

JOINT RESEARCH COMMITTEE OF THE BOARD FOR INTERNATIONAL FOOD AND AGRICULTURAL DEVELOPMENT

#### Meeting

Pursuant to Executive Order 11769 and the provisions of Section 10(a)(2), P.L. 92-463, Federal Advisory Committee Act, notice is hereby given of the second meeting of the Joint Research Committee of the Board for International Food and Agricultural Development on August 25 and 26, 1977.

The purpose of this two-day meeting is: To review research priorities; consider a proposal for a nutrition project; review other project proposals; review the status of Planning Grants; review and update the list of eligible universities under Title XII; and review the questionnaire to be sent to the university community to expand the information on university interests and capabilities for participating in the Title XII program.

The meeting will begin each day at 9 a.m., will adjourn at 5:30 p.m., and will be held at the Ramada Inn, Rosslyn, 1900 Fort Myer Drive, Arlington, Virginia. The meeting room designation will be posted in the lobby of the Inn on each day of the meeting. The meeting is open to the public. Any interested person may attend, may file written statements with the Committee before or after the meeting, or may present oral statements in accordance with procedures established by the Committee, and to the extent the 3. EFFECTIVE DATE OF ESTAB- the Committee, and to the extent the LISHMENT AND DURATION: The Ad- time available for the meeting permits.

Dr. Erven J. Long, Associate Assistant Administrator of the Technical Assistance Bureau, is designated as A.I.D. Advisory Committee Representative at this meeting. It is suggested that those desiring further information write to him in care of the Agency for International Development, State Department, Washington, D.C. 20523, or telephone him at 701-235-9001.

Dated: July 30, 1977.

ERVEN J. LONG, Aid for International Development, Advisory Committee Representative, Joint Research Committee, Board for International Food and Agricultural Development.

[PR Doc.77-22615 Filed 8-4-77;8:45 am]

# INTERSTATE COMMERCE COMMISSION

[Notice No. 452]

## ASSIGNMENT OF HEARINGS

AUGUST 1, 1977.

Cases assigned for hearing, postponement, cancellation, or oral argument appear below and will be published only once. This list contains prospective assignments only and does not include cases previously assigned hearing dates. The hearings will be on the issues as presently reflected in the Official Docket of the Commission. An attempt will be made to publish notices of cancellation of hearings as promptly as possible, but interested parties should take appropriate steps to insure that they are notified of cancellation or postponements of hearings in which they are interested.

No. MC 116677 (Sub-No. 3), Sheridan Travel Bureau, Inc., now being assigned November 2, 1977 (3 days), at Buffalo, N.Y., in a hearing room to be later designated.

MC 116519 (Sub-No. 40), Frederick Transport Ltd., now being assigned November 7, 1977 (3 days), at Buffalo, N.Y., in a hearing room to be later designated.

MC 139579 (Sub-No. 4), George H. Golding, Inc., now being assigned November 10, 1977 (2 days), at Buffalo, N.Y., in a hearing room to be later designated.

MC-F-13140, Marty's Express, Inc.—Purchase—Kruse Trucking Co., MC 39349 (Sub-No. 19), Marty's Express, Inc., now being assigned October 3, 1977 (1 day), for hearing in Philadelphia, Pa., in a hearing room to be later designated.

MC 141776 (Sub-No. 4), Foodtrain, Inc., now being assigned October 4, 1977 (1 day), in Philadelphia, Pa., in a hearing room to be later designated.

MC 142785 (Sub-No. 1), Brotherly Love Express, Inc., now being assigned October 5, 1977 (1 day), in Philadelphia, Pa., in a hearing room to be later designated.

MC 143173, Dependable Delivery Service, Inc., now being assigned October 6, 1977 (2 days), in Philadelphia, Pa., in a hearing room to be later designated.

> H. G. HOMME, Jr., Acting Secretary.

[FR Doc.77-22595 Filed 8-4-77;8:45 am]

INo. 365894

CHICAGO, MILWAUKEE, ST. PAUL & PA-CIFIC RAILROAD CO. AND UNIT-TRAIN-SHIP, INC.

Joint Petition—Experimental Piggyback
Train Service

Present: Dale W. Hardin, Commissioner, to whom this matter has been assigned for action thereon.

By petition filed May 12, 1977, the Chicago, Milwaukee, St. Paul & Pacific Railroad Co. (Milwaukee) and Unit-Trainship, Inc. (UTI) seek an order declaring the lawfulness of the proposed innovative service described in their petition or a finding of exemption under section 12(1) (b) of the Interstate Commerce Act. Replies were filed June 9, 1977, by the Burlington Northern, Inc., and June 13, 1977, by the Union Pacific Railroad Co.

The Milwaukee and UTI propose to establish dedicated nonstop piggyback train service on a round-trip basis with established mutually agreed upon schedules and subject to existing railroad rates applicable on Freight, All Kinds, between Chicago, Ill., on the one hand and Seattle/Tacoma, Wash., and Washington/Portland, Oreg., on the other. UTI, acting in the capacity of a broker, contractually undertakes to provide the Milwaukee with a minimum of 60 loaded or empty trailers or containers three times a week in each direction. In return for the guarantee of a minimum fixed amount of revenue for each 60 unit train, UTI receives a commission equal to 10 percent of the applicable tariff rate applying to the revenue traffic carried by the Milwaukee pursuant to the agreement. UTI receives 20 percent of the tariff rate for revenue traffic offered but not accommodated by the Milwaukee, excluding empty units tendered by UTI to satisfy minimum guarantees. In addition to promoting traffic UTI will prepare a comprehensive manifest for the Milwaukee and take over billing and collecting. Each Monday the Milwaukee will submit a statement of charges and UTI will undertake to remit payment within the specified period less its compensation and credit for each unit not accommodated and for failure to accommodate units

The Milwaukee on its part agrees to provide and control all rail services necessary to accommodate traffic generated by UTI in accordance with the terms of the agreement. It will pay penalties for its failure to accommodate the agreed upon level of generated traffic, and it will retain sole liability for traffic tendered to it through the agreement that is lost, damaged, stolen or delayed.

Because of the novelty of the experimental proposal, interested persons are urged to participate in the development of a record in this proceeding. All statements should address the underlying lawfulness of the proposal with respect to the applicable provisions of the Interstate Commerce Act, 49 U.S.C. 1 et seq., the Elkins Act, 49 U.S.C. 41(1), and

such issues as: (1) the status of UTI, (2) whether the proposal constitutes a special service such as would require tariff publication, and (3) the penalty aspect of the agreement.

It is ordered: Pursuant to section 5(e) of the Administrative Procedure Act, 5 U.S.C. 554(e), and in the exercise of the Commission's sound discretion thereunder, this petition for a declaratory order is granted to determine the lawfulness of the proposed arrangement between petitioners.

Petitioners, the Burlington Northern Inc., and the Union Pacific Railroad Co. are made parties to this proceeding. All other persons desiring to participate shall make such fact known by notifying the Office of Proceedings, Room 5342. Interstate Commerce Commission, Washington, D.C. 20423, on or before August 25, 1977. As soon as practicable the Commission will serve a list of the names and addresses of all persons whom service of statements under the Commission's modified procedure shall be made and the schedule to be followed.

A copy of this order shall be served upon petitioners, the Burlington Northern Inc., and the Union Pacific. Copies shall also be deposited in the Office of the Secretary, Interstate Commerce Commission, Washington, D.C., and given to the public along with a copy of the petition and the attached draft agreement by delivery to the Director, Office of the Federal Register for publication.

Dated at Washington, D.C., this 27th day of July, 1977.

By the Commission, Commissioner Hardin.

> H. G. Homme, Jr., Acting Secretary.

[FR Doc.77-22594 Filed 8-4-77;8:45 am]

[Section 5a Application No. 58 (Amendment No. 2)]

# MACHINERY HAULERS ASSOCIATION Agreement

JULY 27, 1977.

The Commission is in receipt of an application in the above-entitled proceeding for approval of amendments to the agreement therein approved.

Filed July 20, 1977 by: Charles W. Singer, Singer & Sullivan, 2440 East Commercial Blvd., Ft. Lauderdale, Fla., 33308 (of Counsel).

The Amendments involve: (1) Proposed broadening of commodity scope jurisdiction to include general commodities, (2) expansion of rate committee's jurisdiction regarding consideration of matters jointly with foreign carriers, (3) increasing the size of the Rate Committee, (4) granting the Association power to meet and otherwise cooperate and confer with other organizations of common carriers, approved under section 5a by this Commission, with reference to specified matters, and (5) other incidental changes.

The complete application may be inspected at the Office of the Commission, in Washington, D.C.

Any interested person desiring to protest and participate in this proceeding shall notify the Commission in writing on or before September 6, 1977. As provided by the General Rules of Practice of the Commission, persons other than applicants should fully disclose their interest, and the position they intend to take with respect to the application. Otherwise, the Commission, in its discretion, may proceed to investigate and determine the matters involved in such application, without further or formal hearing.

H. G. Homme, Jr., Acting Secretary.

[FR Doc.77-22593 Filed 8-4-77;8:45 am]

[Notice No. 96]

#### MOTOR CARRIER TEMPORARY AUTHORITY APPLICATIONS

AUGUST 2, 1977.

The following are notices of filing of applications for temporary authority under Section 210a(a) of the Interstate Commerce Act provided for under the previsions of 49 CFR 1131.3. These rules provide that an original and six (6) copies of protest to an application may be filed with the field official named in the FEDERAL REGISTER publication no later than the 15th calendar day after the date the notice of the filing of the application is published in the FEDERAL REGISTER. One copy of the protest must be served on the applicant, or its authorized representative, if any, and the protestant must certify that such service has been made. The protest must identify the operating authority upon which it is predicated, specifying the "MC" docket and "Sub" number and quoting the particular portion of authority upon which it relies. Also, the protestant shall specify the service it can and will provide and the amount and type of equipment it will make available for use in connection with the service contemplated by the TA application. The weight accorded a protest shall be governed by the completeness and pertinence of the protestant's information.

Except as otherwise specifically noted, each applicant states that there will be no significant effect on the quality of the human environment resulting from approval of its application.

A copy of the application is on file, and can be examined at the Office of the Secretary. Interstate Commerce Commission, Washington, D.C., and also in the ICC Field Office to which protests are to be transmitted.

MOTOR CARRIERS OF PROPERTY

No. MC 531 (Sub-No. 34TTA), filed July 6, 1977. Applicant: YOUNGER BROTHERS, INC., 4904 Griggs Rd., P.O. Box 14048, Houston, Tex. 77021. Applicant's representative: Wray E. Hughes (same address as applicant). Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Sodium salt solutions, in bulk, in tank truck vehicles, from plantsite of Merichem Co. and/or storage facilities of Merichem Co. in Houston, Tex., to all points in Alabama, Florida, Georgia, Louisiana, Mississippi, and Oklahoma for 180 days, Supporting shipper(s): Merichem Co., 1914 Haden Road, Houston, Tex. 77015. Send protests to Mensing District Supervisor, Interstate Commerce Commission, 515 Rusk, Room 8610, Federal Bldg., Houston, Tex. 77002.

No. MC 43038 (Sub-No. 463TA), filed July 19, 1977. Applicant: COMMERCIAL CARRIERS, INC., 10701 Middlebelt Rd., Romulus, Mich. 48174. Applicant's representative: Paul H. Jones, 29725 Shacket Ave., Madison Heights, Mich. 48071. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Motor vehicles, excluding trailers, in secondary movements, in truckaway service between Sharonville, Cementdale, and Cincinnati, Ohio, on the one hand, and on the other, points in the states of Virginia and West Virginia. Restriction: The operations authorized herein are restricted to the transportation of shipments manufactured, assembled, imported, or distributed by Ford Motor Co., for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper: Ford Motor Co., K. R. Hammond, Manager, Vehicle Transportation Dept., Ford Division General Office, P.O. Box 1529B Dearborn, Mich. 48121, Send protests to: Interstate Commerce Commission, Bureau of Operations, Erma W. Secretary, 604 Federal Bldg. and U.S. Courthouse, 231 West Lafayette Blvd., Detroit, Mich. 48226.

No. MC 43038 (Sub-No. 464TA), filed July 19, 1977. Applicant: COMMERCIAL CARRIERS, INC., 10701 Middlebelt Rd., Romulus, Mich. 48174. Applicant's representative: Paul H. Jones, 29725 Shacket Ave., Madison Heights, Mich. 48071. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Motor vehicles, excluding trailers, in secondary movements in driveaway service, between Sharonville, Cementdale, and Cincinnati, Ohio, on the one hand, and on the other, points in the States of Virginia and West Virginia. Restriction: The operations authorized herein are restricted to the transportation of shipments manufactured, assembled, imported, or distributed by Ford Motor Co., for 180 days.

Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper: Ford Motor Co., K. R. Hammond, Manager, Vehicle Transportation, Ford Division General Office, P.O. Box 1529B, Dearborn, Mich. 48121. Send protest to: Erma W. Gary, Secretary, Interstate Commerce Commission, Bureau of Operations, 604 Federal Bidg. and U.S. Courthouse, 231 West Lafayette Blvd., Detroit, Mich. 48226.

No. MC 43268 (Sub-No. 65TA), filed July 11, 1977, Applicant: WELLS CAR-

GO, INC., 1775 East 4th St., Reno, Nev. 89512. Applicant's representative: David N. Inwood, P.O. Box 1511, Reno, Nev. 89505. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: General commodities (except household goods as defined by the Commission in bulk and Classes A and B explosives), between Carson City, Nev., and Ridgecrest, Calif., via U.S. Highway 395, serving all intermediate points, restricted against traffic originating south of Ridgecrest, Calif., on the one hand, and points and places within 4 miles of the California-Nevada state line on the other, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): There are approximately 57 statements of support attached to the application which may be examined at the Interstate Commerce Commission in Washington, D.C. or copies thereof which may be examined at the field office named below. Send protests to: William J. Huetig, District Supervisor, Interstate Commerce Commission, 203 Federal Bldg., 705 North Plaza St., Carson City, Nev. 89701.

No. MC 47171 (Sub-No. 94TA), filed July 22, 1977. Applicant: COOPER MO-TOR LINES, INC., P.O. Box 4255, Greenville, S.C. 29608. Applicant's representative: Harris G. Andrews, P.O. Box 4259, Greenville, S.C. 29608. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Floor coverings and materials, equipment, and supplies used in the installation, manufacture, packaging, and sale of floor coverings, from Lyerly, Ga., to points in Connecticut, Delaware, Massachusetts, Maryland, New Jersey, New York, North Carolina, Pennsylvania, Rhode Island, South Carolina, Virginia, and Washington, D.C., for 180 days, Supporting shipper(s): Bigelow-Sanford, Inc., P.O. Box 3089, Greenville, S.C. 29602, Send protests to: E. E. Strotheid. District Supervisor, Interstate Commerce Commission, Room 302, 1400 Building, 1400 Pickens St., Columbia, S.C. 29201.

No. MC 69492 (Sub-No. 55TA), filed July 22, 1977, Applicant: HENRY ED-WARDS d.b.a. HENRY EDWARDS TRUCKING CO., P.O. Box 97, Clinton, Ky. 42031. Applicant's representative: Mr. Walter Harwood, Attorney, P.O. Box 15214, Nashville, Tenn. 37215. Authority sought to operate as a common carrier. by motor vehicle, over irregular routes, transporting: Rubber, rubber products, and such other commodities as are manufactured and/or dealt in by rubber manufacturers from the plantsite and warehouse facilities of The General Tire & Rubber Co., at or near Mayfield, Ky., and the plantsite and warehouse facilities of The Goodyear Tire & Rubber Co., at or near Union City, Tenn., to points in Michigan on and east of U.S. Hwy. 27 from the Indiana-Michigan State Line to its junction with Michigan Hwy. 21, thence on and south of Michigan Hwy. 21 eastwards to Port Huron, Mich., for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of oper-

ating authority. Supporting shipper(s):
The General Tire & Rubber Co., One
General St., Akron, Ohio 44329; The
Goodyear Tire & Rubber Co., 1144 East
Market St., Akron, Ohio 44316. Send protests to: Mr. Floyd A. Johnson, District
Supervisor, Interstate Commerce Commission, 100 North Main Bidg., Suite
2006. 100 North Main St., Memphis,
Tenn. 38103.

No. MC 106956 (Sub-No. 4TA) July 19, 1977. Applicant: SYLVESTER TRUCKING CO., 2930 Gradwohl Rd., Toledo, Ohio 43617. Applicant's representative: Wilhemina Boersma, 1600 First Federal Bldg., 1001 Woodward Ave., Detroit, Mich. 48226. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Scrap iron, between Kripke Tuschman facilities in Toledo, Ohio, on one hand, and on the other, Wayne County. Branch County, Kalamazoo County, Genesse County, and Ingham County in Michigan, and DeKalb County and Allen County in Indiana, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper: Kripke-Tuschman Ind., Inc., 5000 North Detroit Ave., Toledo, Ohio 43612, Send protests to: Keigh D. Warner, District Supervisor, Bureau of Operations, Interstate Com-merce Commission, 313 Federal Office Bldg., 234 Summit St., Toledo, Obio 43604.

No. MC 197496 (Sub-No. 1992TA), filed July 22, 1977. Applicant: RUAN TRANS-PORT CORP., 3200 Ruan Center, 666 Grand Ave., Des Moines, Iowa 50309, Applicant's representative: E. Check (same as applicant). Authority sought to operate as a common carrier, by motor vehicle, over irregular routes transporting: Water reducing admixtures, in bulk, in metered tank vehicles from North Judson, Ind., to points in Indiana, Illinois, Michigan, Wisconsin, Iowa, Minnesota, Ohio, Kentucky, West Virginia, Missouri, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting ship-per(s): Penn-Dixie Chemical Co., 2 Porete Ave., North Arlington, N.J. 07032. Send protests to: Herbert W. Allen, District Supervisor, Bureau of Operations, Interstate Commerce Commission, 518 Federal Bldg., Des Moines, Iowa 50309.

No. MC 111401 (Sub-No. 494TA), filed July 21, 1977. Applicant: GROENDYKE TRANSPORT, INC., 2510 Rock Island Blvd., P.O. Box 632, Enid, Oka. 73701. Applicant's representative: Victor R. Comstock, P.O. Box 632, Enid, Okla. 73701. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: lasses, liquid feeds and liquid feed supplements, in bulk, in tank vehicles from the facilities of Cargill, Inc. at Garden City, Kans. to points in Colorado, Kansas, Nebraska, New Mexico, Oklahoma. and Texas, for 180 days. Supporting shipper(s): Cargill, Inc., P.O. Box 9300. Minneapolis, Minn. 55440. Send protests to: Transportation Assistant Kathy

Henson, Rm. 246, Old Post Office Bldg.. 215 Northwest Third St., Oklahoma City, Okla. 73102.

No. MC 114569 (Sub-No. 184TA), filed July 12, 1977. Applicant: SHAFFER TRUCKING, INC., P.O. Box 418, New Kingstown, Pa. 17072. Applicant's representative: Herbert R. Nurick, P.O. Box 1166, Harrisburg, Pa. 17108. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Confectionery, cocoa. chocolate, and products related thereto (except in bulk), and materials, supplies, equipment, and machinery, used in the manufacture, production, distribution or sale of confectionery, cocoa, chocolate, and products related thereto: From the facilities of Hershey Foods Corp., at or near Oakdale, Calif.; to Chicago, Ill.; Cincinnati, Ohio; Detroit, Mich.; East St. Louis, Ill.; Derry Township, Dauphin County, Pa.; Kansas City, Mo.; Milwaukee, Wis.; St. Paul, Minn.; and their commercial zones, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper: Hershey Foods Corp., Hershey, Pa. 17033. Send protests to: Charles F. Mayers, Dist. Supv. Bureau of Operations, Interstate Commerce Commission, 278 Federal Building, P.O. Box 869, Harrisburg, Pa.

No. MC 114632 (Sub-No. 118TA), filed July 21, 1977. Applicant: APPLE LINES, INC., 212 Southwest Second St., Madison, S. Dak. 57042. Applicant's representative: Robert Gisvold, 1000 First National Bank Bldg., Minneapolis, Minn. 55402. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Meats. meat products and meat byproducts and articles distributed by meat packinghouses as described in sections A and C of Appendix I to the report in descriptions in motor carrier certificates, 61 MCC 209 and 766 (except hides and commodities in bulk) from Dodge City. Kans., to points in Connecticut, Delaware, District of Columbia, Iilinois, Indiana, Iowa, Kentucky, Maine, Maryland, Massachusetts, Michigan, Minnesota, Missouri, Nebraska, New Hampshire, New Jersey, New York, North Dakota, Ohio, Oklahoma, Pennsylvania, Rhode Island, South Dakota, Tennessee, Texas, Vermont, West Virginia, and Wisconsin, for 180 days. Supporting shipper(s): Hyplains Dwssed Beef, Inc., P.O. Box 539, Dodge City, Kans. 67801, Max Kline, Assistant Manager, Beef Department. Send protests to: J. L. Hammond, District Supervisor, Interstate Commerce Commission, Bureau of Operations, Room 455, Federal Bldg., Pierre, S. Dak. 57501.

No. MC 119555 (Sub-No. 16TA), filed July 14, 1977. Applicant: OIL & INDUSTRY SUPPLIERS, LTD., P.O. Box 3500, Calgary, Alberta, Canada. Applicant's representative: Ray F. Koby, 314 Montana Bldg., Great Falls, Mont. 59401. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Muriatic acid, in

bulk, in tank vehicles, from the United States-Canada International Boundary Line, located at or near the port of entry of International Falls, Minn., to Duluth and St. Paul, Minn., restricted to traffic originating in Ontario, Canada, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): Van Waters & Rogers, 2313 Wycliff St., St. Paul, Minn. 55114. Send protests to: Paul J. Labane, District Supervisor, Interstate Commerce Commission, 2602 1st Ave., North Billings, Mont, 59101.

No. MC 120761 (Sub-No. 23TA), filed July 12, 1977. Applicant: NEWMAN BROS. TRUCKING COMPANY, 6559 Midway Road, P.O. Box 13302, Fort Worth, Tex. 76118. Applicant's representative: Clint Oldham, 1108 Continental Life Bldg., Fort Worth, Tex. 76103. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Roofing materials, composition shingles, rolled roofing. roofing compounds, and accessories thereto, from the plantsite and storage facilities of ELK Corp., at or near Stephens and Camden, Ark., to points in Alabama, Kentucky, Louisiana, Mississippi, Oklahoma, Tennessee and Texas for 180 days. Applicants has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): ELK Corp., P.O. Box 37, Stephens, Ark. 71764. Send protests to: Robert J. Kirspel, District Supervisor, Room 9A27, Federal Building, 819 Taylor Street, Fort Worth, Tex. 76102.

No. MC 121664 (Sub-No. 22 TA), filed July 12, 1977. Applicant: G. A. HORN-ADY, CECIL M. HORNADY, AND B. C. HORNADY, A PARTNERSHIP, d.b.a. HORNADY BROTHERS TRUCK LINE. P.O. Box 846, Drewry Road, Monroeville, Ala. 36460. Applicant's representative: W. E. Grant, 1702 First Avenue. South Birmingham, Ala. 35233. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Lumber, lumber products, particleboard and plywood, between points in Alabama, Florida, Georgia, Kentucky, Mississippi and Tennessee, restricted to shipments originating at or destined to a facility of Moore-Hadley, Inc., for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): Moore-Hadley, Inc., P.O. Box 2607, Birmingham, Ala. 35202. Send protests to: Clifford W. White, District Supervisor, Bureau of Operations, Interstate Commerce Commission, Room 1616, 2121 Building, Birmingham, Ala. 35203.

No. MC 123233 (Sub-No. 77TA), filed July 20. 1977. Applicant: PROVOST CARTAGE, INC., 7887 Grenache St., Ville d'Anjou, Quebec, Canada, H1J 1C4. Applicant's representative: J.P. Vermette (same address as applicant). Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Corn syrup, in bulk, in tank vehicles, from Chicago, Ill., and its

Commercial Zone, to the Ports of Entry on the International Boundary Line, between the United States and Canada located in Michigan and New York. Restricted to the transportation of traffic in foreign commerce destined to points in Quebec, Canada, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): Clinton Corn Processing, Inc., Subsidiary of Standard Brands, Inc., P.O. Box 340, Clinton, Iowa 52732. Send protests to: District Supervisor, David A. Demers, Interstate Commerce Commission, P.O. Box 548, 87 State St., Montpelier, Vt. 05602.

No. MC 123387 (Sub-No. 8TA), filed July 7, 1977. Applicant: E. E. HENRY. an individual, 1923 Sparrow Rd., Chesapeake, Va. 23320. Applicant's representative: William P. Jackson, Jr., 3426 North Washington Blvd., P.O. Box 1267. Arlington, Va. 22210. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Alcoholic liquors (except in bulk), from Laredo, Tex., to points in North Carolina, Virginia, West Virginia, Maryland, Delaware, and the District of Columbia, for 180 days. Supporting De-Mar Internationale, shipper(s): Ltd., Donald P. Eden, Ex Vice President, 1285 North King St., Hampton, Va. 23669. Send protests to: District Supervisor, Paul D. Collins, Bureau of Operations, Room 10502, Federal Bldg., 400 North 8th St., Richmond, Va.

No. MC 123765 (Sub-No. 10TA), filed July 21, 1977. Applicant: BARRY TRANSFER & STORAGE CO., INC., 120 E. National Avenue, Milwaukee, Wis. 53205. Applicant's reprsentative: Wm. C. Dineen, 710 N. Plankinton Avenue, Milwaukee, Wis. 53203. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Commodities manufactured, shipped, or received by Outboard Marine Corp., its subsidiaries and divisions (except commodities in bulk), between Milwaukee, Manawa and Beloit, Wisconsin. and Waukegan and Galesburg, Ill. Restriction: Restricted to traffic orginating at and/or destined to, the plantsites, warehouses and distribution facilities of Outboard Marine Corp., its subsidiaries and divisions, for 180 days. Supporting shipper(s): Outboard Marine Corp., 180 Sea Horse Drive, Waukegan, III. 60085 (Roland Ronshausen). Send protests to: Gail Daugherty, Transportation Asst., Interstate Commerce Commission, Bureau of Operations, U.S. Federal Building & Courthouse, 517 East Wisconsin Avenue, Room 619, Milwaukee, Wis. 53202.

No. MC 129951 (Sub-No. 3TA), filed July 15, 1977. Applicant: HARLEY I. KEETER, 6379 Valmont Drive Boulder, Colo. 80301. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Ore and ore concentrates, from Boulder County, Colo., to ports of entry located at the International boundaries in the States of Washington and Idaho for 90

days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): Hendricks Mining Co., Inc., 3000 N. 63rd, Boulder, Colo. 80301. Send protests to: Roger L. Buchanan, District Supervisor, Interstate Commerce Commission, 492 U.S. Customs House, 721 19th Street, Denver, Colo. 80202.

No. MC 135082 (Sub-No. 49TA), filed July 21, 1977. Applicant: Bursch Trucking, Inc., d.b.a., Roadrunner Trucking, Inc., P.O. Box 26748, 415 Rankin Road, NE., Albuquerque, N. Mex. 87125. Applicant's representative: D. F. Jones, President, P.O. Box 26748, Albuquerque, N. Mex. 87125. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Forest products, from Prescott, Ariz., to points in California, Colorado, Oklahoma, Nevada, New Mexico, Texas, and Utah, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): Southwest Forest Industries, P.O. Box 7548, Phoenix, Ariz. 85011. Attention: Homer Davenport, General Traffic Manager. Send protests to: Darrell W. Hammons, District Supervisor, 1106 Federal Office Building, 517 Gold Avenue, SW., Interstate Commerce Commission, Bureau of Operations, Albuquerque, N. Mex. 87101.

No. MC 136008 (Sub-No. 84TA), filed July 22, 1977. Applicant: Joe Brown Co., Inc., 8005 South I-35, Suite 102, Oklahoma City, Okla. 73149. Applicant's representative: John Tipsword, 8005 South I-35, Suite 102, Oklahoma City, Okla. 73149. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Gypsum, gypsum products and building materials from the plantsite of United States Gypsum Co., Southard, Okla. to Colorado, Kansas, Missouri, and Texas, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): U.S. Gypsum Co., 101 S. Wacker Drive, Chicago, Ill. 60606, Send protests Transportation Assistant Kathy Henson, Rm. 240, Old Post Office Bldg., 215 Northwest Third St., Oklahoma City, Okla, 73102.

No. MC 138861 (Sub-No. 5TA), filed July 6, 1977, Applicant: C-LINE, INC., Tourtellot Hill Road, Chepachet, R.I. 02814. Applicant's representative: Ronald N. Cobert, 1730 "M" Street NW., Washington, D.C. 20036. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Automotive oil and lubricants, in containers, from Bay Way, N.J., and Bradford and Emlenton, Pa., to points in Connecticut, Florida, Indiana, Kentucky, Massachusetts, Maine, Maryland, New Hampshire, New Jersey, New York, North Carolina, Ohio, Pennsylvania, Rhode Island, Tennessee, Vermont, Virginia and West Virginia for 180 days. Supporting shipper(s): Shore Sales Co., 97 River Street, Beverly, Mass. 01915. Send protests to: Gerald H. Curry District Supervisor, 24 Weybosset Street, Room 102, Providence, R.I. 02903.

No. MC 139360 (Sub-No. 8TA), filed July 19, 1977. Applicant: RAEMARC, INC., 153 Taylor, Racine, Wis. 53403. Applicant's representative: Daniel C. Sullivan. 10 S. LaSalle St., Chicago, Ill. 60603. Authority sought to operate as a contract carrier, by motor vehicle, over irregular routes, transporting: (1)
Materials and supplies (except commodities in bulk), used in the manufacture of agricultural, industrial and contruction machinery and equipment; (2) parts and attachments for agricultural, industrial and construction machinery and equipment between the manufacturing and storage facilities of I. Case Co., at or near Racine, Winneconne, and Wausau, Wis.; Bettendorf and Brulington, Iowa; Terre Haute, Ind.; and Rock Island, Ill.; on the one hand, and, on the other, points in Illinois, Indiana, Iowa, Michigan, Minnesota, Missouri, Ohio and Wisconsin, under contract with J. I. Case Company, for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper: J I Case Co. 700 State St., Racine, Wis. 53404. Send protests to: Gail Daugherty, Transportation Asst. Interstate Commerce Commission, Bureau of Operations, U.S. Federal Building & Courthouse, 517 East Wisconsin Ave., Rm. 619. Milwaukee, Wis. 53202.

No. MC 139577 (Sub-No. 7TA), filed July 20, 1977. Applicant: ADAMS TRAN-SIT, INC., P.O. Box 338, 204 East Winnebago St., Friesland, Wis. 53935. Applicant's representative: Wayne W. Wilson, P.O. Box 8004, Madison, Wis, 53708. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Containers, container closures and ends from Friesland, Wis. to Fairmont, Minn., Scottville, Mich. and Hart, Mich., for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper(s): Stokley-Van Camp, Inc., 941 North Meridian St., Indianapolis, Ind. 46206. Send protests to: Ronald A. Morken, District Supervisor, Interstate Commerce Commission, 139 W. Wilson St., Room 202, Madison, Wis. 53703.

No. MC 140216 (Sub-No. 4TA), filed July 18, 1977. Applicant: JOHN E. WAY, JR., doing business as WAY MESSEN-GER SERVICE, 205 East King St., Lancaster, Pa. 17602. Applicant's representative: John M. Musselman, P.O. Box 1146, Harrisburg, Pa. 17108. Authority sought to operate as a common carrier, by motor vehicle, over irregular routes, transporting: Printed matter between Lancaster. Pa., on the one hand, and, on the other, Baltimore, Md., and Washington, D.C.; and between York, Pa., on the one hand, and, on the other, Baltimore, Md., New York, N.Y., and Washington, D.C.; service to be limited to expedited pickups and expedited deliveries by messenger drivers of minimum gross weight straight trucks, for 180 days. Applicant has also filed an

underlying ETA seeking up to 90 days of operating authority. Supporting shipper: There are statements of support attached to the application, which may be examined at the Interstate Commerce Commission, in Washington, D.C., or copies thereof which may be examined at the field office named below. Send protests to: Charles F. Mayers, Dist. Supv. Bureau of Operations, Interstate Commerce Commission, 278 Federal Bldg. P.O. Box 869, Harrisburg, Pa. 17108.

No. MC 140849 (Sub-No. 12TA), filed July 18, 1977. Applicant: ROBERTS TRUCKING CO., INC., P.O. Drawer G. U.S. Highway 271 South, Poteau, Okla. 74953. Applicant's representative: Prentiss Shelley (same address as applicant, Authority sought to operate as a contract carrier, by motor vehicle, over irregular routes, transporting; (1) Fabrics, piece goods, materials, and supplies used in the manufacture of clothing; (2) clothing. from: (1) Pauls Valley and Idabel, Okla.; to Roswell, Ga.; Hopkinsville, Ky.; and Nashville, Tennessee: (2) from Roswell, Ga.: Hopkinsville, Ky.; and Nashville, Tenn, to Pauls Valley and Idabel, Okla.; under a continuing contract with Kellwood Co., for 180 days. Supporting shipper: Kellwood Co., P.O. Box 656, Pauls Valley, Okla. 73075, Send protests to: District Supervisor, William H. Land, Jr., 3108 Federal Office Bldg., 700 West Capitol, Little Rock, Ark. 72201.

No. MC 143423 (Sub-No. 1TA), filed July 12, 1977. Applicant: WILLIAM T. AUSTIN, d.b.a. AUSTIN TRUCKING CO., Route 5, Box 249, Decatur, Ala. 35601. Applicant's representative: D. H. Markstein, Jr., 512 Massey Bldg., Birmingham, Ala. 35203. Authority sought to operate as a contract carrier, by motor vehicle, over irregular routes, transporting: (1) Electrical wire cable and insulated copper wire, from the plantsite of Cerro Wire & Cable Co., Division of Cerro Corp., at Hartselle, Ala., to points in Georgia, Indiana, Louisiana, Mississippi, Missouri, North Carolina, Tennessee, Texas, Illinois, South Carolina, Florida, Kentucky, Michigan, Pennsylvania, and New York; (2) sugar from Gramercy, La., to Decatur, Ala., from Decatur, Ala., to points in and east of North Dakota, South Dakota, Kansas, Oklahoma, and Texas, under contract with VIPCO. Inc., Decatur, Ala., for 180 days. Supporting shipper(s): Cerro Wire & Cable Co., 201 Cedar Cove Rd., Hartselle, Ala. 35640; Vipco, Inc., 1302 Southfield Industrial Park, Decatur, Ala. 35602. Send protests to: Clifford W. White, District Supervisor, Bureau of Operations, Interstate Commerce Commission, Room 1616, 2121 Building, Birmingham, Ala. 35203.

No. MC 143471TA filed July 6, 1977. Applicant: SHERIDAN HEIGHTS, INC., doing business as KNECHT TRANS-PORT, 301 Mount Rushmore Rd., Rapid City, S. Dak. 57701. Applicant's representative: J. Maurice Andren, 1734 Sheridan Lake Rd., Rapid City, S. Dak. 57701. Authority sought to operate as a contract carrier, by motor vehicle, over irregular routes, transporting: Such commodities

NOTICES

processors or distributors of building materials and supplies; hardware; plumbing supplies; electrical supplies; cement materials and landscaping materials: I. Between the plant sites, warehouses, stores, outlets or other facilities of Knecht Industries, Inc., including their wholly owned, unincorporated divisions of Building Material Distributors, Big K Cash & Carry, Mastercraft Factory, Homes by Knecht, Mastercraft Homes and Knecht Lumber Co., located in Campbell County, Wyo., and Pennington County, S. Dak., on the one hand, and on the other hand, points in California, Colorado, Idaho, Iowa, Minnesota, Montana, Nebraska, Nevada, North Dakota, Oregon, South Dakota, Utah, Washington and Wyoming; and II. Between con-struction sites and dealers of Knecht Industries, Inc., including their wholly owned, unincorporated divisions of Building Materials Distributors, Big K Cash & Carry, Mastercraft Factory, Homes by Knecht, Mastercraft Homes, and Knecht Lumber Co., located in Montana, Nebraska, North Dakota, South Dakota and Wyoming, on the one hand, and on the other hand, points in California, and Wyoming, on the one hand, and on the other hand, points in California. Colorado, Idaho, Iowa, Minnesota, Montana, Nebraska, Nevada, North Dakota, Oregon, South Dakota, Utah, Washington and Wyoming. Both parts under a continuing contract or contracts with Knecht Industries, Inc. and their wholly owned, unincorporated divisions of Building Materials Distributors, Big K Cash & Carry, Mastercraft Factory. Homes by Knecht, Mastercraft Homes and Knecht Lumber Co. Restriction: Against commodities in bulk in tank vehicles in both parts for 180 days. Applicant has also filed an underlying ETA seeking up to 90 days of operating authority. Supporting shipper: Knecht Industries, Inc., 301 Mount Rushmore Rd., Rapid City, S. Dak. 57701, Keith Danley, Director of Purchasing. Send protests to: J. L. Hammond, District Supervisor, Interstate Commerce Commission, Bureau of Operations, Rm. 455, Federal Building. Pierre, S. Dak. 57501.

# PASSENGER APPLICATION

No. MC 143475TA, filed July 7, 1977. Applicant: Potomac Valley Transit Authority, One Virginia Ave., Petersburg, W. Va. 26847. Applicant's representative: Kenneth W. Dyche, (same address as applicant). Authority sought to operate as

as are handled or dealt in by wholesalers, regular routes, transporting: Passengers to Winchester; and return over the same warehousers, retailers, manufacturers, and their baggage, in the same vehicle with passengers, over regular routes, turn over the same route. Passengers limited to transportation of no more than 20 passengers in any one vehicle, between Romney, W. Va.; and Cumberland Md.; via Springfield and Ridgely, W. Va.; and serving all intermediate points as follows: From Romney over West Virginia Route 28 to Cumberland. and return over the same route. Between Pledmont, W. Va.; and Cumberland, Md.; via Westernport, Md.; and Keyser, W. Va.; and serving all intermediate points as follows: From Piedmont, over Maryland Route 36 to Westernport: thence from Westernport over Maryland Route 135 to junction with U.S. Route 220; thence from the junction over U.S. Route 220 to Keyser; thence from Keyser over West Virginia Route 46 to junction with West Virginia Route 28; thence from the junction over West Virginia Route 28 to Cumberland; and return over the same route. Between Romney, W. Va.; and Cumberland, Md.; via Keyser, W. Va.; and serving all intermediate points as follows: From Romney over U.S. Route 50 to junction with U.S. Route 220; thence from the junction over U.S. Route 220 to Keyser; thence from Keyser over West Virginia Route 46 to junction with Mineral County Route 9; thence from the junction over County Route 9 to junction with West Virginia Route 28; thence from the junction over West Virginia Route 28 to Cumberland; and return over the same route. Between Petersburg, W. Va.; and Winchester, Va. via Moorefield and Wardensville, W. Va.; and serving all intermediate points; as follows; From Petersburg over U.S. Route 220 to Moorefield; thence from Moorefield over West Virginia Route 55 to Wardensville; thence from Wardensville over West Virginia Route 259 to junction with U.S. Route 50; thence from the junction over U.S. Route 50 to Winchester; and return over the same route. Between Moorefield, W. Va.; and Harrisonburg, Va. via Petersburg and Franklin, W. Va.; and serving all intermediate points; as follows: From Moorefield over U.S. Route 220 to Franklin; then from Franklin over U.S. Route 33 to Harrisonburg; and return over the same route. Between Petersburg, W. Va.; and Winchester, Va.; via Moorefield and Romney and Capon Bridge, W. Va.; and serving all intermediate points as follows:

From Petersburg over U.S. Route 220 to junction with U.S. Route 50; thence a common carrier, by motor vehicle, over from the junction over U.S. Route 50

route. Between Moorefield, W. Va.; and Harrisonburg, Va.; via Baker, W. and serving all intermediate points as follows: From Moorefield over West Virginia Route 55 to Baker; thence from Baker over West Virginia Route 259 to junction with Interstate Route 81; thence from the junction over Interstate Route 81 to Harrisonburg; and return over the same route, Between Keyser, W. Va.; and Piedmont, W. Va. via Westernport, Md.; and serving all intermediate points as follows: From Keyser over U.S. Route 220 to junction with Maryland Route 135; thence from the junction over Maryland Route 135 to Westemport; thence from Westernport over Maryland Route 36 to Piedmont; and reand their baggage, in the same vehicle with passengers, in round-trip special and charter operations; limited to transportation of no more than twenty (20) passengers in any one vehicle as follows: Authority is requested for operations beginning and ending at places and points in Grant, Hampshire, Hardy, Mineral and Pendleton Counties of West Virginia and extending to places and points in the States of Maryland, Virginia, Pennsylvania, Delaware, New Jersey, The District of Columbia, and that part of the State of New York that is south of Interstate Highway 84, including all of Long Island, for 180 days. Supporting shipper: Region 8 Planning & Dev. Council, P.O. Box 887, Petersburg, W. Va. 26847. J. A. Niggermyer, District Supervisor, Interstate Commerce Commission, 416 Old Post Office Building, Wheeling, W. Va.

39735

By the Commission.

H. G. HOMME, Jr., Acting Secretary.

[FR Doc 77-22596 Filed 8-4-77;8:45 am]

[Volume No. 26]

APPLICATIONS, FINANCE (INCLUDING TEMPORARY PETITIONS. MATTERS AUTHORITIES), RAILROAD ABANDON-MENTS, ALTERNATE ROUTE DEVIA-TIONS. AND INTRASTATE APPLICA-TIONS

Correction

In FR Doc. 77-20881 appearing at page 37468 in the issue for Thursday, July 21, 1977, on page 37474, in the third column, the paragraph beginning "No. MC 11274 (Sub-No. 25)" should begin "No. MC 111274 (Sub-No. 25) ".

# sunshine act meetings

This section of the FEDERAL REGISTER contains notices of meetings published under the "Government in the Sunshine Act" (Pub. L. 94-409), 5 U.S.C. 552b(a)(3).

	Item
Civil Aeronautics Board	3,4
Equal Employment Opportunity	
Commission	8
Federal Deposit Insurance Cor-	
poration	9
Federal Maritime Commission	1
Federal Reserve System	2
National Science Board	10
Securities and Exchange Commis-	-

CONTENTS

1

# FEDERAL MARITIME COMMISSION.

TIME AND DATE: 10 a.m., August 10, 1977.

PLACE: Room 12126, 1100 L Street, NW., Washington, D.C. 20573.

STATUS: Parts of this meeting will be open to the public. The rest of the meeting will be closed to the public.

#### MATTERS TO BE CONSIDERED:

Portions open to the public:

 Monthly Report of Managing Director's Actions Pursuant to Delegated Authority (June).

 Agreement No. 9975-8: Petition to extend the duration of the Japan-East Coast U.S.A. Containership Service Agreement to August 22, 1980.

3. Agreement No. 10117-3: Modifications of the U.S. North Atlantic Spain Rate Agreement, regarding voting procedures, self-policing features, and amount of financial guarantee.

4. Agreement No. 16107-2; Application for Intermodal Authority by Trans-Pacific Freight Conference and eight noncon-

ference competitors.

5. Agreements Nos. 8718-5 and 9731-7: Extensions of the Japanese Flag Containership Service Agreements in the trade between Japan and California and between Japan and California, Hawaii and Alaska, to August 22, 1980.

6. Docket No. 77-10—Agreement Nos. 10072 nd 10072-1 (Establishment of Conference of Passenger and Cruise Lines). Determination whether to Review Discontinuance by the Administrative Law Judge.

# Portions closed to the public:

 Agreement No. 10286; Italy-U.S.A. North Atlantic Pool Agreement, establishment of cargo and revenue pooling agreement.

2. Docket No. 75-3-Chevron Chemical Company v. Mitsui O.S.K. Lines, Ltd., (Preight Overcharge Claim): Petition for Reconsideration of Adoption of Initial Decision.

 Docket No. 77.4—Agreements Nos. 9902-3, 9902-4, 9902-5, and 9902-6 (Modification of Euro-Pacific Joint Service Agreement). Petition for Reconsideration and Motion for Stay of the Commission's Order of Conditional Approval of Agreement No. 9902-5 Pendente Lite.

4. Docket No. 77-21—In the Matter of Continued Qualification for Independent Ocean Freight Forwarder License No. 1744R-Orlando A. Puig d.b.a. Houston Export International; Determination whether to Review Discontinuance by Administrative Law Judge.

5. Docket Nos. 77-27—Trailer Marine Transport Corp., and 77-28—Guif Caribbean Marine Lines, Inc., General Increases in Rates—Motion to Vacate Suspension.

CONTACT PERSON FOR MORE IN-FORMATION:

Joseph C. Polking, Acting Secretary (202-523-5727).

(S-1045-77 Filed 8-2-77;2:33 pm)

2

FEDERAL RESERVE SYSTEM.

"FEDERAL REGISTER" CITATION OF PREVIOUS ANNOUNCEMENT: Sent to FEDERAL REGISTER July 27, 1977.

PREVIOUSLY ANNOUNCED TIME AND DATE OF THE MEETING: 10 a.m., Wednesday, August 3, 1977.

CHANGES IN THE MEETING: Addition of the following open item to the meeting:

 Board consideration of temporary suspension of Regulation Q (Interest on Deposits) penalty for early withdrawal of time deposits for Johnstown, Pennsylvania disaster area.

#### Previously announced open items:

(a) Proposed amendments to Regulation H (Membership of State Banking Institutions in the Federal Reserve System) and Y (Bank Holding Companies), to require certain municipal securities dealers to file with the Board information about persons associated with them as municipal securities principals or municipal securities representatives. (Proposed earlier for public comment; docket no. R-0090).

(b) In connection with the proposed amendments to Regulations H and Y, establiahment of a system of records under the Privacy Act of 1974 dealing with persons who are or seek to be municipal securities principals or representatives and are associated with certain financial institutions. (Proposed earlier for public comment; docket no, R-0091).

2. Possible amendments to the Board's Rules Regarding Delegation of Authority to delegate to the Director of the Division of Banking Supervision and Regulation certain authority in connection with administration of Municipal Securities Rulemaking Board rules.

3. A possible Board interpretation of Regulation B (Equal Credit Opportunity) to clarify the meaning of § 202.8, Special Purpose Credit Programs, by defining the phrase "expressly authorized by law". (Proposed earlier for public comment; docket no. R-0100).  Proposal by the Federal Reserve Bank of Atlanta for approval of conceptual design for the Miami Branch building.

Note.—The initial notice for this meeting was received by the Program Recustres on July 26, 1977, seven days in advance of the meeting.

CONTACT PERSON FOR MORE IN-FORMATION:

Mr. Joseph R. Coyne, Assistant to the Board: 202-452-3204.

Dated: August 2, 1977.

GRIFFITH L. GARWOOD, Deputy Secretary of the Board.

[S-1046-77 Filed 8-3-77;1:52 pm]

3

CIVIL AERONAUTICS BOARD.

TIME AND DATE: 10 a.m., August 2. 1977.

PLACE: Room 1027, 1825 Connecticut Avenue NW., Washington, D.C. 20428.

SUBJECT: 5. TXI Subpart M Application (Not. No. 7139-B) Doc. 30887.

STATUS: Open.

PERSON TO CONTACT:

Phyllis T. Kaylor, the Secretary, (202-673-5068).

SUPPLEMENTARY INFORMATION: On May 16, 1977, Texas International Airlines (TXI) filed an application requesting the removal of certain restrictions on its certificate of public convenience and necessity in order to permit certain additional service. Order 77-5-167, dated May 31, 1977, directed TXI to file additional material for the Board's consideration of this applica-tion. On July 15, 1977, TXI filed a supplement to its application with the required information and on July 25, 1977. supplemental statements requesting dismissal were filed by American and Delta. On July 26, 1977, the Bureau of Operating Rights submitted its recommendation to the Board.

TXI's application was filed under Subpart M of the Board's Rules of Practice (14 CFR § 302.1301-1315). Under the provisions of Subpart M the Board was required to stay or dismiss TXI's application by July 29, 1977, or the application would have automatically moved forward to the next procedural stage.

forward to the next procedural stage.
On July 29, 1977, the Board voted to dismiss the application and to so notify TXI and to issue the order after discussion of the language to be used in the order. So that the order can issue with minimum delay the following Board Members have voted that agency busi-

ness requires the addition of this Docket 30887, Application of Texas International Airlines, Inc., for amendment of its certificate of public convenience and necessity under Subpart M procedures, to the agenda of its August 2, 1977, meeting and that no earlier announcement of the addition was possible:

Chairman Alfred E. Kahn Vice Chairman Richard J. O'Melia Member G. Joseph Minetti Member Lee R. West

[S-1047-77 Filed 8-2-77;4:15 pm]

4

CIVIL AERONAUTICS BOARD.

TIME AND DATE: 10 a.m., August 9,

PLACE: Room 1027, 1825 Connecticut Avenue NW., Washington, D.C. 20428,

1. Ratification of items adopted by nota-

2. Docket 26573, application of American Airlines for renewal of its temporary Islip-Chicago authority (Memo. No. 7307, BOR). 3. Docket 29554, Airwest's application for

authority between Las Vegas-Albuquerque-El Paso-Midland/Odessa-San Antonio-Corpus Christi and Houston and between Phoenix/Tueson-San Antonio-Houston and New Orleans and motion for immediate hearing (Memo, No. 7314, BOR, BLJ).

4. Docket 25342, complaint of aviation consumer action project against the local service carriers (Memo. No. 6563-A, BOR, OCCR,

OGC)

5. Docket 20826 et al., bush route phase of Alaska service investigation (Memo, No. 7302,

6. Dockets 25427 and 25457, Laker Airways, Ltd., "Skytrain" Service Order 77-6-68, target date for Board ruling on petition for reconsideration filed by Laker Airways (Memo. No. 7316, OGC).
7. Docket 28807, Trans International Air-

lines, enforcement proceeding, notice of change of target date (Memo. No. 7309, OGC).

8. Docket 31103, American's application for exemption to provide free transportation to cargo agents from various points to Fort Worth, Tex., and return (Memo. No. 7308,

9. Docket 31122, fare-effectiveness rule proposed by United (tariff sched, for eff. 8-15) (Memo. No. 6967-A, BE).

10. Dockets 15529 and 27589, baggage and liability rules case and domestic baggage liability rules investigation (Memos. Nos. 7132-A and 7132-B, BE)

11, Request of Air B.V.I., Ltd., to use a notice of deliberate overbooking of its own wording (Memo. No. 7313, BE).

 Docket 30756. Pan American's request for permission to hold discussions on North/ Central Pacific cargo rates (Memo. No. 7310,

13. Docket 29100, subsidy rate amendment two to Order 76-12-159, class rate VIII (Memo. No. 5972-J, BE, OC).

14. Docket 31160, complaint of TWA against Prontier's proposed assembly group fares in various Las Vegas markets (sched. for eff. B-15) (BE).

15. Docket 31201, complaint of Century Air Freight, an air freight forwarder, against addition by Airlift of container loading service provisions offering service by the carrier on its premises, applicable to traffic moving to/from San Juan, P.R. (sched. for eff. 8-15)

16. Docket 30985, application of Pan American for extension of embargo on all cargo destined for Caracas not holding confirmed reservations due to lack of storage facilities

(BE), 17. Docket 30332, disposition of various IATA agreements dealing with currency matters, and cargo matters, Agreements CAB 26701, 26703, 26704, and 26707 (BE).

18. Docket 31031, disposition of complaint by Seaboard against a cargo tariff filed by

KLM (BE).

19. Docket 27813, petition by Qantas for reconsideration of Order 77-5-133 dealing with combination fares over Pago Pago, Agreement CAB 25711 (BE).

20. Docket 31161, disposition of complaint by Seaboard against a cargo tariff filed by

TWA (BE).

21. Dockets 31202 and 31203, disposition of complaints by Seaboard against cargo tar-ins filed by KLM (BE).

STATUS: Open.

PERSON TO CONTACT:

Phyllis T. Kaylor, the Secretary (202-673-5068).

[S-1048-77 Filed 8-2-77 4:16 pm]

5

SECURITIES AND EXCHANGE COM-MISSION.

TIME AND DATE: August 2, 1977, 9:35 a.m.

PLACE: Room 812, 500 North Capitol Sreet, Washington, D.C.

STATUS: Closed meeting.

SUBJECT MATTER: Consideration of litigation matter. Chairman Williams, Commissioners Loomis, Evans, and Pollack determined that Commission business required consideration of this matter and that no earlier notice thereof was possible.

AUGUST 2, 1977.

[S-1049-77 Filed 8-3-77 9:04 am]

6

SECURITIES AND EXCHANGE COM-MISSION.

FEDERAL REGISTER CITATION OF PREVIOUS ANNOUNCEMENT: 42 FR 38689, July 29, 1977.

PREVIOUS ANNOUNCED TIME AND DATE: August 5, 1977, 10 a.m.

PLACE: Room 825, 500 North Capitol Street, Washington, D.C.

CHANGES IN THE MEETING: The following additional matter will be considered by the Commission at the open

Consideration of a proposed rule which would prohibit mutual funds from implementing arrangements involving the use of their assets to finance the distribution of shares as an interim measure.

The following additional matters will be considered by the Commission at the closed meeting, immediately following the open meeting: Institution of administrative proceedings; settlement of injunctive actions.

Chairman Williams and Commissioners Evans and Pollock determined that Commission business required consideration of these matters and that no earlier notice thereof was possible.

August 2, 1977.

[S-1050-77 Filed 8-3-77;9:04 am]

7

SECURITIES AND EXCHANGE COM-MISSION.

Notice is hereby given, pursuant to the provisions of the Government in the Sunshine Act, Pub. L. 94-409, that the Securities and Exchange Commission will hold the following meetings during the week of August 8, 1977, in Room 825, 500 North Capitol Street, Washington, D.C.

A closed meeting will be held on Tuesday, August 9, 1977, at 10 a.m. An open meeting will be held on Wednesday, August 10, 1977, at 10 a.m.

The Commissioners, their legal assistants, the Secretary of the Commission. and recording secretaries will attend the closed meetings. Certain staff members who are responsible for the calendared matters may be present.

The General Counsel of the Commission, or his designee, has certified that, in his opinion, the items to be considered at the closed meetings may be so considered pursuant to one or more of the exemptions set forth in 5 U.S.C. 552b(c) (4), (8), (9) A, and (10) and 17 CFR 200.402(a) (4), (8), (9) (i), and (10)

Chairman Williams and Commissioners Evans and Pollack voted to hold the aforesaid meetings in closed session.

The subject matter of the closed meeting scheduled for Tuesday, August 9, 1977

Formal order of investigation. Institution of injunctive actions. Settlement of injunctive actions.

Settlement of administrative proceedings. Referral of investigative filed to Federal, State, or self-regulatory authorities.

Regulatory matters arising from or bearing enforcement implications.

Preedom of Information Act appeals. Request for confidential treatment. Personnel matter.

Application for reinstatement.

The subject matter of the open meeting scheduled for Wednesday, August 10, 1977, will be:

- Consideration of proposals submitted by CBOE, Amex. PHLX, MSE, and PSE to amend their respective rules setting forth option exercise price intervals.
- 2. Consideration of a proposal permitting self-regulatory organizations to establish criteria on which to base recommendations for permanent extensions of time for the

The ratification process provides an entry the Board's minutes of items already adopted by the Board through the written notation process (memoranda circulated to the Members sequentially). A list of items ratified at this meeting will be available in the Board's Public Reference Room, Room 710, 1825 Connecticut Avenue NW., Washington, D.C. 20428, following the meeting.

filing of Part I of Form X-17A-5, which is the monthly report submitted by clearing firms.

3. Consideration of the adoption of Rule 2a-5 under the Investment Company Act of 1940. Rule 2a-5 would declare that certain persons would not be deemed interested persons under the statutes of the Act.

4. Consideration of a petition for Com-mission review filed by Mortgage Investors of Washington regarding an extension of time requested by the company to file their annual report for the year ended March 31,

5. Consideration of the publication of releases regarding: (1) Reporting provisions relating to the remuneration received by officers and directors and (2) the disclosure of criminal litigation involving taxation.

6. Consideration of an application filed by Air Pollution Industries, Inc., pursuant to section 12(h) of the Securities Exchange Act of 1934 for an order which would exempt the company from certain reporting require-

FOR FURTHER INFORMATION CON-TACT:

Edward B. Horahan at 202-376-8072 or Howard B. Scherer at 202-755-1280.

August 2, 1977.

IS-1051-77 Filed 8-3-77;9:04 am]

EQUAL EMPLOYMENT OPPORTU-NITY COMMISSION.

TIME AND DATE: 9:30 a.m. (eastern time), Wednesday, August 10, 1977.

PLACE: Chairman's Conference Room No. 5240, on the fifth floor of the Columbia Plaza Office Building, 2401 E Street N.W., Washington, D.C. 20506.

STATUS: Parts of the meeting will be open to the public. The rest of the meeting will be closed to the public.

MATTERS TO BE CONSIDERED:

Portions open to the public:

(1) Freedom of Information Act Appeal 77-2-FOIA-40. Request, on behalf of an employer charged with discrimination, for documents in the Commission's investigative

Freedom of Information Act Appeal 77-2-FOIA-134. Request by an attorney representing an employer charged with discrimination, against whom a lawsuit has been

filed by the Commission, for the internal memorandum upon which the Commission based its vote to commence litigation

(3) Designation of 706 Agency. Recom-mendation that the Sioux Falls, S. Dak., Human Relations Commission be designated as one to which the Commission may defer charges.

(4) Washington State Human Commission. Recommendation that the Commission allocate additional fiscal year 1977 funds to this State agency for the proc-

essing of more charges.

(5) Revisions of EEOC's procedural regu-

Portions closed to the public:

(1) Litigation authorization; Counsel recommendations. Matters closed to the public under sec. 1612.13(a) of the Commission's regulations (42 FR 13830, March 14, 1977)

(2) Proposed procurement; processing of EEO-4 survey. This matter is continued from the meeting of August 2. Commission approval is requested to contract with an or-ganization to process the State and local government information report, identified as

the EEO-4 survey.
(3) Proposed budget request for fiscal year

Nore .- Any matter not discussed or concluded may be carried over to a later meeting.

CONTACT PERSON FOR MORE IN-FORMATION:

Marie D. Wilson, Executive Officer, Executive Secretariat, at 202-634-6748.

This notice issued August 3, 1977. [S-1052-77 Filed 8-3-77;10:19 am]

9

DEPOSIT INSURANCE FEDERAL CORPORATION.

Pursuant to the provisions of the Government in the Sunshine Act (5 U.S.C. 552b), notice is hereby given that, in addition to the matters already on the agenda for consideration at the meeting of the Federal Deposit Insurance Corporation's Board of Directors scheduled for 10:45 a.m. on Monday, August 8, 1977 (public notice of which was given on August 1, 1977), the Board of Directors will also consider an application from

the Anchor Savings Bank, New York (P.O. Brooklyn), New York, an insured mutual savings bank, for consent to merge under its charter and title with North New York Savings Bank, White Plains, N.Y., also an insured mutual savings bank, and to establish the five offices of North New York Savings Bank as branches of the resultant bank.

The time, date, place, and closed status of the meeting remain unchanged.

Dated: August 2, 1977.

FEDERAL DEPOSIT INSURANCE CORPORATION. ALAN R. MILLER. Executive Secretary

[S-1053-77 Filed 8-3-77;10:47 am]

10

NATIONAL SCIENCE BOARD.

TIME AND DATE: August 19, 1977. Closed session: 8:30 a.m.; open session: 2:00 p.m.

PLACE: National Science Foundation, Room 540, 1800 G Street NW., Washington, D.C. 20550.

STATUS: Parts of this meeting will be open to the public. The rest of the meeting will be closed to the public.

MATTERS TO BE CONSIDERED:

Portions open to the public:

- 1. Minutes-191st meeting.
- 2. Chairman's report.
- 3. Director's report.
- 4. Board committee reports.
- NSF advisory groups.
- 6. Site visits.

Portions closed to the public:

- 1. Minutes-closed session, 191st meeting. 2. Report of ad hoc committee on NSF staff
- and NSB nominees. 3. Possible options for RANN organization
- and leadership. 4. Grant and contracts-proposed awards.
  - 5. NSF budget for fiscal year 1979.

CONTACT PERSON FOR MORE IN-FORMATION:

Vernice Anderson (202-632-5840). [S-1054-77 Filed 8-3-77;11:23 am]

FRIDAY, AUGUST 5, 1977
PART II



# DEPARTMENT OF HEALTH, EDUCATION, AND WELFARE

Food and Drug Administration

LAETRILE

Commissioner's Decision on Status

# DEPARTMENT OF HEALTH. EDUCATION, AND WELFARE

Food and Drug Administration

[Docket No. 77N-0048] .

#### LAETRILE

# Commissioner's Decision

AGENCY: Food and Drug Administration.

ACTION: Notice.

SUMMARY: The Commissioner of Food and Drugs announces that he has compiled a comprehensive administrative record containing information about the drug Lactrile in general and, specifically, about two issues concerning Laetrile's "new drug" status: (1) Whether Laetrile is generally recognized by qualified experts as a safe and effective cancer drug, and (2) whether Laetrile is exempt from the premarket approval requirements for new drugs by virtue of the "grandfather" provisions of the Federal Food, Drug. and Cosmetic Act. The Commissioner concludes, after careful review of this administrative record, including oral argument presented at a public hearing, that: (1) Laetrile is not generally recognized by qualified experts as a safe and effective cancer drug, and (2) Laetrile is not exempt from the premarket approval requirements for new drugs by virtue of the "grandfather" provisions of the act. Distribution of Laetrile in interstate commerce is thus illegal and subject to regulatory activity by the Food and Drug Administration, Conclusions on other issues related to the controversy concerning Lactrile are also set out.

EFFECTIVE DATE: August 5, 1977.

ADDRESSES: The transcript of oral argument presented at the public hearing, affidavits, written testimony, and all other submissions compiled as the administrative record for this proceeding may be seen in the office of the Hearing Clerk (HFC-20), Food and Drug Administration, Rm. 4-65, 5600 Fishers Lane, Rockville, MD 20857, between 9 a.m. and 4 p.m., Monday through Friday. In addition, one copy of the administrative record (except two videotapes of interviews with cancer patients who had been treated with Lactrile) is available for public examination at the following Food and Drug Administration offices during regular business hours: 850 Third Ave., Brooklyn, NY 11232; 880 W. Peachtree St., Altanta, GA 30309; 433 W. Van Buren St., Chicago, IL 60607; 1009 Cherry St., Kansas City, MO 64106; 1521 W. Pico Blvd., Los Angeles, CA 90015; 909 First Ave., Seattle, WA 98104.

FOR FURTHER INFORMATION CON-TACT:

Tenny P. Neprud, Compliance Regulations Policy Staff (HFC-10), Food and Drug Administration, Department of Health, Education, and Welfare, 5600 Fishers Lane, Rockille, MD 20857, 310-44-3480.

SUPPLEMENTARY INFORMATION: In a notice published in the FEDERAL

REGISTER of February 18, 1977 (42 FR 10066), the Commissioner announced that he was initiating a rulemaking proceeding to comply with the opinion of the Court of Appeals in Rutherford v. United States, 542 F.2d 1137 (10th Cir. 1976), and the order of the District Court in Rutherford v. United States, 424 F. Supp. 105 (W. D. Okla, 1977). In those proceedings, the Food and Drug Administration (FDA) was ordered to develop an administrative record concerning the following two issues:

1. Whether the product Laetrile (also known as vitamin B-17 and amygdalin) is a "new drug" within the meaning of section 201(p) of the Federal Food, Drug, and Cosmetic Act (21 U.S.C. 321(p)) in that it is not generally recognized, among experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs, as safe and effective for use in the cure, mitigation, treatment, or prevention of cancer in man ("the new drug issue").

2. Whether, if Lactrile is a "new drug" within the meaning of the act, it is exempt from the premarket approval requirements of section 505 of the act (21

U.S.C. 355) in that:

(a) At any time before June 25, 1938, it was subject to the Food and Drugs Act of 1906, as amended, and at such time its labeling contained the same representations concerning the conditions of its use as its present labeling ("the 1938 grandfather issue," 21 U.S.C. 321 (p)(1)); or

(b) It meets each of the following conditions: (1) On October 9, 1962, it was commercially used or sold in the United States: (2) On October 9, 1962, it was generally recognized, among experts qualified by scientific training and experience to evaluate the safety of drugs, as safe for use in the cure, mitigation, treatment, or prevention of cancer in man; (3) On October 9, 1962, it was not covered by an effective new drug application (NDA) under section 505 of the act (21 U.S.C. 355); (4) It is currently intended solely for use under conditions prescribed, recommended, or suggested in its labeling on October 9, 1962 ("the 1962 grandfather issue," Pub. L. 87-781, section 107(c)(4)).

The February 18, 1977, notice included detailed information regarding the submission of testimony. In response to the notice, over 400 submissions totaling more than 5,500 pages were received. These submissions, representing the views of both proponents and opponents of Laetrile, came from cancer patients, consumers, experts in drug testing and cancer therapy, physicians, State governments, universities, hospitals, and interested organizations.

The District Court, in directing FDA to develop an administrative record, suggested that the agency invite the following individuals to participate in the administrative proceeding: Dr. Dean Burk, Ernst Krebs, Jr., Mike Culbert, Edward Griffin, and Mike Spencer, Rutherford v. United States, supra, 424 F. Supp. at 108. The three individuals whose addresses could be obtained were specifically invited to give their views (R 3, 4, 5), (Kenneth Coe, attorney for plaintiff Glen Rutherford, who had proposed that FDA be required to invite the five named individuals, could not provide the ad-dresses for Griffin and Spencer and agreed that invitations to the three individuals whose addresses he could supply would suffice.) Mr. Griffin did receive notice of the proceeding and he participated (see R 404). Written submissions were received from Dr. Burk (R 302) and Mr. Griffin (Tr. Ex. 1) and from plaintiff Glen Rutherford (R 258).

The Bureau of Drugs, FDA, presented evidence probative of the new drug and grandfather status of Laetrile. For the purposes of the administrative proceeding, separation of functions requirements were observed between the Commissioner and persons advising him and the Bureau of Drugs and persons advising it (see

21 CFR 10.55).

The February 18, 1977 notice stated that oral argument would be held in Kansas City on May 2. A subsequent notice published in the FEDERAL REGISTER of March 25, 1977 (42 FR 16191) set forth the exact time and place: beginning at 9 a.m. on May 2 at the Radisson Muehlebach Hotel, Kansas City, MO. Dr. John Jennings, Associate Commissioner for Medical Affairs, presided over the oral argument. Approximately 40 persons filed written requests to make oral presentations; others took advantage of the opportunity to speak without the filing of such a request as time allowed. Every person who wished to participate in, and who was present at, the oral argument was given a opportunity to express his or her views. In all, 47 persons made presentations. The transcript of the oral argument has been made a part of the record of the administrative proceeding.

Individuals named by the District Court were again notified of the exact time and place of the argument (R 253-55, see also R 247). Oral presentations were made by Edward Griffin (Tr. at 11), Michael L. Culbert (Tr. at 35), Ernst T. Krebs, Jr. (Tr. at 228) and Dr. Dean Burk (Tr. at 401), In addition, plaintiff Glen L. Rutherford and his attorney, Kenneth Coe, Esq., spoke at oral argu-

ment (Tr. at 297, 442).

Written submissions presented at the time of oral argument were made part of the record and considered despite the fact that they were received at a date later than the one set forth in the February 18, 1977, notice. The record of this proceeding was, however, closed at the conclusion of the oral argument. Submissions received thereafter have been docketed with the FDA Hearing

Submissions to the record are referred to by the number assigned to them upon filing by the Hearing Clerk. When exhibits or attachments accompany a submission, they follow the record number of that submission, e.g., "R 12, Ex. A". References to the transcript of the oral argument are cited as "Tr. at" with the applicable page number supplied. Written submissions presented to the energy at the time of oral argument are referred to as transcript exhibits, e.g., Tr.

Clerk but have not been considered as part of the record. The Commissioner's opinion is based entirely upon the administrative record and does not reflect information brought to FDA's attention subsequent to the closing of that record.

No legal memoranda were solicited by the Commissioner in this proceeding. One such memorandum was submitted by the American Cancer Society and has been made a part of this docket in the

Hearing Clerk's office.

In the Commissioner's opinion, the use of Laetrile in the United States has become a genuine public health problem. Increasingly, doctors dealing with cancer patients are finding that the patients are coming to legitimate therapy too late, having delayed while trying Laetrile. It seems clear that another substantial group of persons afflicted with cancer is avoiding effective therapy altogether and using Laetrile instead. The question has become one of life and death for these patients and for others who may be convinced to use Laetrile in the future. For this reason the Commissioner has con-sidered not only the evidence in the record addressed to the specific legal issues remanded to FDA by the courts, but also the great amount of evidence submitted by both proponents and opponents of Lastrile regarding other issues of importance to the controversy over the use of the drug. Since the Commissioner's discussion of these issues is necessarily detailed, he is setting forth, for the reader's convenience, an outline of that discussion as follows:

#### I. LARTRILE

#### A. DEFINITION OF CANCER

- B. COMPOSITION AND IDENTITY OF "LAETRILE"

- What is Amygdalin? What is Laetrile (with a capital L)?
- What is lastrile (with a small 1)?
- 5. What is Sarcarcinase?

#### C. CLAIMS FOR LAUTRILE

- Treatment (Cure or Mitigation of Cancer.
- Analgesic (Pain Killer).
- Prevention of Cancer.
  Facilitation of Other Cancer Therapy,
- Hemoglobin Index. Reduction of Odor Associated with Malig-
- Sickle Cell Anemia.
- 8. Parasitic Diseases.
  9. Regulating Intestinal Flora.
  10. Hypotensive Effect.
- - D. THEORIES OF LABTRILE'S ACTION
    - II. THE "NEW DRUG" ISSUE
  - A. GENERAL RECOGNITION OF EFFECTIVENESS
- 1. Objective Evidence of Effectiveness.
  - (a) What are the Required Studies? (b) The Need for Controlled Studies.
  - (c) The "Evidence" of Lactrile's Effectiveness
  - Case Reports.
     Animal Testing of Laetrile.
- 2. Testimony of Experts.
  - (a) Experts Opposed to Lactrile.
  - (b) Supporters of Lastrile.
  - B. GENERAL RECOGNITION OF SAFETY
- 1. Lack of Adequate Testing.
- 2. Testimony of Experts.

III. THE "GRANDFATHER" ISSUE

A. THE 1938 GRANDFATHER CLAUSE

B. THE 1962 GRANDFATHER CLAUSE

- 1. Composition.
- 2. Investigational Use.
- 3. Conditions of Use in Labeling.
- 4. Lack of General Recognition of Safety in 1062

(a) The Prerequisites.

(i) Lack of General Knowledge of Use.

(ii) Lack of General Knowledge of Formulation.

(iii) Lack of General Knowledge of Conditions of Use Suggested.

(iv) Lack of Safety Data in Scientific Litecature.

(v) Lack of Showing of Safety by Adequate Testing.

(b) Statements by Experts.
(c) Lack of General Recognition of Effectiveness in 1962.

IV. THE POPULARITY OF LABIRILE

- A. LAETRILE AND OTHER UNPROVEN REMEDIES
- 1. The History of Cancer Quackery in the
- United States Similarities Between Laetrile Promotion and That of Other Recent "Unproven" Cancer Remedies.

(a) Mantle of Science.

- (b) Attacks on the "Establishment."
- Claimed Parallel with Scientific Pioneers.
- Reliance on Testimonials. Lack of Scientific Publication. (d) (0)
- Nonexpert Supporters.

Simplistic Theories of Causation (g) and Reliance on Diet.

- A Painless Cure.
- (1) Only Proponents can Effectively Use the Drug.
- B. WHY DO PEOPLE USE LAETRILE?
- 1. The Emotional Reaction to Discovery of
- 2. The Role of Loved Ones.
- 3. Methods of Promotion of Lactrile.
- 4. The Sampson Survey.

C. THE LAETRILE TESTIMONIALS

- V. OTHER ISSUES REGARDING LARTRILE
- A. USE OF LARTRILE OUTSIDE THE UNITED STATES
- B. CLAIMS THAT LAETRILE IS A VITAMIN OR FOOD
- A Vitamin or Food May Be a Drug As Well.
- Is Laetrile a Vitamin?
  - (a) Proponents' Claims.(b) Vitamin Experts' Position.
- 3. Dangers of Ingestion of "Vitamin B-17."
- - C. FREEDOM OF CHOICE
- 1. Balancing Freedoms.
- 2. The Choice Is Not Free.

D. ALLEGATIONS OF BIAS

- E. LIMITING USE TO TERMINAL PATIENTS
- Who Is Terminal.
   Effects on Other Patients.
  - F. USE CONCURRENTLY WITH OTHER THERAPY

VI. CONCLUSIONS

#### I. LARTRILE

# A. DEFINITION OF CANCER

Laetrile has been, over the years, recommended for use in the treatment of cancer. An understanding of the issues concerning the drug requires that the term "cancer" be defined. Cancer has been stated to include "\* \* \* all malignant neoplasms regardless of the tissue of origin including malignant lymphoma, Hodgkins disease, and leukemia" (R 173, Att., "Laws and Regulations Relating to the Diagnosis and Treatment of Cancer." State of California, Division 2, Chapter 7, Section 1705). For an almost identical definition by the American Cancer Society, Inc., see R 173, Att., "American Cancer Society, Inc., Medical Affairs Department, (State) Cancer Remedy Act." A neoplasm is a rew and abnormal growth such as a tumor. Malignant neoplasms are "neoplasms that are characterized by unregulated, uncontrolled, and unrestrained growth and proliferation" (R 190 at 14; R 194 at 3; R 195 at ( 11). It has been said that the "distinguishing feature of cancer is its ability to invade, erode and to metastasize to more or less distant parts" (R 183, Att. 7 at 15), Metastasis, in relation to cancer, means the transfer or spread of the cancer from one site to another, usually through the blood stream or the lymphatic system. In this process, cells may travel throughout the body; when they finally lodge they begin to grow as a new cancer.

There are more than 100 different entitles involved in the disease known as "cancer." These many forms of clinical cancer differ materially in terms of the factors which cause them as well as in terms of populations they affect, their prognosis, and the ease with which they may be treated (Tr. at 144; cf. R 190 at 14; R 186 at 7-9). The cause of cancer is not a question addressed in this proceeding. It should be noted, however, that the record includes indications that various cancers are associated with chronic irritation (e.g., a high frequency of cancer of the groin in textile workers whose jobs required straddling a metal shaft) (R 318 at 37), with cancer-causing chemical substances called carcinogens (e.g., a high frequency of lung cancer in chimney sweepers and coal miners probably cauced by inhaling dust particles) (id.), with irradiation (id. at 38), with virus (Tr. at 223; R 318 at 38), and with hereditary effects (R 318 at 38).

Two novel theories, neither of which has gained acceptance by the scientific community, have been at various times espoused by Laetrile's proponents to explain how cancer is caused. The first of these theories is said to have been first developed by Professor John Beard of Scotland. In 1902, Professor Beard announced his "findings" that the cancer cell and the "trophoblast" cell were one and the same (R 318 at 60). Trophoblast cells are present during pregnancy and they prepare a niche in the uterine wall where the fertilized egg can nestle (R 318 at 56). According to Beard, they share several characteristics with cancer cells: Both are invasive, erosive, corrosive and can be carried through the blood stream to other parts of the body (R 318 at 57). Beard believed that trophoblast cells could be expected to develop at various places in the body from precursor cells distributed throughout the body during the embryonic stage. If the pancreas gland were functioning properly it

would, in his theory, produce enzymes which destroy these trophoblast cells. If such enzymes were not produced, cancer would occur (R 318 at 58-59)

The proponents of Laetrile have more recently taken the position that cancer is a metabolic deficiency or dietary disease (cf. R 302, Ex. H at 76-77). It is claimed that cancer is a "\* \* \* systemic, chronic, metabolic deficiency dis-ease" (Tr. at 348) or " \* a chronic or metabolic disease that is not caused by some mysterious virus" (Tr. at 234). The "nutritional deficiency" theory is refuted by at least one expert (Tr. at 223).

#### B. COMPOSITION AND IDENTITY OF "LAETRILE"

Several terms, such as "Laetrile," "laetrile(s)," "amygdalin," "nitriloside(s)," "vitamin B-17," and "Sarcarcinase" have in many instances been used interchangeably by both proponents and op-ponents of Laetrile. There are, however, distinctions among these terms, which must be understood in order to deal with the issues in this proceeding, The Commissioner will discuss in detail what the record indicates about the meaning of the terms "amygdalin," "Laetrile," 'laetrile" and "Sarcarcinase."

#### 1 Glossary

The following glossary provides a deflnition or description of the above terms and others that will be used in this opinion:

Amygdalin: A specific chemical entity

having the chemical formula:

D-mandelonitrile-bets-D-glucueldo-6-hets-D-glucuride (Merck Index, 9th Md. at 81).

beta-Cyanogenic glucosides; beta-cyanophoric glucosides; beta-cyanogenetic glucosides: Terms used interchangeably in the record. In general, they are used in the record to include compounds which can break down to yield cyanide and glucose. The terms are used to refer to amygdalin, a glucoside present in kernals or seeds of practically all fruits (see, e.g., R 302, Ex. H at 75; R 173, Att., California Administrative Code, Section 10400.1 at 16; Tr. at 272), beta-Cyanogenic glucosides belong to the large class of chemicals known as beta-cyanogenic glycosides. (See definition of glucosides and glycosides below.)

beta-Głucosidase: An enzyme present in plants that participates in the metabolism of glucosides. The enzyme has been identified in apricot and peach kernels and catalyzes the breakdown of amygdalin to free two molecules of glucose and a molecule of mandelonitrile. The enzyme is found only in trace amounts in animal tissues (R 173, Att., "The Vitamin Fraud in Cancer Quackery" (hereinafter "Vitamin Fraud") at 345).

beta-Glucuronidase: An enzyme present in animal tissues that participates in the metabolism of glucuronic acid derivatives (also called glucuronosides). The enzyme reportedly catalyzes the breakdown of "Laetrile" (1mandelonitrile-beta-glucuronic acid) to free glucuronic acid and mandelonitrile.

Enzymes: Chemical compounds (all of them proteins) produced by living or-ganisms that serve as catalysts in metabolic reactions. The suffix "-ase" is given to most enzymes.

Glucoside: A term applied to any glycoside having glucose as its sugar con-

stituent.

Glucuronide; glucuronoside; Terms used in the record to refer to chemical derivatives of glucuronic acid. As an example, Laetrile is identified as 1mandelonitrile-beta-glucuronic acid" in the Merck Index 9th Ed., at 702 and as "laevo-mandelonitrile-betaglucuronoside" in the book Control for Cancer (R 318 at 73).

Glycosides: A broad term which encompasses glucosides. Not all glycosides are glucosides (cf. The Condensed Chemical Dictionary, 7th Ed. at 455). Lastrile: A specific chemical entity having the chemical formula:

I-mandelomitrile-beta-glucurnels anid (Merck Index, Sth

Id. at 702). The same "Lastrila" was perpertuity assigned to this compound by Ernst T. Ereby, Jr. (8 318 at 73).

lactrile: A term used interchangeably with "Lactrile," "amygdalin," "nitriloside," and "vitamin B-17" (R 302, Ex. A; R 183, Att. 10c). The term is also used to include a number of compounds, in which case it may appear as "laetriles."

Mandelonitrile: A specific chemical entity having the chemical formula:

(Merck Index, 9th Ed. at 743).

Nitriloside: A term proposed by Ernst T. Krebs, Jr., for all cyanophorioglycosides of dietary significance (R 302, Ex. H. at 75).

Prunasin: A specific chemical entity having the chemical formula:

Mandatonivella Cloronide (Marck Index, 9th Ed. at 743),

Sarcarcinase: The name given to an enzyme preparation developed by Dr. E. T. Krebs, Sr., and described by him in a 1933 patent application as a mixture of the following enzymes-amygdalase, prunase, oxynitrilase, catalase, peroxydase, and a proteolytic enzyme. He also suggested the presence of isomaltase and a lipase and perhaps other enzymes (R 424)

Vitamin B-17: Described as a group of compounds which include water-soluble, essentially nontoxic, sugary compounds found in over "800 plants" (R 302, Ex. H at 75) used interchangeably with "nitriloside," "Laetrile," "amygdalin," "beta-cyanogenic glucosides" and "cynophoric glucosides" (e.g., R 302, Ex. H at 75; R 302, Ex. A).

# 2. What is Amygdalin?

Amygdalin was reportedly first isolated from bitter almonds by the French chemists Robiquet and Burton-Charland in 1830 (R 173, Att., "Vitamin Fraud" at 345). The name "amygdalin" was derived from the word "amygdala", Greek for almond (R 302, Ex. L at ¶ 9). Amygdalin is a chemical compound composed of two glucose molecules and one molecule of mandelonitrile. Mandelonitrile is a chemical in which cyanide is combined with benzaldehyde (cf. R 173, Att., "Vitamin Fraud" at 345). The German chemists Liebig and Wohler observed that an enzyme preparation (later called emulsin) from bitter almonds was capable of hydrolyzing amygdalin, i.e., breaking it down into the two glucose molecules, the benzaldehyde molecule and a hydrogen cyanide molecule (id.). It was later shown that this hydrolysis occurs through the action of two enzymes (beta-D-glucosidase and betaoxynitrilase) which are present in emulsin (id.). The beta-D-glucosidase hydrolyzes the beta-D-glucoside bond and thus frees the two glucose molecules from the mandelonitrile. beta-Oxynitrilase is the catalyst for the breakdown of mandelonitrile into benzaldehyde and hydrogen cyanide (id.).

Amygdalin may be extracted from apricot kernels (id.), and is present in seeds of other members of the rose family (R 416 at § 31A). The Commissioner concludes that amygdalin, a cyanogenic glucoside, is a chemical having the chemical name D-mandelonitrile-beta-D-glucosido-6-beta-D-glucoside (Merck Index. 9th Ed. at 81, compounds 630; R 183, Att. 10b). The chemical structure of amygdalin is:

# 3. What is Laetril (With a Capital L)?

The term "Laetrile" has been used interchangeably with "amygdalin," "laetrile," "vitamin B-17," "nitrilosides," and "beta-cyanogenetic glucosides" (R 173, Att., Laws and Regulations Relating to the Diagnosis and Treatment of Cancer 10400.1 at 16; R 302, Ex. A; Tr. at 405, 272, 465). It appears, however, that the term "Laetrile" (with a capital L) has been used by the drug's proponents to refer to a particular substance. There are essentially two versions of what that substance is:

(a) The term has been used to refer to a specific chemical compound which was prepared in 1952 by Ernst T. Krebs, Jr., who is said to have derived the name "Laetrile" from the compound's chemical name: laevo-mandelonitrile-betaglucuronoside (R 318 at 73; see also R 262; R 183, Att. 16 at 1 and 2). This chemical is related to, but is distinctly different from, amygdalin. It is claimed that the laveo-mandelonitrile-beta-glucuronoside was derived by Ernst T. Krebs, Jr., while working with the apricot extract his father had prepared and studied some 20 years earlier (R 318 at 70-73). The chemical structure of this material is depicted in several places (R 318 at 154, 157, 162) and is in agreement with the chemical name. The chemical structure of this version of Laetrile is:

(b) In a 1965 affidavit, however, Dr. Krebs, Sr., stated that the name "Laetrile" was devised in 1949 by his son, Ernst T. Krebs, Jr., for a form of amygdalin which Dr. Krebs, Sr., was producing at that time (R 183, Att. 13). As can be seen from the diagrams set out in this opinion, the chemical structure of amygdalin is different from that of Laetrile as described by Mr. Krebs, Jr.

In the second version of the identity of Laetrile, the source of the name is stated as follows: "Because this apricot preparation was 'Laevorotatory' (lefthanded) to polarized light, and because Amygdalin was chemically a 'mandelonitrile,' Krebs, Jr., united the first and the last syllables to invent a name for the

new cancericidal drug-LAETRILE" (R 183, Att. 7 at 24).

The confusion about the meaning of the term "Laetrile" is long-standing and may in part be the result of a desire on the part of promoters to continue to use drugs containing amygdalin while justifying the use of the drugs by theories associated with the Laetrile of Krebs, Jr. For example, in a February 17, 1953 letter to Dr. Ian Macdonald, the Chairman of the Cancer Commission of the California Medical Association, Ernst T. Krebs, Jr., advised that he was forwarding "\* \* \* samples of the biosynthetically degraded amygdalin in which one dextrose was removed by prunasin and the resulting compound, in the presence of platinum black, was oxidized to the corresponding glucuronoside" (R 183, Att. 14), (Note: the compound thus obtained should have been 1-mandelonitrile-beta-glucuronoside or "Laetrile" as described by E. T. Krebs, Jr., R 318 at 73.) The Cancer Commission of the California Medical Association, in its 1953 report, stated: "Chemical analyses done independently for the Commission have identified in the product distributed as Laetrile only the presence of a natural laetrile termed amygdalin" (R 378, Att. 15 at 326).

The question of the identity of the material distributed as "Laetrile" arose again in the early 1960's when the Cancer Advisory Council of the State of California was gathering information on Laetrile in order to enforce the 1959 California law dealing with cancer quackery. In 1963 the Council reported that the California State Department of Public Health had examined different varieties of Laetrile and found that the products were markedly different in composition but did contain varying percentages of amygdalin (R 183, Att. 16, App. 7 and 8). A Canadian Medical Association report published in 1965 found that the United States and Canadian versions of the drug were different-a larger percentage of the Canadian version than of the United States version was found to be made up of amygdalin (R 189; see also R 378, Att., "Supplementary Report by the Cancer Advisory

Council" at 1-2). The record reveals a number of references by Laetrile proponents which use the terms "Laetrile" and "amygdalin" interchangeably (see, e.g., Tr. at 238 and 246, and the book Control for Cancer (R 318)). Even some labels for the drug use the terms synonymously: one identifies the product as "Laetrile (Amygda-lin) 400 mg capsules" (R 183, Att. 10a; see also R 183, Att. 10d). The National Cancer Institute, in its October 1975 "Background Statement on Laetrile" notes the fact that supporters of the drug have used the names "Laetrile" and "amygdalin" interchangeably. The report then correctly identifies amygdalin as mandelonitrile - beta - gentiobioside (this chemical name for amygdalin is listed in the Merck Index, 9th Ed. at 81, compound No. 630) and states that the compound actually tested in 1957, 1960. 1969, 1973, and 1975 by the National Cancer Institute was amygdalin (R 173, Att., 'NCI Testing of Laetrile (Amygdalin)")

Yet it is not possible simply to conclude that the many references to Laetrile as a specific substance are a hoax on the part of Mr. Krebs, Jr., and that Lactrile as used is simply amygdalin. A number of reports by Dr. Manuel Navarro of the Philippines stated that the Laetrile of Ernst T. Krebs, Jr., was in use (R 318 at 155, 161)), as did a report by Dr. John A. Morrone of New Jersey (R 318 at 205).

Additional confusion is added to the record by an article "Nitrilosides (Laetriles) Their Rationale and Clinical Utilization in Human Cancer (December 1962)," by Ernst T. Krebs, Jr., and Dr. N. R. Bouziane, in which the authors report on their use of "non-toxic nitrilosides (Laetrile), to which the trophoblast is susceptible, on terminal cancer cases for two years in Canada \* \* " (R 318 at 187). While the article refers to "cyanophoric glucosides and cyanophoric glucuronosides" (id. at 189), the authors do not identify the "Laetrile" they had been using and about which they were reporting. The authors cite a chemical compound in their discussion (id. at 190) that is not amygdalin nor is it "Laetrile" as described and named by Ernst T.

Krebs, Jr. (id. at 73)

While the prevailing confusion over the true identity of material called "Laetrile" would be a severe drawback to anyone seeking to show through testing that Laetrile was safe and effective, this lack of uniformity has been adopted by Laetrile proponents as a means of discounting data showing the drug to be ineffective and thus unsafe. An example is found in the statement by Ernst T. Krebs, Jr., that "The \* \* \* single negative report on Laetrile, which is based upon the observations of unidentified investigators in unidentified institutions administering a purported Laetrile not obtained from the only source of the material, is to be found in California Medi-cine, 78:320 (1953)" (emphasis added) (R 318 at 251). It should be recalled that at least some of the material supplied to the California Cancer Commission was sent by Ernest T. Krebs, Jr., himself (R 183, Att. 14). In May of 1971, Mr. Mc-Naughton of the pro-Laetrile McNaughton Foundation, in a meeting with FDA's Ad Hoc Committee of Oncology Experts, stated that data obtained prior to 1968 are frequently not valid because of the variability of Laetrile formulations (R. 173, Att. "Report of the Ad Hoc Committee of Oncology Consultants" at 1). See also statement of Robert Bradford, President of the Committee for Freedom of Choice in Cancer Therapy, Inc.: "As an aside, an important aspect of animal tests, and indeed, of human tests has been from time to time the availability of amygdalin which did not meet the specified identification criteria, that is, for its use. Tests with defective materials, as Sloan-Kettering found out, will not be efficacious. Defective material likewise will not be effective in humans' (Bradford, Tr. at 350).

In its 1971 report to FDA, the Ad Hoc Committee of Oncology Consultants agreed that uncertainty about the identity of the drug tested makes the test results obtained questionable. The Committee stated that because of the variability in composition of early preparations, doubt was cast on the bulk of the 1970 McNaughton Foundation Notice of Claimed Investigational Exemption for a New Drug (IND) for Laetrile, which was based almost exclusively on such early material (R 173, "Report of the Ad Hoc Committee of Oncology Consultants" at 1). The Committee further suggested that any protocol for study contain a full description of the drug (formulation, stability etc.) (id. at 4).

There is, quite simply, no one answer to the question "What is Laetrile?". In the glossary to this opinion, the chemical composition of Laetrile is considered to be that described by Ernest T. Krebs, Jr. Yet if some other substance is being used to treat cancer patients, testing of that "Laetrile" would be of no relevance.

Because different persons have used the terms "Laetrile" and "amygdalin" to mean different substances, uniformity of definition will not be possible in discussing the evidence in the record. For this reason, the Commissioner will not, in quoting or citing parts of the administrative record, attempt to define or to determine the identity of the material under discussion but will simply use the term as it appears in that portion of the record. Attempts to identify the material referred to will be made only when necessary for a rational resolution of an issue, e.g., a reference to Sarcarcinase as amygdalin or as Laetrile will not be accepted blindly.

# 4. What is lastrile (With a Small I)?

As noted in the glossary, the term "laetrile" (with a small 1) has been used interchangeably with or synonymously for: nitriloside, Laetrile, vitamin B-17, and amygdalin (e.g., R 302, Ex. A; R, 183, Att. 10c; R 378, Att. 6). The term has, however, also been used to describe a class or group of compounds. For example, amygdalin and prunasin are described as "two common Laetriles" (R 173, Att., "Vitamin Fraud" at 345). It has been stated that: "The term LAE-TRILE is used to designate the laevorotatory containing glucosides in general and the corresponding glucuronoside in particular. The former are found in plants whereas the latter are synthetic" (italics in original) (R 318 at 155). (See also, R 318, at 240 ¶ 9 which defines natural laetriles as beta-cyanogenetic glucosides and glucuronosides.) The Commissioner concludes that the term "laeis an imprecise term and that it does not imply a specific chemical compound. The term is, rather, a broad or generic term for a group of compounds of unknown number.

# 5. What is Sarcarcinase?

Dr. Ernst T. Krebs, Sr., claims to have developed a product called "Sarcarcinase" in 1926 (R. 183, Att. 13). "Sarcarcinase" is stated to be a registered trademark in the United States and 10 other countries (with registrations dating from

March 1933 to January 1934) (see R. 260; R 259). The process for preparing the product is stated to be patented in 15 countries including the United States (see R. 260). It is also reported that Sarcarcinase was used in Japan in 1934 and in Czechoslovakia in 1935 (R. 259). Other references in the same timespan refer to an "enzyme preparation" or "enzyme injection" used within the United States as well as several foreign countries (id.).

Ernst T. Krebs, Jr., stated at oral argument that as early as 1932 his father "\* \* observed the use of amygdalin, or laetrile; made it available across the country and abroad under the term 'Sarcarcinase' to physicians, to researchers" (Tr. at 238). He also stated, \* the first amygdalin was used-1932-it was labeled as 'Sarcarcinase' " (Tr. at 246). Sarcarcinase was not, however, amygdalin nor did it contain amygdalin in any quantity. Rather, in the words of Dr. Krebs, Sr.'s 1933 patent application, Sarcarcinase was "an enzyme for treatment of malignant growths." The patent application actually describes the product as "an enzyme complex" containing "amygdalase, prunase, oxynitrilase, catalase, peroxydase and proteolytic enzyme" plus a suggestion of "isomaltase and a lipase with possibly others" (Patent application attached to R 424 and to R 259).

Amygdalin is not an enzyme. As will be discussed in more detail below, enzymes are chemicals which catalyze the breakdown of other chemicals and which are often named by attaching the ending "-ase" to the chemical which they attack. Thus, "amygdalase," stated by Dr. Krebs, Sr., to be part of his enzyme complex, may have been meant to describe an enzyme which would break down amygdalin.

It has been argued that Sarcarcinase contained some quantity of amygdalin (R 183, Att. 13, and Att. 7 at 23), An expert chemist has stated, however, that much or all of the small amounts of amygdalin in the apricot kernels used in making Sarcarcinase would be destroyed by enzyme action when the kernels are ground up and that only a small fraction of any that remained would survive the rest of the process (R 424). It should be noted that, even if there were any amygdalin in Sarcarcinase, that would not make that drug equivalent to a drug made up of amygdalin either totally or in greater proportions either in a scientific sense or in a legal sense.

There is some indication that Dr. Krebs, Sr., had abandoned Sarcarcinase even at the time when the patent applications were being obtained. One submission by a Laetrile proponent states that Dr. Krebs, Sr., \* \* resigned himself to the fact that there was no sense continuing this particular research to identify the toxic element or elements in the apricot extract he prepared (sometime after 1926) until he acquired the additional knowledge necessary to understand the mysteries that were occurring in his test tubes. He put his extract aside and returned his books"

(R 318 at 42). Krebs himself states that in 1936 he developed a new product. whose "active principle" was amygdalin of 66 percent purity (R 183, Att. 13). (The inactive ingredients of this preparation are not identified.) It is not clear whether it is Sarcarcinase or this new product about which Krebs, Sr., speaks when he states that his "apricot extract" was "so toxic that he and colleagues who were experimenting with him were reluctant to continue its use except in dire circumstances" (R 183, Att. 7 at 23). It was apparently these toxicity problems that led Krebs, Jr., to seek to improve his father's work (id.). Since "Laetrile" was not developed until 1952 by Krebs, Jr., any statement that "Laetrile" was sold as Sarcarcinase in the 1930's is patently erroneous.

#### C. CLAIMS FOR LAETRILE

Laetrile (or amygdalin) has been recommended over the years primarily for use in the treatment and, more recently, "control" of cancer. The claims appear to vary in relation to the sophistication of the intended audience. Thus in the 1962 new drug application (NDA) for Laetrile submitted by Ernst T. Krebs, Jr., to FDA, the drug was claimed to be a palliative (i.e., a drug that mitigates the symptoms of a disease without curing it) to be used with other recognized therapies (R 201, Ex. B at 102). By contrast, in a pamphlet in use in 1965, apparently addressed in part to prospective patients, it is stated that "Laetrile does not palliate, it acts chemically to kill the cancer cell selectively \* \* \*," and use of other cancer therapies concurrently is discouraged (R 201, Ex. C. # III). The following claims have been made for Laetrile (amygdalin):

#### Treatment (Cure or Mitigation) of Cancer

The pamphlet discussed above and others obtained from Dr. Krebs, Sr., by FDA investigators at the same time recommended Laetrile for treatment of cancer (see, generally, R. 201, Ex. C). Dr. Krebs, Sr., in a pamphlet entitled "The Treatment of Breast Cancer with Laetrile by Iontophoresis" promotes the drug for treatment of a number of cancers (R 183, Att. 7 at 26-27 and 30).

A label for 400 mg capsules of "Laetrile (Amygdalin)" claims that the "non-toxic cyanide glucoside is used for specific treatment of cancer by physicians or under directions of a physician" (R 173, Att. 102). Amygdalin has been promoted (as an ingredient of "Bitter Food Tablets") for the cure, mitigation, and treatment of cancer in man (R 173, Att., United States v. Spectro Foods, Civ. No. 76-101 (D.N.J., Jan. 28, 1976) Findings 16 through 23).

As noted above, some claims are limited to palliation. (See, e.g., R 216 at 348; R 318 at 175; Tr. at 238.) Recently, Lactrile has been touted as a "control" for cancer. Proponents of Lactrile making this claim assert that no "cure" for cancer exists (see Tr. at 303). Control for Cancer is also the title of a paperback book on Lactrile (R 318). It is not entirely clear

from the record whether "control" means palliation. Laetrile therapy is said to be responsible for increased appetite and weight gain and an increased "sense of well-being" among treated cancer patients (R 318 at 158 and 165).

# 2. Analgesic (Pain Killer)

An information booklet for physicians about amygdalin makes the claim that the product is a nontoxic analgesic that is highly effective in relieving the pain of terminal cancer (R 183, Att. 10b). The booklet claims that the oral route is the most convenient route of administration for both patient and physician.

Note.—The Commissioner points out that both proponents and opponents have warned against oral use of amygdalin or Laetrile.

See R 318 at 167: "\* \* it (Laetrile) should never be given by mouth because the HCl (of the stomach) is capable of hydrolyzing the Laetrile"; see also R 318 at 158. Compare R 173, Att., Interview with Robert C. Eyerly, American Cancer Society; "Taken orally, it (Amygdalin) is decomposed in the intestinal tract by beta-glucosidase into highly lethal hydrogen cyanide." "Orally it (Laetrile) is extremely toxic due to the release of hydrogen cyanide on contact with the hydrochloric acid of the gastric juice (R 318 at 205).

A label for "The Original Laetrile" claims that the product "relieves pain due to malignancy" (R 183, Att. 9). For another claim that Laetrile reduces cancer-connected pain, see Tr. at 44. It has also been asserted that the hydrogen cyanide and benzaldehyde liberated by hydrolysis of Laetrile are potent analgesics (R 318 at 164).

# 3. Prevention of Cancer

With the advent of their theory that cancer is a deficiency disease and that that deficiency can be overcome by their product, either characterized as a provitamin for vitamin B-12 (R 201, Ex. C, No. IV), or as new vitamin B-17 (see R 183, Att. 10c), proponents of Lactrile have promoted it as a preventative for cancer (see the above references and R 198, Ex. 2 (transcript of the film World Without Cancer); cf. Tr. at 465). (See also R 173, Att., United States v. Spectro Foods, supra, Findings 16 through 23.) While proving that Laetrile (or amygdalin) did not prevent cancer would be extremely difficult, the record does contain evidence that at least one person taking it as a preventative did contract cancer (Tr. at 120).

#### 4. Facilitation of Other Cancer Therapy

While, as noted above, some labeling recommends against use of other cancer therapies with Laetrile, it has been stated that "\* \* if you combine toxic chemotherapy with Laetrile, you can give very high doses of toxic chemotherapy with no side effects, physical and no effects on the blood. That is, you don't get neucophenia (leukopenia?) and you don't get chromositophenia (chromocytompenia?)" (Tr. at 480).

#### 5. Hemoglobin Index

One set of labeling for "Laetrile (Amygdalin)," which appears at two points in the record, recommends the product "for raising hemoglobin index and red count \* \* \*" (R 183, Att. 9; R 201, Ex. C. No. I).

# 6. Reduction of Odor Associated with

It is also claimed that topical application of Laetrile relieves fetor (odor) resulting from the secondary infection of ulcerated carcinoma and that parenteral administration takes care of fetor associated with internal cancers. This action is ascribed to the "antiseptic" properties of HCN and benzaldehyde, which is converted by the cells to benzoic acid (R 318 at 158 and 165).

#### 7. Sickle Cell Anemia

It is theorized that nitriloside (Laetrile) might be of value in the treatment of sickle cell anemia because of the release of cyanide and the subsequent formation of thiocyanates (R 217, article by R. G. Houston). (See also Tr. at 465.) This claim is reportedly refuted by experts in sickle cell hemoglobin (R 416 at 23).

#### 8. Parasitic Diseases

The Houston article also references a report by Navarro and others of the clinical control of schistosomiasis (a snall-borne infection) with nitriloside (Laetrile) (R 217, Houston article at 58). The possibility of using Laetrile to treat parasitic diseases such as schistosomiasis or malaria is discussed in the book Control for Cancer but there are no reports of actual use in the book (R 318 at 111-12).

#### 9. Regulating Intestinal Flora

It has also been suggested that amygdalin has some utility in regulating intestinal flora (Tr. at 476).

# 10. Hypotensive Effect

It has also been claimed that use of Laetrile causes a hypotensive effect (i.e., it reduces blood pressure), at least in cancer patients (R 318 at 165; cf. Tr. at 465).

In addition to these claims by Laetrile proponents (developers, distributors, and promoters), numerous comments from interested citizens contained references to or claims for its therapeutic effects as a cancer cure or as a preventative. There are also references to relief, attributed to Laetrile, from other ailments unrelated to cancer, e.g., arthritis (R 391).

#### D. THEORIES OF LAETRILE'S ACTION

A thorough understanding of the manner in which a compound achieves its therapeutic or beneficial effects is highly desirable. A cancer drug which had been shown to be safe and effective would not, however, be denied marketing approval simply because its action could not be explained. Experience has shown that a good theory to explain or predict the action of a chemical in the body.

does not assure success; neither does a weak theory, or even what turns out to be a totally incorrect theory, mean certain fallure.

Some cancer patients may be turning to Laetrile in the mistaken belief that its use is supported by a respectable -even if not widely accepted-scientific theory. (Unproven remedies throughout the years have benefited from the use of the type of "scientific" theories associated with Laetrile (see, generally, R 413).) The Commissioner finds from the record that the theories advanced for Laetrile's supposed action are based on false or questionable assumptions. An understanding of these theories, furthermore, points up important differences between the "Laetrile" whose use is "justified" by the theories of Krebs, Jr., and the amygdalin-containing products actually being used.

Since a large part of the Lactrile theory of action deals with enzymes, the Commissioner believes that a few brief introductory comments about enzymes would be useful. Enzymes are protein molecules manufactured in the cells of the body which help the cells perform chemical reactions involving other compounds. As an example, trypsin, a common enzyme, aids in the metabolism of proteins in food by breaking these large molecules into smaller, easier-to-handle pieces. Enzymes generally are very specific in the types of chemicals they will attack. Frequently, the name of an enzyme is derived from the compounds that enzyme will break down. The ending "-ase" is often used to indicate an

The chemical "beta-glucosidase" appears frequently in the Lactrile record. The name of this chemical indicates that it is an enzyme (-ase) and, furthermore, the name indicates that it attacks glucose-containing compounds (glucosides) from which it will liberate glucose molecules. As an example, beta-glucosidase liberates two molecules of glucose from amygdalin. If the chemical compound does not contain glucose molecules, beta-glucosidase will not attack it. In a similar vein, beta-glucuronidase will attack chemical compounds that contain glucuronic acid. (These chemical compounds are called "glucuronides" or "glucuronosides" or "glucuronic acid derivatives.")

The original theory of Ernst T. Krebs, Jr., for Laetrile's action involved two enzymes, rhodanese and beta-glucosidase (R 318 at 72). Krebs claimed that normal cells produced these two enzymes, while cancer cells were deficient in rhodanese. In cancerous areas, the theory continues, the beta-glucosidase accumulates in great quantities (R 318 at 72)

According to the theory, when Laetrile comes into contact with the cancerous areas it is hydrolyzed by the enzyme beta-glucosidase to liberate cyanide and benzaldehyde. In normal cells, the enzyme rhodanese converts the liberated cyanide to the less toxic thiocyanate. Cancer cells, lacking rhodanese, are said to be killed by the liberated cyanide

when it reacts with cellular components. Rhodanese from normal cells cannot protect cancer cells because, it is claimed, cancer cells produce chorionic gonadotropic hormone that effectively blocks the action of rhodanese (R 318 at 153). In later versions of the theory, the benzaldehyde is also considered to be responsible for killing the cancer cells, either alone or in concert with the cyanide. According to this theory, benzaldehyde is normally converted by a cell to benzoic acid by oxidation. Cancer cells are said to oxidize the benezaldehyde at a slower rate than normal cells, making it toxic to cancer cells and nontoxic to normal cells. (See Krebs, Jr., "The Nitrilosides (Vitamin B-17) - Their Nature, Occurrence Metabolic Significance (Antineoplastic Vitamin B-17)" (R 183, Att. 10c at 80) )

In fact, it has been reported that only traces of beta-glucosidases have been found in animal tissues and even less in experimental tumors than in such organs as liver and spleen (R 173, Att., "Vitamin Fraud" at 345). Apparently for this reason. Krebs, Jr., at one time modified his theory. In the modified version it is betaglucuronidase rather than beta-glucosidase which is abundant in cancerous areas. This change is reflected in a 1955 pamphlet co-authored by Dr. Krebs, Sr., and Dr. Arthur T. Harris, in which it is stated: "As soon as the Laetrile betaglucuronidase, which bathed the cancer cell, because of its affinity for sugar split the glucoside (or sugar radical) from the Laetrile molecule" (R 183, Att. 7 at 24). (See also R 318 at 151-53.)

The change in theory is important. beta-Glucuronidase hydrodyzes (breaks down) beta-glucuronosides (or "betaglucuronic acids" or "beta-glucuronides") but does not hydrolyze beta-glucosides (R 318, Att. 16 at 24). Thus, beta-glucuronidase will hydrolyze "Laetrile" of the formulation devised by Krebs, Jr. (i.e., laevo-mandelonitrilebeta-glucuronoside), but it will not hydrolyze amygdalin (D-mandelonitrilebeta- D-glucosido-6-beta-D- glucoside) or other "nitrolosides" found in nature. What this means is that amygdalin, which has been sold as "Laetrile," would not be hydrolyzed by the body to liberate cyanide (R 183, Att. 16 at 41).

Recognition of this fact apparently led Krebs, Jr., to formulate his version of Laetrile in the first place. He is reported to have stated in a manuscript that "the natural laetriles have been

\*In his 1933 patent application for Sarcarcinase, Dr. Ernst T. Krebs, Sr., discussed his own theory of cancer, apparently now abandoned by Laetrile proponents. He perceived a malignant protein ("\* \* a so-called abnormal glucosido-protein \* \*") in cancer cells and explained why his enzyme extract, prepared from apricot kernels, should be effective against those cells (R 424). He believed that the enzyme would break up the abnormal gluco-protein and thus be an effective treatment against cancer (R 318 at 40-41). While it is claimed that some positive effects were observed in cancers in mice, the extract proved to be toxic and Dr. Krebs, Sr., discontinued working on the extract (id. at 421).

abandoned for the more specific synthetic laetrile tailored as specific glucuronsidic substrates for the tumor beta-glucuronidase" (R 183, Att. 16 at 14).

The specificity of beta-glucuronidase for glucuronides (or glucuronosides) and its lack of activity against glucsides (such as amygdalin) is rigorously addressed in the record:

Numerous glucuronides are hydrolyzed by beta-glucronidase. Mentyl-alpha-Dglucuronide and alpha-and beta-methyl-Dglucosides are not split by the enzyme.

Further checking of this important point is consistent with the idea that the enzyme in question (beta-glucuronidase) not hydrolyze the beta-glucosides, which are the only Laetriles actually utilized by the Krebs' for human treatment (R 183, Att. 16 at 24).

(It should be remembered that the Cancer Commission of the California Medical Association had determined that the material labeled "Laetrile" was in fact amygdalin—a glucoside and not a glucuronoside (R 183, Att. 15 at 326).)

The conclusion seems justified that the presence of the terminal carboxyl group on position 6 appears to be the important factor in determining a specificity which is markedly different from that of B. Glucosidase \* \* \* (R 183, Att. 16 at 25).

(The Commissioner points out that amygdalin has two glucose molecules but does not have a carboxyl group. Laetrile, as reportedly prepared, described, and named by Ernst T. Krebs, Jr., in the late 1940's or early 1950's (R 318 at 73) does have a carboxyl group on position 6 (the glucuronic acid portion of the molecule).)

Dr. Krebs. Sr., in his 1955 pamphlet on Laetrile appears to recognize that "animal beta-glucuronidase" and beta-glucosidase are different substances. (He characterizes the latter as a "prepared enzyme.") He does claim that the two enzymes react with Laetrile in the same manner (R 183, Att. 7 at 24).

There is some indication that Ernst Krebs, Jr., in later years abandoned his attempt to develop a Laetrile that could be broken down by beta-glucuronidase and began treating beta-glucuronidase and beta-glucosidase as equivalent. In a 1962 letter, Krebs, Jr., seems to refer to the former as an example of the latter: "beta glucosidases (e.g., beta glu-curonidase)" (R 183, Att. 16, App. 12 at 2) and seeks, in describing an experiment with water fleas he had designed. to extrapolate results obtained with beta-glucosidase to results with betaglucuronidase he feels is found in malignant lesions (id. at 4). (See also Krebs' 1970 article "The Nitrilosides (Vitamin B-17) - Their Nature, Occurrence and Metabolic Significance (Antineoplastic Vitamin B-17)" in which he again equates beta-glucosidase with beta-glucuronidase (R 183, Att. 10c at 82).)

Three other problems with this theory are quickly identifiable: (1) there is evidence that beta-glucuronidase is not particularly abundant in malignant tissues. The record shows that "\* \* beta-glucuronidase is found in all tissues of the animal body and in particularly high concentrations in spleen, liver, and en-

docrine organs, as well as in plasma and in tumors arising from estrogen-influenced tissues. Per gram of tissue, the spleen and liver have a higher concentration of beta-glucuronidase than do most tumors," (R 183, Att. 16 at 15 and App. 14). It is further stated, "Such a statement as " \* the malignant cell \* " is virtually an island surrounded by a sea of beta-glucuronidase' is sheer nonsense" (R 183, Att. 16 at 15 and App. 14).

(2) There is no evidence that cancer cells are deficient in the enzyme rhodanese. In reviewing the record, the Commissioner has not found any support for the bald assertion by the Krebs and other Laetrile proponents that cancer cells are deficient in the production of a hydrogen cyanide-inactivating enzyme called rhodanese. If there is any scientific support for that assertion, it is indeed strange that it has never been cited by the Krebs' or otherwise brought to the attention of the scientific community. The record shows, in fact, that: "There is no evidence of pronounced differential between the rhodanese content of comparable normal and cancerous tissue" (R. 378, Att. 9 at 346).

(3) The complete breakdown of Lactrile into cyanide may require an enzyme not found in animal tissues, Hydrolysis of Laetrile by beta-glucuronidase (and hydrolysis of amygdalin by beta-glucosidase) only represents the first step in breaking down the molecules to release hydrogen cyanide and benzaldehyde (which are supposed to kill the cancer cell). The first reaction in each case would yield mandelonitrile plus (for Laetrile) glucuronic acid or (for amygdalin) glucose. Mandelonitrile must then be hydrolyzed or broken down to yield hydrogen cyanide and benzaldehyde (R 416 at 1 8).

Enzymes present in apricot kernels (specifically oxynitrilase) will hydrolyze mandelonitrile to cyanide and benzaldehyde, but this enzyme is not reported to exist in animal tissues (R 399 at [ 7B). Nor does the record show that any other enzyme capable of breaking down the mandelonitrile exists in animal tissues (or in malignant lesions). Thus, if Laetrile were injected into the blood stream and did go to the malignant lesion, even if it were broken down into mandelonitrile and gulcuonic acid, it might never be further broken down to yield hydrogen cyanide and the supposed action of that substance in killing the cancer cell would never take place.

At one time, Ernst T. Krebs, Jr., apparently attempted to deal with some of these problems by separating out the elements of the apricot extract his father had prepared. The fact that both betaglucosidase and oxynitrilase are present in apricot pits and thus in the extract provides the potential for the whole breakdown process to occur in the apricot extract itself at the time when it is prepared. Krebs, Jr., sought to prevent this from happening (and perhaps sought to reduce toxicity) by separating amygdalin from "emulsin" by purifying the apricot extract. Emulsin contains, among

other things, beta-glucosidase and betaoxynitrilase (R 173, Att., "Vitamin Fraud" at 345).

Ernst T. Krebs, Jr., reportedly separated amygdalin from emulsin in 1952 and " \* " advised their administration separately in order to avoid the premature trigger-off of HCN (hydrogen cyanide) from the chemical breakdown in the somatic (or normal) tissue \* \* " (R 183, Att. 7 at 23). It is further stated that by injecting the cyanogenetic glucoside (amygdalin) followed 15 minutes later by the enzyme beta-glucosidase a "high degree of safety" as well as cancerolytic effect was obtained (id. at 23-24).

If the beta-glucosidase preparation preparation reached the same area of the body that the amygdalin had reached, its presence would lead to the breakdown of amygdalin to release mandelonitrile (see R 416 at § 18; R 183, Att. 7 at 31). While Ernst T. Krebs, Jr., apparently recommended that the second injection be of emulsin (R 183, Att. 7 at 23), Dr. Krebs, Sr., states that his second injection would consist only of the enzyme beta-glucosidase (id. at 24). If the second injection did contain emulsin, the presence of the oxynitrilase in that complex might in fact lead, assuming the emulsin caught up with the amygdalin in the body, to breakdown of the mandelonitrile to release hydrogen cyanide (and benzaldehyde). However, there is little reason to believe that this release of cyanide would occur only in or near tumor cells. (In the same article in which Ernst T. Krebs. Sr., explained the process of injecting the beta-glucosidase, he stated his understanding that it was the beta-glucuronidase "which bathed the cancer cell" that acted to break down the amygdalin (id. at 25).) It should be noted that, since the time of the 1955 pamphlet, no evidence has appeared, at least in this record, that two injections, one containing beta-glucosidase, are being used in Lactrile therapy.

In light of the above, one must be concerned that products are being used that contain not only amygdalin but emulsin. As the Krebs themselves recognized, unless emulsin is separated from amygdalin (both of which exist in the apricot extract), there may occur the premature trigger-off of HCN (hydrogen cyanide) from the chemical breakdown in the somatic (or normal) tissue (R 183, Att. 7 at 23). It is this type of cyanide poisoning which has occurred from ingestion of Lactrile and from eating apricot kernels, (See R 378, California Morbidity, Nov. 14, 1975, No. 45.)

It is thus clear that the theory propounded by the promoters of Laetrile is based on faulty and unproven assumptions. The invention of Laetrile as described by Krebs, Jr., and his suggestion that an injection of amygdalin be followed by an injection of emulsin were two different ways to deal with the fact that enzyme beta-glucosidase does not exist in human tissues. What is perhaps most important about the proffered theoretical justification for Laetrile's action

is that, even if they were accepted, they would not justify the administration of amygdalin alone.

#### II. THE "NEW DRUG" ISSUE

The Commissioner will now address the first of the two issues remanded to the agency: Whether Lactrile is a "new drug" within the meaning of the Federal Food, Drug, and Cosmetic Act, 21 U.S.C. 301 et seq. (hereinafter the act). Based upon a careful review of the administrative record, as detailed below, the Commissioner finds that Lactrile is not generally recognized by qualified experts as a safe and effective cancer drug. Accordingly, the Commissioner concludes as a matter of law that Lactrile is a new drug and thus subject to the premarket approval requirements of the act.

The term "new drug" is defined by sec-

The term "new drug" is defined by section 201(p) (1) of the act (21 U.S.C. 321 (p) (1)) as follows:

Any drug \* \* \* the composition of which is such that such drug is not generally recognized, among experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs, as safe and effective for use under the conditions prescribed, recommended, or suggested in the labeling thereof \* \* \*

Although the act defines "new drug", it does not contain a definition of "generally recognized as safe and effective."

In 1973, the Supreme Court, in a series of four cases (Weinberger v. Hynson, Westcott & Dunning, Inc., 412 U.S. 609 (1973); Ciba Corp. v. Weinberger, 412 U.S. 640 (1973); Weinberger v. Bentex Pharmaceuticals, Inc., 412 U.S. 645 (1973); USV Pharmaceutical Corp. v. Weinberger, 412 U.S. 655 (1973)) involving the procedures adopted and utilized by FDA to regulate new drugs pursuant to the Drug Amendments of 1962 (76 Stat. 780), established the legal principles that are applicable and controlling here. In Hynson, the Court discussed "general recognition" as it pertains to the effectiveness of a drug as follows:

The thrust of § 201(p) is both qualitative and quantitative. The Act, however, nowhere defines what constitutes "general recognition" among experts. \* \* We agree with FDA, however, that the statutory scheme and overriding purpose of the 1962 amendments compel the conclusion that the hurdle of "general recognition" of effectiveness requires at least "substantial evidence" of effectiveness for approval of an NDA. In the absence of any evidence of adequate and well-controlled investigation supporting the efficacy of (a drug), a fortiori (that drug) would be a "new drug" subject to the provisions of the Act, 412 U.S. at 629-30.

We accordingly have concluded that a drug can be "generally recognized" by experts as effective for intended use within the meaning of the Act only when that expert consensus is founded upon "substantial evidence" as defined in § 505(d), 412 U.S. at 632.

The term "substantial evidence" is defined in the last sentence of section 505 (d), 21 U.S.C. 355(d), as:

evidence consisting of adequate and wellcontrolled investigations, including clinical investigations, by experts qualified by scientific training and experience to evaluate the effectiveness of the drug involved, on the basis of which it could fairly and responsibly be concluded by such experts that the drug will have the effect it purports or is represented to have under the conditions of use prescribed, recommended, or suggested in the labeling or proposed labeling thereof.

In Bentex, based upon its discussion of the term "general recognition" in Hynson, the Court concluded:

Whether a particular drug is a "new drug" depends in part on the expert knowledge and experience of scientists based on controlled clinical experimentation and backed by substantial support in scientific literature, 412 U.S. at 652.

It accordingly held that "the reach of scientific inquiry under both section 505 (d) and under section 201(p) is precisely the same" (id.).

The requirements that a drug have not only controlled clinical investigations but also publication of the studies concerning it in the scientific literature are designed to assure that the community of qualified experts in general is aware of the data concerning the drug. Thus, one could not obtain general recognition just by doing the required studies without publishing them in the scientific literature, making them available to other scientists. (Studies submitted to scientific publications must undergo peer review before they are published. A study published in a scientific journal is thus more likely to form a basis for expert recognition than is one published by the lay press.) A practical effect of the statutory system, as the Supreme Court acknowledged in Hynson, supra, is that drugs will have accumulated for themselves sufficient scientific evidence to justify approval of an NDA "long before they are in a position to drop out of active regulation by ceasing to be a 'new drug' " (412 U.S. at 631).

Under the Supreme Court's authoritative interpretation of the act, therefore, general recognition of the safety and effectiveness of Laetrile depends upon two criteria: (1) Controlled clinical investigations conducted by qualified experts establishing the safety and effectiveness of the drug and published in the scientific literature, and (2) expert consensus, based upon that evidence, that the drug is safe and effective. Both requirements must be met in order for Laetrile to escape the need for premarket approval under the act; however, a finding of a failure to meet either set of requirements is sufficient to classify the drug as a new drug.

With respect to the first criterion, the safety of Laetrile must be established by adequate tests by all methods reasonably

<sup>\*</sup>Section 505(d) of the act (21 U.S.C. 355 (d) sets forth the standards applicable to obtain marketing approval of a new drug. With respect to reports of investigations which are required to be submitted concerning the safety of a drug, the act provides that such reports must include "adequate tests by all methods reasonably applicable to show whether or not such drug is safe for use under the conditions prescribed, recommended, or suggested in the proposed labeling thereof".

applicable (see 21 U.S.C. 355(d); 21 CFR 314.111(a)(1)). In addition, the effectiveness of Laetrile must be established "substantial evidence," which the statute (21 U.S.C. 355(d)) defines as evidence consisting of adequate and wellcontrolled clinical investigations. (Clinical investigations are studies involving human beings as test subjects.) The requirements for an adequate and wellcontrolled clinical investigation are set forth in 21 CFR 314.111(a) (5) (see discussion below). Both types of testing must be available to the community of experts in the evaluation of drug safety and effectiveness by means of publication in the scientific literature.

For satisfaction of the second criterion. a showing must be made of recognition among the qualified experts which is "general." It has been held that a genuine difference of opinion among experts on the question of general recogni-tion is sufficient to show that such recognition of a drug's safety does not exist (see United States v. An Article of Drug. Etc., 294 F. Supp. 1307, 1311 (N.D. Ga. 1968) aff'd 415 F. 2d 390 (5th Cir. 1969); United States v. 354 Bulk Cartons, Etc., 178 F. Supp. 847, 853 (D.N.J. 1959); Merritt Corp. v. Folsom, 165 F. Supp. 418, 421 (D.D.C. 1958)). This interpretation of "general recognition" has been criticized as requiring "unanimous" recognition (see United States v. 7 Cartons, More or Less, Etc., 293 F. Supp. 660, 662-63 (S.D. III. 1968) aff'd 424 F. 2d 1364 (7th Cir. 1970). For purposes of completeness, the Commissioner in his opinion will consider "general recognition" to require, as the 7 Cartons Court "extensively, suggested, recognition though not universally; most frequently, but not without exception" (id.),

# A. GENERAL RECOGNITION OF EFFECTIVENESS

#### 1. Objective Evidence of Effectiveness

The Courts thus have determined that, as a matter of law, no "general recognition" of a drug's effectiveness can exist absent adequate and well-controlled clinical investigations and substantial support in the scientific literature. There are no clinical investigations of Laetrile's effectiveness, published or otherwise, which are even arguably adequate and well-controlled. (See, e.g., R 185 at § 19; R 186 at §12; R 390 at § 19). For this reason, Laetrile cannot escape "new drug" status as "generally recognized" as safe and effective. It is thus a new drug without an approved new drug application whose sale or distribution, where interstate commerce is involved, is illegal.

There is, however, an apparent public lack of understanding of what the required studies consist of and why they are required. The Commissioner will thus include in this opinion a discussion of what adequate and well-controlled studies are and why they are needed. He will then discuss the deficient "evidence" of effectiveness submitted by Laetrile's proponents.

(a). What Are the Required Studies.
"(A) dequate and well-controlled clinical investigations," as those terms are used

in the act (21 U.S.C. 355(d)) are defined in detail by regulation (21 CFR 314.111 (a) (5) (ii)). These regulations, discussed with approval by the Supreme Court in Weinberger v. Hynson, Westcott & Dun-ning, Inc., supra, 412 U.S. at 617-19, were upheld in Upjohn v. Finch, 422 F.2d 944 (6th Cir. 1970) and Pharmaceutical Manufacturers Ass'n v. Richardson, 318 F. Supp. 301 (D. Del. 1970). Simply stated, such investigations are designed to determine whether an improvement noted after administration of a drug is in response to the drug or whether it is caused by some other factor. To do this, patients as nearly identical in their disease state as possible are divided into groups and treated, insofar as possible, exactly the same in all respects except one: One group receives the test drug; the other receives a placebo (a substance that looks just like the test drug but is not a drug). Since a patient might feel better through knowledge of receiving the test drug, and since the investigator might subconsciously record better re-sults because of the knowledge that he or she were administering the test drug, experiment is "double-blind" Neither the patient nor the investigator knows until after the experiment which patient is getting the test drug and which the placebo. If, at the end of the investigation, the patients receiving the drug did better than those not receiving it, one can be fairly certain that it was the drug and not some other factor that caused the improvement.

(b) The Need for Controlled Studies. In 1962, the Congress of the United States, after extensive hearings, concluded that testimonial evidence of a drug's effectiveness—even including testimonials and illustrative "case histories" by physicians—was simply not reliable. It passed the law requiring that effectiveness be shown by "adequate and well-controlled clinical investigations" which is discussed elsewhere in this opinion.

The Supreme Court examined this issue closely in 1973 and determined that Congress' decision and FDA's enforcement of that decision were supported by the evidence elicited at the congressional hearings:

(The FDA's) strict and demanding standards, barring anectodal evidence indicating that doctors "believe" in the efficacy of a drug, are amply justified by the legislative history. The hearings underlying the 1962 Act show a marked concern that impressions or beliefs of physicians, no matter how fervently held, are treacherous. (Emphasis added.)

Weinberger v. Hynson, Westdott & Dunning, Inc., supra, 412 U.S. at 619. It noted:

the conclusion of Congress, based upon hearings, that clinical impressions of practicing

physicians and poorly controlled experiments do not constitute an adequate basis for establishing efficacy.

(id., 412 U.S. at 630).

Since both the Congress and the Supreme Court have spoken on this question, no further discussion by the Commissioner of the need for adequate and well-controlled studies," rather than reliance on testimonial evidence, to show a drug's effectiveness is legally necessary. However, there is a continued public belief in testimonial or anecdotal evidence, fostered even by the lawyer of one of Laetrile's supporters. (See oral argument of Kenneth Coe, in which he contends that the safety and effectiveness of Laetrile has been shown by "anecdotes of people who have been diagnosd as terminal with cancer, anecdotes of people who have been cured of cancer, anecdotes of people who are walking around today, that are here today-well" (Tr. at 453). For this reason, the Commissioner will discuss the evidence in the record illustrating the need for scientific studies to show a drug's effectivness.

In his affidavit (R 175, Ex. A at § 3) Dr. William Beaver, an expert in drug testing, notes that critics of well-controlled studies "often point out the undisputed fact that great strides have been made in therapy in the past without the benefit of this experimental device, but simply on the basis of the uncontrolled observations of astute clinicians \*. (However), these critics often fall to mention the thousands of drugs which, on the basis of 'clinical experience,' were once accorded an 'indispensable place' in therapy, and which are now known to be useless." Dr. Beaver states (id. at \*4): "The function of the controlled clinical trial is not the 'discovery' of a new drug or therapy. Discoveries are made in the animal laboratory, by chance observation, or at the bedside by an astute clinician. The function of the formal controlled clinical trial is to separate the relative handful of discoveries which prove to be true advances in therapy from a legion of false leads and unverifiable clinical impressions, and to delineate in a scientific way the extent of and the limitations which attend the ef-fectiveness of drugs." See also affidavit of Bryant L. Jones (R 431 at 18): "Most medical mistakes of past centuries were a direct result of beliefs that were predicated on conviction rather than evidence. Most medical advances in modern times can be traced directly to the scientists insistence on valid scientific evidence to support use of today's drugs."

Because of the insidious nature of cancer, it is all the more important that the effectiveness of a drug purported to be useful in the treatment of cancer be demonstrated by well-controlled clinical studies and not solely by testimonials or anecdotes. Cancers in humans vary greatly in their behavior, i.e., their rate

See, e.g., Hearings on S. 1552 before the Subcommittee on Antitrust and Monopoly of the Senate Committee on the Judiciary, 87th Cong., 1st Sess., pt. 1, at 195, 282, 411-12. For a detailed discussion of the Congressional decision in 1962 to require adequate testing of drugs' effectiveness, see Pharmaceutical Manufacturers Ass'n v. Richardson, supra, 318 F. Supp. at 306 et seq.

fit should be noted that other types of testing are required to show a drug's safety, some of which must be completed before clinical investigations to show effectiveness can begin. (See 21 U.S.C. 355(d).)

of growth, pattern of spread, effects on the normal organs of the person, and the types of clinical symptoms or signs they produce. There is wide variability in the pattern of spread and the outcome for an individual. The effects of cancer on an individual have an element of randomness, that is, an element of chance. Physicians are therefore simply unable to predict the outcome of any cancer at any stage of development with great accuracy. Patients with terrible and widely spread cancer will occasionally have miraculous or unexpected remissions of the disease. Untrained clinical investigators who administer any remedy to a large enough group of patients with cancer will ultimately observe a miraculous outcome. This apparently miraculous outcome may well mislead the untrained investigator into belief that the remedy was responsible for the result (R 390 at 114-15).

It has been noted above that, in an adequate and well-controlled study, nelther the patient nor the investigator is told that the test drug is being administered because, if they knew, they might report results based merely on high expectations. This problem of assigning improvement to a drug when the improvement was simply a result of high expectations is known as the problem of the "placebo effect." The placebo effect is particularly common in cancer patients. A study of the placebo effect among 288 cancer patients undergoing controlled trials of oral analgesics showed that 112 patients received 50 percent or greater relief from placebo (i.e., non-drug formulations (R 186 at 1 13)

In his affidavit (R 185 at 5 20f) Dr. Daniel S. Martin states: "Humans are very susceptible, particularly when ill and desperate with hope, to the power of positive suggestion-namely, when given a 'drug' by an authority figure (e.g., a physician) with the firm statement and promise they will now begin to feel better, to have pain relief, to eat better, and to get well, these hopeful patients frequently do just what they have been told to expect." This, Dr. Martin states "is termed the placebo effect." Dr. Martin also explains that "Cancer is a chronic disease which some patients can live with for years before dying of the disease. During this slow death there are periods of 'ups' as well as 'downs,' and it is not surprising to have a Laetrile patient ascribe the 'up' to Laetrile, when it was merely coincidental timing" (id.). Similarly, Dr. Carl M. Leventhal states (R 184 at 1 7): "(P) sychogenic responses, popularly known as the placebo effect, are well documented and have been shown to occur from 30 to 70 percent of patients who are treated for pain."

The need for controlled testing as opposed to testimonial or ancedotal evidence of effectiveness is well-recognized by experts in the evluation and use of drugs. As Dr. Bayard H. Morrison states: "The problem is the anecdote doesn't permit you to know what happened yesterday. It doesn't permit you to know what is really going on today and cer-

tainly it doesn't give you any insight at all in what will happen to a given patient tomorrow, next week, or next year. To really know what a drug, any treatment, does to a patient you have to be able to evaluate him in the context of a large group whose disease you can follow carefully over a considerable period of time" (Tr. at 150).

(c) The "Evidence" of Laetrile's Effectiveness. (i) Case Reports.-The proponents of Laetrile (or amygdalin) have not submitted anything to the record that could be characterized as an adequate and well-controlled clinical study of Laetrile. In the regulation which defines adequate and well-controlled clinical investigations, it is clearly stated that: "Isolated case reports, random experience, and reports lacking the details which permit scientific evaluation will not be considered," even as corroborative evidence for adequate and wellcontrolled studies (21 CFR 314.111(a) (5)(ii)(c)). Yet that kind of report is the only "evidence" of Laetrile's effectiveness which has been submitted.

Because of the possible public belief in this kind of report, the Commissioner will discuss those submitted to this record. In addition, evaluations, submitted to the record, of earlier "case histories" or testimonials relating to use of Laetrile will be discussed.

# (a) Reports Submitted to This Record

#### (1) Dr. Binzel

Phillip E. Binzel, Jr., M.D., Scientific Advisor to the Committee for Freedom of Choice, and in private practice as a family physician since 1955, appeared at the hearing to make an oral presentation (Tr. at 360-364) and to submit written testimony (Tr. Ex. 13). Dr. Binzel's submission at the hearing (id.) was a report of a study he conducted on more than 200 cancer patients to whom he administered a nutritional program, including Laetrile. Dr. Binzel stated that he had excluded from his report the following patients: "1: Those who were alive but who had been under treatment for less than 4 months. 2: Those who had died within the first 3 months of treatment. These are the patients whose disease was already too far advanced for any form of treatment to be beneficial \* \* \*. 3: Those on whom there is not sufficiently adequate follow-up information to know for certain what their present condition is" (id.).

After the above exclusions, there remained 107 patients in Dr. Binzel's study who had been treated between 4 months and 2½ years and who, according to Dr. Binzel, "are spread pretty equally throughout those time periods" (id.). He reported that 57 patients had primary carcinoma, and 50 patients had proven metastatic carcinoma at the time they started "nutritional therapy" (id.).

Dr. Binzel's exclusion from his study of patients for whom adequate followup information could not be obtained can be expected to exclude those patients who were dissatisfied with Laetrile treatment and left his care. That exclusion, together with the exclusion from con-

sideration of patients who died within 3 months of the first treatment, would be expected to bias the study in favor of effectiveness.

Dr. Binzel states that he "did not attempt to differentiate between those patients who have had surgery and/or cobalt and/or chemotherapy and those who had none of these 'conventional' treatment" (Tr. Ex. 13, "Nutrition and the Cancer Patient" at 2). This, as Dr. Binzel notes, presents a "very valid question." Without knowing whether the patients in whom he saw improvement had had other, recognized effective, treatments, his conclusions cannot be evaluated (cf. 21 CFR 314.11(a) (5) (ii) (a) (2) (iii)).

Dr. Binzel's three-page "study," which was submitted without supporting documentation, simply lacks the details necessary to permit scientific evaluation and would not, for that reason, be considered by experts in drug evaluation even as corroborative of adequate and well-controlled studies if such studies existed (21 CFR 314.11(a) (5) (ii) (c)).

# (2) Dr. Richardson

Edward Griffin, at the oral argument in this proceeding, submitted page proofs of a book entitled Laetrile Case Histories, The Richardson Cancer Clinic Experience by John A. Richardson and Patricia Griffin (Tr. Ex. 1.) Griffin stated: "Previously the opponents of Laetrile have said that there is no evidence that Laetrile works. There has been evidence of course, known to those of us who have been close to the subject. But admittedly, there has not been a great deal of medically documented evidence open to the public. And I believe that with the publication of this book at least we will be able to put an end once and for all to this claim of there not being any evidence' (Tr. at 15-16). Dr. Richardson does not claim to have conducted a research program and his case histories in no way even approximate an adequate and wellcontrolled clinical investigation.

Dr. Richardson's license for the practice of medicine has been revoked by the State of California (R 183 at 13) because he was found to have discouraged patients from seeking conventional therapy and to have practiced a type of treatment of cancer patients characterized as "an extreme departure from the standard practice of medicine" (R 179, Ex. B at 5). The California State Board of Medical Quality Assurance stated that these two findings established "gross negligence" on the part of Dr. Richardson (id. at 10). Dr. Richardson did not choose to appear in the administrative proceeding in which his license was ordered revoked (id. at 1-2).

A number of obvious questions are raised by Dr. Richardson's book: (1) There is no indication in this book—nor is there in the other reports in this section—of the chemical composition of the "Laetrile" which was used. Since there are variations in the composition of the drugs called by the name "Laetrile," this fact leaves the reader with no certainty

as to what substance is claimed to be effective.

(2) The technique for selecting patients for reporting is hardly scientific. The book states: "Out of (a group of approximately 4,000 cancer patients), we selected a cross-section of about 500 for our study. We were able to establish contact and a working relationship with only about 250 of these. The cases with the weakest medical histories were discarded, as were those which were overly repetitious. The remainder (62) are contained in the study; but by no means do they represent our entire files" (Tr. Ex. 1

It is absolutely incredible that anyone would expect to show the effectiveness of a drug by describing 62 out of over 4,000 patients with a selection process of the type Richardson describes. The Commissioner has no means of knowing what happened to the other 3,938 or more patients. No details are given to show how the 500 patients representing a "crosssection" of the 4,000 were chosen. The failure "to establish contact and a working relationship" with half of the patients that were chosen illustrates a serious lack of followup. Logic suggests that those patients who were not benefited by Lactrile would be less likely to be willing to develop a "working relationship" with Dr. Richardson's office. Clearly patients who had died would not be available for such a relationship. The discarding of weak medical histories has never been an accepted practice in the study of any drug. What constitutes the weakness of a medical history is not explained.

(3) There is some question whether what Richardson claimed to be positive effects were in fact positive. The authors admit that one of the weaknesses of the study is the shortage of cases involving 5-year survival or longer (Tr. Ex. 1 at

Indeed, the case histories section of the book (Tr. Ex. 1 at 126-276) list for each of the 13 different groupings of cancer the expected death rate for those cancers in terms of percentage survival over a set number of years, usually 5 years. Many of the patients simply had not survived long enough at the time of the book's writing to constitute successes.

For example, six cases of female breast cancer are reported (Tr. Ex. 1 at 126-137). According to Dr. Richardson, these women have received metabolic therapy, including Lactrile, for periods varying from 13 to 32 months with an average of less than 21 months. Since Dr. Richardson states (Tr. Ex. 1 at 126); "Two out of every 3 patients with cancer of the breast who do not use Leatrile but choose instead to submit to orthodox therapies will be dead within five years," the fact that he has six patients who have survived 13 to 32 months with a mean of less than 21 months does not provide any evidence of the effectiveness of Dr. Richardson's treatment.

Dr. Richardson recognizes that some of the patients whom he has selected may not have had cancer. See, e.g., Tr. Ex. 1 at 148, patient B1381 where Dr. Richardson, in discussing the chest x-rays of a

patient, states that the period of May 73 to January 77 represented: "A three-and-one-half-year remission of probable cancer of the lung." (Emphasis added). For several other patients, Dr. Richardson does not quote from the pathology report but merely reports that such a report was positive for cancer (see, e.g., patient P131B (Tr. Ex. 1 at 167)).

(4) Dr. Richardson relies in many instances upon what patients have told him about their medical histories, either orally or in writing. Some patient reports upon which he relies are hardly credible. (See, e.g., patient C106MA (Tr. Ex. 1 at 146): "The patient states the local doctors strongly recommended removal of both lungs and the permanent hospitalization of the patient, who then would be forever dependent on machines to do her breathing.")

(5) Some patient reports are so sketchy as to provide no basis for any conclusion. See, e.g., patient B144J (Tr. Ex. 1 at 202-203: There is no information regarding how the diagnosis of cancer was made. There is no indication that Dr. Richardson ever characterized the size of the tumor or whether he relied on the patient's description. There is no indication whether the patient had received any Laetrile since January 1970. There is no indication that there has been any contact with the patient since February 1976.

(6) As noted above, for each of the 13 groups of cancer which Dr. Richardson has used in his book he cites the anticipated fatality rates for patients receiving only orthodox treatment. As discussed elsewhere in this opinion, cancers are very different in their behavior, i.e., their rate of growth, their pattern of spread, their effects on the normal organs of the person and the types of clinical symptoms or signs that they produce (R 390 at § 14). Experts in the testing of cancer drugs stress that the effects of cancer on a person have an element of randomness and that the ability to predict the outcome of any cancer at any stage of development varies (id.). In light of the regularity with which cancer patients' diseases vary from their expected courses, it would have been surprising if Dr. Richardson were unable to report that 62 out of 4,000 (c. 11/2 percent) of patients he saw had remissions for periods of up to, but often much shorter than, 5 years.

Thus, the Richardson book is not only not the kind of adequate and wellcontrolled clinical investigation necessary to show the effectiveness of a drug, it is not even on its face a particularly credible recounting of medical case his-

#### (3) Dr. McDonald

Lawrence Patton McDonald, M.D., a urologist, and a member of the United States House of Representatives, submitted a statement in which he reported the following observations after treating almost 200 cancer patients (R 509 at 3):

"(1) Most patients had proven cancer and had had surgery and radiation and/or chemotherapy. Most cases would

have been hopeless by routine medical standards.

"(2) Perhaps 30-35% received minimal to no benefit from the program.

"(3) Approximately 40-45% received notable benefits from the program such as improved appetite, improved interest in life, weight gain, lessening or cessation of pain. This category ultimately died but were individually pleased with their improvements.

"(4) About 20% were in the category of marked improvement. In some cases this has been miraculous with these same patients doing very well today."

Dr. McDonald provided no details whatsoever other than those quoted above. Dr. McDonald's report was not, nor does it appear to have been submitted as, an adequate and well-controlled clinical investigation.

#### (4) Dr. Soto

Although Dr. Mario H. Soto appeared and testified (Tr. at 478-481) at the oral argument and was given an opportunity (Tr. at 481) to submit for the record data from his treatment of cancer patients with Laetrile, no data have been received from him.

#### (b) Case Reports Evaluated Previously

Much of the evaluation of case reports has been done in California, where Laetrile originated and was, along with a number of other "unproven" remedies. responsible for the 1959 passage of a State law aimed at cancer quackery. In 1952, the Cancer Commission of the California Medical Association collected information on 44 patients treated with Lactrile, all of whom either had active disease or were dead of their disease, with 1 exception. In some instances, the members of the Cancer Commission had the opportunity of seeing the patients thus treated. The conclusions of the Cancer Commission were that, of those alive with disease at the time of the study, no patient had been found with objective evidence of control of cancer under treatment with Laetrile. Nine patients who died from cancer after treatment with Laetrile were autopsied. Histological studies done for the Commission by five different pathologists showed no evidence of any chemotherapeutic effect (R 378, Att. 15 at 320-326; R 183. Att. 16 at 2-19 and App. 2-3).

In June 1962, the Cancer Advisory Council of the State of California Department of Public Health examined a total of 35 case histories of cancer patients treated with Lactrile. The Council unanimously judged these cases inadequate for any critical evaluation of Lactrile. "Many of the cases had received orthodox treatment; objective evidence of benefit was absent or insufficient, most of the documentation (dealt) with subjective improvement; some contained no pathological proof of malignancy; many were 1961 cases without followup; the duration of treatment was frequently unknown because the data reported the period of hospitalization only and often discharge dates were not shown" (R 183.

Att. 16 at 30-31).

Thirty-six clinical records translated into English from the French were evaluated by the Cancer Advisory Council in December 1962. The Council reached the following conclusions:

1. The records failed to indicate that the patients treated with Lactrile secured either palliation or regression of their cancerous affliction as a result of the therapy.

2. In several instances, there was absolutely no evidence presented as to the response of

the patient to the therapy.

3 In other instances, objective evidence documenting the statement of benefits, was

not provided.

4. In one group of 17 of these cases, sufficient followup was absent. The longest period of followup was 14 months, 12 days and 23 days of hospitalization. The next longest was 381 days and the last was 127.

5. Results which were reported as "improved" were without meaning since no criteria, subjective or objective, were pro-

6. The evidence presented lends no credence to the alleged efficacy of Laetrile and Vitamin B<sub>10</sub> in the treatment, durative or palliative, of advanced cancer (R 183), Att.

In preparing its 1963 report, the Cancer Advisory Council also reviewed about 16 pounds of documentary material delivered by Laetrile proponents. The material contained a total of 63 case histories, 15 of which were submitted by two doctors in the United States (Dr. Ray Evers, Allusia, Ala., and Dr. John R. Morrone, Jersey City, N.J.).

The opinion [of the Cancer Advisory Council] on review of these cases is that they give no credence to the claimed curative effects of Lactrile in human cancer nor in those animal cancers where it had been investigated. The evidence of palliative response, both subjective and objective, is tenuous and poorly documented. Except in the cases in which death intervened and one or two others in which there was questionable diagnosis of cancer, no followup has been re-corded, with the result that the final outcome of the cases is not recorded (R 183, Att. 16 at 35-36).

In January 1963, the McNaughton Foundation and the North End Medical Center, both in Montreal, Canada, submitted to the Cancer Advisory Council of the California State Department of Public Health a total of 14 clinical records on patients treated with Laetrile. These were not complete records but were abstracts furnished by various hospitals in Canada to the McNaughton Foundation. The Cancer Advisory Council appointed a committee of three physicians highly qualified and actively engaged in the treatment of cancer to review and evaluate these records. Each physician made an independent evaluation.

The committee reported: "These 14 records provided by the McNaughton Foundation were examined and fail to indicate that the patient treated with Laetrile secured either palliation or regression of their cancerous affiliction as a consequence of the therapy. In several intances, there is absolutely no evidence presented as to the response of the patient to the therapy and in other instances objective evidence which documents claims of benefit is not provided. It is concluded from careful review of these records that they are inadequate as reports of therapeutic use of Laetrile, and they do not indicate that therapeutic benefit resulted from treatment with Laetrile, and do not indicate that this agent is of value in the treatment, cure, or palliation of cancer. In only one instance is there a statement by the examining physician indicating that a definite beneficial effect from Laetrile might have occurred" (emphasis in original) (R 378, Att. 14 at 26).

In 1971-72, the FDA, together with the National Cancer Institute, investigated and evaluated 12 clinical histories submitted by Dr. Ernesto Contreras of Mexico covering his experience with Laetrile in the treatment of cancer (see R 184, Ex. 3). FDA was able to obtain documentation covering the full course of the disease in 7 of the 12 case reports. All seven patients whose records were reviewed had received treatment other than Laetrile, including surgery, chemotherapy, or radiotherapy, or more than one of these approaches, either before, after, or concurrently with Laetrile therapy (R 184, Ex. 3; R 198 at 9-10).

Most of the alleged improvements stated, in the 7 case reports which could be evaluated, to be associated with Laetrile treatment have been found to be associated with one or more of the following events in the patient's disease (see R 183, Att. 16 at 10-11):

a. Subjective improvement was interpreted as being evidence of the agent's affecting the neoplasm, rather than being due to the general effect on the host, whether by metabolic or psychologic reasons.

b. Phases in the natural history of malignant neoplasm not infrequently observed in patients who are receiving no treatment whatever were interpreted as being due to the therapy employed (emphasis in original).

\* \* (For example,) occasional patients

with widespread peritoneal carcinomatosis will exhibit regression of their disease following simple exploratory procedures.

c. Patients reported as showing regression of cancer with Lastrile were either receiving concurrent treatment by other methods, or had in their recent past been treated by some (orthodox therapy) and were exhibiting a degree of control of their disease entirely attributable to the previous treatment (Emphasis in original).

d. A few of the patients treated did not have proof of the presence of cancer in the form of histological diagnoses, the evidence being more or less inferential, as radio-graphic observation of lesions in the lung, or a surgeon's diagnosis of a lesion as cancerous on observations of gross pathology at operation, without confirmation with

biopsy.

e. Very few of the clinical records to which the Cancer Commission had access contained any sort of satisfactory evidence as to objective, accurate evaluation of the progress of the primary neoplasm or its metastases while under treatment.

(ii) Animal Testing of Laetrile.—As indicated elsewhere in this opinion, general recognition of Laetrile's effectiveness among experts in the evaluation of drug effectiveness could only be based upon adequate and well-controlled clinical (i.e., human) investigations. Thus, even if Laetrile had been shown to be effective in animal test systems, and the

Commissioner concludes it has not, that fact would not remove Laetrile from the category of "new drug."

Nevertheless, in the interest of providing the public with all the information available in the record concerning this drug, the Commissioner will discuss the animal tests done with amygdalin about which there is controversy. Amygdalin has been extensively tested in animal systems. From the tests done, Dr. Dean Burk, president of the Dean Burk Foundation, Inc., has selected three tests done in the United States as showing a positive effect (R 302). In each case the laboratories which ran the tests found them to be negative. Dr. Burk also cites two foreign reports, one of which was not published (id.). His contentions, and the evidence relating to each in the record, will be discussed point by point and other animal testing done with the drug will be noted.

#### (a) Tests Claimed to Show a Positive Effect

#### (1) Sloan-Kettering

Dr. Burk includes the following in his list of animal studies showing a positive effect for amygdalin:

Sloan-Kettering Cancer Center (New York), with CD<sub>e</sub>F, mice bearing spontaneous mammary carcinomas: Inhibition of formation of lung metastases, inhibition of growth of primary tumors, and greater health and appearance of animal hosts, upon treatment with 1-2 gm crystalline amygdalin/kg body weight/day (Report of K. Sugiura, June 13, 1973) (R 302, Ex. A at 15).

Regarding the studies conducted at Sloan-Kettering, C. Chester Stock, Ph. D., Vice President and Associate Director for Administrative and Academic Affairs, Sloan-Kettering Institute for Cancer Research, stated in his affidavit (R 195) that:

We have tested amygdalin at high doses, 1000 mg/k/day, in over a dozen transplantable tumor systems and one induced tumor system without seeing any action against the tumors. The chemotherapeutic agents effective in clinical cancer have had or would have had their activities detected in one or more of those systems.

In spite of demonstrated utility of transplanted experimental animal tumor systems. some individuals believe that use of spontaneous animal tumors is more appropriate for seeking drugs for use in man. It was considered that this would be true of the advocates of the use of Lactrile who believe it needs to be used for relatively long periods of time.

Consequently, Dr. K. Sugiura in my laboratory looked for the effects of amygdalin on the growth of spontaneous mammary tumors in CD,F, mice and also on metastatic spread to lungs of the hosts. Early observations of Sugiura featured an apparent inhibition of the appearance of metastases in the lungs of mice given daily (except Sunday) doses of 2000 mg/k of amygdalin in his 3 initial ex-periments. The treated mice showed lung metastases in 20% while 80% of the controls had metastases. The mice had been injected until death or until the primary tumors were over 2.5 cm in diameter. The data from these experiments were leaked to the press unfortunately before they could be checked adequately. Subsequent experiments, in some of which Dr. Sugiura participated, some conducted with Dr. Daniel Martin of the Catholic Medical Center of Brooklyn and Queens and some which were independent by other investigators in our Institute, showed that the initial results were not consistently observable. In some experiments there were more metastatic mice in the treated than in the control mice. In the latest experiment in which Dr. Sugiura read the lungs of the mice without knowing what treatment they had received, there was essentially no difference found between the treated and control groups (R 195 at § 10).

In his affidavit (R 185), Daniel S. Martin, M.D., states that: "My laboratory's tests with Laetrile demonstrated Laetrile to be without effect (on spontaneous tumors in experimental animals). Further, these negative tests on these animal tumors were confirmed by three other investigators at Memorial Sloan-Kettering Cancer Center in New York. One of the latter investigators (Dr. K. Sugiura) reported his initial experiments to demonstrate Laetrile to have anti-cancer activity, but his subsequent results were negative. A degree of variability in results is common in biological research, and the final opinion is based on whatever the majority of the findings are. In this instance, the totality of the data clearly and unequivocally reveals Laetrile to be without anti-cancer activity" (R 185 at 7 21 (c-d)).

#### (2) Southern Research Institute

#### Dr. Burk's citations continue:

Southern Research Institute (Birmingham, Alabama) for the National Cancer Institute, in a majority of 280 BDF, nice bearing Lewis lung cancers, treated with up to 400 mg crystalline amygdalin per kg body weight, with respect to increased life span (Report, December 3, 1974) (R 303, Ex. A at 15).

The results of two studies conducted by Southern Research Institute for the National Cancer Institute were published in the scientific literature in 1975. One of the studies involved an experiment "in which four transplantable rodent tumors (L1210 lymphoid leukemia, P388 lymphocytic leukemia, B16 melanoma, and Walker 256 carcinosarcoma) were used to investigate the antitumor activity of amygdalin MF \* \* \* alone and in combination with beta-glucosidase" (R 184, Ex. 3b at 939). No antitumor activity was observed in any of the four tumor systems tested with amygdalin alone or in combination with beta-glucosidase (id.; see also R 173, Att., Memorandum, March 12, 1973).

The second study, in which amygdalin MF (i.e., amygdalin provided by the McNaughton Foundation) was evaluated alone or in combination with beta-glucosldase against three transplantable rodent tumors (Ridgeway osteogenic sarcoma, Lewis lung carcinoma, and P388 leukemia), showed that amygdalin alone or in combination with beta-glucosidase did not demonstrate antitumor activity against any of these three tumor systems (R 184, Ex. 3C at 952-53).

At the oral argument, Bayard H. Morrison, M.D., Assistant Director at the National Cancer Institute stated that the Institute:

has sponsored—other organizations have conducted—tests of Lactrile at various dosage levels in a variety of animal tumor systems, probably exceeding 15 or more, probably closer to 20.

This indeed really is about the most extensive that NCI and other laboratories in the aggregate have tested of essentially a non-active substance. For in all of these tests which include tumors ranging from carcinomas, sarcomas, lymphomas, any kind of tumor which parallels to a large degree the human type of tumor, the results have been essentially negative. There have been occasional, marginal evidences of activity which have not been reproducible.

So, in balance, Lastrile has failed the test of demonstrating activity in the preclinical animal tumor systems that we know now predict for activity in human cancer.

And I should add that of the 30 or 40 drugs that are now regularly available and known to have effect in certain forms of human cancer, all of these drugs have demonstrated activity, significant activity, in one or more of these animal tumor systems (Tr. at 146A-47).

The proponents of Lactrile question the statistical controls and experimental design employed in the studies conducted by Southern Research Institute (see R 302, Ex. E). They suggest the utilization of methods of statistical analysis developed for use in judging results obtained with physical, rather than biological, systems. One of the research scientists at the National Cancer Institute responsible for the studies conducted by Southern Research Institute points out that "[t]he variation in all biological systems is far greater than that involving physical phenomena" (R 438 at 1). He suggests that it is, for that reason, not possible to use the internal statistical analyses suggested by the proponents of Lactrile (id.).

#### (3) Scind Laboratories

#### Dr. Burk's third reference is:

Scind Laboratories, University of San Francisco, 400 rats bearing Walker 256 carcinoma (200 treated with amygdalin, 200 controls), with 80 percent increase in life span at optimum dosages (500 mg amygdalin/kg body weight) (October 10, 1968). Cf. FDA-IND application No. 6734, pp. 247-8, 00080-00093 (E 303, Ex. A at 15).

The Scind Laboratory data were submitted to FDA in support of the Mc-Naughton Foundation's IND for amygdalin in 1970. An ad hoc committee of cancer experts evaluated these data during its review of the IND. In its report, the Committee stated: "We are particularly cognizant of the lack of adequate evidence of in vivo antineoplastic characteristics. The Scind Laboratory data in the initial submission of IND 6734, April 6, 1970, concerning two experiments with a Walker 256 system are considered unacceptable because of inadequate documentation of status of animals, percentage of tumor take, rate of growth, and accounting for acute deaths[.] and the other substantial lack is a statistical analysis. Scind Laboratory, in a letter dated October 18, 1968, filed with [an] October 31, 1970, amendment, states 'Laetrile, when administered without Beta glucosidase has little or no effect upon transplanted rodent tumor systems tested.' (emphasis theirs [i.e., Scind Laboratory's])" (R 184, Ex. 2 at 1).

#### (4) Pasteur Institute

#### Dr. Burk's fourth reference is:

Pasteur Institute (Paris), with human cancer strain maintained in mice treated at optimal dosage of 500 mg Amygdalin Marsan/kg body weight/day; increased life span and delayed tumor growth up to 100 percent (Dec. 6, 1971 report by M. Metianu) (R 302, Ex. A at 15).

In a sworn affidavit (R 422), a medical officer in the Bureau of Drugs, who is trained in medicine and experienced in scientific research and who is fluent in both French and German, commented on the cited studies conducted at the Pasteur Institute in Paris and the Institute von Ardenne in Dresden (see discussion below).

The medical officer, through the American Embassy in Paris, learned that the report entitled, "Anti-Tumor Toxicity and Activity" was written on the letterhead of the Institute Pasteur and was an internal report of the Institute that has never been published in any scientific journal. In the affidavit, the medical officer states that the fact that the report "represents preliminary work only and has not been published in any scientific journal since it was prepared six years ago raises my suspicions that the preliminary results obtained could not be reproduced" (R 422 at § 6).

#### (5) Institut von Ardenne

The fifth reference cited by Dr. Burk

Institut von Ardenne (Dresden, Germany), H strain mice bearing Ehrlich ascites carcinoma treated with bitter almond amygdalin ad libitum in addition to the regular chow diet: increased life span and decreased rate of cancer growth, treatment beginning 15 days before cancer inoculation (Arch. Geschwulstforsch 42, 135-7, 1973) (R 302, Ex. A at 15).

After reviewing the article published in the Arch. Geschwulstforsch, the Bureau of Drugs medical officer made the following comments:

The author's terms (in the summary section) "feeding with bitter almonds," "prolongation of survival" (due to feeding with bitter almonds), and "inhibition of tumor growth" are not adequately defined in the subsequent text or by the content of the text and thus are uninterpretable.

The description of the methodology is deficient for a number of reasons. It falls to provide information whether the mice were kept singly or caged in groups. It fails to provide information on the techniques for demonstrating whether and how much of the bitter almonds had been eaten by each experimental mouse. It falls to inform on the origin, quality, and composition of the bitter almonds with respect to the latter alleged role of "amygdaline" and HCN. Due to these failings it is not possible to draw conclusions from any differences of events between experimental and control animalsif such differences could be demonstrated at all. The author also fails to give the ago of the mice and the body weight of each individual mouse of each group and at each weighing date. The use of sole mean values in this paper is potentially misleading. (The scientific evaluation of data requires implementation of variabilities of the individual measurements.)

The author makes statements on "tumor growth" which are based on implications, indirect deductions, and on arbitrary assumptions.

The term "tumor growth" is potentially misleading for the Ehrlich ascites cancer which consists of a cancer cell suspension in the peritoneal fluid. The study fails to use precise methods of measuring the number of cancer cells present in each mouse.

In my opinion, this article falls to provide any evidence that bitter almonds are effective in inhibiting the growth of tumors (R 422 at (7).

#### (b) Other Tests

As noted above, the two tests done by the Southern Research Institute and the Sloan-Kettering studies now completed have demonstrated conclusively, in the view of most experts, that amygdalin, either alone or in conjunction with the enzyme beta-glucosidase, exhibits no antitumor effect. These results are in accord with the negative findings of three earlier animal studies commissioned by the National Cancer Institute. Those tests are summarized in the record as follows:

1957: Amygdalin was tested with three transplanted mouse tumor systems used at the time by the NCI Cancer Chemotherapy National Service Center (CCNSC) to screen compounds for anti-cancer activity. Amygdalin produced no significant inhibition or growth of the carcinoma 775 or sarcoma 180 tumors, and produced no significant increase in the lifespan of mice with leukemia Li210 tumors.

1960: Material from a different source was tested against the same three mouse tumors. The compound failed to show antitumor aclivity.

1969: Amygdalin was tested alone and in combination with beta-glucosidase against leukemia Li210 in mice. Amygdalin was inactive against the tumor, alone and in combination with the enzyme. Toxic side effects increased when the drug and enzyme were given together (R 178, Att., "NCI Testing of Lastrile (Amygdalin)"):

A study, entitled "Failure of Amygdalin to Arrest B16 Melanoma and BW5147 AKR Leukemia," Hill et al., Cancer Research, 36:2102-07, June 1976, appears as Exhibit 3 to R 170. The December 9, 1976 report of yet another animal test of amygdalin is Exhibit 3D to R 184. The drug was found not to be active against human breast and colon tumor xenografts in athymic mice.

The failure of Laetrile (or amygdalin) to show any effect in animal systems is important because those systems have shown an ability to predict effectiveness in humans. See the statement of Dr. James F. Holland (R 396): "No drug has been proved active in human cancer which does not show anti-cancer activity in experimental animals, Human cells are not so different from other mammalian cancer cells that an active drug does not act on at least one other mammalian system \* \* \*. Laetrile is completely inactive against animal cancers. It has been repeatedly tested in reputable laboratories against a broad spectrum of rodent neoplasms. Inasmuch as no drug has been found active against cancer which isn't active in the screening tumors, there is no basis to consider Laetrile a candidate chemotherapeutic compound against human cancer \* ... No scientifically accepted data whatever have been presented indicating evidence of benefit from Laetrile."

See also the statement of Dr. Bayard H. Morrison, Assistant Director of the National Cancer Institute: [Olf the 30 or 40 drugs that are (now) regularly available and known to have effect in certain forms of human cancer all of these drugs have demonstrated activity, significant activity, in one or more of these animal tumor systems." (Tr. at 147).

One comment theorized that the reason why animal tests do not show Lastrile to have any anticancer activity is because the laboratory animals are bred to have defective immune rejection systems (R 235 at 7). This theory assumes that Laetrile is hydrolyzed by an enzyme that is in greater concentration at the cancer site than at other locations in the body. The comment explains that: "If, because of a defective immune system, laboratory animals produce no hydrolyzing enzyme at the cancer, (sic) site, that fact alone would explain why Laetrile doesn't work on laboratory animals. It can't work on any organism that has a defective immune system." (See R 235 at 7-8). No evidence has been submitted to support the comment's theory.

The Commissioner concludes that the animal studies conducted to date fail to show that Laetrile (or amygdalin) has anticancer activity in laboratory animals. As has been noted previously, even if these tests showed that the drug had anticancer activity in laboratory animals, such findings would not be relevant to the question whether it is generally recognized by qualified experts as a safe and effective anticancer drug, since general recognition must be based upon testing in human beings. The lack of positive effect in test animals is of some importance, since a clear showing of success in animals might suggest the propriety of clinical testing in humans. The failure of amygdalin to produce an anticancer effect in animals is added reason for skepticism concerning the claims that it is effective in humans.

#### 2. Testimony of Experts

(a) Experts Opposed to Laetrile. The evidence that experts in the evaluation of drug safety and effectiveness do not "generally" recognize Laetrile as effective for any therapeutic use is overwhelming. (It should be remembered that for recognition to be general it must be shown that most qualified experts recognize the drug's safety and effectiveness. The fact that a few persons claiming expertise believe the drug safe and effective is thus not sufficient.) The record contains statements that Laetrile is not considered as an effective cancer therapy from several organizations with members who are experts in cancer drug evaluatione.g., the American Cancer Society, the

American Medical Association, the Committee on Neoplastic Diseases of the American Academy of Pediatrics—and from a large number of the Nation's most eminent and well-qualified experts in the area of cancer drug evaluation. It is difficult to conceive of a clearer showing of a lack of "general" recognition of a drug's effectiveness than the expression of the views of these many experts.

The Commissioner will describe the qualifications of some of these experts and either quote from or summarize the views which they have expressed. Each of the submissions referred to contains a great deal of information concerning Laetrile and the consensus of expert opinion about it, and the following excerpts are meant only to be illustrative of the views each expert expressed:

Arthur I. Holleb, M.D., Senior Vice President for Medical Affairs, American Cancer Society, Inc., submitted an affidavit (R173). His curriculum vitae lists his membership in and leadership of several professional societies, which include the James Ewing Society and the American Radium Society. He also serves on the Cancer Commissions of the American College of Surgeons and the American College of Radiology and is editor-in-chief of CA, a cancer journal published by the American Cancer Society.

Dr. Holleb stated (R 173 at 13) that he had reviewed three basic documents attached as exhibits to his affidavit and supporting documents for these basic documents and that the information contained therein was true and correct. Submitted as an attachment to Dr. Holleb's affidavit is a "Statement Concerning Laetrile" by Frank J. Rauscher, Jr., Ph. D., Former Director, National Cancer Program, National Cancer Institute. Dr. Rauscher states, "There is no evidence that Laetrile works. Over the last decade, or more, NCI has repeatedly conducted tests of Lactrile in a variety of animal tumor systems. Most have been com-pletely negative. The others have shown only marginal levels of activity which could not be reproduced. The animal systems used are those which have detected the active properties of the scores of drugs which, unlike Laetrile, have proven to be of demonstrable value in patients with many forms of cancer. The therapeutic benefits as well as the attending side effects of these materials have been clearly and amply documented in clinical literature based on carefully conceived, meticulously conducted and monitored clinical trials. The same cannot be said for Laetrile where clinical reports are largely anecdotal and unsubstantiated. Thus, there is no laboratory or clinical evidence of the effectiveness of Laetrile" (id. at 2).

See also testimony of R. Lee Clark, M.D., President, American Cancer Society, in which he states: "The American Cancer Society views Laetrile as having no proven value in the treatment of human cancers. The Society has made a continuing review of all the literature and other information available and finds no evidence that treatment with Laetrile results in objective benefits to

patients with cancer. Since 1956, the National Cancer Institute, in conjunction with the cancer research centers of America, has reviewed over 300,000 drugs, chemicals, antibiotics, and other agents including Laetrile to evaluate them in regard to their usefulness in cancer treatment. From this research, more than forty specific agents have been found to have an effect against cancer in animal and in man. Although several trials have been made with Laetrile, it has never been proved effective in cancer in any way whatsoever" (R 307 at 1).

Frederick N. Silverman, M.D., Chairman, Committee on Neoplastic Diseases, American Academy of Pediatrics, submitted testimony (R 233 and 317) which included the following comments from his committee: "Laetrile has never been shown to exhibit any efficacy in the treatment of neoplastic disease in children. It cannot be regarded as safe if it is used in lieu of drugs currently employed either as accepted treatment or in carefully designed investigative treatment protocols."

William R. Barclay, M.D., testified (Tr. 269-281) for the American Medical Association (AMA). The AMA supports the "FDA's contention that laetrile is a new drug and is neither generally recognized by experts as safe and effective for its purported use nor should (it) be distributed in interstate commerce until such time as its safety and efficacy for the treatment of cancer have been established through controlled preclinical and clinical studies" (Tr. at 272).

Dr. Barclay discussed (Tr. at 274) a May 1965 report in the Canadian Medi-Association Journal which "concluded that laetrile could not be considered as a palliative in cancer therapy on the basis of the biological rationale advanced by the manufacturer." He further states that: "the American Cancer Society has long pointed out through its continuous reviews of the scientific literature that laetrile is not a proven or generally recognized treatment for cancer. The American Medical Association likewise views laetrile as ineffective in the treatment of cancer" (Tr. at 275-276). At its 1976 Clinical Convention, the AMA adopted the following resolution pertaining to the profession's view of Laetrile: "Resolved: That the American Medical Association continue to inform the public of the danger of delay in the diagnosis and treatment of malignancies by methods not generally recognized by the medical profession as beneficial and effective; and be it Further resolved, That the American Medical Association inform the public that the safety and efficacy of amygdalin for the treatment or palliation of malignancies is unproven and that the use of amygdalin in such cases exploits the victims of malignancies and their families by preying upon the emotions of the hopelessly ill, in some cases for the profit of the unscrupulous." Dr. Barclay (Tr. at 276) states: "We believe that experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs are virtually unanimous in their recognition of the ineffectiveness of laetrile for the treatment of cancer." The AMA testimony concludes (Tr. 280): "It is clear that laetrile is not generally recognized by experts qualified to evaluate the safety and effectiveness of drugs as safe and effective."

W. Sherwood Lawrence, M.D., a Medical Officer of the State of California Department of Health, Public Health Division, Food and Drug Section, serves as the Executive Secretary of the State of California Cancer Advisory Council. He is the custodian of the records of the Council and is knowledgeable of the work of the Council and the study that the Council has conducted (R 183 at 1). Dr. Lawrence states (R 183 at 17): "The evaluation by the Council of all the clinical data available here and in Canada has failed to establish any evidence of clinical efficacy. The proponents have never published competent well-designed controlled clinical studies demonstrating the slightest efficacy of Laetrile in the cure, amelioration or control of cancer. Laetrile (amygdalin) is not generally recognized by qualified experts as either safe or effective in cancer therapy."

Jonathan E. Rhoads, M.D., is National Chairman of the National Cancer Advisory Board, a surgeon, former President of the American Cancer Society and a member of a number of organizations focusing on research, including the American Association for Cancer Research and the American Institute on Nutrition. He made a presentation at the oral argument (Tr. at 109-115), on behalf of himself, as a citizen, and the American Cancer Society. Dr. Rhoads stated (Tr. at 114): "The position of the American Cancer Society is that Laetrile can be toxic in some doses and some modes of administration. But that more importantly, it is unsafe because its effectiveness has not been demonstrated scientifically so that reliance on it may lead patients to forego better treatment. Laetrile certainly has not been proven effective as a cancer treatment or cure and is not generally recognized by qualified experts as safe and effective for cancer.

Jesse L. Steinfeld, M.D., is Dean of the School of Medicine, Medical College of Virginia, Richmond, Va.; he formerly Deputy Director of the National Cancer Institute; United States Surgeon General; Deputy Assistant Secretary for Health and Scientific Affairs, Department of Health, Education, and Welfare; and Chairman of the Department of Oncology and Director of the Comprehensive Cancer Center at the Mayo Clinic. His professional experience includes over 20 years of involvement in cancer research, particularly with respect to the metabolic effects in cancer patients that occur as cancers grow and metastasize (R 194). Dr. Steinfeld was recognized as an expert in the evaluation of the safety and effectiveness of cancer drugs by the Court in Durovic v. Richardson, 479 F. 2d 242, 248 (7th Cir.), cert. denied 414 U.S. 944 (1973). He states that neither amygdalin nor any other cyanogenic glycoside is generally recognized by himself or by experts generally, to be safe and effective for any medical purpose (id. at 5). Dr. Steinfeld also states: "I have reviewed the clinical records of a number of patients who have received laetrile as treatment for cancer, while I was in California. In that review, there was no evidence to support the view that lactrile was of value to cancer patients. I have reviewed the volumes of material submitted to the FDA in 1970, requesting an Investigational New Drug Application (IND) for lactrile. The application was not approved because of serious flaws or deficiences in both the animal and hu-man trials" (id. at 8).

Richard H. Lange, M.D., is Chief, Sec-tion of Nuclear Medicine, Ellis Hospital, Schenectady, N.Y. He submitted verified testimony (R 385) in which he cited his experience in the field of internal medicine, nuclear medicine, and his particular interests in the problem of cancer. Dr. Lange states: "When one reviews the extensive information presently available from leading experts on cancer, there is no evidence to suggest that Laetrile is in any way an effective cancer drug. \* \* \* The theory that Laetrile is effective because it destroys cancer cells by producing a release of cyanide has never had any scientific support, nor has the newer claim in the prior approach that cancer is caused by a vitamin B-17 deficiency and that Laetrile is vitamin B-17. No scientific group has recognized Laetrile as a vitamin. Evidence of an antitumor effect in animals must be suggested or proven before a drug can be used in human clinical trials. Without such proof of effectiveness, the concept of scientific investigation would be altered; the gates would be open to all sorts of quacks and utter confusion would result. Placebo effects and personal testimonials must be separated from competent objective scientific investigation which is free from bias, personal prejudice or emotional involvement" (R 385 at 1-2).

Michael B. Shimkin, M.D., is Professor of Community Medicine and Oncology, School of Medicine, University of California, San Diego, and has had 40 years of experience in cancer research, teaching, and clinical treatment of patients. He has authored or co-authored over 280 publications on clinical and laboratory cancer research (R 192), Dr. Shimkin states: "My knowledge about amygdalin ('Laetrile') spans some 30 years. At no time, nor now, has there been evidence that this material is useful in the prevention or treatment of cancer in man or in experimental animals. I know of no expert of cancer in chemotherapy who has evidence of usefulness of amygdalin in the treatment of cancer, nor of any recognized journals or textbooks in medicine that indicate such usefulness. I know of no laboratory or clinical studies of amygdalin that demonstrate scientifically any significant, repeatable benefit in animals or in man" (id. at ¶ 12).

Bernard C. Korbitz, M.D., is Chief of the Chemotherapy Section, Department of Oncology at the Radiologic Center, NOTICES

Inc., Nebraska Medical Hospital, Omaha, Nebr. He has been involved in various aspects of cancer research and cancer therapy since approximately 1954. His professional training and experience includes the authorship or co-authoriship of approximately 40 articles relating to cancer hematology and internal medi-cine (R. 181). Dr. Korbitz states: "To date, there has been no bona fide or substantiated evidence that Laetrile has any significant anti-tumor effect in any of the rodent animal systems evaluated. I have reviewed reports by Dr. Navarro in the Philippine Medical Journal who purported to have produced good results in cancer patients using larger doses. In reviewing his studies there is no objective evidence to support these claims that Laetrile is effective in any dose range against cancer" (id. at 3). Dr. Korbitz also states, "There is no objective evidence of any sort from pre-clinical or sketchy clinical reports to indicate that Lastrile has any benefit in the treatment

of cancer patients" (id. at 4).
Susan J. Mellette, M.D., is Associate
Professor of Medical Oncology, Medical College of Virginia and has had over 20 years experience in a private practice essentially limited to patients with metastatic malignant diseases. In 1975 she was President of the American Association for Cancer Education, an organization of medical and dental school faculty interested in cancer teaching in professional schools (R 420). Dr. Mellette states, "My views on the substance Laetrile are based on reports of the ineffectiveness of amygdalin which have been published in the standard scientific literature and also on two books and other printed materials put out by proponents of Lactrile which I have read. In the latter, I have found only unsubstantiated testimonials and hearsay in the patient reports and so-called scientific arguments which reach unwarranted conclusions without appropriate experimental methodology" (R 420 at 1).

Daniel S. Martin, M.D., has been involved in general cancer research since 1946. Since 1950, he has worked in cancer chemotherapy, and since 1958 in cancer immunology as well. His professional bibliography includes over 100 publications, the vast majority of which resulted from research in cancer immunology and chemotherapy (R 185). Dr. Martin states, "The proponents of Laetrile claim that their clinical studies in cancer patients demonstrated that Lactrile often reduced the size of a malignant tumor and caused some tumors to completely regress. Evidence-none; i.e., no objective evidence support such a claim. No 'hard' patient data, no tumor measurements of the progress of the disease state, no biochemical data, no survival data, etc. The pro-Laetrilists do not present any competent scientific evidence that Laetrile is effective for the treatment of cancer. Only testimonials-'anecdotal' evidence-are presented that the Laetrilecancer patients and their doctors 'believe' in its efficacy. Belief, however, is not adequate for reliance of drug efficacy. Only strict scientific standards should be employed; namely, adequately documented scientific, well-controlled, evidence of objective antineoplastic effects in humans. The fact that a great many cancer patients have received Laetrile and attest to its benefits is not evidence. Mere clinical experience per se is not a substitute for lack of appropriate objective documentation of clinical efficacy" (emphasis in original) (id. at § 20e).

Harold James Wallace, Jr., M.D., is Director of Cancer Control and Rehabilitation at Roswell Park Memorial Institute, Buffalo, N.Y. He has had extensive training in the clinical pharmacology of cancer drugs and has participated directly in the clinical testing of a number of new anti-cancer drugs. He has, over the past 20 years, conducted and published the results of controlled clinical trials of drugs, radiation therapy, and other treatments of cancer (R 199). He is a cured cancer patient (Tr. at 170). Dr. Wallace states: "There is no evidence in either animal models or in the large numbers of patients who have received amygdalin that it is effective in any way in preventing cancer, causing a regression or remission of cancer, or improving the life expectancy of the cancer patient. Neither has there been any evidence that it decreases the symptoms of pain, weakness, or depression from cancer in any direct way. It is not analgesic or antiemetic in character. The anecdotal evidence claimed by amygdalin proponents has not been presented to me or to any scientific forum for critical review and these claims have not been substantiated by documentation in medical records available for review" (R 199 at \$14).

John T. P. Cudmore, M.D., is a Boardcertified surgeon whose professional experience includes the practice at oncology for the past 20 years. Dr. Cudmore stated that his work requires him to be acquainted with the literature related to drugs used in the treatment of cancer published in professional journals, and that he regularly attends meetings of experts at which drugs used in the treatment of cancer are discussed and evaluated (R 178). Dr. Cudmore states (id. at [ 10): "In my practice of oncology in San Diego since 1956. I have examined numerous patients after their treatments with amygdalin or Lactrile in nearby Tijuana, Mexico. I have never seen any evidence of cure or palliation with Laetrile. I can conclude from my personal experience that Lactrile or amygdalin is ineffective in the treatment and prevention of cancer." In support of these statements, Dr. Cudmore discusses in his affidavit the case histories of nine patients who have received Laetrile, all of whom in his opinion received no benefits therefrom. Dr. Cudmore states (id. at ¶ 8): "The composition of amygdalin is such that I do not recognize it, nor is it generally recognized by experts qualified through scientific training and experience to evaluate the safety and effectiveness of drugs, as safe and effective for the use in treatment or prevention of cancer."

Sidney Weinhouse, Professor of Biochemistry at Temple University School of Medicine, is a researcher in the field of cancer for the past 30 years, editor of the journal, Cancer Research, and a member of the Board of Directors of the American Cancer Society (R 384). He made the following statement regarding Lactrile: "Although widely touted for its curative effects on cancer for many years, there is no shred of evidence from any reputable cancer researcher that this substance has any therapeutic value. I know of no reputable scientist who has published evidence for the effectiveness of Laetrile" (id. et 1).

Bryant L. Jones, M.D., is a Medical Director in the Commissioned Corps of the United States Public Health Service (R 431). Since 1960, first in the pharmaceutical research industry and then in government, he has worked with the design and evaluation of clinical investigations, the purpose of which were to determine the safety and effectiveness of drugs (id. at § 2). Dr. Jones states, "I am presently responsible for the review and evaluation of protocols and reports concerning the use of drugs subject to New Drug Applications (NDA's) and Notices of Claimed Investigational Exemption for New Drugs (IND's). I review such protocols and reports for the purpose of determining whether or not they provide scientifically acceptable standards of safety and effectiveness. I would estimate that I have reviewed several thousands such reports, most of which were and are directly related to and involve drugs intended for use in the field of oncology. which is the management of cancer" (id. at [3). Dr. Jones states, "I have made a careful review of the statements which are part of the record in this proceeding identified as:

- 1. Comments; C0001 through C0247.
- Testimony: TS 01 through 14.
   Letters; Let No. 1 through 49.

4. Oral arguments: OR 01 through 11. (When submissions were received by the Hearing Clerk, they were assigned both a number-letter code (used here by Dr. Jones) and an R number, utilized for purposes of citation in this opinion.) I have evaluated each statement and report which purports to show that Laetrile, amygdalin, or any of the cyanogenetic glycosides are safe or effective in the treatment of cancer, as a palliative, as an analgesic, or for any medical purpose. None of the statements or reports are adequate, well-controlled scientific studies. The reports I have examined fail to measure up to the principles applicable to adequate, well-controlled scientific studies in every particular. The studies not only fail to measure up to minimum standards applicable to adequate, wellcontrolled scientific studies, but also fail to present any scientifically acceptable. objectively documented clinical evidence of safety and effectiveness for amygdalin, Laetrile, or any cyanogenic glycoside" (id. at 16).

George J Hill, II., M.D., is Professor and Chairman of the Department of Surgery and Associate Dean for Clinical Affairs of the Marshall University School of Medicine, Huntington, West Virginia (R 170). His professional duties involve the medical management of cancer and require that he be familiar with drugs that are generally recognized as safe and effective in treating cancer. He keeps abreast of the consensus of informed opinion by reading medical literature concerning cancer and its management, by attending meetings of experts where methods of treatment that are recognized as safe and effective are described and discussed, and through teaching, conducting research, and exchanging views with his colleagues who are experts in the field (id. at [ 12). He has himself conducted studies on amygdalin, the reports of which have been published and are attached as Exhibits 2 and 3 to his affidavit. He states, "In the course of my investigation of amygdalin's potential as an antitumor agent, an extensive review of both popular and scientific literature relating to it and Laetrile was conducted. Most reports in the scientific literature supporting Laetrile have appeared in foreign medical journals. Only one preliminary report purporting to support use of Laetrile was found in an American journal. The favorable reports concerning clinical use of Laetrile or amygdalin were testimonials based on individual case reports. There were no adequate, well-controlled clinical studies which demonstrated or purported to demonstrate that amygdalin or Laetrile were safe and effective for use in the medical management of cancer, Neither were there any favorable clinical reports in which an attempt was made to measure any objective parameters for adequate periods of followup to determine any possible drug-induced effect. The literature also contains reports concerning a limited number of carefully monitored clinical cases in which use of amygdalin failed to result in any objective benefit in the management of can-cer" (id. at ¶ 9-10). Dr. Hill also states, "The composition of amygdalin is such that I do not recognize it, nor is it generally recognized among experts qualified through scientific training and experience to evaluate the safety and effectiveness of drugs, as safe and effective for use in cancer management, or for any other known medical use. I know of no medical school where use of amygdalin for treatment or prevention of cancer is taught. I know of no medical expert qualified through scientific training and experience in the control of cancer who advocates use of amygdalin" (id. at 1 13).

Vincent T. DeVita, Jr., M.D., is Professor of Medicine at the George Washington University School of Medicine, Washington, DC, and is a diplomate of the American Board of Internal Medicine, with subspeciality certification in Hematology and Medical Oncology (R 169). Dr. DeVita has been Director of the Division of Cancer Treatment, National Cancer Institute, since 1974. His job requires that he regularly attend meetings of experts at which drugs used in the treatment of cancer are discussed and evaluated and that he be acquainted with

the literature published in professional journals relating to drugs used in the treatment of cancer (R 169 at § 1-16). Dr. DeVita states (id. at § 18-19), "The composition of amygdalin is such that I do not recognize it, nor is it generally recognized by experts qualified through scientific training and experience to evaluate the safety and effectiveness of drugs, as safe and effective for use in the treatment or prevention of cancer. I know of no adequate, well-controlled clinical study which shows it to be safe and effective for use in the treatment or prevention of cancer in humans. I know of no medical school where use of amygdalin is taught, and of no recognized medical text which prescribes, recommends, or suggests its use. Neither do I know of any expert in cancer chemotherapy who is of the opinion that it is useful in the treatment or prevention of cancer, or that there is evidence that it is useful in the medical management of cancer."

R. L. Meckelnburg, M.D., is Director, Department of Nuclear Medicine, Wilmington Medical Center, Wilmington, Delaware, and a physician concerned with the care and treatment of cancer patients by means of chemotherapy (R 154). He described his limited experience with treating patients with Laetrile in the years 1963-1967. Although Dr. Meckelnburg states (and the Commissioner agrees) that the study was not a clini-cally controlled series, he reports that "the results of these treatments were uniformly unsuccessful." Dr. Meckelnburg noted, "The individuals who were most interested in promoting the use of Laetrile failed to administer the drug in a manner consistent with good clinical investigative methodologies, particularly the use of the double blind control and crossover models of study." He concluded, "The promulgation of the drug as a preventive for cancer in the light of today's knowledge is totally absurd" (id.).

Robert C. Eyerly, M.D., is a physician and surgeon on the staff of the Geisinger Clinic in Danville, Pennsylvania, and a diplomate of the American Board of Surgery (R. 167), He currently serves as Chairman of th Committee on Unproven Methods of Cancer Management, American Cancer Society. This Committee's chief concern is, "With methods that are promoted as having established value in diagnosis, prevention, treatment, or control of cancer, despite a lack of competent scientific evidence to support claims made for them. The Committee reviews material assembled by its staff, mostly from published sources, to find out what kind of claims are made, and they evaluate scientific literature to see if it contains evidence to support such claims" (id. at § 5-7). Dr. Eyerly states, "The composition of amygdalin is such that I do not recognize it, nor is it generally recognized by experts qualified through scientific training and experience to evaluate the safety and effectiveness of drugs as safe and effective for administration to humans for the treatment or

prevention of cancer, or for any other purpose" (id. at § 10).

Several experts qualified by scientific training and experience in the field of cancer research and cancer treatment submitted similar statements attesting that they knew of no cyanogenic glycoside that is generally recognized as safe and effective for the treatment, prevention, or cure of cancer, for the relief of pain associated with cancer, or for any medical purpose. They also stated that the composition of these cyanogenic glycosides, in general, and of amygdalin, in particular, is such that they do not recognize them, and the cyanogenic glycosides are not generally recognized among experts qualified through scientific training and experience to evaluate drugs, as safe and effective for the treatment of cancer, for prophylaxis against cancer, for relief of pain associated with cancer, or for any medical use. These experts further stated that the scientific literature contains no reports of adequate, well-controlled, scientific studies or other evidence upon which recognition of safety and effectiveness may be predicated. They did not know of any recognized medical text in which the use of amygdalin or any other cyanogenic glycoside is recommended for the treatment of cancer. They did not know of any medical school where use of these substances for such purpose is taught. They did not know of any expert in cancer chemotherapy who is of the view that there is evidence that these substances have any useful effect in treating cancer. They did not know of any report in the scientific literature describing an adequate, well-controlled study which demonstrates that amygdalin or any other cyanogenic glycoside is safe and effective (Dr. Daniel S. Martin, R. 185; Dr. Joseph F. Ross, R 190; Dr. Charles G. Moertel, R 186; Dr. Jesse L. Steinfeld, R 194; Dr. C. Chester Stock, R 195; Dr. Harold James Wallace, R 199; Dr. Peter H. Wiernik, R 200; Dr. Emil J. Freireich, R 390; Dr. David T. Carr, R 176). The qualifications of the individuals not previously discussed are set forth in the following paragraphs:

Joseph F. Ross, M.D., is Professor of Medicine at the University of California School of Medicine at Los Angeles, California, and Director of the United States Public Health Service-funded Research Training Program in Hematology and Hematologic Oncology at UCLA. He submitted an affidavit (R 190) in which he described his educational background and experience in teaching medical students and physicians. He is actively involved in the medical care of cancer patients. Dr. Ross listed his membership in several societies and councils which deal with cancer treatment and his membership on the editorial boards of several scientific publications.

Charles G. Moertel, M.D., is Chairman of the Department of Oncology at the Mayo Clinic, Director of the Mayo Comprehensive Cancer Center, and Professor of Medicine at the Mayo Medical School, Rochester, Minnesota. He described his educational background and experience

which have included serving as Editor of Cancer Yearbook, Associate Editor of Cancer Medicine, serving on the editorial board of Cancer, serving on a number of cancer committees, being involved in clinical research in pharmacology concerning cancer chemotherapy and clinical oncology, and publishing as author or co-author over 200 articles, abstracts, and editorials in recognized medical and scientific journals (R 186).

C. Chester Stock, Ph.D., is Vice President and Associate Director for Administrative and Academic Affairs of the Sloan-Kettering Institute for Cancer Research, New York, New York, and Professor Emeritus in Biochemistry at Cornell University. He described his educational background and experience as including serving as a member of several boards and societies concerned with cancer research and serving as Chief of the Division of Experimental Chemotherapy at Sloan-Kettering, where for many years he has had a major responsibility in the development of new drugs for the treatment of cancer (R 195).

Peter H. Wiernik, M.D., is Professor of Medicine, University of Maryland School of Medicine and Chief, Clinical Oncology Branch, National Cancer Institute, Baltimore Cancer Research Center. His educational background and experiences include duties as a reviewer for 9 medical-scientific journals, co-editor of 2 journals and the author or co-author of over 140 articles, editorials, and abstracts which have appeared in medical-scientific literature most of which deal directly with cancer (R 200).

Emil J. Freireich, M.D., is Head of the Department of Developmental Therapeutics and Professor of Medicine, and Chief, Division of Oncology at the University of Texas Medical School at Houston, Texas. His educational background and experience includes membership in several societies and committees concerned directly with cancer treatment. In addition, Dr. Freireich serves as a member of editorial boards of medical and scientific journals concerned with cancer research and, as such, reviews and evaluates scientific papers relating to the causes, treatments and control of cancer. He has published in internationally recognized journals over 250 articles. the majority of which have been concerned with cancer (R 390).

David T. Carr, M.D., is Professor of Medicine at the Mayo Medical School, Associate Director for Cancer Control and Community Relations of the Mayo Comprehensive Cancer Center, and a member of several professional societies and committees concerned with the treatment of cancer. His professional education and experience include the responsibility for a program of public education about cancer. His special interests are internal medicine and medical oncology, and he is regularly engaged in the medical management of cancer (R 176 (see also Tr. at 180–89)).

Several other experts submitted testimony in which they stated that, in their experience, Laetrile was not effective in the treatment of cancer. Their qualifications are set out in the following paragraphs:

James F. Holland, M.D., Professor and Chairman, Department of Neoplastic Diseases; Chief, Division of Medical Oncology; and Director, the Cancer Center, Mount Sinai School of Medicine, is a physician who has worked exclusively in cancer medication for over 26 years, specializing in cancer chemotherapy. He states, in his verified testimony, that he is thoroughly familiar with the action of drugs on cancer (R 396).

Carl M. Leventhal, M.D., is Deputy Director of the Bureau of Drugs, Food and Drug Administration. He holds the rank of Medical Director in the Commissioned Corps of the Public Health Service and is Assistant Professor of Neurology and Pathology at Georgetown University. As Deputy Director of the Bureau of Drugs, he participates in meetings in which the status of Laetrile is discussed and evaluated (R 184).

William A. Nolen, M.D., is Chief of Surgery at the Meeker County Hospital, Litchfield, Minnesota. He has served on the board of editors of the Minnesota State Medical Journal and has written a number of articles, editorials, and books on subjects of public health interest, including a book entitled Healing: A Doctor in Search of a Miracle, in which he describes his personal experience with a patient who lost her life because she chose Laetrile for treatment of an early cancer, thereby delaying conventional medical treatment (R 188).

Thomas H. Jukes, Ph.D., is Professor of Medical Physics and Research Biochemist at the University of California, Berkeley, California. He is a member of several professional societies, serves on the editorial boards of several scientific publications, has written three books and over 250 articles in scientific journals and has conducted research in the vitamin and cancer fields (R 416 (see also R 41)).

Robert S. K. Young, M.D., Ph.D., is a physician and has a doctorate in pharmacology. He serves as adjunct Assistant Professor of Pharamacology at Georgetown University School of Medicine and Dentistry and is group leader for the Oncologic Drug Class, Bureau of Drugs, Food and Drug Administration (R 201 (see also R 430)).

(b) Supporters of Laetrile.—In contrast to the great amount of evidence that experts in drug evaluation do not generally recognize Laetrile (or amygdalin) as effective, the evidence in the record to the contrary is meager. The Commissioner will outline qualifications of those persons who claim any modicum of training or experience in the area of drug evaluation whose support for the use of Laetrile appears in the record. The submissions of the following three physicians are discussed under II.A.1.c. above, "The 'Evidence' of Laetrile's Effectiveness":

John A. Richardson, M.D., stated in testimony (Tr. at 462-463) that he had been in general practice for 25 years and since 1971 had been engaged in nutritional using Laetrile, amygdalin, or vitamin B-17 and that he had treated, over the past 6 years, between 4,000 and 5,000 cancer patients. Dr. Richardson made no claim to special training or board certification in the area of oncology or of training or experience in the evaluation of the safety and effectiveness of drugs. Dr. Richardson's license to practice medicine has been revoked (R 183 at 13).

Philip E. Binzel, Jr., M.D., stated that he has been a family physician since 1955 and currently serves as scientific advisor to the Committee for Freedom of Choice (Tr. Ex. 13). He stated that he has treated over 400 patients in the last 3 years with a "metabolic therapy" for cancer (Tr. at 360-361). No special training in oncology or in the evaluation of drug safety or effectivenes is claimed.

Lawrence (Larry) Patton McDonald, M.D., who stated that he has been a urologist since 1963, is a former member of the State of Georgia Medical Education Board, and is currently a member of the U.S. House of Representatives, He stated that he was a member of several societies and associations, including the American Society of Clinical Urologists. the Southeastern Section of the American Urological Association, and the American Association of Physicians and Surgeons (R 509). No showing has been made that Dr. McDonald has a specific expertise in cancer treatment or in the evaluation of the safety and effectiveness of drugs.

Dr. Edward M. Arana, who spoke at the oral argument in this proceeding, identified himself only as a practicing dentist in Carmel, California (Tr. 472– A).

Ernst T. Krebs, Jr., spoke at oral argument in this proceeding (Tr. at 228-248). While he is referred to as Doctor, he did not complete his medical training and is a doctor only by virtue of an honorary degree. No special training in the area of cancer therapy or in the evaluation of safety and effectiveness of drugs has been shown for Mr. Krebs, Jr.

Paul Hart, M.D., spoke at oral argument. He described himself as having been employed at a pathology laboratory that dealt with the effects of radiation from atomic bombing in Japan and that was associated with Deaconess Hospital in Boston (Tr. at 457-58), and as being a diplomate of the National Board of Medical Examiners (Tr. at 457). He stated that he has an interest in "the Carl O. Simonton, M.D., psychotherapeutic approach to cancer therapy " " "" (Tr. at 458). While he indicated a personal respect for various Laetrile proponents, he did not give an opinion as to whether or not Laetrile is generally recognized by experts qualified to evaluate the safety and effectiveness of drugs as safe and effective for use in cancer therapy.

Dr. Mario Soto spoke at oral argument. He described himself as being former head of the chemotherapy departments of two different Mexican hospitals. He stated that he is an independent investigator for the National Cancer Institute and a conventional oncologist and chemotherapist. He is medical direc-

ico (Tr. at 478-479).

Dean Burk, Ph. D., who spoke at oral argument and provided a written submission (R 302), is a biochemist and is president of the Dean Burk Foundation, Inc. He stated at oral argument that he had spent 50 years in research on cancer and vitamins, 35 of which were with the National Cancer Institute (Tr. at 402). His position is that Laetrile is not a drug but a vitamin.

An affidavit of Chauncey D. Leake, Ph. D., prepared for another proceeding, was submitted to this record (R 302, Ex. The affidavit indicates that he is Senior Lecturer in Pharmacology at the University of California School of Medicine, San Francisco, His curriculum vitae showed that he has a history of teaching, participation in organizations dealing with medicine and pharmacology, editing of journals, and authorship of books and articles.

An affidavit of Charles Gurchot, also apparently prepared for another proceeding, was submitted as Exhibit L to R 302. His degree is in chemistry and physiology. Now semi-retired, he has taught pharmacology, biochemistry and chemistry at several schools of medicine and is a member of a number of scientific societies.

James Cason, Ph. D., submitted a statement in which he states that he has been a chemistry professor for some 35 years, has published over 100 research papers in scholarly journals and has served on the editorial boards of Organic Syntheses, and the Journal of Organic Chemistry (R 217), He is currently a professor of chemistry at the University of California, Berkely (id.). While he states his opinion that a diet high in nitrilosides leads to a low incidence of cancer, he gives no opinion as to whether or not Laetrile (or Amygdalin) is generally recognized by qualified experts to be a safe and effective cancer drug.

The statute requires that "general recognition" be among experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs (21 U.S.C. 321 (p) (1)). Few of the proponents of Laetrile who have made submissions to this record possess the necessary training and experience to qualify them as such experts. The Commissioner has, however, for purposes of completeness, considered as coming within the category of "experts" for this purpose, persons, including those listed above, who have exhibited even a small modicum of scientific experience or experience in the area in which they have offered submissions. The weight to be given the testimony of such persons, of course, must correspond to their expertise, cf. United States v. 1,048,000 Capsules, More or Less, Etc., 347 F. Supp. 768, 771 (S.D. Tex. 1972), aff'd 494 F.2d

1158 (5th Cir. 1974). The Commissioner concludes that, the lack of adequate and well-controlled clinical investigations published in the scientific literature aside, the record clearly demonstrates that the overwhelming majority of experts in the

tor of a Laetrile clinic in Tijuana, Mex- evaluation of the safety and effectiveness of drugs do not recognize Laetrile as effective. Even the proponents of Laetrile, while they may argue that the majority is wrong, could hardly be heard to argue this point. Laetrile is thus a new drug within the meaning of the act.

#### B. GENERAL RECOGNITION OF SAFETY

As noted above, for a drug to be exempt from new drug status under 21 U.S.C. 321(p) (1) it must be recognized by "experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs" to be both safe and effective under the conditions of its intended use. While lack of such recognition of Laetrile's effectiveness has already been shown, the Commissioner will discuss in addition the evidence on the question of general recognition of the drug's safety. He finds that such recognition does not exist.

#### 1. Lack of Adequate Testing

As has been discussed above, for a drug to be generally recognized as safe it must have accumulated at least the amount of evidence of safety that would be required for approval of a new drug application and that evidence must be generally available to the community of experts through publication in the scientific literature. In order for a new drug application for a drug to be approved. there must exist as to that drug "adequate tests by all methods reasonably applicable" that show the drug's safety (21 U.S.C. 355(d); cf. 21 CFR 314.111 (a) (1)),

An attempt to show that Laetrile had been proven by adequate testing to be safe for use in man was made in 1970 when the McNaughton Foundation submitted to FDA a notice of claimed investigational exemption for a new drug (IND) for Laetrile. The FDA terminated that exemption because of a lack of evidence of safety. Subsequent to the termination, the IND was referred to an Ad Hoc Committee of Oncology Con-sultants. The report of this committee is submitted with R 184 as Exhibit 2. This report states. "The Committee concurs with the action of the Commissioner in termination of IND 6734." Addressing the toxicity question, the Committee concluded (id. at 2), "Although it is often stated in the IND that amygdalin is non-toxic, data to demonstrate this lack of toxicity are absent, particularly with respect to the oral route."

The animal studies done to show Laetrile's safety did not justify use of the dosage suggested in the IND. "[T]he sponsor wishes to begin oral studies in patients at 2.95 mg/kg (oral); this is to be compared with a documented safe oral dose in dogs of 7.5 mg/kg daily for 6 months \* \* . On the basis of docu-mented data, if substantiated, then a proper starting dose that might be considered in man, would be 1/10 of 7.5 mg/kg or 0.75 mg/kg (oral). The proposed starting dose of 2.95 mg/kg is 1/100 of the oral acute LD in mice. It is considered to be dangerous to base the starting dose for a chronic (6 + weeks) study in man on a single dose

study in mice. It is also dangerous to initiate human studies while the nature of the toxicity has not been elucidated in large animal species. No documented data are presented in the IND to permit a higher starting dose" (id. at 3-4).

Dr. W. Sherwood Lawrence, Executive Secretary for the State of California Cancer Advisory Council states (R 183 at 17), "An extensive review of the world's scientific literature has been made by the Council. The evidence available for the determination of the recognition of safety of the compound is characterized by the lack of a body of scientifically sound information such that experts qualified by experience and training to make such determinations are unable to do so. In the absence of such a determination by qualified experts Laetrile (amygdalin) cannot be considered to be generally recognized as safe."

There is thus an absence of scientifically sound data upon which experts qualified by training and experience to evaluate the safety and effectiveness of drugs could base an opinion that Laetrile is safe for use in man. In the absence of such data the Commissioner must conclude that the safety of use of Laetrile in man has never been, and is not now, "generally recognized" by experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs.

## 2. Testimony of Experts

The Commissioner's conclusion of lack of general recognition among experts of Laetrile's safety is supported by testimony of such experts in the administrative record. (The qualifications of the experts whose statements are quoted have been discussed above.)

As one expert notes, while the toxicity of injected Laetrile has not been studied, there is evidence that amygdalin when ingested (eaten) is harmful to humans, evidence that has led to cautions on the part of Laetrile promoters themselves: In discussing the use of Laetriles, Dr. W. Sherwood Lawrence states (R 183 at 16), "There has never been a formal evaluation of the safety of these compounds to determine their safety. Although the proponents claim Laetriles are non-toxic, there are bonafide reports of clinical toxicity on oral intake (California Morbidity Reports, Ankara); and even fatality (Bitter Almond Poisoning, Zmedizinfrul). Furthermore the proponents themselves are aware of this toxicity as evidence by the proposed labeling submitted in an application for an exemption for an Investigational New Drug which warned: 'CAUTION: Laetrile is not to be taken orally. It is extremely toxic by this route of administration

\* \* \*.' There are no studies adequately showing the distribution, activity and metabolic fate of the parenterally administered compound. Chronic effects are not studied and reported in depth. There is cause for the concern as other similar beta-cyanogenetic compounds cause serious toxic effects when ingested on a chronic basis, e.g., Tropical Atataxic Neuropathy in Nigeria (Attachment 20). Effects of industrial poisoning from chronic low concentration exposure are well-known." (See also, for discussion of toxicity of Laetrile, section V.B.3. below, "Dangers of Ingestion of 'Vitamin B-17'".)

Dr. Robert C. Eyerly reports, "One of the unproven remedies for cancer management which the committee (on Unproven Methods of Cancer Management) investigated is the drug Laetrile. Other names for Laetrile include amygdalin and prunasin. These compounds are classified as cyanogenic glycosides. Cyanogenic glycosides, including amygdalin, are generally regarded as toxic substances, rather than as foods, because when they break down they liberate hydrogen cyanide, one of the most toxic substances known" (R 167 at § 8).

Dr. Robert S. K. Young states (R 430 at § 4, 5, and 7), "The FDA does not have authenticated or validated data on the toxicity of amygdalin in humans. It does not have scientific studies or the data upon which scientific studies might be constructed, in humans receiving amygdalin which would allow it to define the toxic effects of this drug. There are no such reports in the medical literature, Nevertheless, Dr. Nepier from Germany has reported that amygdalin causes hypotension and hemoglobinurea in humans. There have been reported cases of cyanide poisoning in humans who ate apricot kernels. The symptomatology includes dyspnea, cyanosis, vomiting, prostration, convulsions, stupor, and paralysis. Since these toxic effects are caused by the cyanide, which is a constituent part of amygdalin, amygdalin could cause the same toxic effects. Although it is possible that amygdalin can be given to humans in doses which are non-toxic in man, the drug is unsafe for use in humans. There is no scientific evidence that the drug can cure or is effective as a treatment for any human cancer." Dr. Young also states (id. at 15 2-3), "There is a difference between a drug's toxicity and a drug's safety. The toxic effects of a drug are those effects which are not beneficial to the person taking the drug, but are deleterious. The safety of a drug is defined by the context of its use and includes consideration of issues such as a disease for which the drug is intended, the alternative remedies which are available and their efficacy and safety, and the possible abuse of the drug by those who do not have the disease for which the drug is intended. Acute toxicity tests of amygdalin have been carried out in animals. It appears that relatively large quantities of amygdalin can be given parenterally. When given by the oral route, however, the toxicity of amygdalin is greatly increased (by a factor of approximately twenty-five times).

James F. Holland, M.D., indicated that he does not accept the theory of proponents of Laetrile that patients should be allowed to take Laetrile since, even if it is ineffective, it cannot hurt. He states (R 396 at 1), "It can hurt by interfering with patients' acceptance of indicated therapy in the mistaken and false hope of potential benefit from Laetrile. Delayed

operation, refused radiotherapy, skipped chemotherapy all risk an increasing cancer morbidity and mortality because of the premise that Laetrile is active. This is a very dangerous side effect, indeed."

In addressing whether or not Laetrile is safe, Dr. Carl M. Leventhal states (R. 184 at [ 19), "The question of whether Laetrile is now, or ever was, generally recognized as safe goes beyond the absence of any evidence indicating the lack of toxicity of the drug. The safety of a drug for human use depends, in large measure, on the therapeutic effectiveness of the particular drug. When patients forego effective forms of therapy and turn instead to worthless potions and nostrums, their disease may progress while effective therapies are foresaken. In the case of cancer, treatment with an ineffective drug will inevitably and inexorably lead to the patient's death. Seen in this light, an ineffective cancer drug is inherently unsafe and even lethal. because of the patient deaths which will necessarily ensue."

Dr. Harold J. Wallace, Jr., states (Tr. at 174) that: "The safety of the various forms of amygdalin has not been tested by the usual scientific methods of clinical pharmacology. There is evidence that the crude oral form can and has caused toxicity in humans and may cause death. There has been no documentation of the usual parameters that we require of drugs when used in a clinical situation. We don't have blood levels achieved, activation, clearance, metabolism, distribution or excretion of amygdalin compounds in man, as is usualy required in the preclinical and clinical evaluation testing of chemotherapeutic compounds or other drugs."

Dr. Frank Rauscher, a former Director of the National Cancer Institute (NCI), and currently associated with the American Cancer Society, said in his statement concerning Laetrile, while he was Director, NCI: "Assertions of the non-toxic nature of Laetrile have not been demonstrated in vigorous clinical studies. Even if this claim is true, there is no basis whatsoever for recommending the clinical use of any non-toxic agent if it cannot be expected to produce objective clinical benefits" (R 173, Att. "Statement Concerning Laetrile" at 2-3).

Dr. Thomas H. Jukes states (R 41 at 1): "Laetrile is not generally recognized by experts as safe. In the presence of the enzyme beta-glucosidase, Laetrile is hydrolized to glucose and mandelonitrile. Mandelonitrile readily decomposes with the liberation of hydrocyanic acid, which is extremely poisonous at low levels. The enzyme beta-glucosidase is widely distributed in materials of plant origin. The potential danger that laetrile may be decomposed with liberation of hydrocyanic acid makes it unsafe."

Dr. George J. Hill, II, after noting that ineffective remedies for cancer can lead to delay in treatment and "needless and untimely death," states: "In the absence of scientific evidence of effectiveness, no drug intended for use in treating cancer can be regarded as safe" (R 170 at ¶ 11),

Dr. Joseph F. Ross noted that that delay in cancer therapy because of use of Laetrile "results in loss of life, tragic suffering, and shortened life span" (R 190 at 8) and that use of the drug is "hazardous to the health of cancer patients" (id. at 7). He states: "Additionally, the use of 'Laetrile,' Vitamin B-17, 'Aprikern' and other such amygdalin containing materials when ingested presents a definite health hazard. The action of gastrointestinal fluids and enzymes releases the C=N (cyanide) radical from the compound and this may produce acute cyanide poisoning" (id. at 8).

Several additional experts submitted affidavits in which they state that neither amygdalin nor any other cyanogenic glycoside has ever been generally recognized as safe (Dr. Charles G. Moertel, R 186 at § 12; Dr. Jesse L. Steinfeld. R 194 at 5-6; Dr. Peter G. Wiernik, R 200 at § 16; Dr. Emil J. Freireich, R 390 at § 19; and others).

#### III. THE "GRANDFATHER" ISSUE

Because Laetrile is not generally recognized by qualified experts as safe and effective (see discussion above), it is subject to the Act as a "new drug" unless it is exempted from the statute's provisions under either of the two "grandfather clauses." These two exceptions, described in more detail below, limit the protection provided to the public with respect to certain drugs that fulfill a number of carefully defined conditions. Accordingly, the courts have recognized the narrowness of the exceptions. United States v. Allan Drug Corp., 357 F. 2d 713, 718 (10th Cir. 1966) cert. denied 385 U.S. 899 (1966): "Since we are dealing with a Grandfather Clause exception, we must construe it strictly against one who invokes it."; Durovic v. Richardson, supra, 479 F. 2d at 250 n. 6; United States v. An Article of Drug "Bentex Ulcerine" \* \*, 469 F. 2d 875, 878 (5th Cir. 1972), cert. denied 412 U.S. 938 (1973); United States v. 1,048,000 Capsules, More or Less, Etc., supra, 347 F. Supp. at 770.

The Court in Bentex Ulcerine held, 469 F. 2d at 878, that any party seeking to show that a drug comes within the grandfather exemptions "must prove every essential fact necessary for invocation of the exemption," Accordingly, the Commission concludes that Laetrile will not qualify for grandfather clause exemption unless each of the essential facts has been proved by evidence submitted in the record.

In the February 18, 1977 Federal Register notice initiating this proceeding, proponents of Laetrile were informed of their obligation to bring forth evidence that would support their claim that Laetrile qualifies for this exception. The notice set forth the provisions of a regulation (21 CFR 314.200(e)(2)) that detailed the format to which submissions directed to the grandfather clause exceptions should conform (42 FR 10069). Failure to submit formulas, labeling and

evidence of marketing in that format was stated to constitute a waiver of any contention that Laetrile was exempt from new drug provisions of the act. Failure to submit evidence in the format has resulted in such a waiver.

Despite the waiver, the Commissioner, in order to fully address the issue remanded by the courts, has culled the entire record for evidence that might arguably be relevant to the grandfather status of Laetrile. He has considered this evidence in determining whether the essential facts necessary to invoke the grandfather clause exemptions have been proved. Moreover, against the chance that it should later be held that those contending that Laetrile's use is illegal must prove the nonexistence of the essential facts necessary for the invocation of the grandfather clause exceptions, the Commissioner has considered the evidence in the record in light of this possibility.

The essential facts necessary to invoke the two exemptions are discussed, together with the evidence relevant to each, below. The Commissioner's conclusions on these issues may be summarized

as follows:

 Contentions that Lactrile qualifies for either grandfather clause exception are waived.

(2) Evidence presented does not prove the existence of each essential fact necessary to the invocation of either grandfather clause.

(3) While it is of course not possible to prove a negative with regard to the existence of each of the essential facts involved, the record assembled contains substantial evidence, constituting a clear preponderance of the evidence submitted, that these essential facts do not exist.

#### A. THE 1938 GRANDFATHER CLAUSE

To qualify for exemption from the definition of a new drug under the 1938 grandfather clause, it must be shown that the drug "at any time prior to the enactment of this chapter [1938] \* \* was subject to the Food and Drugs Act of June 30, 1906, as amended, and \* at such time its labeling contained the same representations concerning the conditions of its use; \* \* \* 21 U.S.C. 321(p)(1).

Thus, to qualify for this exemption, it must be proved that (1) the identical drug (2) bearing labeling containing the identical representations concerning the conditions of its use (3) was introduced into interstate commerce in the United States (or was manufactured in a Federal territory or the District of Columbia) after June 30, 1906 and prior to the enactment of the act in 1938. The exemption applies only to drugs whose labeling with respect to representations as to conditions of use has undergone no changes whatsoever from the labeling utilized prior to the passage of the 1938 act, and whose composition is completely identical to its composition prior to this passage. If any change in representations for conditions of use in labeling or any change in composition has occurred since the enactment of the 1938 act, such

change precludes the applicability of the 1938 exemption. The proof required would necessarily involve the production of quantitative formulas, labeling, and evidence of marketing both for the pre-1938 use and for the present use. While submissions to the administrative record contained a number of references to use of Laetrile or its predecessors before 1938, no proof was submitted to show that what was termed "Laetrile" or "amygdalin" as used before 1938 was the same drug which is now being marketed. Nor is there any indication whatever that the labeling of the various drugs claimed to have been marketed before 1938 contained representations concerning conditions of use which are identical to the representations associated with the presently marketed drug. It should be noted that the term "labeling" is defined in the act to include not only "all labels" but also, "other written, printed, or graphic matter (1) upon any article or any of its containers or wrappers, or (2) accompanying such article," 21 U.S.C. 321(m). This definition has been given a broad interpretation, see, e.g., Kordel v. United States, 335 U.S. 345, 347-50 (1948); United States v. Urbutcit, 335 U.S. 355, 357 (1948).

A number of submissions in the record referred to use of substances claimed to be related to Laetrile or amygdalin in ancient times. While each of these statements was hearsay unsupported by any sort of corroborating evidence and thus cannot be considered trustworthy, it is apparent that even if accepted at face value these claims would not justify invocation of the 1938 grandfather clause. Examples of such claims in the record include those found in a Mc-Naughton Foundation article entitled "Information for Physicians: Amygdalin, The Non-Toxic Analgesic" which cites use by the Chinese 3,500 years ago as well as use by the Greeks and the Romans (R 183, Att. 106 at ¶ 1). See also Hanson v. United States, 417 F. Supp. 30, 36 (D. Minn.) aff'd 540 F.2d 947 (8th Cir. 1976) (copy of opinion attached to R 173), noting that plaintiffs in that court action had introduced hearsay concerning the use of amygdalin by ancient Egyptians, evidence upon which the court did not rely; affidavit of Chauncey Leake, Ph.D .- recommendation of almonds for various purposes (not to cure cancer) by a first century Greek surgeon (R 302, Ex. K); statement by John J. O'Conner, Jr., in support of a Maryland State bill on Laetrile-bitter almond used by Chinese for the treatment of tumors 3,500 years ago: use by Greeks and by Romans (Tr. Ex. 4, Att. at 22).

Other submissions mentioned that amygdalin was prepared by two Frenchmen in 1830 and analyzed by two Germans shortly thereafter. (See R 183, Att. 10b at § 1; R 302, Ex. K; R 168, Att. at 345; R 80, Att.) No claims were made that these 19th century experimenters used amygdalin to treat cancer. There is a claim, however, that a Russian physician used amygdalin for that purpose in 1845. The reference is to a report in

the Gazette Medicale De Paris, Tome XIII, Samedi, Le 13 Septembre 1845, by Dr. Inosemtzeff. This article is referred to in a number of submissions. (See, e.g., R 259, Att. at 1.) It was not, however, itself submitted and in thus not part of the record available for analysis by the Commissioner. According to the description in the "Listing of Documents Relative to \* \* \* Laetrile" attached to R 259, a submission by Mr. Wynn Earl Westover, the author of the 1845 article was a professor of surgery at the Imperial University of Moscow, and his article described two cases of cancer apparently successfully controlled for 11 years and 3 years, respectively, by the use of amygdalin (for other references to this article, see R 260, Att. at 1; R 302, Ex. Lat ( 5).

As is obvious, this "evidence" relating to ancient and 19th century use is irrelevant to the 1938 grandfather clause issue for the following reasons: (1) It does not indicate that the drug was used in the United States after June 30, 1906 and before 1938. (2) It gives no indication that the drug used was the same as Laetrile. Most of the references in fact indicate that the substances used were either some extract of almonds, or, as in the case of the alleged Russian physician, simple amygdalin. (3) no suggestion that any drug was to be used in accordance with the indications now associated with treatment with Laetrile may be found in these references, Again, where the submissions go into detail concerning the historical uses of almonds or amygdalin, it is apparent that different conditions of use are involved.

A number of claims purporting to be relevant to the 1938 grandfather issue dealt with the appearance of almond extract in various Pharmacopeia, (See, e.g., Tr. at 250; R 302, Ex. K and Ex. L, 7 9-12 Tr. Ex. 9.) The opinion in Hanson v. United States, supra, indicates that plaintiffs in that case relied upon a listing of amygdalin in the Merck Index of 1896. The references in the Pharmacopeia involve in each case some sort of almond extract. There is no indication that that extract was to be used as an injection to cure, control, or prevent cancer. The references to the Pharmacopeia, as is the case with other general, unsupported references indicating use in cancer patients in previous centuries (see, e.g., R 509 at 2) are for the reasons detailed simply not probative of grandfather status.

Of more direct relevance to possible grandfather status is the information in the record regarding the work done with what was apparently Laetrile's predecessor by Dr. Ernst Krebs, Sr. A great deal of conflicting information regarding the dates of Dr. Krebs' work was submitted to the record. There are numerous instances in the record of statements that Dr. Krebs developed a drug related in some way to modern day Lactrile either in 1920 or shortly thereafter, while a new and allegedly nontoxic form of Laetrile was developed by Ernst T. Kerbs, Jr., in 1952 or in the early 1950's. See e.g., the American Cancer Society Committee on

Unproven Methods of Cancer Management's article on Lactrile (R 167, Ex. 2) : A Report on the Treatment of Cancer Beta-Cyanogenetic Glucosides ("Laetriles") by the Cancer Advisory Council, State of California (1963) at 2 (R 183, Att. 16). The latter report may be the genesis of the 1920 date, though the former indicates that it is reporting the date "According to" Dr. Krebs. At any rate, the 1920 date appears in or is alluded to in a number of submissions (see R 173, Att., "Questions Most Frequently Asked," and Att., "Laetrile: The Making of a Myth," FDA Consumer (Dec. 1976-Jan. 1977) at 6; R 184 at 1 6; Tr. at 272; Tr. at 41; R 250 at 2-3; R 170, Ex. 3 at 2104; R 258, Ex. 16; R 386; Tr. Ex. 10; R 183, Ex. 3 at 33). There may have been some basis for the original statement that Dr. Krebs had begun to work or had achieved results on a substance containing amygdalin in 1920 or in the early 1920's, but nothing has been submitted to indicate what that basis is. The apparent manner in which one submission has relied upon another on this question illustrates the undersirability of relying on hearsay accounts to prove a fact of this kind. None of these statements indicate, in any case, that the materials with which Dr. Krebs was experimenting were identical to, and were used under conditions indicated in labeling which were identical to, the composition and indications for present day Lactrile.

Michael L. Culbert, representing the Committee for Freedom of Choice in Cancer Therapy at the oral hearing, stated: "Dr. Krebs, Sr., both publicly and privately and in numerous different ways, has published not only results but some labels of material that goes back to the 1920's when Dr. Stohl in Switzerland and a number of Japanese scientists and Dr. Krebs, Sr., and others were working with the original extract" (Tr. at 41). If Mr. Culbert or his group have in their possession such publications, they have not submitted them.

A document submitted which would seem, questions of credibility aside, to be the most reliable on the question of the dates of Dr. Krebs' work and that of his son is an affidavit signed and sworn to by Dr. Krebs on April 28, 1965. This affidavit, taken by an FDA employee, appears as Exhibit 6 to R 184 and as attachment 13 to R 183. Since this affidavit is under oath and is by the person most likely to know of the dates in question, the Commissioner concludes that where the dates in the affidavit are different from those appearing elsewhere, chief reliance should be placed on the affidavit. In the relevant paragraphs, Dr. Krebs says:

2. In 1928, I made an extract from apricot kernels which I called Sarcarcinase. This extract contained Amygdalin and I-glucoaldase. When I injected this product into rats it was toxic and killed some of them.

3. In 1936, I changed the composition of the preparation resulting from the extract of apricot kernels so that the only active principle which remained was Amygdalin.

 During the period between 1936 and 1980, I perfected the purification process so that the purity of the Amygdalin rose from 66 percent in 1936 to 99.8 percent by 1960.

 In 1955, I began to lyophilize the Amygdalin and I have been lyophilizing it in its final form ever since when I produce it in my laboratories.

6. In 1949, my son, Ernst T. Krebs, Jr., gave the name Laetrile to the Amygdalin I was producing and I have used the name of Laetrile ever since that time for the final form of the Amygdalin which I produce.

7. As early as 1926 and up through 1962, I first began to ship and have done so continuously thereafter the Sarcarcinase extract (cf 2), then the amygdalin (cf 3), then the purified amygdalin (cf 4), then the purified and lyophilized amygdalin (5), and then since 1949 (cf 6) the latter under the name of Laetrile to persons in other States outside of the State of California and in many other countries. Many of these persons have reported their studies in scientific and medical journals and in private communications over several decades. The above shipments were for investigational use only.

As the dates cited by Dr. Krebs illustrate, the substance with which he experimented in the 1920's and 1930's was not the same substance as that which he was using in 1962. The pre-1938 use is, for that reason alone, not sufficient to qualify Laetrile for exemption from coverage of the act under the 1938 grandfather clause.

In another document upon which the Commissioner would ordinarily place reliance, a December 15, 1962 report by FDA inspectors describing their conversations with Dr. Krebs, Sr. (R 184, Ex. 5), Dr. Krebs is reported to have stated that "he began experimenting some 10 months ago with the extraction of Cyanogentic Glucoside from a mixture containing apricot pits. The purification of this glucoside was effected in the laboratories of Dr. Krebs and used in the treatment of his patients with, according to him, satisfactory results. This material assertedly liquefies malignant growths by the release of cyanide in the area. Injections are made around the area and the case of lung cancer injections are made in the apex of the trapezei," It may be that Dr. Krebs in his statement to the inspectors was speaking of his efforts to purify Amygdalin, referred to in paragraph 4 of his affidavit. On the second page of the inspectors' report, they indicate that "E. T. Krebs, Jr., stated that Dr. Harry Pincus Jacobson, M.D., was the first to use 'Laetrile' on humans and that this was in June 1952. Up to the present time (December 1952) he has used the product on approximately 14 cases."

This last quotation from the inspection report comports with statements elsewhere indicating that in 1952 Mr. Krebs, Jr., developed a new product, related to the products with which his father had been working, which he called Laetrile. While the failure of the affidavit of Dr. Krebs, Sr., to mention this "improvement" by Mr. Krebs, Jr., might lead one to question whether such an improvement had taken place, an article by the senior Krebs and Dr. Arthur Harris, copyright 1955, entitled, "The Treatment of Breast Cancer with Laetrile by

Iontophoresis" (R 183, Att. 7) at 23-24, states as follows:

In 1952 the senior author's blochemist son, Ernst T. Krebs, Jr., became interested in the preparation his father had used on cancer for so many years in his laboratory—the John Beard Memorial Poundation—he tore the drug apart and came to the conclusion that it was not only the glucoside but more particularly the cyanogenetic glucoside that had benefited cancer patients. He succeeded in separating the enzyme Emulsin from the cyanogenetic glucoside and advised their administration separately, in order to avoid the premature trigger-off of HCN from the chemical breakdown in the somatic (or normal) tissue, for this gas—HCN—was the active agent in destroying the cancer cell.

Again the senior author tried each purified preparation—the cyanogenetic glucoside and the enzyme Beta-glucosidase—separately. He administered the cyanogenetic glucoside parenterally (by injection) and followed it in fifteen minutes or so with an injection of the enzyme Beta-glucosidase. The cancer victims so treated tolerated both the drug and the enzyme excellently—and were immeasurably improved. Using the chemical and the enzyme separately, therefore, gave a high degree of safety as well as enhancing its cancerolytic effect.

Because this apricot kernel preparation was "Laevorotatory" (left-handed) to polarlezel light, and because Amygdalin was chemically a "mandelonitrile," Krebs, Jr., united the first and last syllables to invent a name for the new cancerleidal drug—LAETRILE.

Krebs Jr. uncovered the vital link that united Lastrile with the Unitarian or Trophoblastic Thesis of Cancer. In the previous chapter we emphasized the known fact that most malignant lesions are focally characterized by high concentrations of the enzyme Beta-glucuronidase-one of the main attributes common to both the trophoblast the cancer cell. The Betacell and glucuronidase of the animal kingdom is the equivalent of Emulsin in the vegetable kingdom, and Emulsin is the very enzyme that Krebs, Jr. separated from Amygdalin to make the empirical apricot formula safe for par-(injection) administration

This was an epochal milestone. Krebs, Jr., worked and experimented feverishly now: he was on the brink of cataclysmic discoverers, discoveries which, if substantiated, could mean victory over invincible Cancer!

He found that when he added Emulsin to Lastrile and incubated the mixture, hydrocyanic acid gas (HCN), one of the deadliest of gaseous poisons, was given off. He found that when he added animal Betaglucuronidase for prepared enzyme Betaglucusidase) to Lastrile and incubated the mixture, HCN was again given off. This, he knew then, was the reaction that took place within the body—IN THE CANCER CELL!

The need for Krebs, Jr.'s improvement was related to the lack of safety of his father's original preparation. As this article co-authored by the senior Krebs states, the original "preparation proved so toxic that he and his colleagues who were experimenting with him were reluctant to continue its use, except in direcircumstances" (id. at 23; cf R 167, Ex. 2; R 170, Ex. 3 at 2014; R 386, Att. at 2-3). Again, it is apparent that the improvement in the substance used by the Krebs (father and son) after 1938 made the drug different in composition, and

also in indications for use, from the drug which was-used before 1938. The 1952 date appears at several points in the record. (See e.g., R 167, Ex. 2; R 173, Att. "Laetrile: The Making of A Myth" supra; R 173, Att. 3; R 184 at 16; R 189; The at 272; R 250 at 2.3; The Ex. 10 at 3)

Tr. at 272; R 250 at 2-3; Tr. Ex. 10 at 3.1 In the submission of Mr. Wynn Earl Westover (R 259) (see also R 260), in a document entitled "Listing of Documents Relative to the Krebs Enzyme Extracts Later Known as Laetrile," at 13, there is a list of registrations of trademarks and issuances of letters patent allegedly granted for Sarcarcinase during the years 1930 through 1935. These documents have not all been submitted. Apparently submitted as representative of the patents is a patent specification from the Government of Ireland. As the above discussion indicates, the material covered in these patents is different from the material now known as Laetrile. A submission by Eric E. Conn, Professor of Biochemistry at the University of California, Davis (R 424) discusses this patent application and states that if the procedure set out in the patent is followed, "much or all of the amygdalin in the intact kernels may be destroyed by enzymes set in action by the grinding of the kernels to produce the extract. Most of the amygdalin remaining would be lost in processing. The extract produced "would be a mixture of glycerides, esters, certain pigments and other fatsoluable compounds that might or might not also contain a small amount (less than 5 percent) of the amygdalin remaining in the finely ground kernels." Compare the claims by Robert W. Bradford. President of the Committee for Freedom of Choice in Cancer Therapy, Inc., at the oral argument, that "Lactrile was first offered for sale in a trademark assigned in 1934, that it was sold at that time in three different forms: tablets, capsules, and injectables, (and that it) pharmaceutically was the same substance used today in cancer therapy. There can be no disagreement on this point" (Tr. at 346). Mr. Bradford submitted nothing to support his claim, which is at odds with the factual information submitted in the record and discussed above.

The Westover submission (R 259) also includes copies of a number of letters by various doctors who indicate that they have used the senior Krebs' formulation in the treatment of tumors or cancer. The letters bear dates in the 1930's. Mr. Westover's submission claims that there are a large number of other letters, not submitted, which are of generally the same type. Ernst T. Krebs, Jr., appearing at the oral argument, testified that amygdalin had been used as early as 1932. He indicated that the product then in use was labeled Sarcarcinase. (See Tr. at 232, 238, 246.) Sarcarcinase is the name of the product which was, according to Mr. Westover, granted a United States trade-mark in 1934. (See also Tr. at 446-48.) Two affidavits, apparently prepared for some court action, by Charles Gurchot, Ph.D., and Chauncey Leake, Ph.D., indicate that the affiants were involved in the

treatment of patients with Krebs, Sr.'s product in the 1930's (R 302, Ex. K and L).

The Commissioner has carefully surveyed the entire administrative record brought together for this proceeding. While it appears that Dr. Krebs, Sr., was utilizing some substance, which apparently had the trademark name of Sarcarcinase, before 1938, there is no evidence that that substance is identical in its formulation, or in its indications for use, to present day Laetrile (cf R 416 at ¶ 27(I) (7) (pg. 23)). In fact, as discussed above, the record is clear that the substance with which Dr. Krebs, Sr., experimented in the 1930's is different from the drug now being used by Laetrile proponents. The evidence suggests that the substance used by Dr. Krebs, Sr., in the 1930's was too toxic for general use. This toxicity appears to have been the reason for the work of Mr. Krebs, Jr., which, apparently, culminated in a substantial change in the formulation around 1952.

The Commissioner thus concludes that (1) no proof has been offered which shows that Laetrile was used and labeled before 1938 in a manner identical to its present use and labeling, and that (2) the evidence in the record demonstrates that present day Laetrile was not developed until after 1938. Thus, regardless of where the burden of proof lies in an administrative proceeding of this type, the Commissioner must conclude that Laetrile is not eligible for exemption from the protection to the public provided by the new drug provisions of the act because of use prior to 1938 involving identical labeling as to conditions of use.

#### B. THE 1962 GRANDFATHER CLAUSE

The provision that has been characterized as the "1962 grandfather clause" is set forth at section 107(c) (4) of Pub. L. 87-781 (note following 21 U.S.C. 321):

(4) In the case of any drug which, on [October 9, 1962] the day immediately preceding the enactment date, (A) was commercially used or sold in the United States, (B) was not a new drug as defined by section 201(p) of the basic Act as then in force, and (C) was not covered by an effective application under section 505 of the Act, the amendments to section 201(p) made by this Act shall not apply to such drug when intended solely for use under conditions prescribed, recommended, or suggested in labeling with respect to such drug on that day.

The "basic Act as then in force" read in relevant part as follows:

Sec. 201. For the purposes of this Act-

(p) The term "new drug" means-

(1) Any drug the composition of which is such that such drug is not generally recognized, among experts qualified by scientific training and experience to evaluate the safety of drugs, as safe for use under the conditions prescribed, recommended, or suggested in the labeling thereof, except that such a drug not so recognized shall not be deemed to be a "new drug" if at any time prior to the enactment of this Act it was subject to the Food a d Drugs Act of June 30, 1906, as amended, and 'at such time its labeling contained the same representations concerning the conditions of its use; or

(2) Any drug the composition of which is such that such drug, as a result of investigations to determine its safety for use under such conditions, has become so recognized, but which has not, otherwise than in such investigations, been used to a material extent or for a material time under such conditions.

The Commissioner has previously admitted that one of the conditions for 1962 grandfather status does exist-Laetrile (or amygdalin) was not covered by an effective NDA on October 9, 1962. (See 42 FR 10069). The Commissioner concludes on the basis of the information in the administrative record that Laetrile (or amygdalin) fails to meet all of the other requirements for qualifying for the 1962 grandfather clause exemption: (1) No showing has been made that a drug was used or sold on October 9, 1962 which has the same composition as a drug used or sold, or sought to be used or sold, today. (2) The record is clear that any use of drugs called "Laetrile" or "amygda-lin" in cancer therapy in 1962 was for investigational use. Investigational use can not provide the basis for exemption from new drug status on October 9, 1962 (see section 201(p) (2) of the act as then in force, set forth above) and of course does not constitute commercial use or sale. (3) No showing has been made that conditions of use recommended in labeling of a drug used or sold on October 9, 1962 are the same as those now recommended in labeling for the same drug. In fact, neither present labeling nor labeling in use on October 9, 1962 has been submitted to the record. Review of labeling available for "Laetrile" before and after October 9, 1962 reveals substantial changes in the prescribed conditions of use. (4) Laetrile (or amygdalin) was not generally recognized, by experts qualified by scientific training and experience to evaluate drug safety, as safe for use in cancer therapy on October 9, 1962.

As has been noted above, the Commissioner has concluded that the proponents of the proposition that Laetrile is exempt from the act because of "grandfather" status must bear the burden of proving that it is exempt. As has also been previously noted, however, the Commissioner has made a determination based on the alternative theory that the Government must prove that at least one of the essential facts leading to exemption does not exist. Proof of a negative is obviously more feasible in some instances than in others. The record leaves no doubt that use of Laetrile (or amygdalin) on October 9, 1962 was for investigational purposes and that use of the drug in cancer therapy was not "generally recognized" by qualified experts to be safe on that date. Since neither the present composition nor the present labeling of the drug appears in the record. it may not be conclusively determined that that composition and the conditions of use suggested in that labeling are not the same as the composition and suggested conditions of use of some drug in 1962. Nonetheless, the Commissioner is able to conclude, based upon substantial evidence which constitutes the preponNOTICES

39791

derance of the evidence in the record, that neither of those essential facts (i.e., identical formulation and identical conditions of use) do exist as to Laetrile (or amygdalin).

## 1. Composition

Clearly, for the 1982 grandfather clause to apply, the identical drug must have been used or sold in 1962 as is presently used or sold. The fact that a drug with the identical name (or names) was being used is irrelevant. Similarly, the fact that a drug in commercial use on October 9, 1962 has ingredients (such as amygdalin) in common with drugs in use today would not be sufficient under the grandfather clause if any of the ingredients of the drug, or the proportions in which those ingredients appeared in the drug, had changed (see generally 21 CFR 310.3(h)). Even a change in an inactive ingredient will make a drug a "new drug," (see 21 CFR 310.3(h) (1): United States v. Article of Drug "Entrol-C Medicated," 513 F.2d 1127, 1130 n. 7 (9th Cir. 1975)).

The discussion earlier in this opinion of the identity of drugs characterized at different times as "amygdalin" or as "Lactrile" illustrates the wide variation in composition of these drugs. There is no evidence in the record of the present formulation of Laetrile or of amygdalin medication. Neither is there evidence of the composition of such a drug on October 9, 1962. The Commissioner thus concludes that there has been no showing that Laetrile or amygdalin as presently constituted was in use on October 9, 1962. The Commissioner also concludes, based upon the evidence of wide variation in the drugs' composition, both before and after 1962 (discussed below), that the 1962 versions and the versions of the drugs currently in use are not identical.

It should be noted that the Commissioner's decision on this point is in accord with a statement by Andrew Mc-Naughton of the pro-Lactrile McNaughton Foundation, discussed earlier, that data on Laetrile obtained prior to 1968 are frequently not reliable because of the variability in composition of early preparations (R 173, Att., "Report of Ad Hoc Committee of Oncology Consultants"). Other evidence on the question of composition of the drugs consists of (1) analyses done of Laetrile products and (2) representations made as to the products' composition. (The discussion of the 1938 grandfather issue sufficiently catalogues and disposes of claims that drugs similar to Laetrile or amygdalin were marketed prior to 1938, and this section will thus discuss evidence postdating 1938.)

Analyses. As discussed previously, the results of analyses of drugs called "Laetrile" have often been at variance with their labeled composition. Analyses by Canadian investigators, reported in 1965, found that two versions of the drug, one manufactured in the United States and one manufactured in Canada, had different compositions. The American version contained 98±2 percent amygda-

lin plus .5 percent phenol. The Canadian version contained 87±2 percent amygdalin, 5 percent di-isopropylammonium iodide, and 8±2 percent sucrose (R 189, Att., "Laetrile: A Study of its Physiochemical and Biochemical Properties" at 1059).

Analyses done in 1961 and 1962 for the California Cancer Advisory Council of samples of Laetrile from various different sources-samples obtained in 1951 and 1953, samples obtained from Hale Laboratories, and samples obtained from Dr. Krebs, Sr.-also showed a variation in composition among the drugs (R 183, Att. 16 at 27), Similarities to commercial amygdalin were revealed in some tests. (See, e.g., R 183, Att. 16 at 28 and App. 8.) "The old Lastrile (1951 and 1953) was similar but not identical to amygdalin in the (infra-red examination), while the new Laetrile exhibited certain similarities and certain dissimilarities to both the old Laetrile and to amygdalin" (R 183, Att. 16, App. 8 at 1). Some samples were found to contain inorganic iodine; others did not contain that substance, (See, generally, R 183, Att. 16, App. 7 and 8.)

Claims. A new drug application submitted to FDA by Ernst T. Krebs, Jr., on October 3, 1962, lists the composition of Laetrile as:

L-mandelonitrite-diglucoside [amyg-dalin] 1,000 mg.

N, N-diisopropylammonium iodide 50 mg.

Inactive saccharides, principally sucrose 176-250 mg.

The drug was to be reconstituted with a sterile isotonic solution (R 201, Ex. B at 101-102).

Dr. Krebs, Sr., in his 1965 affidavit, stated that his preparation contained amygdalin as its only active principle and that that amygdalin, by 1960 at least, was lyophilized and 99.8 percent pure. (See R 183, Att. 13.) It is not clear whether other, inactive, ingredients were a part of the drug he prepared.

A 1953 letter from Ernst T. Krebs, Jr., to California medical authorities states that he was forwarding to Dr. Macdonald "samples of biosynthetically degraded amygdalin in which one dextrose was removed by prunasin and the resulting compound, in the presence of platinum black, was oxidized to the corresponding glucuronoside" (R 183, Att.

In 1965, FDA investigators obtained examples of labeling utilized by the senior Krebs' laboratory. The labeling indicates that Laetrile is "cyanide glucoside type amygdalin." Additional labeling, a pamphlet entitled "Laetrile: Directions for the Administration of Laetrile," states that the drug is to be reconstituted with water, a non-isotonic solution (see R 201 at § 16a and Ex. C).

Mr. Krebs, Jr., in a 1970 article in the Journal of Applied Nutrition (R 183, Att. 10c) in which he explained his theory that Laetrile and similar substances make up Vitamin B-17, suggested the drug use of a substance clearly different from all other Laetrile drugs previously in use. While what had been used pre-

viously had apparently been a manufactured drug containing either the "Laetrile" of his own formulation or amygdalin in a more or less purified form, in this article he advised that "one gram of defatted apricot seed or kernel carries about 30 milligrams of nitriloside. Six or seven teaspoonsful will supply what our clinical investigators consider an adequate oral dose-one gram. It is best that the (beta) -glucosidase enzyme be completely heat inactivated in such material" (id. at 84). As discussed previously, this advocacy of the use of apricot kernels rather than a manufactured drug represents a change in the formulation of the product which is of particular importance because of the danger of toxicity associated with oral ingestion of apricot kernels.

The Commissioner concludes that drugs called variously Laetrile and amygdalin have no set composition, their makeup varies depending upon the manufacturer and the time of manufacture. It thus appears that any drug in use on October 9, 1962 was different in composition from Laetrile as used, or proposed to be used, today.

#### 2. Investigational Use

The record is clear that use of Laetrile (amygdalin) on October 9, 1962 was for investigational, not commercial, pur-poses. This fact is borne out by legal documents concerned with each of the two major figures in its development-Dr. Ernst T. Krebs, Sr., and his son, Ernst T. Krebs, Jr.—and by other information in the record. Much of the evidence relating to the 1962 grand-father issue, like that relating to the 1938 grandfather issue, is not of the type which would be considered reliable evidence in a court of law. In many cases the "evidence" consists of hearsay which is not substantiated by any documentation. The record does contain, however, a sworn affidavit of Ernst T. Krebs, Sr. In this affidavit (R 183, Att. 13 at 57) Dr. Krebs describes his shipment in interstate commerce of various versions of his cancer cure, including amygdalin which he stated to have been sold after 1949 under the name of Laetrile, "up through 1962." Dr. Krebs states, "The above shipments were for investigational use only."

Ernst T. Krebs, Jr., and the John Beard Memorial Foundation were convicted in 1962, upon pleas of guilty, of charges of introducing and delivering for introduction in interstate commerce a new drug without an approved new drug application (R 183, Att. 16, App. 17). The drug involved there was another unproved remedy, called by Mr. Krebs "pangamic acid" or "Vitamin B-15." Sentence of imprisonment on those charges was suspended and the defendants placed on probation for 3 years on the condition that they not manu-facture, sell, offer for sale, hold for sale, or deliver or give away any "new drug." Mr. Krebs, Jr., obtained a special order which allowed him to ship 400 vials of Laetrile to the McNaughton Foundation in Canada, "for investigational use" pro-

vided the Canadian Food and Drug Directorate acquiesced in that shipment. In a supplemental order of June 28, 1962, Ernst T. Krebs, Jr., was permitted, under certain detailed conditions, to deliver Laetrile to experts qualified by scientific training and experience to investigate the safety of drugs. The drug was not to be administered to any patient except one with extensive malignancy who was receiving Laetrile under Kreb's direction as of June 15, 1962. Thus, if Laetrile were in commercial use on October 9, 1962, and if the Lactrile involved were supplied by Mr. Krebs, Jr., he was in violation of this court order. Copies of the court papers involved in this criminal prosecution are found at appendix 17 to attachment 16 to R 183.

Use of a drug is investigational, as contrasted with commercial, when that use is for the purpose of determining whether, or demonstrating that, the drug in question is safe and effective. The record contains no evidence to suggest that, contrary to the affidavit of Dr. Krebs, Sr., or to the court order binding Ernst T. Krebs, Jr., Laetrile (or amygdalin) was being used on October 9, 1962 for other than investigational uses.

"In 1953 the Cancer Commission of the California Medical Association investigated the claims made for the use of Laetrile in cancer treatment and condemned its use" (R 168, Att. "Vitamin Fraud"). The activities that led to the Medical Association action were apparently based upon Dr. Krebs' use of Laetrile. As shown by labeling collected during a 1952 FDA inspection, the Laetrile then in use was labeled, "Caution: New Drug limited by Federal Law to investigational use." (See R 184, Ex. 5.)

A submission to the record which contains much information about the use of Laetrile at about the crucial date of October 9, 1962 is the 1963 report of the California State Cancer Advisory Council entitled "Treatment of Cancer with Beta-cyanogenetic Glucosides ('Laetriles')" (R 183, Att. 16). While the Council concluded that use of Laetrile was not warranted in any context, its report does not contradict Krebs' claims that use was investigational at that

Other references to the use of Laetrile prior to 1962 do not specify whether or not the use mentioned was investigational, see R 183, Atts. 5, 6; R 307 at 1; R 64; R 174, Ex. 2; Tr. at 81-82.

A pamphlet entitled Information for Physicians, Amygdalin The Non-Toxic Analgesic provides information about what it states to be the experience of various doctors around the world in administering amygdalin to patients. Some of the statements indicate use by doctors before 1962—that use appears to be investigational and was not, with the exception of 10 cases reported by a New Jersey doctor, in the United States (R 183, Att. 10 (b)).

The article by Levi et al., "Laetrile: A study of its Physiochemical and Biochemical Properties," discussed above, refers to Laetrile as "a drug manufactured and distributed until recently for

clinical trial in Canada and the United States to determine its value as a palliative in cancer therapy" (R 189, Att. at 1057).

In 1970 the McNaughton Foundation submitted an IND for Laetrile, which was disapproved by FDA (R 184 at ¶ 9). An IND is a notice, filed by persons interested in the development of a drug product, which seeks permission to distribute an unapproved new drug for the purpose of conducting clinical investigations of it in humans. Such clinical investigations must be completed to form the basis for an NDA for the drug. The fact that Laetrile's proponents were still seeking to investigate its use in 1970 is additional evidence that any use of the drug on October 9, 1962 was investigational.

# 3. Conditions of Use in Labeling

In order to qualify for the 1962 grandfather clause, Laetrile (or amygdalin) would need to "be intended solely for use under conditions prescribed, recommended, or suggested in labeling with respect to such drug on" October 9, 1962 (section 107(c) (4) of Pub. L. 87-781). Conditions of use include, among other things, what the drug is recommended for, how it is to be administered, and in what quantities it is to be administered. Under the statute, any change in those conditions from October 9, 1962 to the present disqualifies the drug from exemption. (See United States v. Allan Drug Corp. supra.) Here, no submission contains either labeling now in use or proposed for use, or labeling used on October 9, 1962.

Since no labeling in use on October 9, 1962 has been submitted, the indications found in labeling in use in years prior to that time will be discussed as illustrative of the variation in proposed conditions of use apparent in the record. The Commissioner will then review labeling from dates after October 9, 1962. As the following discussion demonstrates, not only do the proposed conditions for use of Laetrile (amygdalin) vary from before October 9, 1962 to after that date, no two sets of labeling propose the same conditions.

# Before October 9, 1962

A new drug application (NDA) submitted to FDA by the John Beard Memorial Foundation and Ernst T. Krebs, Jr., on October 3, 1962 indicates that Laetrile was a lyophilized water-soluble powder for use in the palliation of human cancer, Excerpts from the NDA, attached as Exhibit B to R 201, provide information concerning its intended uses: It was to be administered by injections of 1 gram each, which were to be either every day or every second day and either intravenously or intramuscularly. Intravenous administration was stated to be preferred. The average administration was stated to be every other day for a total of 10 injections. Apparently, a total of 20 grams of Laetrile were expected to be administered. Dosages, frequency, and route of administration are described as varying widely with each individual case. The

application indicated that Laetrile often produces a temporary hypotensive reaction shortly after injection, especially in hypertensive patients. Laetrile is not indicated for use to the exclusion of surgery, radiation, or other chemotherapeutic substances where those find any indication.

The proposed labeling in the NDA is, of course, not an example of labeling in commercial use at the time of the NDA's submission. The NDA does, however, state the indications which Mr. Krebs, Jr., thought to be most appropriate for the use of Laetrile at that time. Thus, if Laetrile had been commercially used at that time, it is reasonable to believe that the indications proposed in the NDA would be the ones proposed in any labeling used for such a commercial product.

An article by Dr. Krebs, Sr., and Dr. Arthur T. Harris, entitled "The Treatment of Breast Cancer With Laetrile By Iontophoresis" (copyright 1955 by the John Beard Memorial Foundation) (R 183, Att. 7) proposes three different methods of utilizing Laetrile. At page 30, the three main methods of administering Laetrile and its auxiliary Beta-glucosidase" are described as: (1) Parenteral administration (injection into the muscle), (2) iontophoresis, discussed below, and (3) tamponade.

Perhaps the most bizarre of the proposed methods of administration for Laetrile is "Iontophoresis". This new procedure developed by the senior Krebs for treatment of cancers "especially in the breast" is described as "infinitely more effective and thorough." The procedure is described as follows:

It is to force by galvanic current the Lacirile through the skin and into as well as between the individual cancer cells. The apparatus we use is a simple galvanic instrument (or, preferably, one of the modern instruments with resistors instead of tubes). The positive pole lead goes to the tumor site-the breastnegative to the back. The solution of Lactrile is soaked in gauze and covered by a block tin electrode, then positioned firmly over the tumor. The negative pad, well moistened, is positioned on the back, and the current turned on. Slowly the amperage is raised to 10 milliamperes then 15, never more than 20 except in a very thick chest wall. In fifteen to thirty minutes, depending on the size of the growth, the pad has become almost dry Laetrile has been driven into-not around-the cancer cells (id. at 26-27).

The action of iontophoresis is described in more detail on pages 30-31. Apparently it is expected that the iontophoresis therapy will liquify the tumor mass, and a physician will thus be able to draw out, with an aspirating needle, the "cancer-juice" before administering the next iontophoresis treatment (id. at 32). Iontophoresis therapy involves administration every 2 to 5 days (id. at 31).

This article, which has been quoted and referred to previously, explains some of the history and theories of Laetrile's use. The article promotes the "Howard Beard Anthrone Test" for the diagnosis of cancer (id. at 34). This test involves analysis of the urine of the patients (see id. at 16-19). The authors recommend against biopsies to determine whether tumors are malignant (id. at 27-28). The

39793 NOTICES

authors state their opposition to surgery prior to "control" of the cancer by Laetrile (id. at 35).

Laetrile (amygdalin) is apparently currently in use as an oral medication. Nothing in the record, other than conclusory statements of the most general kind, indicates that any version of the drug was in use as an oral medication on October 9, 1962. The only statement that such a drug was ever used orally before that date which purports to be based on first-hand knowledge is the statement of Charles Gurchot, Ph.D. (R 302, Ex. L at ( 8), that between 1933 and 1934 a Dr. Lewis administered amygdalin orally as well as intramuscularly and intravenously. Dr. Gurchot states that use in California, in which he participated between 1934 and 1945, involved administration intramuscularly and intravenously (id. at ¶ 14). As discussed above in the section on the 1938 grandfather clause, the "amygdalin" Dr. Gurchot states he was involved in using is different from that used at later dates.

### After October 9, 1982

Variations in the conditions for use of Lactrile (or amygdalin) proposed in its labeling continued after the critical October 9, 1962 date. In 1965, an FDA inspection of Krebs' Laboratories produced labeling for Laetrile which suggested a new set of conditions for its use (see, generally, R. 201, Ex. C.). The laebls on the packages of the drug stated "For raising hemoglobin index and red count[:1 relieves pain due to malig-nancy." (Similar labels were obtained by California State health officials in 1971 (R 183, Att. 9).)

In a pamphlet published by Krebs' Laboratories, obtained in the 1965 inspection, injections at various sites were indicated for various types of cancerbrachial vein for cancer of lungs; brachial vein and innominate artery for breast cancer; external carotid or one of its branches for cancer of the neck. thyroid, face, and temple area; brachial vein for cancer of liver, gastro-intestinal tract and the spleen; the vault of the vagina, the abdominal aorta, or the internal iliac arteries for cancer of the uterus and ovaries; the scrotal sac for cancer of the prostate and testicle (R

201, Ex. C. II)

Two pamphlets obtained in the 1965 inspection are in fact inconsistent with each other in some instances, though the similarities in printing style indicate that they were printed at about the same time. One states the dose of Laetrile to be administered to be "(g) enerally speaking 10 mgs, per pound of patient's weight, with "occasionally" 15 mgs. per pound and "very rarely" 20 mgs. per pound (id.). The second states that: The usual daily dose of Laetrile now is 20 mgs, of the glucoside Amygdalin for every pound of the patient's weight, or even twice this, particularly in bone cancer." Three gms. are recommended for a 150-pound person and 4 gms. for a 175pound patient, i.e., over 20 mgs. per pound (id. at Ex. C. III). (While no labeling indicating such conditions was

submitted, it should be noted that Dr. Binzel, at oral argument, talked of injections of from 9 to 15 grams of amygdalin at one time (Tr. at 363). The page proofs of Dr. Richardson's book indicate that he uses intravenous injections of "6-9 gms, or more" of Laetrile during the first month of treatment with intravenous or intramuscular injections of 3 grams thereafter (Tr. Ex. 1 at 124).)

More important, however, are the difference between the conditions recommended in the labeling collected in 1965 and those in that submitted with the 1962 NDA. In the 1962 NDA, Laetrile was to palliate, not to cure; in the 1965 labeling it is stated: "Laetrile does not palliate, it acts chemically to kill the cancer cells selectively without injury to the normal tissues of the body" (R 201, Ex. C). While the 1962 NDA stated that Laetrile was not indicated to the exclusion of other recognized cancer therapies, the labeling collected in 1965 states: "The less drugs and medicines given, during the Laetrile treatment the better. What should be especially avoided is sulphur and sulphur drugs and other cancer therapies, \* (emphasis added) (id.). Even more frightening to those who are concerned that utilization of therapies of proven effectiveness will be delayed until too late because of use of Laetrile is the statement in the pamphlet in use in 1965 that: "Being harmless \* \* \* Laetrile should be used first instead of last as generally has been done when everything else has been tried and hope is gone" (id.). An affidavit submitted by Dr. Robert S. K. Young describes the medical importance of the numerous variations between the 1962 labeling and that of 1965 (see ¶ 11 of R 201).

The labeling discussed, which bears the name of Krebs Laboratories and of Dr. Krebs, Sr., appears as Ex. C to R 201. It should be noted that there is no copyright or other date on the labeling that was found in Dr. Krebs' establishment in 1965. One of the phamplets, that which contains some of the statements quoted above, is described as a "pre-1963 pamphlet" in the affidavit of Dr. Sherwood Lawrence (R 183 at 4). It appears as attachment 8 to that affidavit.

A pamphlet published by the Mc-Naughton Foundation suggests intravenous dosages of amygdalin of from 3 to 6 grams a day administered over a 24-

hour period (R 183, Ex. 10b at 5). That pamphlet, which cites references dated May 11, 1970 and thus must have been published thereafter, described the use of amygdalin as an analygesic, yet also

indicates that the drug inhibits the growth of malignancies (id. at 1).

The record contains labeling for Laetrile (or amygdalin), which was in use after October 9, 1962, which clearly rec-ommends oral administration of the drug. See R 183, Att. 10a-capsules, 400 mg.; R 183 Att. 4c-capsules, 400 and 500 mg.; R 183, Att. 10b-amygdalin tablets which may be broken up and added to drinking water or food (1/2 to 2 grams per day recommended); R 183, Att. 10d-"Magydalin" capsules with 500 mgs. of "pure crystalline LAETRILE (amyg-dalin)".

While in 1962 Laetrile was proposed in the NDA as a palliative, the labeling in the record makes clear that it has been touted since that time as a treatment for cancer (see R 183, Att. 10a.; see also R 201, Ex. C, discussed above), Mr. Krebs, Jr., claims Vitamin B-17, which may be or may contain Laetrile, to be "antineoplastic" and to be instrumental in therapy" for cancer (Journal of Applied Nutrition; Vol. 22, "The Nitrilosides (Vitamin B-17)—Their Nature, Occurrence and Metabolic Significance (Antineoplastic Vitamin B-17)," at 75, 81 (R. 183, Att. 10c).

In a transcript dated November 18, 1974, prepared by FDA, of a film entitled "World Without Cancer", produced by the proponents of the use of Laetrile, the claim is made that 15 percent of persons with advanced metastasized cancer will be saved by "vitamin therapy," which from the context includes vitamin B-17 (Lactrile). The film claims that, of those with cancer diagnosed early, at least 80 percent will be saved by vitamin therapy. Of those who are healthy with no clinical evidence of cancer, the film's narrator states that close to 100 percent can expect to be free from cancer as long as they utilize vitamin B-17. The use of the term "vitamin B-17" indicates that the film was made after 1962, since Laetrile was not claimed to be a "vitamin" until after that time (see, generally, exhibit 2 to R 198).

As discussed above, to qualify for exemption from the "new drug" definition of 21 U.S.C. 321 (p) pursuant to the "1962 grandfather clause," the proponents of Laetrile (or amygdalin) would need to show among other things that the drug in question is now "intended solely for use under conditions prescribed, recommended, or suggested in labeling with respect to such drug on" October 9, 1962. No evidence in the record shows either that the drug was used or that any conditions of use were recommended for it on that date. Evidence in the record indicates that conditions of use recommended prior to the critical date not only conflict with each other, but also conflict with recommendations after that date, which themselves conflict with each other. The Commissioner concludes, on the basis of the evidence in the record, that Laetrile as now known is not intended solely for use under conditions recommended in labeling on October 9.

### 4. Lack of General Recognition of Safety in 1962

As discussed above, a drug could not escape new drug status under the "1962 grandfather clause" if it were a "new drug" on October 9, 1962. To have been exempted from new drug status on that date, Laetrile (or amygdalin) would have to have been "generally recognized among experts qualified by scientific training and experience to evaluate the safety of drugs, as safe for use under the conditions prescribed, recommended, or suggested in the labeling thereof" (21 U.S.C. 321(p)(1) (1962)), and that general recognition would have to be based upon use other than investigational use

(21 U.S.C. 321 (p) (2) (1962)). The Commissioner has elsewhere discussed the evidence that demonstrates that Lactrile and amygdalin (to the extent that they are different) are not now generally recognized by qualified experts as safe for use under the conditions prescribed, recommended or suggested in their labeling. While the present lack of general recognition of the substances would not necessarily demonstrate that they were not so generally recognized in 1962, that fact does provide evidence of the earlier lack of recognition.

The evidence in the record provides a number of independent grounds upon which the Commissioner concludes that Laetrile (or amygdalin) was not generally recognized by experts in drug safety evaluation as safe on October 9. 1962. That conclusion is supported (1) by the proven lack of a number of prerequisites to such general recognition: lack of knowledge among such experts generally of Laetrile's use, of Laetrile's formulation, and of the proposed conditions of Laetrile's use; lack of data published in the scientific literature supporting Laetrile's safety as a cancer drug; and lack of scientific testing sufficient to show safety: (2) by statements in the record by experts in the evaluation of the drug safety that Laetrile was not generally recognized as safe as a cancer drug by themselves and their peers on October 9, 1962; and (3) by abundant evidence that Lastrile was not generally recognized by appropriately qualified experts to be effective in cancer therapy on October 9, 1962. The Commissioner concludes that the showing in the record on each of these points is itself sufficient to demonstrate that Laetrile (or amygdalin) was a new drug in 1962.

(a) The Prerequisites. (i) Lack of General Knowledge of Use.—A number of submissions to the administrative record indicated that the use, and the details of the use, of Lactrille or amygdalin were simply not generally known to the community of experts in the safety evaluation of drugs on October 9, 1962. Thus, there could not be any sort of "general" recognition of the substances' safety in 1962. (See, e.g., the oral testimony of Dr. Rhoads, the national chairman of the National Cancer Advisory Board (Tr. at 110–111; Tr. Ex. 5); oral testimony of Dr. Carr, professor in medicine at the Mayo Medical School (Tr. at 181).)

(ii) Lack of General Knowledge of formulation.—The variability in, and uncertainty about, the composition of the drug in use at that date (discussed in detail above) means that "general recognition" of the drug's safety by experts in drug safety evaluation would be impossible. The fact was recognized in Durovic v. Richardson, supra, 479 F. 2d at 251, in which another unproven cancer remedy was ruled not to be exempted from regulation by the 1962 grandfather clause.

(iii) Lack of General Knowledge of Conditions of Use Suggested.—Equally important, the variation in and uncertainty about the conditions of use suggested in the labeling of Lactrile (or amygdalin) on October 9, 1962, also discussed in detail above, means that such general recognition could not have existed. The law as of that date is clear that general recognition must be of safety "for use under the conditions prescribed, recommended, or suggested in the labeling" of the drug (21 U.S.C. 321(p) (1) (1962)). If experts throughout the country could not have known of those conditions of use, recognition of safety by them could not have existed.

(iv) Lack of Safety Data in Scientific Literature.—The existence of published data available in the scientific literature on the safety of a drug is a prerequisite to general recognition by experts of that drug's safety within the meaning of 21 U.S.C. 321(p). Weinberger v. Bentex Pharmaceuticals, Inc., supra, 412 U.S. at 652; see United States v. 41 Cases, More or Less, 420 F. 2d 1126, 1130 (5th Cir. 1970); United States v. 1,048,000 Capsules, More or Less, supra, 347 F. Supp. at 771. The record lacks any reference to any such published data available to experts on October 9, 1962.

In fact, the record demonstrates that, while data showing the lack of Laetrile's effectiveness have been published in the scientific literature, data upon which an expert in the evaluation of drug safety could make a judgment that Laetrile was safe for use in cancer therapy do not exist in the scientific literature available to ex-

perts generally even today.

ty) Lack of Showing of Safety by Adequate Testing.-As noted in the sections of this opinion dealing with the new drug issue, the Supreme Court has held in Weinberger v. Bentex Pharmaceuticals, Inc., supra that "general recognition," as those terms are used in 21 U.S.C. 321(p), requires the same type of showing of safety and efficacy necessary for approval of an NDA pursuant to 21 U.S.C. 355(d). For approval of an NDA prior to October 9, 1962, the application was required to contain "adequate tests by all methods reasonably applicable to show whether or not such drug is safe" for its intended uses, and those tests were required to in fact show that the drug was safe (21 U.S. 355(d) (1962)). It appears from the record that no such tests existed for Laetrile (or amygdalin) on October 9, 1962. Were there a question about the lack of such studies, that question could be resolved by the fact that, at approximately the time in question, NDA's for Lactrile and for a combination of Lactrile and iodomine were submitted to the FDA. Both applications were declared to be incomplete because of the lack of required data to show safety and effectiveness. (See, generally, the letter from John L. Harvey, FDA Deputy Commissioner, to K. F. Ernst, M.D., April 30, 1963 (R 183, Att. 16, App. 18).)

(b) Statements by Experts. Even setting aside the above important prerequisites to general recognition, the evidence in the record that Lactrile was not generally recognized as safe by experts in the evaluation of drug safety on October 9, 1962 is extremely strong.

The plethora of statements of experts in drug evaluation that Laetrile (or amygdalin) is not now generally recog-

nized as safe is discussed elsewhere. Some of the experts focused upon the October 9, 1962 date. (See affidavit of Dr. Emil J. Freireich; "Neither amygdalin nor any other cyanogenic glycoside was generally recognized as safe for any (use in the treatment of cancer or prophylaxis against cancer or relief of pain associated with cancer, or for any medical use) on October 10, 1962" (R 390 at § 19; accord affidavit of Dr. Daniel T. Carr. (R 178 at § 15).) For a similar statement that Laetrile was not generally recognized as safe by appropriately qualified experts in 1962, see affidavit of Dr. Carl M. Leventhal (R 184 at § 13).

Even more compelling evidence on this question can be gleaned from statements of experts in drug safety evaluation made near the October 9, 1962 date. Fortunately, at just about that date the State of California Cancer Advisory Counsel was polling just that type of expert concerning Laetrile (R 183, Att. 16 at 37-38). (Since the Krebs Laboratory was located in California, it would seem that experts in the California area would be most likely to be aware of recognition of Laetrile or amygdalin's safe us as cancer therapy.) The experts polled, representing each of the medical schools in the California university system, were asked about the drug's efficacy rather than their views on the question of the safety of Laetrile's use in cancer therapy. Clayton G. Loosli, M.D., Dean of the University of Southern California School of Medicine, speaking for the members of the school's faculty, indicated that Laetrile, while extensively investigated, was in the unanimous opinion of the faculty without value in the treatment of human cancer. He stated that "further, we consider its use not only not valuable even as a placebo but harmful in that use of Laetrile prevents patients from receiving what otherwise might be an effective modality of treatment" (id., App. 10). J. B. deC. M. Saunders, M.D., Dean of the University of California School of Medicine at San Francisco, speaking for the clinical staff of his medical school, gave their opinion that the use of Laetrile was of no value in the treatment of cancer. He said "(1) t may not only delay or interfere with conventional therapy (surery and radiation) but indeed could seriously jeopardize whatever chances the patient may have for cure. The unscrupulous use of unproven cancer 'remedies' such as Laetrile tragically increases the human suffering already associated with cancer (id.).

The only evidence submitted by proponents of Laetrile that experts qualified to evaluate the safety of drugs generally recognized the drug as safe when used in cancer therapy were two affidavits by Charles Gurchot, Ph.D., and Chauncey D. Leake, Ph.D., (R 302, Exs. K and L). The Gurchot affidavit states in paragraph 14 that amygdalin in liquid and solid form was used prior to 1962 (between 1934 and 1945) by Gurchot, under the supervision of five named medical dostors at the University of California Medical School at San Francisco (R 302, Ex. L). This amygdalin

was, according to his statement, administered "on patients intramuscularly and intravenously" (id.). At the same time, the amygdalin preparation he used, he states, was used by "about a dozen physicians throughout California through the University of California Medical Schools and as recommended by members of the Hospital Staff of the University of California Medical School at San Francisco" (id.). Gurchot states that these physicians were qualified by medical and scientific training and professional experience to evaluate the safety of substances such as amygdalin and that they recognized it as safe (id.). The Gurchot affidavit should be compared, in the first instance, to the 1962 statement, already discussed, of the Dean of the University of California Medical School at San Francisco, which states, "Laetrile is not, nor has it been, in clinical use or in experimental trials in this institution \* \* \*" (emphasis added) (R 183, Att. 16, App. 10 at 8).

Even were the Commissioner to credit Dr. Gurchot's statement, he would have to conclude that, whatever had happened in the years 1934-1945, that experience did not form a basis for general recognition by qualified experts of safety in 1962, since even the faculty of the medical school in which Gurchot claimed the experiments had taken place had no knowledge of them. It should also be noted that, as discussed in the section on the 1938 grandfather clause, the "amygdalin" Gurchot could have been using would not have been the same substance in use today. His evidence, in addition, speaks only of investigational, as opposed to commercial, use of the drug-an improper basis for "general recognition" (see discussion above). In light of these facts, and of the other information in the record on this issue, Gurchot's statement in paragraph 16 of his affidavit indicating his belief that the general recognition of safety requirement for exemption from new drug status did exist for amygdalin on or prior to October 10, 1962 must be

The affidavit of Chauncey D. Leake. Ph.D., indicates that he is familiar with Dr. Gurchot's use of amygdalin in the mid-1930's and 1940's at the University of California Medical School Hospital in San Francisco. He states that at that time, i.e., in the 1930's, "it was generally held by physicians and other scientists familiar with it, that amygdalin was safe when used in the treatment of cancer as well as in its use as an expectorant or cough suppressant" (R 302, Ex. K at § 6). This conclusion does not, however, indicate recognition by anybody in 1962; it does not, as demonstrated elsewhere, deal with the drug presently being used (see affidavit. of Dr. Krebs, Sr., (R 183, Att. 13)); it refers only to physicians and scientists "familiar with it", thus not addressing the question of whether recognition was

(c) Lack of General Recognition of Effectiveness in 1962.—Experts in the

evaluation of the safety of a drug do not conclude that a drug is safe, if that drug is intended for the treatment of a life-threatening disease, if it has not been shown to be effective. The record illustrates a broad consensus of cancer researchers and physicians that Laetrile presents a grave danger to patients who might be helped by orthodox therapy. The concern is that such patients may be, induced to turn instead to this ineffective drug, their disease may progress while effective therapies are foresaken. and the use of the ineffective cancer drug will inevitably and inexorably lead to the patient's death. (See, e.g., R 396 at 1; R 384; R 170 at ¶ 11; R 183 at 18; R 266, Ex. 3 at 865; R 192 at § 14; R 193 at 1; and R 195 at § 13.) Thus, even if it were shown, as it has not been, that experts in 1962 generally were aware of the drug, its formulation, its conditions of use, and of toxicity data concerning it published in the scientific literature, the alleged nontoxicity of Laetrile (or amygdalin) would not form a sufficient basis for general recognition of safety in 1962.

At the time of the 1962 amendments to the act, it was made clear that, where drugs utilized for life-threatening diseases are involved, evidence of effectiveness is essential to proof of safety. The Senate report on the amendments stated.

The Food and Drug Administration now requires, in determining whether a "new drug" is safe, a showing as to the drug's effectiveness where the drug is offered for use in the treatment of a life-threatening disease, or where it appears that the "new drug" will occasionally produce serious toxic or even lethal effects so that only its usefulness would justify the risks involved in its use.

(S. Rep. No. 1744, 87 Cong. 2d. Sess., 1962 U.S. Code Cong. Ad. News 2884, 2891.) The report made it clear that the amendments were "in no way intended to affect any existing authority of the (FDA) to consider and evaluate the effectiveness of a new drug in the context of passing upon its safety" (emphasis added) (id. at 2892).

As the Court held, when dealing with a similar unproven cancer remedy, in Durovic v. Richardson, supra, lack of general recognition of the effectiveness of a drug intended for treatment of a life-threatening disease on October 9, 1962 means that general recognition of its safety could not have existed:

(A) drug offered for use in the treatment of cancer is now, and was before the amendments, a new drug unless it has achieved general recognition among the experts as safe and effective for such use (479 F. 2d at 250).

The evidence in the record overwhelmingly demonstrates that, among experts in the evaluation of the safety and effectiveness of drugs, Laetrile (or amygdalin) is not recognized as effective (see discussion above). It is a fair inference, absent any indication to the contrary, that a drug not recognized as effective now was not so recognized on October 9, 1962.

Again, however, the record supplies evidence of opinions of such experts given at almost exactly the time in question. As noted above, the medical schools in California were asked their opinions of Laetrile's effectiveness. Each of the medical schools contacted stated that Laetrile was never used in their institutions and that they concluded that it was not effective and had not been shown by testing to be effective. (See, in addition to the letters discussed above, letters from Dean David B. Hinshaw, Dean of Loma Linda University Medical School, Dr. M. H. Simmers, Coordinator, Cancer Training, California College of Medicine, Sherman M. Mellinkoff, Dean. University of California, Los Angeles Medical School.) These letters are printed as appendix 10 to R 183, Att. 16; see also R 183, Att. 16 at 38. The report states that Robert H. Alway, M.D., Dean of Stanford University School of Medicine, also indicated that Laetrile was of no value in cancer treatment and was not part of the treatment program at his medical school (R 183, Att. 16 at 38). The report also states that two other professors involved with cancer therapy and research concurred in this evaluation (id.). The report of the California Cancer Advisory Counsel itself constitutes convincing evidence that at about the time of the crucial date. October 9, 1962, experts did not generally recognize Laetrile as safe for the treatment of cancer, in particular because it was considered to be a worthless treatment for a life-threatening disease. As has been pointed out elsewhere, no adequate and well-controlled clinical investigations, the prerequisite for general recognition by experts of a drug's effectiveness (Weinberger v. Hynson, Wescott and Dunning, Inc., supra) exist as to Laetrile. Thus, even were the testimony in the record on this question less conclusive than it is, it would be necessary to find that there was no general recognition by experts that Laetrile was safe for use for any purpose on October 9, 1962.

The Commissioner thus concludes that Laetrile (or amygdalin) does not qualify for exemption from the new drug provision of the act by virtue of compliance with the 1962 grandfather clause.

### IV. THE POPULARITY OF LAETRILE

### A. LAETRILE AND OTHER UNPROVEN REMEDIES

Laetrile, as far as is known, has nothing in common scientifically with any of the other "unproven" cancer remedies of the past. Yet the method of promotion of the drug and the arguments advanced for its use are markedly similar to those of past cancer frauds.

### 1. The History of Cancer Quackery in the United States

Through the ages there have been literally thousands of supposed remedies

<sup>&</sup>lt;sup>a</sup> Dr. Ernst Krebs, Sr., though he thought people should be allowed to use his Laetrile, and a number of other remedies since forgotten by the public, is on record as stating that he could see no rationale for Krebiozen, the last of the highly publicized "unproven" cancer remedies (R 183, Att. 14 at 2).

for cancer, generally so outlandish that it seems incredible that people once believed in them. One historian of health quackery pointed out that the promotion of "unproven" cancer cures has a long history in this country:

Cancer quackery appeared in America during colonial times, one example being the alleged "Chinese Stones" vended by a pur-Frenchman, Francis Torres, hawked his cures from town to town. During the nineteenth century, an alert physician, cer quack, (each of whom offers) his "secret specific" to the panicked citizenry who, "like a drowning person grasping at straws seize upon the frail hope that is offered by the hand of ignorant charlatanry!" "Dr. Johnson's Mild Combination Treatment for Canoffered the first serious legal challenge to the 1906 Pure Food and Drug Act, requiring the Congress to enact the Sherley Amendment of 1912. At this same time, Dr. Arthur J. Cramp of the American Medical Association devoted fifty pages in his first Nostrums and Quackery volume to a detailed account of ten major cancer 'cures' deceiving the American people. Compiling a third volume in 1936, Dr. Cramp pointed to twenty-nine purported cancer cures, stating that "hardly a week has passed when the Bureau of Investigation of the American Medical Assoclation has not received one or more letters in which the writers stated that they had discovered, or had in their possession, a 'sure cure' for cancer."

Nor has cancer quackery diminished as the twentieth century has progressed. Indeed, with the decline of contagious diseases, due mainly to the chemotherapeutic revolution, and the consequent rise of cancer into second place as a cause of death, cancer quackery has expanded. The 1971 edition of Unproven Methods of Cancer Management, published by the American Cancer Society, described fifty-four promotions offering hope to cancer sufferers but deemed devoid of value by ACS. The 1976 edition of Unproven Methods of Cancer Management cites in its appendix seventy-one such methods (R 400 at 1-2; see also R 400, Ex. 2).

Evaluation of approximately 60 of these methods may be found as attachments to R 400. Ex. 2.

Each decade seems to have an unproven cancer remedy that is promoted so effectively that it attracts a large following and becomes a cause celebre. In the 1940's and early 1950's, the Koch Antitoxins were heavily promoted as a specific cure for cancer. The Koch Antitoxins thesis, promoted by William F. Koch, M.D., advanced "the theory that cancer is caused by a microorganism resembling the spirochete of syphilis, which could be destroyed by a differen-tial poison of his invention" (R 183, Att. 3 at 43). "The Koch medications, known collectively as Koch's Synthetic Antitoxins or oxidation catalysts, were individually packaged in 2 ml ampules. Malonide and glyoxylide (were) claimed to be present in a concentration of one part in a trillion parts of water, and parabenzoquinone one part in a million parts of water" (id.). "Glyoxylic acid, of which glyoxylide is the anhydride (the resulting element after water is removed), is a normal constituent of the human body. About two grams are formed daily-at any given time there are about five milligrams in the human body, whether healthy or diseased. It would take a tril-

lion 2 ml ampules of Koch's glyoxylide to equal the amount produced daily by the body, and two and one half billion ampules to equal the amount present in the body at any one time" (id at 44). Even so, cancer patients paid as much as \$300 per injection for this worthless remedy (R 400, Ex. 2, ACS "Koch Antitoxins").

Another unproven cancer remedy whose promotion reached substantial proportions in the 1950's was the medications of Harry Hoxsey. Two liquid mixtures played the central role in the Hoxsey remedy. The "brownish black liquid" contained potassium iodide and "some of all of the following inorganic substances as the individual case may demand: Licorice, red clover, burdock root, stillingia root, berberis root, poke rot, cascara, Aromatic USP 14, prickly ash bark, (and) buckthorn bark (R 400, Ex. 2, ACS "Hoxsey Method"), The "pink liquid" was composed of lactate of pepsin and other ingredients (R 416, Ex. 6 at 368).

Hoxsey and his spokesmen were frank to confess that they did not completely know why his colored mixtures cured cancer. They asserted that they had been kept too busy treating cancer patients and fighting court battles to keep their clinic open "to spare the time, personnel, and facilities for objective study" (id. at 369). Hoxsey's hypothesis "held that a major chemical imbalance in the body caused normal cells to mutate into a cancerous form, and his medicines restored the original chemical environ-ment, checking and killing the cancerous cells" (id. at 369). The proponents of the Hoxsey remedy, like the Laetrile proponents of today, condemned the only treatments then recognized as having value in cancer therapy. The Hoxsey proponents held that "X-ray and radium (had) no place in the treatment of cancer \* \* \*. They further upset basic cell metabolism rather than do anything to correct it" (id. at 369).

Harry Hoxsey promoted his unproven cancer remedy for more than 30 years until 1960, when after years of numerous local, State, and federal court actions, the sale of the Hoxsey medicines was stopped in the United States. At the time of the 1960 permanent injunction banning the sale of Hoxsey remedy at the Taylor Clinic, more than 10,000 patients were receiving the remedy. (See, generally, R 400, Ex. 2, ACS "Hoxsey Method"; R 416, Ex. 6.)

In 1964 a California State government report stated that, at that time, "Possibly no other unproven treatment for cancer has received so much public attention or approbation as Krebiozen. This agent has been the subject of intense scrutiny by scientists and government officials, and loudly discussed by the press and by the general public. The events surrounding the introduction of Krebiozen as a potential cancer cure and the subsequent trials to test its capabilities produced an air of notoriety seldom seen in the medical world" (R 183, Att. 3 at 59).

Unlike Harry Hoxsey's backwoods herb remedy, Krebiozen, the most heavily pro-

moted improven cancer remedy of the 1960's, had an aura of high scientific prestige. The drug's principal proponent. in the United States was Dr. Andrew C. Ivy, then Vice-President in charge of the Chicago Professional Colleges, Distinguished Professor of Physiology and Head of the Department of Clinical Science. University of Illinois (id.) Krebiozen was reportedly produced originally in Argentina by Stevan Durovic, M.D., a Yugoslavian physician, and brought to the United States in 1949 (R 400, Ex. 2, ACS "Krebiozen and Carcalon"). "According to Dr. Durovic, the original 2 grams of powder, from which he said 200,000 doses were prepared, was obtained as an extract of the blood of 2,000 Argentinian horses which had previously been injected with a sterile extract of Actinomyces bovis, a microorganism which causes a disease called 'lumpy jaw' in cattle" (id.) . "Food and Drug Administration analyses of Krebiozen ampules (showed) that those sold before 1960 (were) different from those sold in 1963, and that neither contain(ed) any of the powder identified in July 1963 by Dr. Stevan Durovic as Krebiozen, and found to be creatine monohydrate, which will not dissolve in min-eral oil. \* \* \* analyses of Krebiozen ampules shipped before 1960 showed they contained nothing but mineral oil, while ampules shipped since then contained mineral oil plus minute amounts of amyl alcohol and 1-methylhydantoin, a derivative of creatine which will dissolve in mineral oil" (id.).

In 1963, a committee of 24 cancer experts was appointed by the Director of the National Cancer Institute to review clinical records on 504 patients treated with Krebiozen, and to recommend whether the Institute should spensor clinical trials of Krebiozen. The committee unanimously concluded that Krebiozen was an ineffective cancer drug and strongly urged that no clinical trial be undertaken (id.).

In November 1964, Drs. Ivy and Durovic and other proponents of Krebiozen were indicted on 49 counts for violations of the Federal Food, Drug, and Cosmetic Act, mail fraud, mislabeling, making false statements to the government, and conspiracy. All of the defendants were acquitted in January 1966, after a 9-month jury trial (id.). Although the acquittal meant that the government did not prove its case beyond a reasonable doubt, it did not have any bearing on the question of whether Krebiozen was a safe and effective cancer drug. As an unapproved new drug, its distribution in interstate commerce remained illegal. In spite of the acquittal, the Krebiozen boom collapsed shortly thereafter.

Similarities Between Lactrile Promotion and That of Other Recent "Unproven" Cancer Remedies

The promotion of Laetrile in the 1970's is completely in character with the historical pattern of the promotion of other unproven cancer remedies such as the Koch Antitoxins, the Hoxsey method, and Krebiozen. These characteristics include the following:

 The proponents "don the mantle of science while at the same time traducing the reputable scientists of their day" (R 400 at

(2) The proponents claim that "prejudice of organized medicine hinders their efforts" and they "challenge established theories and attack prominent scientists with bitter criticism" (R 400, Ex. 2 "Unproven Methods of Cancer Management—1976" at 3) (hereinafter cited as "Unproven Methods").

(3) The proponents "cite examples of physicians and scientists of the past who were forced to fight the rigid dogma of their day"

(id.).

(4) The proponents rely mainly on testimonials and anecdotes as evidence that their remedy is a safe and effective cancer thera-

peutle agent (see R 400 at 4).

(5) The proponents "do not use regular channels of communication (current, reputable scientific journals) for reporting scientific information" (R 400, Ex. 2 "Unproven Methods" at 2-3). The main channels of communication are the mass media, popular journalism, and word of mouth (see R 400 at 4-5).

(6) The proponents' "chief supporters tend to be prominent statesmen, actors, writers, lawyers, even members of state or national legislatures—persons not trained or experienced in the natural history of cancer, the care of patients with cancer, or in scientific methodology." (See R 400, Ex. 2 "Unproven Mathedical 2)

(7) The proponents often offer a simplistic theory for causation of the disease frequently involving claims that dietary management can counteract virulent pathologic processes

(R 266, Ex. 3 at 865).

(8) The proponents' remedy is "easy and pleasant, compared with the frightening therapies wielded by orthodoxy, the surgical knife, harsh chemical drugs, poisonous radiation" (R 400 at 8).

(9) The proponents claim that the mode of administration of the drug and the method of treatment can only be learned from them (R 400, Ex. 2, "Unproven Methods" at 3).

The record illustrates the remarkable conformity of the Lactrile promotion to this pattern:

(a) Mantle of Science.—Throughout history, promoters of unproven cancer remedies have couched the explanation for the remedies in pseudoscientific terms. "Impressive and plausible to the layman, such arcane explanations, to true scientific specialists, came off as nonsensical balderdash" (R 400 at 3). The promoters of Laetrile have presented a series of shifting theories to explain the alleged anticancer activity of Laetrile. These theories have been examined in detail above. (See, generally, R 318.)

(b) Attacks on the "Establishment."—
The proponents of Laetrile have often accused government agencies and organized medicine of making untruthful and irresponsible statements regarding the experimental evidence of Laetrile's anticancer activity. (See, e.g., R 302, Ex. A at 14-16; R 509 at 3-4.) In other instances, proponents of Laetrile have chastized the orthodox methods of cancer treatment and management, i.e., surgery, radiation, and chemotherapy. (See, e.g., Tr. at 16-27, 297-318, and 417-426).

The most vocal arguments challenging established orthodox treatments have been concerned with the issue of freedom of choice, discussed elsewhere in this opinion, These arguments, many of them

from cancer patients or their relatives and friends, hold that the "bureaucracy" has no right to interfere with the physician-patient relationship by withholding from them a treatment in which they believe and which they want. (See, generally, Tr. at 55–56, 255–256, and 454–456.)

(c) Claimed Parallel with Scientific Pioneers.—To combat criticism from the established medical societies and government agencies that Laetrile had not been shown to be safe and effective, its proponents compare the originators of the drug and physicians who prescribe Laetrile with earlier scientists who were persecuted and ostracized for their scientific theories: Copernicus, Newton, Freud, Galileo, and Semmelweiss. (See, e.g., R 318 at 61-63; R 198, Ex. 2 at 3-5).

(d) Reliance on Testimonials.-As previously discussed, the proponents of Laetrile rely on testimonials and anecdotes as evidence that the drug is safe and effective in the treatment of cancer. In reviewing the administrative record the Commissioner has not encountered even one study that meets the legal and scientific standards for making a determination that Laetrile is safe and effective. Proponents claim that physicians using Laetrile are too busy treating patients to be able to maintain the records needed to document adequately the case histories they present. (See, e.g., Tr. Ex. 1 at 117.)

(e) Lack of Scientific Publication .-Good science demands that evidence that a drug is safe and effective be presented in a manner whereby that evidence can be reviewed and evaluated by other scientists. Usually this evidence is published in scientific journals and presented for discussion at symposia and other meetings. Historically, "the main reliance of unorthodox promoters rests on the anecdotal evidence of testimonials from laymen, and the main channel for reaching an audience is through the mass media. In earlier days newspaper advertising trumpeted the promise of cancer cures, bolstered by the faces and words of grateful testifiers, not infrequently already dead of the disease" (R 400 at 4-5). The proponents of Laetrile have relied heavily on popular journalism, advertisements, radio and television, "health" organizations and word of mouth to spread their claims that Laetrile is a safe and effective anticancer drug. (See, e.g., R 318; R 302, Ex. A and H; R 198, Ex. 2.) A number of experts active in the management of cancer have submitted testimony stating that the scientific literature contains no reports of adequate, wellcontrolled studies upon which Laetrile can be regarded as generally recognized as safe and effective. (See, e.g., R 185 at 5; R 186 at 4; R 390 at 6.)

(f) Nonexpert Supports.—The proponents of Lactrile are well-organized and, through organizations such as the Committee for Freedom of Choice in Cancer Therapy, have conducted active campaigns to move the discussion of the safety and effectiveness of the drug from the scientific to the political arena. These organized efforts have encouraged cancer patients and others to write their

local, state, and congressional representatives demanding that Laetrile be "legalized." These efforts are addressed not to discussions of the scientific merits of Laetrile as a cancer drug, but rather to Laetrile as a cancer drug, but rather to the issue of "freedom of choice" discussed elsewhere in this opinion. Such action on the part of the Laetrile proponents is typical of other unproven cancer remedies. Failing to win acceptance in the established medical community, proponents seek sympathetic allies in places of political power. (See R 400 at 6-7.)

(g) Simplistic Theories of Causation and Reliance on Diet .- The latest claims being made for Lactrile are that it is a "vitamin," and that cancer is a vitamin deficiency disease. The basis for these claims is discussed elsewhere in this opinion. It is sufficient to note here only that this simplistic theory of cancer prevention and treatment is common to other unproven cancer remedies. Cancer patients are told that they can cure or control their cancer by strict adherence to a special diet that includes a special "vitamin" even through this "vitamin" is not recognized by nutritional experts. (See R 266, Ex. 3 at 865-66.)

(h) A Painless Cure.—Laetrile, like other unproven cancer remedies, is promoted as a harmless cancer remedy free of the side effects associated with orthodox methods of treatment such as radiation and chemotherapy. Many of the statements submitted by cancer patients and their relatives and friends reflect the proponents' claims that Laetrile is free of side effects. (See, e.g., R 17; R 48; R 137.)

(i) Only Proponents Can Effectively Use the Drug.—In common with the supporters of other unproven cancer remedies, the proponents of Laetrile stress, as did Robert W. Bradford of the Committee for Freedom of Choice in Cancer Therapy, that "you" do not and cannot expect to get results from Laetrile treatment unless you are a trained metabolic physician" (Tr. at 349). These arguments are used to explain why orthodox physicians (i.e., those not trained in the proper use of Laetrile) do not see any evidence of Laetrile's effectiveness as a cancer drug.

### B. WHY DO PEOPLE USE LAETRILE?

Throughout history persons afflicted with cancer have turned away from the medical establishment to a series of what most euphemistically might be called "unproven remedies." Laetrile is the most recently publicized of these remedies. but, as the discussion above illustrates, it follows on the heels of other widely publicized therapies such as Krebiozen and the Hoxsey cure. Thoughtful persons have questioned the reasons for this troubling phenomenon. Why do people bet their lives, or the lives of their loved ones, on a therapy which is rejected by almost everyone trained and experienced in cancer research and treatment?

Much evidence in the record addresses this question. The answer lies in the fear that cancer engenders—and that proven therapies for cancer engender—and the need of patients and families for hope in a situation where the hope offered by the legitimate therapies is often modest. The use of "unproven remedies' is, in the opinion of observers, in large part attributable to the loved ones of the cancer victim, in whom both fear and the need for hope are magnified by sympathy and by the guilt that one feels at being unable to relieve the suffering of a person one loves. This situation is, unfortunately, skillfully exploited by the purveyors of "unproven" cancer remedies, of which Laetrile is only the most publicized.

### 1. The Emotional Reaction to Discovery of Cancer

"[W]hen cancer afflicts an individual, he is frequently faced with a circumstance which is virtually without hope. First of all, the cancer patient must be terrified by the diagnosis \* \* \*. It would be enough to terrify any lay person to simply be told that he has cancer. But more important than that is the fact that once he is told that he has cancer, he is told by he doctor that the treatments that we have available are very often disfiguring; they can be painful; they can be unpleasant; they can

even be risky" (Tr. at 204).

The cancer patient must thus cope with two wounds simultaneously. The first is to the body itself (R 423 at 1). "The other wound is to the psyche, reflected in the loss of the feeling of being invulnerable, a feeling which is basic to ordinary day by day living" (id.). The cancer patient senses suddenly that the future is limited. Social and work mobility are seen as curtailed; so are the patient's functional role in the family and the community. In addition, the patient senses a new dependence on others and may fear that he or she will become a burden on the family (id. at 2). "The initial psychological status of the patient and family is characterized by disorientation, anxiety, guilt, fear of pain and suffering" (R 421 at 1).

Dr. Robert C. Eyerly, Chairman of the American Cancer Society's Committee on Unproven Methods of Cancer Management, states that, "Indeed, we've found that the major reason cancer patients use Laetrile is fear \* \* \* fear that the disease is incurable, that surgery or other therapy is mutilating, and that the medical profession is not to be trusted" (R 173, Att. "Laetrile: Focus on

the Facts").

In this climate of anxiety and fear, the medical establishment-which, unlike the proponents of "unproven" remedies, feels an obligation to be honest with the patient and his family-cannot always offer hope: "[P]robably the most important factor (explaining why cancer patients choose to use Laetrile) has been the failure of modern medicine and technological advances to cure or adequately control some cancers. These unfulfilled expectations lead patients to disappointments in standard medicine and to attempt a cure of their disease by pseudoscientific methods" (R 398 at 19).

Physicians, trained in the saving of life and the alleviation of suffering but unable, in some cases, to do either with

cancer patients, may contribute to the frustration. "Many patients sense a feeling of frustration and hopelessness conveyed, perhaps unconsciously, by the physician who tells them the nature and probable outcome of their disease-a natural feeling on the part of the physician who is discouraged by his recognition that he cannot cure the patient. Patients sensing this hopelessness frequently are unwilling to 'abandon hope' and therefore seek (unorthodox therapies)" (R 190, Ex. 4, Editorial at 327) Glen W. Davidson, Ph.D., Chairman of the Department of Medical Humanities, Southern Illinois University School of Medicine, testified that, "\* \* when primary emphasis for treatment is placed on 'cure' and the physician's abilities, rather than on 'coping' and the patient's abilities, the patient is placed in an inappropriate and ineffective dependency relationship. When the physician can no longer promise 'cure' and then attempts to refer the patient out of his practice, or leaves the patient to institutional care of others, the patient feels abandoned. The patient has already had his coping abilities undermined. And many patients react to unfulfilled expectations and violated trust with anger and panic" (R 387 at 2).

A patient facing cancer and the lack of positive assurance from the physician that the cancer can be cured may simply give up hope that what the physician can do for the patient can work. This lack of confidence in proven remedies is tragic in an era when, in the case of many cancers, a significant percentage of patients can be cured or have their lives extended. See, e.g., R 173, Att. ACS, 1977 Cancer Facts at 3. Laetrile's proponents expend great efforts to encourage this feeling. Much of the oral argument of Laetrile proponents in this proceeding was addressed not to the effectiveness of Laetrile but to the ineffectiveness of proven remedies (see., e.g., Tr. at 16 et seq.; Tr. at 228). With real hope extingished, the use of Laetrile or other unproven remedies is a way of avoiding an acceptance on a conscious level of the consequences of the disease: "The decision to use Laetrile indicates that, at the subconscious level, patients and their families have given up on conventional therapy and, in fact, have accepted the inevitability of death. On the more superficial level, patients choosing Laetrile are persons who believe that they do not require the use of sophisticated, anti-cancer treatments. This reflects an ambivalence which many patients feel at the time they are required to make decisions about cancer therapy. If patients can maintain denial about the seriousness of their cancer, then they can permit themselves to experiment with a bizarre apricot-extract, such as Laetrile" (R 433 at | 13), "Human beings have become accustomed to using the psychological techniques of denial in dealing with real problems" (R 390, Ex. 3 at 386). "The decision to use Laetrile is, in essence, an attempt 'magically' to avoid the reality of cancer" (R 433 at 19).

2. The Role of Loved Ones

Patients with a diagnosed malignancy frequently encounter ostracism in their private, social, and vocational roles (R 387 at 1). At this point, the caring of loved ones and their sympathetic willingness to continue to associate with and to share the suffering of the cancer patient assume great importance to the patient. This caring relationship, quite understandably, leads to a dependence by the patient on the loved one and a corresponding feeling of responsibility in the nonpatient: "Many patients in their initial response to cancer diagnosis surrender control to those closest to them, further complicating the issue of informed choice. Highly anxious relatives with little or no medical understanding of cancer as a disease entity fall prey to the emotional appeal of the proponents of Laetrile" (R 421 at 2). "Cancer patients are most vulnerable to the manipulations of others when they feel they are (1) being abandoned, (2) unable to control pain, and (3) unable to maintain a 'sense of dignity' by being able to make decisions for themselves. Attempts at guarding oneself from all three fears are often incompatible. Many cancer patients feel they are in a 'double-bind' If they don't follow their physician's treatment plan, the disease process won't be arrested. If they don't follow the competing, and often contradictory advice from relatives and friends, they will be abandoned. And if they assert their own feelings they will be ostracized by others at the very time they most need support from others" (R 387 at 1-2). Thus, some patients pay the price of what benefits are available from orthodox treatments in order not to be abandoned by family and friends-"a psychological analogue to the theological concept of being 'cast into Hell' \* \* \*" (id. at 2). In many cases, it is family and friends who, amplifying the patient's feelings, try to get their anger and panic under control by manipulating the patient into use of medically unacceptable remedies. (See id. at 2.) The families of cancer patients, particularly parents of children with cancer, are understandably desperate for anything that will cure cancer. They often are beset by irrational feelings of guilt, and seek to assuage these feelings with the assurance that " \* \* \* 'we did everything for our child' even to the point of foolishness in going after an unproven cure \* \* " (R 394 at 2).

The shared responsibility of the loved oner of cancer patients for the patients' involvement with Laetrile (or other unproven remedies) helps to explain why these families have been among the most vociferous proponents of Laetrile. "This reaction can be understood because such persons, whether they are family members or friends, have to justify the deceased's use of Laetrile by suggesting that the patients were considerably helped by the drug, that their lives were prolonged to a significant extent, or, at the least, that they did not suffer a great deal of pain during treatment with the drug. To do otherwise would require them to acknowledge that they made a mistake

and misled the patients or that they went along with decisions which were clearly erroneous. Living with that kind of guilt is very difficult and the advocacy of Laetrile is a way of avoiding it" (R 433 at § 15). It is only those family members who did not participate in, or dissented from, the decision to use Laetrile who, after the patient's death, raise their voices against the drug's use (see, e.g., R 47; R 429; R 300; R 348).

### 3. Methods of Promotion of Lastrile

As is obvious from the above discussion, the cancer victim and his or her family are extremely vulnerable to the kind of persuasion used so skillfully by Laetrile's promoters. This persuasion may take the form of highly polished and thus convincing films and books (see Tr. at 331) or of personal visits. The fact that many persons involved in Laetrile promotion believe strongly in the drug makes their presentations, because sincere, all the more compelling. In his affidavit, one cancer patient, speaking from his own experience, stated that "immediately after a diagnosis of cancer, most patients and family members are susceptible to something such as Laetrile. which offers a painless treatment with certain results" (R 388 at 2). The patient also stated that, somehow, the names of cancer patients in his area had been obtained by certain persons helping to spread the Laetrile theory. He indicated that Laetrile proponents exerted constant pressure on him and his wife to quit orthodox medical treatment and try Lactrile. Testimonials from patients who spoke in glowing terms of their recovery or successful treatment with Laetrile were offered to supported the proponents' claim (id.). Laetrile promoters are diligent in searching out persons with reported cancer to offer their product. One physician noted that he had a patient who, within 24 hours of his being diagnosed as having lung cancer, received information in the mail telling him he ought to take Laetrile and where and how to get it (Tr. at 184).

Lactrile proponents are keenly aware of the involvement of family members and friends in decisions to accept unproven remedies and actively seek to persuade them of the drug's benefits. One woman who had had surgery and chemotherapy for treatment of breast cancer commented: "My biggest problem has been coping with well-meaning relatives and friends who swallow this propaganda of unprofesisonals and then try to make me feel guilty because I don't take their advice \* \* \* " (R 96).

Lastrile proponents play upon the victim's frustration with a medical establishment that cannot offer the certainty of a cure. Some patients reportedly turn to Laetrile precisely because it is "fl-legitimate," behavior that appears to be an anger reaction toward legitimate medicine" (R 387 at 3). This antagonism toward the medical establishment is fanned by Laetrile proponents (as it has been by the purveyors of previous "unproven" remedies) to a pitch that most observers would consider absurd. When a speaker at the oral argument asked the audience, which consisted predomi-nantly of Laetrile supporters, if "you really think that a quarter of a million physicians across the country can let people die because they want to make a profit off of thems", the audience response was a loud chorus: "Yes" (Tr. at

Laetrile proponents also play upon and build the cancer patient's fear of legitimate cancer therapies. (See R 421 at 2: "The promise of a painless cure through Laetrile, as opposed to ordothox medical methods with their side effects capitalizes on the fear of pain and suffering".)
"Slash (or cut), burn, and poison" are the code words of the Laetrile supporters for the proven remedies of surgery, radiation and chemotherapy (see, e.g., Tr. at 291, 357, 463). A videotape of an interview with a cancer patient (R 419, Ex. B; see also R 197 at (7) that is part of the record shows graphically the costs of this sort of propaganda. The victim is a woman who, at the time her breast cancer was discovered, was given a reasonably good prognosis of recovery after surgery. Out of fear of surgery she tried Laetrile therapy. Though the tumor grew to involve her whole breast she continued to avoid conventional therapy, even trying, after Lactrile did not help, an "asparagus" diet cure, garlic, and finally a fruit and vegetable diet with hot baths. When, nearly at death's door, she returned to the surgeon, it was too late for surgery to be effective. She then was convinced to try radiation therapy, which she testified she had avoided because the negative descriptions of it in Prevention magazine, to which she had long subscribed. The radiation therapy helped reduce the size of her tumor and make her more comfortable, but her expected survival was greatly diminished by her delay in obtaining effective treatment. This kind of disparagement of conventional thearpy, a bulwark of the cam-paigns of Laetrile proponents, is perhaps the most morally reprehensible aspect of the pattern of the drug's promotion.

### 4. The Sampson Survey

While the conclusions about the reasons for use of Laetrile expressed in the record are based upon a multitude of experiences by various witnesses with patients taking Lactrile, it is interesting to note the conclusions of the one attempt to survey Laetrile patients about their reasons for using the drug.

Based upon about 20 interviews with cancer patients who abandoned orthodox therapy in favor of Laetrile, Dr. Wallace I. Sampson, Clinical Associate Professor of Medicine, Stanford University School of Medicine, stated that about 75 percent of the patients reported that they had serious problems with their physicians. About 75 percent believed in Laetrile's therapeutic rationale and effectiveness. About 75 percent of the patients were involved in other methods of therapy that included high doses of Vitamin C, megavitamin therapy, and immunotherapy given by unqualified individuals. Dr. Sampson is of the opinion that the patients receiving Laetrile were involved in other types of unorthodox tally destroyed by, for instance, the 7th

therapy because of their outlook on life (i.e., they seek nonrational, magical solutions to the problems of dread and often uncurable illness) or perhaps because of difficulties in relating to a standard physician. A large majority of the patients believed that there is a conspiracy to keep Lactrile off the market. Less than 10 percent of the patients tried to inform themselves about Laetrile from non-Laetrile sources. (See Tr. at 118-119; R 398 at 4.)

### C. THE LAETRILE TESTIMONIALS

Unproven cancer remedies like Laetrile are invariably supported by numerous testimonials of persons who pronounce themselves satisfied with the results they, or their deceased friends and relatives, have achieved with the drug. The present widespread use of Laetrile as an alternative to remedies of proven effectiveness illustrates the problems to which such "evidence" of a drug's effectiveness leads, and it is a legitimate question to ask why there are so many such testimonials.

The Commissioner does not doubt the honesty or the sincerity of the many testimonials for Laetrile, but many of the positive experiences reported may be accounted for by explanations other than the claimed effectiveness of the drug. The placebo effect discussed above undoubtedly accounts for some of the reports, particularly those claiming decrease in pain and increased sense of well-being. Experts interested in the question have provided other explanations. Most of the patients reporting Lastrile "cures" appear actually to have had the benefit of other, proven effective therapies. Some of those who believe themselves cured may never have had cancer at all. Others may simply not be cured, despite their

Many of the testimonials and anecdotes concerning the effectiveness of Lactrile replay the same scenario. The cancer patient is told he has cancer and agrees to surgery, radiation, and/or chemotherapy. After some time, the patient, feeling nauseous, weak, and general malaise, in desperation turns to Laetrile. Within a few days or weeks after stopping orthodox treatment and starting to use Laetrile, the patient feels better, has an appetite, and is able to move about on his own. The patient in all sincerity attributes his recovery and feeling of well-being to his decision to reject orthodox medical treatment and to choose Laetrile. (See, e.g., R 9; R 35; R 223; R 267; R 315; R 391; R 483.) Many families of deceased cancer patients who had orthodox therapy and who then used Lactrile believe that the patient benefited from the Lastrile and might still be alive if they had turned to Laetrile earlier. (See, e.g., R 19; R 208; R

It is easy to understand how such a situation could develop. A doctor may prescribe 10 applications of a proven cancer drug, perhaps after surgery. The cancer may have been totally removed by the surgery or it may have been toof the 10 applications of the effective drug. Because the physician cannot know this, and because he cannot risk the chance that some cancer remains, he has prescribed the recognized treatment regimen. Use of cancer drugs (referred to as chemotherapy) or of radiation may involve unpleasant side effects. The patient, sickened by the side effects of the drug and importuned by Laetrile proponents, may stop the chemotherapy before the prescribed regimen is completed. As the side effects clear up, the patient feels better. If a full cure has been accomplished, it will be attributed to Laetrile. If it has not, the surviving family may well believe that, since the patient felt better after stopping chemotherapy and starting Laetrile, the therapy was only received "too late." See R 184 at 17:

Testimonials attesting to a feeling of gen-eral improvement and cessation of pain in patients upon abandonment of radiation and chemotherapy in favor of Laetrile treatment do not indicate that Laetrile is effective in curing cancer or in relieving pain. The feeling of well being experienced by these patients derives from two phenomena, one physical and the other psychological. Chemotherapy and radiation treatments produce unpleasant side effects in most patients. When such therapies are stopped, the side effects they produce disappear. This natural physical effect in the case of these patients is reinforced when Laetrile is administered because of the patients' expectation that the treatment will have a beneficial effect.

Dr. John A. Richardson, himself a major proponent of Laetrile therapy, stated that 85 percent of the 4,000 to 5,000 patients treated with Laetrile at his clinic had previously received some type of orthodox medical treatment (Tr. at

Sometimes conventional and Laetrile therapies are administered simultaneously, with any beneficial effects attributed by patients to the latter. Dr. Emil J. Frereich is involved in the development of cancer drugs. He stated that:

(W)e have numerous patients who are receiving developmental therapy drugs which have at the time, real promise, and subsequently prove to be useful and are introduced into practice, who unbeknownst to us, were also taking therapy with laetrile and when their disease responds to therapy, (they) inadvertently ascribe it to the effect tiveness of the unproven remedy, whose administration is revealed to us subsequently. When we compare the responses of patients on a given therapy who have received lastrile at the same time, with those who received none, there is no significant difference, which indicates clearly that those observed re-sponses were due to the cancer chemotherapy drugs which were being administered by us not by the additional use of laetrile-(R 390 at \$ 20.)

For other testimony on the propriety of attributing to Laetrile cures that may be caused by other, proven effective, drugs, see, generally, R 174 at 9 and R 185 at 7 20e.

"Some people who believe that Laetrile cured them never had cancer to begin with" (R 174 § 9). In a number of the "case histories" submitted to show Laetrile's effectiveness, there is no accepta-

ble showing that the patient ever had cancer. (See, e.g., R 183, Att. 16, App. 2; R 184, Ex. 2; R 378, Att. "Supplementary Report," cf. evaluation of case histories above.)

In one 1955 pamphlet, Dr. Krebs, Sr., discouraged biopsy, the procedure often used to determine whether a tumor is malignant (cancerous) (R 183, Att. 7 at 14). He urged instead that a special urine test, not generally accepted by the medical community as useful, be the means for diagnosing cancer (id. at 16). Even where the diagnosis has been done by someone other than a Laetrile proponent, a mistake is possible. Some cancers which are discussed in reference to Laetrile are very difficult to diagnose histologically. Thus, a diagnosis of cancer may often on later review be reversed. (See Tr. at 141.)

"Many cancer patients have given testimonials believing themselves cured, only to discover later that they still have the disease" (R 174 at 19). Since he is involved in the testing of cancer drugs, Dr. Emil J. Freireich is in a good position to follow up on patients who leave his program to use Laetrile. Dr. Freireich reports that "(i)n virtually every instance, (Laetrile patients treated in our department and subsequently followed by our tumor registry, have been) found to have evidence, not only of progressive disease, but to have expired after receiving such unsuccessful treatment, and a significant fraction eventually return to our clinic for more developmental ther-(R 390 at ¶ 20).

An illustration of what, in all likelihood, explains most Laetrile testimonials

appears in the record:

Testimonials fail to provide objective evidence that there has been control or regression of a tumor which is attributable to the use of Lactrile \* \* \*. To illustrate why such data are important, let us examine two typical versions of testimonials from women who state that their cancer of the breast was cured by Laetrile. The first testimonial is from Jane Doe. She discovered a lump in her breast and based upon the urging of friends has consumed on her own initiative a number of Laetrile tablets. It is also possible that she saw a doctor who administered injections and prescribed a special diet. In a month, the lump has disappeared, and Jane Doe sings the praises of Laetrile. "It cured my cancer; I am living proof." This is not credible evidence. The lump detected may have been caused by a variety of conditions. Without laboratory confirmation that a malignant condition existed, there is no basis assume that it was cancer and that Laetrile contributed to its disappearance. The second testimonial is from Dorothy Doe. She had objectively diagnosed cancer, underwent a mastectomy, and postoperative chemotherapy or radiation treatments. The physician informs Dorothy that an additional surgical procedure may be necessary.

Dorothy decides against further unpleasant treatment and takes Laetrile. Now, six months or three years later—time makes little difference-she, too, sings the praises of Laetrile. Dorothy's experience does not constitute evidence. It is possible that her orthodox treatment was successful; it is possible that she still has cancer, but that it will not manifest itself for another year or, indeed, as is sometimes the case, for another dozen years. The point is that there are no objective data upon which to assess Doro-

thy's condition at the time Laetrile was administered and the effects of Laetrile. In the absence of such data, there is no basis for a claim that Lastrile was effective (R 191 at

As another affidavit states,

\* \* It is a certainty that any substance without significant toxic or harmful effect, including mystical activities, faith healing and all other types of non-toxic or nonharmful remedies will be effective in a small fraction of the very large population of patients with hopeless terminal cancer. Those individuals who fall to respond to such treatment, that is, who have the expected outcome, which is progression of their cancer and death, are no longer living and those rare individuals who have the exceptional or miraculous outcomes frequently live for long periods of time. It is obvious that a large number of individuals can be identified who have unusual outcomes. These individuals are of course easily convinced of the effectiveness of such treatments and are free to testify to their effectiveness for as long as their disease remains in control. Such testimonials contribute no significance toward our understanding of the effectiveness of any treatment for cancer. Evidence accumulated in the proven, objective, medical and scientific fashion is the only evidence that can be of use in evaluating the potential of any treatment for influencing the course of malignant disease (R 390 at ¶ 21).

Dr. Melvin Krant, Professor of Medicine and Psychiatry and Director of Cancer Programs at the University of Massachusetts Medical Center, reviewed a number of the testimonials submitted to the record from patients and relatives and friends of patients who have been treated with Laetrile. He stated that the testimonials "do not offer evidence for eifectiveness because frequently the treatments with Laetrile were taken after other treatments such as surgery, radiation, or chemotherapy. At times, the Laetrile was taken in conjunction with other modes of therapy such as chemotherapy. In such instances, it is impossible to know whether the Laetrile added anything to the patient's response. There are no objective ways to measure the patient's response. In many instances, it seems like the main emphasis of the testimonials is on the patient's emotional reaction to being treated. Because the testimonials are not presented in a scientific manner, it is also impossible to determine if there were any side effects from Laetrile administration" (R 453 at

V. OTHER ISSUES REGARDING LAETRILE

A. USE OF LAETRILE OUTSIDE THE UNITED STATES

Laetrile's proponents have sometimes sought to give the impression that Lactrile is in use around the world and that it is only the United States' overly restrictive drug laws or an evil conspiracy among drug companies, physicians, and bureaucrats that is preventing marketing of the drug in this country. (See, e.g., the claim, in a 1963 publication, Control for Cancer by Glenn D. Kittler, that Laetrile was being studied in several countries in addition to the United States: Canada, the Philippines, Japan, England, Belgium, Italy, Union of South Africa,

and Mexico (R 318 at 31).) The book also reported that the drug was registered in Iran in 1962 (id.). (See also the reference to the use of Laetrile (or Amygdalin) in West Germany in the late 1960's (R 302, Ex. G).) (Cf. R 198, Ex. 2 at 24, (transcript of film World Without Cancer); Tr. at 424.)

The record reflects no international recognition or use of the drug. The State Department and the United States Mission to the World Health Organization made an effort to determine whether Laetrile, Amygdalin, Vitamin B-17, or such drug under any other name was known and approved elsewhere in the world. The State Department sent inquiries to all American embassies instructing embassy officials to ascertain the status of the drug in their respective host countries, and the mission to the World Health Organization made telephone inquiries of member states throughout Western Europe. The following information was obtained:

The American Embassy in Mexico advised that in 1974 the Mexican government gave provisional approval, contingent upon the presentation of evidence of Laetrile's effectiveness in treating cancer, to two laboratories in that country to manufacture the drug. This approval was cancelled in late 1976 because no positive results were obtained in research carried out at the Medical Center General Hospital. The decision to ban Laetrile has been appealed by Laetrile proponents and is now in the Mexican courts (R 426; see also Tr. at 430).

The mission to the World Health Organization had been told by some European contacts that "Laetrile" can be "purchased across the counter in Geneva without prescription" (R 426). The American Embassy in Switzerland, upon inquiry, was told that Laetrile is not sold on the Swiss market and is not approved there. One company does sell "small quantities of Laetrile," "exclusively to cancer research scientists" primarily in Western Europe. The company told the American embassy in Bern that "Laetrile is not made available commercially, nor is it sold as a cancer 'cure'" (id.).

In Madagascar, Laetrile is known as Amygdalin and is considered a poison by health authorities. Its use is prohibited. In Chile, Laetrile is also known as azaribina and its use is prohibited under any circumstance. This prohibition followed receipt of Newsletter 172 from the World Health Organization which described the potential dangers of use of the drug. The importation or use of Laetrile (Amygdalin) is illegal in the Republic of Korea (id.).

Health officials in Guyana reported that Laetrile has been used there. The Minister of Health indicated that he was not aware of FDA's prohibition of the use of Laetrile, however, and that, since the United States standards are closely followed in that country, his country would also ban the drug (R 426).

The State Department inquiry drew 69 responses from around the world. Each of the countries not already mentioned responded by indicating that "Laetrile," "Amygdalin," and "Vitamin B-17" were unknown or were not approved for use for treating cancer or any other use. The responding countries included the Philippines, Japan, the United Kingdom, Belgium, Italy, South Africa, and the Republic of Germany, as well as France, Korea, Taiwan, Hong Kong, India, and others from every part of the globe (id.). The United States Mission at the World Health Organization confirmed that Laetrile "is not registered and by definition, unavailable," in any of WHO's member states throughout western Europe (id.).

## B. CLAIMS THAT LAETRILE IS A VITAMIN OR FOOD

Proponents of Laetrile (or amygdalin) have in recent years contended that their product is a vitamin or that it is a natural food substance rather than a drug. These claims are properly irrelevant to the questions this administrative proceeding was intended to address. However, in light of the interest in the vitamin issue demonstrated by the submissions to the record, the Commissioner will take this opportunity to discuss it. The potential safety problems presented by this concept will also be discussed.

### 1. A Vitamin or Food May Be a Drug As Well

This question is irrelevant to the issues in this administrative proceeding because, even if Laetrile (or amygdalin or "laetrile") were a vitamin (or a food), it would still be a drug. Any substance, including a vitamin or food, is a drug and subject to regulation as such if it is intended for use in the "diagnosis, cure, mitigation, treatment, or prevention of disease in man or other animals \* (21 U.S.C. 321(g)(1)), (See Rutherford v. United States, supra, 542 F. 2d at 1140; United States v. General Research Lab-oratories, 397 F. Supp. 197, 200 (C.D. Cal. 1975).7 As the previous discussion illustrates, there is no question that Laetrile or amygdalin has been recommended in the treatment of cancer. In fact, in the very article in which Ernst T. Krebs, Jr., explained his theory that his product was "Vitamin B-17", he promoted it for can-cer treatment. (See R 183, Att. 10c.)

Section 411 of the act (21 U.S.C. 350) deals specifically with vitamins and minerals. Section 411(a)(1)(B) does limit the authority of the Secretary to classify a vitamin as a drug "solely because it exceeds the level of potency which the Secretary determines is nutritionally rational or useful." The vitamin provisions do not, however, affect FDA's authority to classify and regulate vitamins as drugs if they are represented to be for use in the diagnosis, cure, mitigation, treatment, or prevention of disease. The confer-ence committee report states, "Except as spe-cifically provided, the conference substitute alter the drug or food provisions of the Federal Food, Drug, and Cosmetic Act. If a product containing vitamins, minerals or other ingredients is a drug within the meaning of Section 201(g) of the Act, the Secretary may, with regard to such product, exercise his authority under Chapter V of the Act" H. R. Rep. No. 1005, 94th Cong. 2d. Sess. (April 2, 1976); (see also, 122 Cong. Rec. H3244-H3248, April 12, 1976).

It has been suggested that the claims that Lactrile (or amygdalin) is a vitamin or a food are simply an effort to establish that the substance is covered by the food requirements of the Federal Food, Drug, and Cosmetic Act and its regulations rather than those requirements applied to drugs. (See, generally, Tr. at 216, 225, 405; R 173, Att. "Questions most frequently asked about "Laetrile," at 1; R 416 at § 16.) One court has called the attempts by Laetrile proponents to represent Laetrile as something other than a drug, "a patently absurd and transparent attempt to avoid the drug labeling provisions of the Federal Food, Drug, and Cosmetic Act."
United States v. Spectro Foods Corp.,
Civil No. 76-101 (D.N.J., Jan. 29, 1976) (R 173, Att.). The Commissioner does not agree that Laetrile is a vitamin. (See discussion below.) It is clear, however, that even if Laetrile were a vitamin (or a food) it would be subject to the drug provisions of the act.

### 2. Is Laetrile a Vitamin?

This administrative proceeding was not intended to address the issue of whether Laetrile is a vitamin, and testimony on that issue was not solicited. Nevertheless, a considerable amount of evidence in the record addresses this issue. It appears that (a) Laetrile proponents classify amygdalin and certain related substances as a vitamin under their own definition of that term and (b) experts in the vitamin area, utilizing the criteria against which each of the legitimate vitamins have been assessed, conclude that amygdalin and other nitrilosides are not a vitamin.

(a) Proponents' Claims .- The idea that Laetrile could be considered a vitamin first appears in a pamphlet," in use in 1965, published by Krebs Laboratories and entitled "Cancer Is A Deficiency Disease: The Deficiency of Cyanide Sugars" (R 201, Ex. C, No. IV). In that pamphlet, amygdalin and other "cyanogenetic glucosides" are characterized as pro-vitamins for vitamin B-12. This means that they participate in the formation of vitamin B-12. It is stated that "during the process of formation the liver is thoroughly fumigated and rendered sterile (id.). The real anticancer effect of amygdalin is said to be not this formation but the release of cyanide in the cancer cells by the mechanism discussed above under "Theories of Action." It is interesting to note that another Laetrile proponent, Dr. Navarro of the Philippines, states that vitamin B-12 should never be administered to cancer patients (R 318 at 165).

The pro-vitamin theory had apparently been set aside by 1970 when an article by Ernst T. Krebs, Jr., referred to previously, "The Nitrilosides (Vitamin B-17—Their Nature, Occurrence and

The Krebs were no strangers to the "vitamin" area. Ernst T. Krebs, Jr.'s, marketing of another of his inventions, "Vitamin B-15" ("pangamic acid"), led to his plea of guilty to a charge of causing the introduction into interstate commerce of an unapproved new drug in 1962 (R 185, Att. 16, App. 17 at 7).

Metabolic Significance (Antineoplastic Vitamin B-17)," was published (R 183, Att. 10c). In this article, Krebs, Jr., uses the term "vitamin B-17 (nitriloside)" as 'a designation proposed to include a large group of water-soluble, essentially non-toxic, sugary compounds found in over 800 plants, many of which are edible" (id. at 75). He indicates that the compounds "are collectively known chemically as beta-cyanophoric glycosides. They comprise molecules made of sugar, hydrogen cyanide, a benzene ring or an acetone" (id.). These compounds could be hydrolyzed by beta-glucosides to a sugar, free hydrogen cyanide, and benzaldehyde or acetone (id.). He states that amygdalin is one of the most common of the nitrilosides and that it "occurs in the kernels or seeds of practically all fruits" (id.)

Mr. Krebs, Jr., in this article, attempts to build a theory that vitamin B-17 is a specific dietary factor that could be used to prevent and to cure cancer. He explains that prevention and cure occur through the cytotoxic (toxic to cells) compounds of vitamin B-17—hydrogen cyanide and benzaldehyde—through mechanisms discussed elsewhere in this opinion (id. at 80). He concludes that: "In nitriloside or vitamin B-17 we have a new vitamin in which all of us are severely deficient" (id. at 84). The theory that Laetrile (or laetriles) constitutes a vitamin has found another proponent in the person of Dr. Dean Burk, a blochemist and president of the Dean Burk Foundation, Inc. (see, generally, R.

302: Tr. at 401 et seq.).

Clearly, whether or not a given sub-stance comes within the definition of vitamin depends upon the definition chosen. In his affidavit, Burk defines a vitamin as a substance which is "virtually non-toxic, water-soluble, an exogenous nutrient or food factor, and active in relatively small, essentially catalytic, non-calorific amounts, and is essential or beneficial in normal metabolism and/or physiologic functioning to overcome deficiency lesions and symptoms of nutritional disease" (emphasis in original) (R 302 at 4). Dr. Burk continues that in animal experimentation "\* \* \* the deficiency lesions and symptoms of nutritional disease are best illustrated by the action of amygdalin in lengthening of animal lifetime or decreasing development of metastases, or both, and increase in health and well-being \* \* " (id.). (See also Tr. at 408 and 465.)

Thus, even by the Burk definition, the claim that Lactrile is a vitamin depends in large part on the substance's ability to combat cancer, an ability not shown by testing convicing to drug experts in general. Proponents of the vitamin theory claim that the higher the everyday diet is in nitrilosides, the lower the incidence of cancer (R 73 at 36). Others claim that it may not be only high nitril-oside levels that account for this observation but that other dietary elements (e.g., vitamin C) may play a role. (See, generally, R 318). These claims are based upon assertions that in some geographical areas, where the normal diet con-

tains nitrilosides in abundance, cancer does not exist. Evaluation of the prevalence of cancers requires careful studies by conpetent epidemiologists and suitable cancer registries, which contain reports by professional pathologists (R 399 at [9). What evidence does exist in this area indicates a complete lack of the correlation between high nitriloside diet and low cancer incidence that the Vitamin B-17 proponents claim. The record contains citations to numerous reports showing that a variety of cancers do occur in populations consuming nitriloside-containing diets. These include findings that cases of most of the recognized cancers appear in the Kampala Cancer Registry, Uganda (id.). There are also references to published papers from the Ibadon Cancer Registry, Nigeria on Burkitt's lymphosarcoma, Kaposis sarcoma and breast cancer, cancer of the bladder in Kenya, and the cancer incidence in Bantu (id.). Some cancers that are rare or absent in North America and Western Europe occur in the populations which consume high levels of nitrilosides (id.).

In the film "World Without Cancer." the people of the Kingdom of Hunza in the Himalayan mountains are said to eat a diet containing over two hundred times more nitrilosides than the average American diet and to prize above all other foods the apricot seed. It is stated that "[vlisiting medical teams from the outside world report that there never has been a case of cancer in Hunza" (R 198, Ex. 2 at 8). However, in 1955, a Japanese medical expedition studied the Hunza people and reported that they have many diseases, including cancer (R 173, Att. "Questions Most Frequently Asked" No.

6; Tr. at 338). Similarly, the film claims that the Eskimos eat a high nitriloside diet and are "found to be totally free of cancer" (R 198, Ex. 2 at 8). The Eskimos have also been found to have cancer (R 173, Att. "Questions Most Frequently Asked" No. 6: Tr. at 339). In commenting on another reference to "the diets of the cancer-free population" (R 217 at 2), Thomas H. Jukes, Ph.D., states correctly: "There are no cancer-free populations"

(R 416 at ¶ 28(C)).

(b) Vitamin Experts' Position .-Numerous nutrition experts and organizations concerned with nutritional science provide support in the record for the Commissioner's conclusions that "Laetrile," or "amygdalin," or "nitriloside," is not a nutrient or vitamin. (See, e.g., R 173, Att. "Questions Most Frequently Asked" (American Cancer Society); R 168, Att. "The Vitamin Fraud" and R 399 at § 12(C) (David M. Greenberg, Ph.D.); R 416 (Thomas H. Jukes, Ph.D.); R 378, Att. Editorial; Att. American Institute of Nutrition letter; R 169 at ¶ 17 (Vincent T. DeVita, Jr., M.D.); R 191 at § 11 (Philip S. Schein, M.D.); R 227 at 3 (National Council on Drugs).) The steps which are necessary to establish that a substance is a vitamin are described in the record. These steps include the publication in reputable journals of a complete description of the

research procedures, and confirmation, by other scientists, of the results obtained. If the work cannot be repeated. the existence of the vitamin is not recognized (see R 416 at ¶ 15), Additional steps include demonstrating the presence of the purported vitamin in foods, determination of its exact chemical molecular structure, the demonstration of its effectiveness, and its chemical synthesis. After these steps are completed, the Food and Nutrition Board of the National Academy of Sciences sets up Recommended Daily Allowances for the vitamin which then must be adopted by the Food and Drug Administration (id.)

The lack of scientific evidence of any effect, which has prevented Laetrile from being recognized by experts as a safe and effective drug, also prevents recognition of its claimed status as a vitamin. (See e.g., R 416 at 1 16: "(T) here are no data available to show that a disease state is produced or alleviated by the exclusion from (or) addition to the diet of amygdalin".) Other experts emphasize that there is no evidence that (1) laetriles (beta-cyanogenic glucosides, vitamin B-17) are essential nutritional components nor that (2) they promote any physiological process vital to the existence of any living organism. (See R 168,

Att. at 347; R 395.)

A compelling point made by experts in this area is that if there were a vitamin B-17, and if cancer were a vitamin B-17 deficiency disease, then every animal deprived of the vitamin would get cancer while no animal given the vitamin in sufficient amounts would get cancer. It is noted, that "No person given adequate vitamin C, for example, ever gets scurvy (R 198 at 5). Stated another way: "The key to the term 'vitamin' is that the absence of vitamins from the diet in an experimental animal or a human being must lead to the appearance of a nutritional deficiency disease, which is prevented or cured by adding the vitamin to the diet. Laetrile has no such property" (R 416 at § 14).

It is further stated that "(no vitamin has) the property of destroying tissue, such as cancer tissue, that is claimed of Laetrile. Such a property would be incompatible with the action of vitamins" (Tr. at 223). Other experts in this area rejected the claimed vitamin status in part on the grounds, discussed below, that amygdalin is, or can be, harmful to the body. It is pointed out that, amygdalin properly belongs to the class of compounds termed "toxicants occurring nat-urally in foods". (R 416 at \$ 27(A)(1)).

3. Dangers of Ingestion of "Vitamin B-17"

As has been demonstrated above, the presence of beta-glucosidase, oxynitralase, and amygdalin together in apricot pits presents the potential for a combination that would release cyanide and cause poisoning of the individual consuming an extract of the pits. Additionally, though beta-glucosidase is not present in animal tissues, it, and other substances capable of breaking down amygdalin, may be present in the digestive track and thus may break down orally consumed amygdalin to release cyanide in the body. It is therefore with great concern that the Commissioner views the emergence of the theory that human beings should step up their consumption of "naturally occurring" nitrilosides such as amygdalin.

In his vitamin B-17 article, Ernst T. Krebs, Jr., notes that while the "stupidity" of "political power" may keep prepared vitamin B-17 off the market, six or seven teaspoonsful of defatted apricot seed or kernel would supply what is considered to be an adequate oral dose of nitrilosides (R 183, Att. 10c at 84). He suggested, "It is best that the beta-glucosidase enzyme be completely heat inactivated in such material" (id.). He does not indicate how much heat inactivation should be accomplished, nor does he cite any support for the idea that it can be.

There are documented cases of poisoning, some fatal, due to the consumption of apricot pits or kernels. The toxic element in these cases is the hydrogen cyanide which is released from the cyanogenetic glucosides by the action of the enzymes (including beta-glucosidase) present in apricot kernels (R 378, Att. "California Morbidity Reports," Att.
"Hazards to Health"). The suggestion by Mr. Krebs, Jr., that the beta-glucosidase be "inactivated" can be taken as tacittoo tacit-acknowledgement that the kernels and/or pits present a hazard when consumed unless the enzymes are first destroyed. The Commissioner notes that other proponents of Laetrile clearly state that amygdalin products should never be given by mouth because the hydrochloric acid in the stomach is capable of hydrolyzing the drug (R 318 at 158). A 1954 document, "The Rationale and Clinical Evaluation of Laetrile-Beta-Glucosidase Palliative Therapy" states, "CAUTION: Lactrile (1-mandelonitrilebeta-glucuronidase) is NOT TO BE TAKEN ORALLY. It is extremely toxic by this route of administration, since the gastric hydrochloric acid acts to hydrolize the glucoside with the release of hydrogen cyanide" (R 388, Ex. 5).

Dr. Burk seeks to support the idea that Laetrile or amygdalin is a food and, among other things, may be safely con-sumed by an allegation that " \* \* laetrile is listed in the HEW-FDA GRAS list (foods 'Generaly Regarded (sic) as Safe') under the heading of natural extractive from bitter almond, apricot or peach kernels" (R 302 at 3; see also Ex. B). The material to which Dr. Burk refers is "Bitter almond (free from prussic acid)" which does appear on the generally recognized as safe (GRAS) 21 CFR 182.20. (Prussic acid is another name for hydrogen cyanide.) The material on the GRAS list, however, is an oil extracted from peach, almond, or apricot kernels. After cold pressing the oil from its source, it is processed to effect enzymatic hydrolysis of amigdalin. There is no amygdalin present after the hydrolysis step. The final product is essentially benzaldehyde, "Thus the material listed for flavor use \* \* \* is not

amgydalin and thus neither is it Laetrile" (R 415 at 2). Dr. Burk's contention is thus incorrect and has no basis in fact.

The idea that foods containing nitrilosides may be safely consumed is also supported by stories of "non-toxic" nature of nitrilosides which are found in the diets of various peoples. This claimed nontoxicity is not borne out by reality. In some parts of Africa two important human diseases-human ataxic neuropathy and endemc goitre-appear to be associated with high cassava intake (R 183, Att. 20 at 161; R 378, Att. 6). (Cassava contains linamarin, a commonly consumed nitriloside (R 217, Att. "Sickle Cell Anemia" at 51).) It is also reported that cows have been killed by eating large amounts of young millet, which is particularly high in nitrilosides, and intoxication of other animals has been reported. (See R 416, Ex. 5 at 302.) There have been reports in this country of toxicity and, in some cases fatalities, in humans from consumption of amygdalin-containing substances (See, e.g., R 378, Att. "California Morbidity"; see also R 378, Att. "Hazards to Health; Cyanide Poisoning from Apricot Seeds Among Children in Central Turkey, Sayer et al.). Thus there is ample and clear indication that the consumption of "nitrilosides" is not without hazard. To urge the public to consume "apricot pit milkshakes" or similar foods in order to be sure to get an ample supply of amygdalin or "vitamin B-17" or "nitrilosides" is irresponsible and foolhardy.

### C. FREEDOM OF CHOICE

The administrative record contains many comments not directed to the legal issues of Laetrile's "new drug" or "grandfather" status or even to the question of whether the drug is safe or effective for use in cancer therapy. Rather these comments support the proposition that a person should be free to choose his or her own cancer therapy, at least if the drugs involved are not overtly toxic (see, e.g., Tr. at 33; R 231; R 238; R 242; R 209; R 211; R 283; R 500; R 155; R 272). The issue of "freedom of choice" is irrelevant to the issues remanded to the agency by the Rutherford courts. Nevertheless, because of the demonstrated public interest in this issue, the Commissioner has given it, and submissions addressing it, careful consideration.

The very act of forming a government, of course, necessarily involves the yielding of some freedoms in order to obtain others, In passing the 1962 Amendments to the act-the amendments that require that a drug be proved effective before it may be marketed-Congress indicated its conclusions that the absolute freedom to choose an ineffective drug was properly surrendered in exchange for the freedom from the danger to each person's health and well-being from the sale and use of worthless drugs. This is in fact the same decision made by those in government who have decided over the years that only those persons may practice medi-cine who have been certified by experts to be qualified to actually help the patients who would choose to seek their assistance.

Some would argue that the lawmakers' well-considered decision to prohibit the use of drugs not shown to be effective was the wrong one. One alternative suggested is that a drug such as Laetrile should be marketed with labeling which indicates that experts do not consider it to be effective. The present use of Laetrile vividty illustrates the impracticability of such a solution. There can be few patients taking Laetrile in this country today who do not know that the government and most experts consider it worthless. Yet the drug continues to be used, to the detriment of cancer patients who might otherwise be helped by conventional treatment.

39803

The choice to use Laetrile is seldom. in any case, a free one. As the discussion above (Why Do People Use Lactrile?) illustrates, a cancer patient is a person beset by immense stresses, physical, psychological, emotional and societal: and the persuasion that the patient and his family are subjected to by Laetrile proponents is seldom limited to a rational laying out of competing arguments. The information that the proponents of Laetrile provide is false—that the drug cures, palliates, relieves pain, "controls" cancer. As Dr. Sampson's survey, discussed above, indicates, few Lactrile patients make an effort to hear the argument against Laetrile therapy. The idea that a reasoned free choice is involved in the selection of Laetrile rather than legitimate therapy is thus ultimately an illusion, (See R 421 at 1.)

The record contains the views of many persons who have considered the issue of "freedom of choice" in cancer therapy. Each represents a thoughtful attempt to deal with this question, which, while irrelevant to the legal issues which are the subject of this opinion, is troubling to those concerned with the Laetrile problem. These comments may be grouped roughly as supporting the two responses to the "freedom of choice" argument set forth above: (1) The surrender of an absolute freedom to choose among cancer remedies in order to obtain the greater freedom from the suffering associated with use of ineffective remedies is a rational decision; and (2) the "choice" of unproven cancer remedies cannot fairly be characterized as "free."

### 1. Balancing Freedoms

Reverend Allan W. Reed, Director, Department of Pastoral Services and School for Pastoral Care, Massachusetts General Hospital, considered the ethical side of the "freedom of choice" question. He states "The ethical issues of a group of legislating for an individual, thereby threatening the principal (of) freedom of choice, contrasts with the ethical principal of a government protecting its citizens from fraud and abuse. In the case of a drug for which there is no proven efficacy, the ethical weight is on the side of protection of the citizens" (R 148).

Leroy G. Kerney, Chief of the Department of Spiritual Ministry at the Clinical Center, National Institutes of Health, pointed out that, "Freedom of the indi-

vidual is important. But when the freedom to accept any drug for treatment and the freedom to injure oneself collide, a judgement must be made: Stop signs or restrictions on turning at certain corners restrict my freedom in driving. but, at the same time, they protect my freedom from hurting myself and others

in traffic" (R 414 at 3).

J. Philip Wogaman is Dean and Professor of Christian Social Ethics at the Wesley Theological Seminary and past president of the American Society of Christian Ethics. He noted an "initial presumption" in favor of freedom from governmental prohibition but concluded that Lactrile should be banned for three reasons: (1) The ban prevents fraud in the medical marketplace; (2) "[M]isrepresentation in the field of medicine is particularly serious because it undermines public confidence in medicines that are of real value"; (3) "(T) here is a real danger that persons may be led by false hopes in a worthless drug to neglect treatment at a time when it could be most effective" (R 417 at 2-3).

Dr. James F. Holland of the Mount Sinai School of Medicine points out that the freedom achieved by regulation of drug products is often the freedom to live: "For the patient ignorant of the inertness of Laetrile as an anticancer drug, there is an overriding concern that he not be denied his individual freedom by untimely death from cancer from having relied on Lactrile to help. This is a cruel deprivation of individual freedom, since the patient does not get a sec-

ond chance" (R 396 at 2).

James Harvey Young, Ph.D., a historian of health quackery, discussed the past use of the freedom of choice concept and phrased his conclusions concerning the validity of application of that concept to health care in colorful terms. He states that acceptance of the primacy of the freedom to choose medical therapies "leads only toward the license of those ancient days, when 'the toadstool millionaires,' preaching religion and spouting patriotism, operating without restraint, fleeced and often killed their gullible victims. That is a fate from which seven decades of constructive legislation, beginning with the Pure Food and Drugs Act of 1906, has somewhat rescued the nation. Complex, modern, industrial, urbanized society, with standards of medical judgment far more precise than those existing in the nineteenth century, cannot afford to let the nation's health concerns be governed by a distored definition of that great symbol, 'freedom', which would return piratical anarchy to the realm of health" (R 400 at 11-12).

### 2. The Choice Is Not Free

The discussion above of "Why People Use Laetrile?" describes the many pressures that induce cancer patients and their families to make the decision to use Laetrile. Orville Eugene Kelly is a cancer patient who has founded an organization called "Make Today Count," which now has 103 chapters in 30 states, to help other cancer patients and their families

deal with the problems that discovery of cancer entails. He addressed the question of "freedom of choice" in an affidavit submitted to the record (R 389). He notes from personal experience that patients and their families are often susceptible to arguments that a painless drug like Laetrile can cure them (id. at 2) and describes the persistence with which those arguments are made. He himself has tried to present the counter-arguments to other cancer patients. "But it is difficult to convince some of these people that the substance Laetrile is ineffective as a therapy for cancer when they have watched a film, listened to tapes, and heard testimonials from other patients, quite sincere in their beliefs that Laetrile has helped them" (id. at 3). He asks: "(I)s it a fair choice if (the cancer patients) are being pressured by Lactrile proponents?" (id.) .

The constant efforts of Laetrile proponents are emphasized by those dealing with cancer patients. See, e.g., statement of Helene Brown, Executive Director of Community Cancer Control, Los Angeles, "that far from exercising a free and informed choice patients are confronted with enormous pressures to use Laetrile instead of conventional forms of therapy and that representatives attesting to the worth of Laetrile make untrue, misleading and unsubstantiated claims" (R 393

The presentations of the Lastrile proponents are made, as discussed in more detail elsewhere, to patients and families deprived of their normal decisionmaking abilities. See statement of John J. Dawson, M. Div., Director, Patient and Family Support, Mountain States Tumor Institute: "Research conducted at the Mountain States Tumor Institute and elsewhere indicates that the emotional trauma of a cancer diagnosis severely impairs the patient's and families' ability to engage in rational decisionmaking processes" (R 421 at 1).

Other submissions reflect a similar conclusion, see the statement of Rev. Reed: "The ability of (cancer patients and their families) to protect themselves is often severely limited by the emotional situation in which they find themselves" (R 418). (See also R 414 at 4; R 433 at 1 14.)

The Commissioner thus concludes as

(1) To the extent that any freedom has been surrendered by the passage of the legislation which bans from the marketplace drugs that have not been proven to be effective, that surrender was a rational decision which has resulted in the achievement of a greater freedom from the dangers to health and welfare represented by such drugs.

(2) The choice of Laetrile therapy, by persons under the severe stresses associated with discovery of cancer and in response to misinformation presented persuasively by Laetrile's proponents, cannot be regarded as a choice which is

### D. ALLEGATIONS OF BIAS

Several submissions charged that FDA is too biased against Laetrile to conduct

a fair hearing. Aside from general allegations of bias (R 313; R 248; R 353; R 507, R 302; R 73, Att. at 43; Tr. at 444-45), these submissions fall into two general categories: (1) the administrative rulemaking proceeding should have been conducted by someone other than FDA (R 144; R 505; R 222; Tr. at 12, 29, 75, 444-45); and, (2) the drug approval process administered by FDA is wrong (R 235; R 258; R 144; R 509; Tr. Ex. 1)

It is difficult for an agency charged with bias to rebut such charges persuasively. Nonetheless, the Commissioner feels that a complete decision requires some rebuttal of charges that he regards as erroneous and misdirected. Insofar as comments suggesting that the proceedings should have been conducted by someone other than FDA, it should be noted that FDA was required by court order to assemble an administrative record and make appropriate determinations therefrom. The task could not have been delegated to anyone else, and, even had the agency been able to do so, it is not likely that any tribunal chosen by FDA would have satisfied those persons who are convinced that the agency is biased. The FDA is, of course, the agency designated by Congress to evaluate the safety and effectiveness of drugs and, as such, it is the agency with expertise in this area.

The comments charging that the requirements for drug approval administered by FDA are wrong contained statements to the effect that testimonial evidence should be accepted as adequate proof of safety and effectiveness or that the cost of a clinical trial is too great a burden for the proponents of Laetrile to bear. Lactrile proponents place particular emphasis on the cost factors, stating that because clinical trials are expensive. FDA somehow favors only large drug companies.

As has been discussed above, FDA is bound by the requirements of law regarding drug safety and effectiveness. Those requirements have been challenged in court before, by the very drug companies toward whom Laetrile proponents allege FDA has a positive, favorable bias. These "favored" groups did not prevail, and the safety and effectiveness provisions were upheld.

In Pharmaceutical Manufacturers Ass'n v. Richardson, supra, a trade association whose membership includes major drug firms sought to enjoin the FDA regulations establishing the standards of evidence necessary to demonstrate the effectiveness of drug products (21 CFR 314.111). Pointing out that Congress could not have had testimonial evidence. clinical impressions, practical experience, or the unsubstantiated subjective views of medical practitioners in mind when it defined "substantial evidence." the court upheld the regulations. As one witness in the case pointed out, the approach which assumed that a collection of impressions would furnish the truth. "did not prevent doctors from having unbounded faith in the curative powers of leeches for hundreds of years before scientific evaluation became the preferred means of judging efficacy of ther-

apy" (318 F. Supp. at 307).

In Upjohn Company v. Finch, 422 F.2d 944 (6th Cir. 1970), a drug manufacturer sought review of an FDA order revoking marketing approval for seven combination antibiotic drugs. Stating that testimonial evidence was not enough to meet the standard of substantial evidence, the court held that "the record of commercial success of the drugs in question and their widespread acceptance by the medical profession, do not, standing alone, meet the standards of substantial evidence, prescribed by 21 U.S.C. 355(d)" (422 F. 2d at 954). Although the cost of developing the proper scientific evidence of safety and effectiveness is high, placing this burden upon those who wish to sell drugs is more than justified by the need to protect the consumer from harmful, useless, and fraudulent drugs.

The Commissioner acknowledges that the FDA is biased in one sense; the agency is committed to requiring that drugs meet the standards of safety and effectiveness required by law. The standards are designed to protect the public from drugs which are not both safe and effective. While the standards are rigorous, they are not mysterious. They are accepted by the scientific community and can be applied by any scientists who seriously wants to prove the value of a drug. The proponents of Laetrile choose to attack the standards. They have not attempted to meet them.

### E. LIMITING USE TO TERMINAL PATIENTS

There has been concern expressed in the submissions to the record that Laetrile might be approved for use by "terminal" cancer patients. Such an approval would be theoretically justified only on the grounds that since such patients might be considered beyond the help of other therapies, Laetrile cannot hurt them. Approval of a drug for use by terminal patients is not possible under the act; however, in light of the interest in this issue the Commissioner will discuss the evidence relating to it.

One submission objected to the possible use of Laetrile by terminal patients on the grounds that approval of such use constitutes sanction of an inhumane fraud upon the patients involved, one which wastes the financial resources of the patients and their families uselessly (R 190 at § 17). Two other arguments were expressed by a number of submitters: (1) there is no such thing as a "terminal" patient and (2) allowing use by a subgroup of cancer patients would lead to increased use by patients who could be helped by legitimate therapy.

### 1. Who is Terminal?

Dr. Peter H. Wiernik, Chief of the Clinical Oncology Branch of the National Cancer Institute's Baltimore Cancer Research Center, states, "One major difficulty in making a particular chemical available for terminal patients only, is that no one can prospectively define the term 'terminal' with any accuracy. A patient can be said to be terminal only after he dies. Many patients who are critically ill respond to modern day management of cancer" (R 200 at § 18).

D. Joseph F. Ross, Professor of Medicine at the University of California School of Medicine at Los Angeles, is actively involved in the medical care of cancer patients. He states, "[T]he distinction of 'terminal' patients from 'nonterminal' patients may not be reliably determined and an assumption that Laetrile may be given to such patients with impunity may deprive such patients of therapeutic measures which could help them" (R 190 at § 17). Cf. R 393, Ex. 1 at 2: "Medical history is full of miracles." "No one knows if and when any patient is going to die." (Helene Brown, Executive Director of Cancer Control/Los Angeles); see also R 173, Att. "Questions Most Frequently Asked

### 2. Effects on Other Patients

Approval for use of Laetrile by "terminal" patients, assuming some way could be found to define that class of individuals so as to exclude all those who might be helped by legitimate therapy, would still pose a risk to other patients who could be helped. This effect would, the evidence in the record shows, occur in two ways. First, approval for even this limited use would encourage illegitimate use of the type now occurring in this country.

Historian James Harvey Young, based upon his study of past "unproven" medical cures, states, "Permitting Lactrile's use in terminal cases gives it a credence among the public at large that will expand its use in early cases, for people will prefer taking a 'vitamin' to con-fronting the surgeon's knife" (R 400 at 11)

Dr. Samuel C. Klagsbrun, a psychiatrist who works with cancer patients at St. Luke's Hospital in New York, states, "Permitting Laetrile to be used by any population of cancer victims would have the correlative effect of creating the misimpression in the minds of other cancer victims that the drugs is, in fact, safe and effective for a broader population" (R 433 at 1 12)

A second danger from such a limited approval of Laetrile is that the limitation would be extremely difficult to enforce. Kenneth A. Durrin, Acting Director, Office of Compliance and Regulatory Affairs, Drug Enforcement Administration, submitted an affidavit describing the detailed and costly regulation of "controlled substances" under the Controlled Substances Act (CSA) and then considered the possibility of approval of Lactrile "for terminal patients only" (R 435). He stated his conclusion as to the practicality of preventing the diversion of Laetrile from "terminal" patients, if approved for such patients, to others who might be helped by legitimate therapy: "Absent the kinds of controls available under the CSA-and indeed even with such controls-it is my opinion that a drug such as Laetrile could not effectively be restricted to a class of terminally ill cancer patients. For example, absent a quota on production, manufacturers would not be limited to producing an amount of Laetrile sufficient only to provide a source of supply for terminally ill cancer patients. Manufacturers would not be restricted in the channels in which they could permissibly distribute thn drug. They would not be required to report transactions in Laetrile. The amount of Laetrile which could be imported into this country would be unlimited.

39805

"Given such unrestricted and unfettered availability of Laetrile, it is my opinion that there would be no practical way of limiting access to the drug to terminally ill cancer patients only. It is completely unrealistic to suggest that any other result would occur" (id. at 118-19)

The Commissioner concludes that approval of Laetrile restricted to "termipatients would lead to needless deaths and suffering among (1) patients characterized as "terminal" who could actually be helped by legitimate therapy and (2) patients clearly susceptible to the benefits of legitimate therapy who would be misled as to Laetriles' utility by the limited approval program or who would be able to obtain the drug through the inevitable leakage in any system set up to administer such a program.

### F. USE CONCURRENTLY WITH OTHER THERAPY

Some persons not familiar with the problem of drug interactions have suggested that Laetrile might be approved for use concurrently with legitimate cancer therapy. This theory would logically extend to allow any worthless drug to be used as long as effective therapy was also utilized. Such a limited use program would, of course, involve the problems of administration discussed in the previous section. Particularly in light of the Laetrile proponents' practice of dissuading patients from what they characterize as the "cut, burn, and poison" techniques of legitimate therapy, any seeming government sanction of Laetrile would inevitably involve encouragement of use of "painless" Laetrile therapy alone and thus would result in needless suffering and loss of life (cf. R 191 at 117).

More important, it simply has not been shown by any sound scientific evidence that the administration of Laetrile along with other therapy may not either make such therapy more dangerous or interfere with its effects, Dr. James F. Holland states (R 396 at 2), "That Laetrile is inert as an anticancer drug does not mean it may not interfere with the metabolism of and compromise the effects from known anticancer treatments. This would require years of study to elucidate, and it is not a worthwhile undertaking since Laetrile itself has no anticancer activity. One does not seek further information on why not to use Laetrile. If there is no good reason to do something, the best reason exists not to do it" (emphasis in original). Thus, the same reasons that justify the law's ban on use of drugs not shown to be effective form an equally strong basis for

the ban on that use where the use will be concurrent with other therapy.

### VI. CONCLUSIONS

The Commissioner, after careful review of the administrative record amassed in this rule making proceeding, makes the following conclusions:

(1) Although the terms "Laetrile," "laetrile," "amygdalin," "Sarcarcinase," "vitamin B-17," and "nitriloside" have been used interchangebly, the chemical identity of the substances to which these terms refer has varied over the years. The identity of material referred to or called by any of those names is often not known. All too frequently terms have been used haphazardly or imprecisely by proponents, as well as opponents, of Laetrile:

"Laetrile," as described by Ernst T. Krebs, Jr., is: 1-mandelonitrile-beta-

glucuronic acid.

"Amygdalin" is: D-mandelonitrilebeta-D-glucosido-6-beta-D-glucoside.

"Sarcarcinase" is the name given by Dr. E. T. Krebs, Sr., to a mixture of 6, possibly more, enzymes extracted from apricot pits.

(2) Neither Lactrile nor any other drug called by the various terms mentioned above nor any other product which might be characterized as a "nitriloside" is generally recognized by experts qualified by scientific training and experience to evaluate the safety and effectiveness of drugs to be safe and effective for any therapeutic use.

(3) Animal studies conducted to date show that Laetrile has no anticancer activity in laboratory animals. Even if such activity were shown, the data would not be relevant to the issue of whether Laetrile is generally recognized by qualified experts as a safe and effective anticancer drug in humans.

(4) Neither Laetrile, amygdalin, nor any other drug called by the various terms set out in conclusion (1) is exempted from the "new drug" definitions of the act (21 U.S.C. 321(p)) by virtue of compliance with either the "1938 grandfather clause" (21 U.S.C. 321(p)(1)) or the "1962 grandfather clause" (section

107(c) (4) of Pub. L. 87–781).

(5) The history and promotion of Laetrile are characteristic of other unproven cancer remedies. Laetrile's popular acceptance by laymen lies not in credible proof of its effectiveness, but rather in the fears of orthodox medical treatment and the false hope, fostered by Laetrile's proponents, that suffering and eventual death can be avoided through Laetrile.

(6) Laetrile is not in general use as cancer therapy anywhere in the world.

(7) There is no evidence that "Vitamin B-17" is generally recognized among experts in the field of nutrition or nutrition research as a vitamin. Even if there were such recognition, "Vitamin B-17" would still be subject to regulation as a drug under the Federal Food, Drug, and Cosmetic Act because of the claims made for its use in cancer therapy.

(8) The safety of ingesting amygdalin, Laetrile and/or apricot or peach kernels or pits has not been established. There is, in fact, evidence of frank toxicity from ingestion of the kernels or pits.

(9) There is no basis in law or in fact for the use of Laetrile or related substances in the treatment of cancer.

The foregoing opinion in its entirety constitutes the Commissioner's findings of facts and conclusion of law. Distribution of Laetrile, amygdalin, or any other substance called by the various terms set out in conclusion (1) in interstate commerce is in violation of the Federal Food, Drug, and Cosmetic Act and subject to regulatory action.

Dated: July 29, 1977.

DONALD KENNEDY, Commissioner of Food and Drugs.

[FR Doc.77-22310 Filed 8-4-77;10:00 am]

FRIDAY, AUGUST 5, 1977
PART III



# DEPARTMENT OF JUSTICE

PAROLE, RELEASE,
SUPERVISION AND
RECOMMITMENT OF
PRISONERS, YOUTH
OFFENDERS, AND
JUVENILE DELINQUENTS

Title 28—Judicial Administration

CHAPTER I-DEPARTMENT OF JUSTICE

PART 2—PAROLE, RELEASE, SUPERVI-SION AND RECOMMITMENT OF PRISON-ERS, YOUTH OFFENDERS, AND JUVE-NILE DELINQUENTS

Paroling, Recommitting and Supervising Federal Prisoners

AGENCY: The United States Parole Commission, Justice.

ACTION: Final rules.

SUMMARY: The Parole Commission has adopted a procedure whereby federal prisoners will be notified of their ultimate release dates at the outset of their terms of imprisonment. This procedure is necessary to reduce the degree of uncertainty with which federal prisoners presently serve their sentences of imprisonment. The purpose of the procedure is to achieve a significant degree of certainty while not foregoing the advantageous features of the parole system.

EFFECTIVE DATE: September 6, 1977, for all prisoners sentenced on that date or thereafter. For prisoners sentenced prior to September 6, 1977, the substantive provisions of the amended rules (setting presumptive release dates) will apply at the next scheduled in-person hearing (initial, review, rescission, or revocation).

FOR FURTHER INFORMATION CONTACT:

Michael A. Stover, Office of the General Counsel, United States Parole Commission, 320 First Street, NW., Washington, D.C. 20537, telephone 202-724-3092.

SUPPLEMENTARY INFORMATION:

(A) THE PROPOSAL AND ITS PURPOSE

On June 10, 1977, the United States Parole Commission published a proposal whereby Federal prisoners would be notified of their ultimate release dates at the outset of their terms of imprison-ment (42 FR 29934). The purpose of this proposal, which is now adopted as a final rule, was to achieve a substantial reduction of indeterminacy in Federal prison sentences (i.e. increasing certainty on the part of the prisoner as to what his total incarceration will be), without foregoing the significant advantages of the present federal parole system. Among the advantages offered by the federal parole system are: (1) Release decisionmaking by a small, independent, collegial body of correctional experts adhering to a national parole policy (promoting reduction of unwarranted disparity in punishments); and (2) the ability to account for intervening factors not foreseeable at the time of sentencing, through systematic review of each prisoner's case (promoting fairness to the individual and avoiding excessive use of confinement).

### (B) PUBLIC COMMENT

The proposal prompted numerous letters from the public, among which were letters from prisoners, families of prisoners, corrections officials, probation officers, one legislator, and one prison ministry organization. The majority of these letters endorsed the proposal because of the certainty it would bring to prisoners and to those awaiting a prisoner's return to society, thus increasing the stability of the prisoner's community support.

Corrections officials also favored the proposal. Chairman Ira Blalock of the Oregon State Parole Board wrote that a similar system adopted in Oregon has been administratively successful and generally beneficial. Chairman Blalock also pointed out that the proposal was consistent with the recommendations of criminologists and of the American Bar Association's Draft Standards Relating to the Legal Status of Prisoners (American Criminal Law Review, January 1977). Assistant Director Roy Gerard of the Federal Bureau of Prisons wrote that the new system will make prison management easier and more efficient, in-mates will be better informed and less anxious, and the processing of parole procedures will be improved.

Finally, Representative Robert W. Kastenmeier, Chairman of the House Subcommittee on the Courts, Civil Liberties, and the Administration of Justice, and a key architect of the Parole Commission and Reorganization Act of 1976, wrote:

I whole-heartedly endorse such a new rule. I believe it is entirely consistent with the intent of Congress that federal prisoners be provided with clear, consistent parole policies which will permit them to know at an early date when they can expect release.

This comment shared the position taken by the 1977 Report of the Senate Subcommittee on National Penitentiaries:

First, the subcommittee must see that the Parole Commission continues to administer the guidelines system in a way that is consistent with the intention that indeterminancy be reduced to the extent consistent with the law.

The legislation attempted to provide inmates with knowledge of their parole status, so that the typical inmate would know the prospective time of his release, plan for this, and not make release plans when he has no hope of early release. The legislation attempts to achieve this without creating procedural requirements that would be the basis for extensive and continuous litigation. At present, prospective parole information is not being given to all the prison population, and this would be of future concern to the Subcommittee. [at page 2 of the Report]

The most common criticism of the proposal from prisoners was its distinction between sentences of less than seven years and sentences of seven years or more. The effects of this demarcation are that: (1) A prisoner with a sentence of seven years or more and a minimum

term of imprisonment must await the completion of his minimum term before receiving his initial hearing, whereas all prisoners with sentences of less than seven years receive an initial hearing at the outset of incarceration; and (2) in all of the longer sentences, a presumptive release date cannot be set if it would result in a date more than four years from the date of the initial hearing.

The seven-year mark as a divider between short and long sentences for the purpose of setting a prisoner's entitlement to hearings is a figure already selected by Congress at 18 U.S.C. 4208(h) That section uses the seven-year mark to distinguish between those sentences in which interim hearings are required every eighteen months and those sentences (of seven years or more) in which interim hearings are required every twenty-four months. Moreover, the Commission decided that, for its present administrative purposes, a four-year effective limit on the setting of presumptive release dates is a practical restriction, as well as one which coincides with the statutory scheme. Whether the limit may be expanded in the future is a question which the Commission reserves for further deliberation.

Other comments urged that the Commission adopt a similar policy with regard to federal parolees serving new federal sentences for crimes committed while on parole, by informing such prisoners at the outset of the total combined length of the new confinement and the consecutive parole violator term (the remaining time on the original sentence). This proposal raises substantial questions beyond the scope of the present rule-making (for example, the problem of federal parolees serving new state sentences, in whose situations the Commission could not set a combined release date). However, the proposal will be taken under study.

One comment suggested that the Commission's plan contained an inherent paradox, stating that "\* \* if the purpose of parole is to determine the extent of rehabilitation and fitness for return to society, how can the I Commission I make such determinations without a longer period of incarceration?" The point this writer missed is that, in the federal system, seriousness of the offense and likelihood of favorable parole outcome are the principal standards by which parole decisions are made. (See 18 U.S.C. 4206). A prisoner's release date is not tied to the outward indicia of his rehabilitative efforts.

### (C) CHANGES FROM THE PROPOSAL

The proposal was adopted substantially as set forth in the Federal Register of June 10, 1977, with one exception. The five-year limit on presumptive release dates in the case of sentences of seven years or more was reduced to four years, in order to coincide with the occurrence of the second interim (statu-

tory) review hearing at forty-eight months from the initial hearing. Thus, in a case in which no presumptive release date was set at the initial hearing, the second interim review hearing would be conducted as a four-year reconsideration hearing pursuant to §§ 2.12(c)(2) and 2.14(c).

The amended rules also make clear that the formal rescission procedures of § 2.34 apply to presumptive parole dates. a point not covered in the proposal. This consistent with the Commission's statement that the intent of the proposal is that release will normally be granted at the presumptive date (42 FR 29934). By the same token, the amended rules also contain a clear statement that once set, a presumptive release date shall not be advanced except under clearly exceptional circumstances.

### (D) SUMMARY OF THE PRINCIPAL AMENDMENTS

Adoption of this proposal required numerous conforming amendments, in addition to the substantive changes. For the convenience of the reader, the Commission's rules (together with changes effected by accompanying documents) are republished in their entirety. A summary of the principal amendments covered by this document follows.

In § 2.1, the term effective date of parole is defined to distinguish that term from the term presumptive parole date. An effective date of parole is a parole date that has been approved following an in-person hearing held within six months of such date, or following a prerelease record review. Thus, a presumptive parole date become an effective date of parole when approved following a prerelease record review, or when approved following an interim hearing which is held within six months of the presumptive parole date. However, the term effective date of parole also includes the familiar grant of parole with a few months delay for the development of a release plan. The term presumptive release date encompasses both presumptive release by parole (a presumptive parole date), as well as presumptive re-lease through the accumulation of good time (mandatory release pursuant to 18 U.S.C. 4163 and 4164).

In § 2.12, the principal features of the proposal (the holding of early initial hearings and the setting of presumptive release dates) are codified. The reader should not fail to note that the setting of presumptive release dates (either by parole or by mandatory release) will be pursuant to the Commission's guidelines at § 2.20 (including decisions above or below the guideline ranges).

In § 2.13, a number of provisions relating to the conduct of the initial hearing as restructured. The only substantive change is the requirement that if a release date is set in excess of six months from the date of the hearing, reasons must be given as in the case of a parole

In § 2.14, the three types of proceedings subsequent to the initial hearing are fully described: Interim hearings pursuant to 18 U.S.C. 4208(h); pre-release reviews; and four-year reconsideration hearings. It is important to note that under no circumstances will a prisoner go without the periodic reviews to which he is entitled by section 4208(h).

In § 2.29, the terms of an effective date of parole are set forth. (The good conduct condition is omitted since it is already contained in § 2.34.)

In § 2.34, the amendment at paragraph (a) (3) permits the Commission to defer consideration of disciplinary infractions until the commencement of the next interim hearing or the pre-release review required by § 21.4(b). Since, as a practical policy, the Commission considers only those disciplinary infractions that have been the subject of formal findings following an Institutional Disciplinary Committee hearing, a delay until the next scheduled review will not operate to the prisoner's disadvantage (through loss of evidence, etc.).

### (E) EFFECTIVE DATE

These amended rules will become effective as follows: (1) In the case of prisoners sentenced on September 6. 1977, or thereafter (including prisoners with one or more multiple sentences imposed on September 6, 1977, or thereafter), all provisions of the amended rules shall apply from the initial hearing onward;

(1) In the case of prisoners sentenced prior to September 6, 1977, the amended rules will apply, excepting the provisions of § 2.12(a), at the first regularly scheduled in-person hearing that is held on September 6, 1977, or thereafter. Thus, following the first hearing (initial, review, rescission, or revocation hearing) that is held on September 6, 1977, or thereafter, each prisoner sentenced prior to September 6, 1977, will be notified of a presumptive release date according to the procedures of \$ 2.12 (c). (d), and (e), and related provisions.

### (F) FURTHER CONSIDERATION OF THESE AMENDED RULES

The Commission intends to evaluate the first four months of the operation of these rules at its meeting in January. 1978. Therefore, public comment by interested persons will continue to be welcome and will be considered at that time.

### (G) CONCLUSION

Accordingly, pursuant to the provisions of 18 U.S.C. 4203(a) (1) and 4204 (a) (6), 28 CFR Chapter I, Part 2, is amended as set forth below to become effective in the manner described above.

Dated: August 2, 1977.

CURTIS C. CRAWFORD, Acting Chairman, Parole Commission.

Sec.

2.1 Definitions.

Eligibility for parole, adult sentences. Same; Narcotic Addict Rehabilitation 2.2 2.3

Same; juvenile delinquents. Same; youth offenders.

Withheld and forfeited good time. 2.6

Committed fines.

Mental competency proceedings.

20 Study prior to sentencing.

2.10 Date service of sentence commences. Application for parole; notice of hear-

ing. Initial hearings: Setting presumptive release dates.

2.13 Initial hearing; procedure. 2.14

Subsequent hearings.

Petition for consideration of parole prior to date set at hearing. 2.15

Parole of prisoner in state, local, or territorial institution.

Original jurisdiction cases.

2.18 Granting of parole. Information considered.

Paroling policy guidelines; statement of general policy. Reparole consideration guidelines.

Communication with the Commission.

Delegation to hearing examiners. 2.24 Review of panel recommendation by the Regional Commissioner.

Regional Appeal.

2.26 Appeal to National Appeals Board. Appeal to original jurisdiction cases.

Reopening of cases. 2 20

Release on parole. 2.30

Palse or withheld information. 2.31 Parole to detainers; statement of

policy, policy to local or immigration 2.32

2.33 Release plans.

2.34 Rescission of parole.

2.35 Mandatory release in the absence of parole.

Same: youth offenders.

Reports to police departments of names of paroless; statement of 2.37 policy.

Community supervision by United States Probation Officers.

2.30 Jurisdiction of the Commission.

2.40 Conditions of release.

2.41 Travel by parolees and mandatory re-Icases.

Probation Officer's Reports to Commissions. 2:43

Early termination of parole. 2.44

Summons to appear or warrant for retaking of parolee. Same: youth offenders.

2.46 Execution of warrant and service of summons. 247 Warrant placed as a detainer and dis-

positional review.

Revocation by the Commission; pre-liminary interview.

Place of revocation hearing. 2.50 Revocation hearing procedure.

2.51 Issuance of subpoena for the appearance of witnesses or production of documents.

2.52 Revocation of parole or mandatory release.

2.53 Mandatory parole.

Reviews pursuant to 18 U.S.C. 11 4203/

2.55 Disclosure of records.

2.56 Special parole terms.

2.57 Prior orders.

2.58 Absence of hearing examiner.

Appointment of Committees.

AUTHORITY: 28 CFR Chapter 1, Part O Subpart I, and (18 U.S.C. 3655, 4164, 4201-4218, 4254-5, and 5005-5041).

### § 2.1 Definitions.

As used in this part:

(a) The term "Commission" refers to the United States Parole Commission.

- (b) The term "Commissioner" refers to members of the United States Parole Commission.
- (c) The term "National Appeals Board" refers to the Vice Chairman of the Commission and two other National Commissioners who are assigned in the headquarters office of the Commission in Washington, D.C. The Vice Chairman shall be the Chairman of the National Appeals Board. In the absence or vacancy of the Vice Chairman the Chairman of the Commission functions as the Chairman of the National Appeals Board. In the absence or vacancy of a member the Chairman of the Commission functions as a member of the National Appeals Board.
- (d) The term "National Commissioners" refers to the Chairman of the Commission and the three members of the National Appeals Board. The Vice Chairman of the Commission shall be the Chairman of the National Commissioners. In the absence or vacancy of the Vice Chairman, the Chairman of the Commission shall be Chairman of the National Commissioners.

(e) The term "Regional Commissioner" refers to Commissioners assigned to the Commission's regional offices.

(f) The term "eligible prisoner" refers to any Federal prisoner eligible for parole pursuant to this Part and includes any Federal prisoner whose parole has been revoked and who is not otherwise ineligible for parole.

(g) The term "parolee" refers to any Federal prisoner released on parole or as if on parole pursuant to 18 U.S.C. 4164 or 4205(f). The term "mandatory release" refers to release pursuant to 18

U.S.C. 4163 and 4164.

(h) The term "effective date of parole" refers to a parole date that has been approved following an in-person hearing held within six months of such date, or following a pre-release record review.

(i) All other terms used in this part shall be deemed to have the same meaning as identical or comparable terms as used in Chapter 311 of Part IV of Title 18 of the United States Code or Chapter I, Part O, Subpart V of Title 28 of the Code of Federal Regulations.

# § 2.2 Eligibility for parole; adult sentences.

(a) Unless otherwise provided by statute, a Federal prisoner confined and serving a maximum term or terms of more than one year may not be released on parole prior to completion of one-third of such term or terms, or prior to completion of ten years of a life sentence or of a sentence of over thirty years (18 U.S.C. 4205(a)).

(b) If the court has designated a minimum term (which may be less than but not more than one-third of the maximum sentence imposed), a Federal prisoner serving a maximum term of more than one year may not be released on parole prior to completion of the court-designated minimum term (18 U.S.C. 4205(b)(1)).

(c) In cases in which the court designates only a maximum term and specifies that the Commission may release on parole at any time, the prisoner may be released on parole in the discretion of the Commission (18 U.S.C. 4205(b) (2)).

(d) A Federal prisoner sentenced to a maximum term or terms of at least six months but not more than one year prior to May 14, 1976 is eligible for parole consideration after service of one-third of such term or terms.

(e) A Federal prisoner sentenced under 18 U.S.C. 924(a) or 26 U.S.C. 5871 for violation of Federal gun control laws is eligible for parole consideration as if sentenced under 18 U.S.C. 4205(b) (2).

(f) A Federal prisoner committed under 18 U.S.C. 3651 for a period of six months or less with a period of probation to follow is not eligible for parole.

### § 2.3 Same; Narcotic Addict Rehabilitation Act.

A Federal prisoner committed under the Narcotic Addict Rehabilitation Act may not be released on parole prior to completion of at least six months in treatment, not including any period of time for "study" prior to final judgment of the court. Before parole is ordered by the Commission, the Surgeon General or his designated representative must certify that the prisoner has made sufficient progress to warrant his release and the Attorney General or his designated representative must also report to the Commission whether the prisoner should be released. Recertification by the Surgeon General prior to reparole consideration is required (18 U.S.C. 4254).

### § 2.4 Same; juvenile delinquents.

A committed juvenile delinquent may be released on parole at any time in the discretion of the Commission (18 U.S.C. 5041).

### § 2.5 Same: youth offenders.

A committeed youth offender may be released on parole at any time in the discretion of the Commission (18 U.S.C. 5017(a)).

### § 2.6 Withheld and forfeited good time.

(a) While neither a forfeiture of good time nor a withholding of good time shall bar a prisoner from receiving a parole hearing. § 4206 of Title 18 of the United States Code permits the Commission to parole only those prisoners who have substantially observed the rules of the institution.

(b) Forfeiture of statutory good time not restored shall be deemed, in itself, to indicate that the prisoner has violated the rules of the institution to a serious degree.

### § 2.7 Committed fines.

In any case in which a prisoner shall have had a fine imposed upon him by the committing court for which he is to stand committed until it is paid or until he is otherwise discharged according to law, such prisoner shall not be released on parole or mandatory release until

payment of the fine, or until the fine commitment order is discharged according to law as follows:

(a) An indigent prisoner may make application to a U.S. Magistrate in the District wherein he is incarcerated or to the chief executive officer of the institution setting forth, under the institutional regulations, his inability to pay such fine; if the magistrate or chief executive officer shall find that the prisoner, having no assets exceeding \$20 in value except such as are by law exempt from being taken on execution for debt, is unable to pay the fine, and if the prisoner takes a prescribed oath of indigency, he shall be discharged from the commitment obligation of the committed fine sentence.

(b) If the prisoner is found to possess assets in excess of the exemption in paragraph (a) of this section, nevertheless if the chief executive officer of the institution or U.S. Magistrate shall find that retention of all such assets is reasonably necessary for his support or that of his family, upon taking of the prescribed oath concerning his assets the prisoner shall be discharged from the commitment obligation of the committed fine sentence. If the chief executive officer of the institution or U.S. Magistrate shall find that retention by the prisoner of any part of his assets is reasonably necessary for his support or that of his family, the prisoner upon taking of the prescribed oath concerning his assets, shall be discharged from the commitment obligation of the committed fine sentence upon payment on account of his fine or that portion of his assets in excess of the amount found to be reasonably necessary for his support or that of his family.

(c) Discharge from the commitment obligation of any committed fine does not discharge the prisoner's obligation to pay the fine as a debt due the United States

### § 2.8 Mental competency proceedings.

(a) Whenever a prisoner or parolee is scheduled for a hearing in accordance with the provisions of this part and reasonable doubt exists as to his mental competency, i.e., his ability to understand the nature of and participate in scheduled proceedings, a preliminary hearing to determine his mental competency shall be conducted by a panel of hearing examiners or other official(s) (including a U.S. Probation Officer) designated by the Commission.

(b) At the competency hearing, the hearing examiners or designated official(s) shall receive oral or written psychiatric or psychological testimony and other evidence that may be available. A preliminary determination of the prisoner's mental competency shall be made upon the testimony, evidence, and personal observation of the prisoner. If the examiner panel or designated official(s) determines that the prisoner is mentally competent, the previously scheduled hearing shall be held. If they determine that the prisoner is not men-

tally competent, the previously scheduled hearing shall be temporarily post-

poned.

(c) Whenever the hearing examiners or designated official(s) determine that a person is incompetent and postpone the previously scheduled hearing, they shall forward the record of the preliminary hearing with their findings to the Regional Commissioner for review. If the Regional Commissioner concurs with their findings, he shall order the temporarily postponed hearing to be postponed indefinitely until such time as it is determined that the prisoner or parolee has recovered sufficiently to understand the nature of and participate in the proceedings and, in the case of a parolee, may order such parolee transferred to a Bureau of Prison's facility for further examination. In any such case, the Regional Commissioner shall require a progress report on the mental health of the prisoner at least every six months. When the Regional Commissioner determines that the prisoner has recovered sufficiently, he shall reschedule the hearing for the earliest feasible date.

(d) If the Regional Commissioner disagrees with the findings of the hearing examiners or designated official(s) as to the mental competency of the prisoner. he shall take such action as he deems

appropriate.

### § 2.9 Study prior to sentencing.

(a) When an adult Federal offender has been committed to an institution by the sentencing court for observation and study prior to sentencing, under the provisions of 18 U.S.C. 4205(c), the report to the sentencing court is prepared and submitted directly by the United States Bureau of Prisons.

(b) The court may order a youth to be committed to the custody of the Attorney General for observation and study at an appropriate classification center or agency. Within sixty days from the date of the order, or such additional period as the court may grant, the Commission shall report its findings to the court (18

U.S.C. 5010(e)).

### § 2.10 Date service of sentence commences.

(a) Service of a sentence of imprisonment commences to run on the date on which the person is received at the penitentiary, reformatory, or jail for service of the sentence: Provided, however. That any such person shall be allowed credit toward the service of his sentence for any days spent in custody in connection with the offense or acts for which sentence was imposed.

(b) The imposition of a sentence of imprisonment for civil contempt shall interrupt the running of any sentence of imprisonment being served at the time the sentence of civil contempt is imposed, and the sentence or sentences so interrupted shall not commence to run again until the sentence of civil contempt is lifted.

(c) Service of the sentence of a committed youth offender or a person committed under the Narcotic Addict Rehabilitation Act commences to run from the date of conviction and is interrupted only when such prisoner or parolee (1) is on bail pending appeal; (2) is in escape status; (3) has absconded from his or her district of supervision; or (4) comes within the provisions of subsection (b) of this section.

### § 2.11 Application for parole; notice of hearing.

(a) A federal prisoner (including a committed youth offender or prisoner sentenced under the Narcotic Addict Rehabilitation Act) desiring to apply for parole shall execute an application form as prescribed by the Commission. Such forms shall be available at each federal institution and shall be provided to each prisoner who is eligible for an initial parole hearing pursuant to § 2.12. Prisoners committed under the Federal Juvenile Delinquency Act shall be considered for parole without application and may not waive parole consideration. A prisoner who receives an initial hearing need not apply for subsequent hearings.

(b) A prisoner may knowingly and intelligently waive any parole consideration on a form provided for that purpose. If a prisoner waives parole consideration, he may later apply for parole and may be heard during the next visit of the Commission to the institution at which he is confined, provided that he has applied at least 45 days prior to the first day of the month in which such visit of the Commission occurs.

(c) A prisoner who fails to submit either an application for parole or a waiver form shall be referred to the Commission's representatives by the chief executive officer of the institution. The prisoner shall then receive an explanation of his right to apply for parole at a later date

(d) In addition to the above procedures relating to parole application, all prisoners prior to initial hearing shall be provided with an inmate background statement by the Bureau of Prisons for completion by the prisoner.

(e) At least thirty days prior to the initial hearing (and prior to any hearing conducted pursuant to \$2.14), the prisoner shall be provided with written notice of the time and place of the hearing and of his right to review the documents to be considered by the Commission, as provided by § 2.55. A prisoner may waive such notice, except that if such notice is not waived, the case shall be continued to the time of the next regularly scheduled proceeding of the Commission at the institution in which the prisoner is confined.

### § 2.12 Initial hearings: Setting presumptive release dates.

(a) An initial hearing shall be conducted within 120 days of a prisoner's arrival at a federal institution, or as soon thereafter as practicable, in the following cases:

(1) A prisoner with no minimum term of imprisonment; and

(2) A prisoner with a minimum term of imprisonment and a maximum term or terms of less than seven years.

(b) In the case of a prisoner with a minimum term of imprisonment and a maximum term or terms of seven years or more, an initial hearing shall be conducted at least thirty days prior to the completion of the minimum term of imprisonment, or as soon thereafter as

(c) Following initial hearing: (1) The Commission shall set a presumptive release date (either by parole or by mandatory release), or set an effective date of parole, in the case of every prisoner with a maximum term or terms of less

than seven years.

(2) In the case of a prisoner with a maximum term or terms of seven years or more, the Commission shall either set a presumptive release date, if such date falls within four years of the initial hearing, or continue the prisoner to a fouryear reconsideration hearing pursuant to § 2.14(c), or set an effective date of parole.

(d) Notwithstanding the above paragraph, a prisoner may not be paroled earlier than the completion of any judicially set minimum term of imprisonment or other period of parole ineligibil-

ity fixed by law.

(e) A presumptive parole date shall be contingent upon a continued record of good conduct and the establishment of a suitable release plan, and shall be subject to the provisions of §§ 2.14 and 2.34. In the case of a prisoner sentenced under the Narcotic Addict Rehabilitation Act, 18 U.S.C. 4254, a presumptive parole date shall also be contingent upon certification by the Surgeon General pursuant to § 2.3 of these rules.

### § 2.13 Initial hearing; procedure.

(a) An initial hearing shall be conducted by a panel of two hearing ex-aminers. The examiners shall discuss with the prisoner his offense severity rating and salient factor score as described in § 2.20, his institutional conduct and, in addition, any other matter

the panel may deem relevant.

(b) A prisoner may be represented at a hearing by a person of his choice. The function of the prisoner's representative shall be to offer a statement at the conclusions of the interview of the prisoner by the examiner panel, and to provide such additional information as the examiner panel shall request. Interested parties who oppose parole may select a representative to appear and offer a statement. The presiding hearing examiner shall limit or exclude any irrelevant or repetitious statement.

(c) At the conclusion of the hearing, the panel shall orally inform the prisoner of its recommendation and, if such recommendation is for denial, of the reasons therefor. Written notice of the official decision, or the decision to refer under § 2.17 or § 2.24, shall be mailed or transmitted to the prisoner within 21 days of the date of the hearing, except in emergencies. If parole is denied, or a release date is set in excess of six

prisoner shall also receive in writing the reasons therefor.

(d) In accordance with 18 U.S.C. 4206, reasons for parole denial may include the following, with further specification as appropriate:

(1) The prisoner has not substantially observed the rules of the institution or

institutions in which confined;

(2) Release, in the opinion of the Commission, would depreciate the seriousness of the offense or promote disrespect for the law; or

(3) Release, in the opinion of the Commission, would jeopardize the public

welfare.

In lieu of, or in combination with, the above reasons the prisoner shall be furnished with a guidelines evaluation statement containing his offense severity rating and salient factor score (including the points credited on each item of such score) as described in § 2.20, as well as the specific factors and information relied upon for any decision to continue such prisoner for a period outside the range indicated by the guidelines.

(e) No interviews with the Commission or any representative thereof, shall be granted to a prisoner unless his name is docketed for a hearing in accordance with Commission procedures. Hearings

shall not be open to the public.

(f) A full and complete record of every hearing shall be retained by the Commission. Upon a request, pursuant to § 2.55, the Commission shall make available to any eligible prisoner such record as the Commission has retained of the hearing.

### § 2.14 Subsequent proceedings.

(a) Interim proceedings. The purpose of an interim proceeding required by 18 U.S.C. 4208(h) shall be to consider any significant developments or changes in the prisoner's status that may have occurred subsequent to the initial hear-

(1) Notwithstanding a previously ordered presumptive release date or fouryear reconsideration hearing, interim hearings shall be conducted by an examiner panel pursuant to the procedures of § 2.13 (b), (c), (e), and (f) at the fol-lowing intervals from the date of the

last hearing:

(i) In the case of a prisoner with a maximum term or terms of less than seven years, every eighteen months (until released).

(ii) In the case of a prisoner with a maximum term or terms of seven years or more, every twenty-four months (un-

til released).

- (2) However, in the case of a prisoner with an unsatisfied minimum term, the first interim hearing shall be deferred until the docket of hearings immediately preceding completion of the minimum term.
- (3) Following an interim hearing, the Commission may:
- (i) Order no change in the previous decision:
- (ii) Advance a presumptive release date, or the date of a four-year reconsid-

months from the date of the hearing, the eration hearing. However, it shall be the policy of the Commission that once set, a presumptive release date or the date of a four-year reconsideration hearing shall not be advanced except under clearly exceptional circumstances:

(iii) Retard or rescind a presumptive parole date for reason of disciplinary infractions. In a case in which disciplinary infractions have occurred, the interim hearing shall be conducted in accordance with the procedures of § 2.34(a).

(b) Pre-release reviews. The purpose of a pre-release review shall be to determine whether the conditions of a presumptive release date by parole have been satisfied.

(1) At least sixty days prior to a presumptive parole dates, an examiner panel shall review the case on the record. including a current institutional progress report.

(2) Following review and recom-mendation, the Regional Commissioner

(i) Approve the parole date;

(ii) Advance or retard the parole date

as provided by § 2.29(c);

(iii) Retard the parole date or commerce rescission proceedings as provided

(3) A pre-release review pursuant to this section shall not be required if an inperson hearing has been held within six

months of the parole date.

(c) Four-year reconsideration hearings. A four-year reconsideration hearing shall be a full reassessment of the case pursuant to the procedures of § 2.13 to determine whether the setting of a presumptive release date would be appropriate at that time.

(1) A four-year reconsideration hearing shall be ordered following initial hearing in any case in which a release

date is not set.

(2) Following a four-year reconsideration hearing, the Commission may:

- (i) Set a presumptive release date, if such date falls within four years of the hearing; or
- (ii) Continue the prisoner to a further four-year reconsideration hearing if no presumptive release date is set.

# § 2.15 Petition for consideration of parole prior to date set at hearing.

When a prisoner has served the minimum term of imprisonment required by law, the Bureau of Prisons may petition the responsible Regional Commissioner for reopening the case under § 2.28 and consideration for parole prior to the date set by the Commission at the initial or review hearing. The petition must show cause why it should be granted, i.e., an emergency, hardship, or the existence of other extraordinary circumstances that would warrant consideration of early parole.

### § 2.16 Parole of prisoner in state, local, or territorial institution.

(a) Any person who is serving a sentence of imprisonment for any offense against the United States, but who is confined therefor in a state reformatory or other state or territorial institution,

shall be eligible for parole by the Commission on the same terms and conditions, by the same authority, and subject to recommittal for the violation of such parole, as though he were confined in a Federal penitentiary, reformatory, or other correctional institution.

(b) Federal prisoners serving concurrent state and Federal sentences in state, local, or territorial institutions shall be furnished upon request parole application forms. Upon receipt of the application and any supplementary classification material submitted by the institution, parole consideration shall be made by an examiner panel of the appropriate region on the record only. If such prisoner is released from his state sentence prior to a Federal grant of parole, he shall be given a personal hearing as soon as feasible after receipt at a Federal institution.

(c) Prisoners who are serving Federal sentences exclusively but who are being boarded in state, local or territorial institutions may be provided hearings at such facilities or may be transferred by the Bureau of Prisons to Federal Institutions for hearings by examiner panels

of the Commission.

### \$ 2.17 Original jurisdiction cases.

(a) A Regional Commissioner may designate certain cases for decision by a quorum of Commissioners as described below, as original jurisdiction cases. In such instances, he shall forward the case with his vote, and any additional comments he may deem germane, to the National Commissioners for decision. Decisions shall be based upon the concurrence of three votes with the appropriate Regional Commissioner and each National Commissioner having one vote. Additional votes, if required, shall be cast by the other Regional Commissioners on a rotating basis as established by the Chairman of the Commission.

(b) The following criteria will be used in designating cases as original juris-

diction cases:

(1) Prisoners who have committed serious crimes against the security of the Nation, e.g., espionage or aggravated subversive activity.

(2) Prisoners whose offense behavior: (i) Involved an unusual degree of 50phistication or planning or (ii) Was part of a large scale criminal conspiracy of a continuing criminal enterprise.

- (3) Prisoners who have received national or unusual attention because of the nature of the crime, arrest, trial, or prisoner status, or because of the community status of the offender or his victim.
- (4) Long-term sentences, Prisoners sentenced to a maximum term of fortyfive years (or more) or prisoners serving life sentences.
- (c) (1) Any case designated for the original jurisdiction of the Commission shall remain an original jurisdiction case unless designation is removed pursuant to this subsection.
- (2) A case found to be inappropriately designated for the Commission's original jurisdiction, or to no longer warrant such

designation, may be removed from original jurisdiction under the procedures specified in paragraph (a) of this section following a regularly scheduled hearing or the reopening of the case pursuant to \$2.28. Removal from original jurisdiction may also occur by majority vote of the Commission considering an appeal pursuant to \$2.27. Where the circumstances warrant, a case may be redesignated as original jurisdiction pursuant to the provisions of paragraphs (a) and (b) of this section.

### § 2.18 Granting of parole.

The granting of parole to an eligible prisoner rests in the discretion of the United States Parole Commission. As prerequisites to a grant of parole, the Commission must determine that the prisoner has substantially observed the rules of the institution or institutions in which he has been confined; and upon consideration of the nature and circumstances of the offense and the history and characteristics of the prisoner, must determine that release would not depreciate the seriousness of his offense or promote disrespect for the law, and that release would not jeopardize the public welfare (i.e., that there is a reasonable probability that, if released, the prisoner would live and remain at liberty without violating the law or the conditions of his parole)

### § 2.19 Information considered.

(a) In making a determination under this chapter (relating to release on parole) the Commission shall consider, if available and relevant:

 Reports and recommendations which the staff of the facility in which such prisoner is confined may make;

- (2) Official reports of the prisoner's prior criminal record, including a report or record of earlier probation and parole experiences:
  - (3) Presentence investigation reports:
- (4) Recommendations regarding the prisoner's parole made at the time of sentencing by the sentencing judge and prosecuting attorney; and
- (5) Reports of physical, mental, or psychiatric examination of the offender.
- (b) There shall also be taken into consideration such additional relevant information concerning the prisoner (including information submitted by the prisoner) as may be reasonably available (18 U.S.C. 4207). The Commission encourages the submission of relevant information concerning an eligible prisoner by interested persons.

### § 2.20 Paroling policy guidelines; statement of general policy.

(a) To establish a national paroling policy, promote a more consistent exercise of discretion, and enable fairer and more equitable decision-making without removing individual case consideration. the United States Parole Commission has adopted guidelines for parole release consideration.

- (b) These guidelines indicate the customary range of time to be served before release for various combinations of offense (severity) and offender (parole prognosis) characteristics. The time ranges specified by the guidelines are established specifically for cases with good institutional adjustment and program progress.
- (c) These time ranges are merely guidelines. Where the circumstances warrant, decisions outside of the guidelines (either above or below) may be rendered.
- (d) The guidelines contain examples of offense behaviors for each severity

level. However, especially mitigating or aggravating circumstances in a particular case may justify a decision or a severity rating different from that listed.

- (e) An evaluation sheet containing a "salient factor score" serves as an aid in determining the parole prognosis (potential risk of parole violation). However, where circumstances warrant, clinical evaluation of risk may override this predictive aid.
- (f) Guidelines for reparole consideration are set forth at § 2.21.
- (g) The Commission shall review the guidelines, including the salient factor score, periodically and may revise or modify them at any time as deemed appropriate.

### Guidelines for decisionmaking

[Customary total time to be served before release (including juit time)]

Offense characteristics severity of offense behavior (examples)	Offender characteristics—parole prognosis (callent factor score) (in months)			
Canada	Very good (11 to 9)	Good (8 to 8)	Fair (5 to t)	Poor (3 to 0)
Adult				
Low:	1			
Escape Jopen Institution or program (e.q., CTC, work				
release)—absent less than 7 d.  Marihuana or soft drugs, simple possession (samil	6-10	8-10	10-14	12-18
quantity for own use).	375757	37.30	4000	Here
Property offenses (theft or simple possession of stolen property) less than \$1,000.	Contract of the Contract of th			
Low moderate:	1			
Alcohol law violatious				
Counterfeit currency (passing/possession less than \$1,000).	1			
Immigration law violations.	1 10000			
Income tax evasion (less than \$10,000). Property offenses (lorgery/fraud/theft from mail/em-	8-12	12-16	16-20	20-28
bezzlement/interstate transportation of stolen or				
forged securities/receiving stolen property with	10000			
Intent to resell) less than \$1,000. Selective Service Act violations.				
Moderate:	1			
Bribery of a public official (offering or accepting)  Counterfeit currency (passing/possession \$1,000 to				
\$19,900).	The same of the sa			
Drugs:				
Maxibuana, possession with intent to distribute/ sale (small scale (e.g., less than 50 lb)).	1 - 2 -			
"Seft drugs", possession with intent to distribute/	163			
sale (less than \$500).  Escape [secure program or institution, or absent 7 d or				
more—no fear or threat used).	12-10	16-20	20-24	24-32
Firearms Act, possession/purchase/sale (single wapon: not sawed-off shotgun or machine gun).	12			
Income tax evasion (\$10,000 to \$50,000)				
Mailing threatening communication(s)				
Misprison of felony Property offenses (theft/forgery/frand/embezziement/				
interstate transportation of stolen or forged securities/	1 1 1 1 1 1			
receiving stolen property) \$1,000 to \$19,999.				
Smuggling/transporting of alien(x).  Theft of motor vehicle (not multiple theft or for resale).				
ligh:				
Counterfeit currency (passing/possession \$20,000 to \$100,000).	- 1 K-			
Counterfeiting (manufacturing)				
Drugs:				
Maribuana, possession with intent to distribute/ male (medium scale (e.g., 50 to 1,990 lb).				
"Soft drugs", possession with intent to distribute/				
sale (\$500 to \$5,000). Explosives, possession/transportation.	10-20	20-26	26-34	34-44
Firearms Act, possession/purchase/sals (sawed-off-shot-				
gun(s), machine gun(s), or multiple weapons).	-			
Mann Act (no force—commercial purposes).  Theft of motor vehicle for resale.	TO GITTE			
Property offences [theft/forgery/fraud/embezzlement/]				
interstate transportation of stolen or forged securities/ receiving stolen property) \$20,000 to \$100,000.				

(examples)	Very good (11 to 9)	(8 to 6)	Fair (5 to 4)	Poor (3 to 0)
Very high:  Robbery (weapon or threat)  Breaking and entering (bank or post office-entry or			HIERON.	
attempted entry to vault).  Drugs: Maribuana, possession with intent to distribute/				
sale (large scale (e.g., 2,000 lb or more)). "Soft drugs", possession with intent to distribute/ sale (over \$5,000). "Hard drugs", possession with intent to distribute/	26-36	36-48	48,60	00-7
sale (not exceeding \$100,000).  Extortion.  Mann Act (force)	20150	W/ 10	10.00	
Property offenses (theft/lorgery/frand/embezzlement/ interestate transportation of stolen or forged securities/ receiving stolen property) over \$100,000 but not exceeding \$500,000. Sexual act (force).				
Greatest: Aggravated felony (e.g., robbery, sexual act, aggravated 'assault)—weapon fired or personal injury. Aircraft hijacking.				
Drugs: "Hard drugs", possession with intent to dis- tribute-sale (in excess of \$100,000). Espionage.	given due t	above-howev	number of ci	asen and the
Explosives (detonation).  Kidnaping.  Willim homicide.	extreme va	riation in sev	erity possible	within the
YOUTH/N	VARA			
Low: Escape (open institution or program (e.g., CTC, work release)—absent less than 7 d).			1	
Marihuana or soft drugs, simple possession (small quantity for own use).  Property offenses (theft or simple possession of stolen property) less than \$1,000.	6-10	8-12	10-14	12-1
Low moderate: Alcohol law violations. Counterfeit currency (passing/possession less than				
\$1,000). Intuigration law violations. Income tax evasion (less than \$10,000).	8-12	- 12-16	16-20	20-2
Property offenses (longery/fraud/theft from mailten- bezzlement/interstate transportation of stolen or forged securities/receiving stolen property with in- tent to resell) less than \$1,000.				
Selective Service Act violations.  Moderate:  Bribery of a public official (offering or accepting)  Counterfeit currency (passing/possession \$1,000 to \$19,000).				
Drugs:  Marihuana, possession with intent to distribute/ sale (amail scale (e.g., less than 50 lb)).  "Soft drugs," possession with intent to distribute/ sale dies than \$500).				
Escape (secure program or institution, or absent 7 d or more—no fear or threat used). Firearms Act, possession/purchase/sale (single weapon:	9-13	13-17	17-21	21-2
not sawed-oil shotgun or machine gum).  Income tax evasion (\$10,000 to \$50,000).  Mailing threatening communication(s).				
Misprison of felony Property offenses (theft/forgery/frand/embezzlement/ interstate transportation of stolen or forged securities, receiving stolen property) \$1,000 to \$19,999.	500			
Sunggling/transporting of allen(s). Theft of motor vehicle (not multiple theft or for resale). High:			1.3	
Counterfeit currency (passing/possession \$20,000 to 100,000).  Counterfeiting (manufacturing)				
Marihuana, possession with intent to distribute/ sale (medium scale (e.g., 50 to 1,999 lbs)). "Soft drugs", possession with intent to distribute/ sale (\$500 to \$5,000).				
sale (\$500 to \$5,000).  Explosives, possession/transportation.  Firearms Acts, possession/purchase/sale (sawed-off shotgun(s), machine gun(g), or multiple weapons).	12-16	16-20	20-26	26-1
Theft of motor vehicle for resale  Property offenses (theft/forpery/fraud/embezziement/				
interstate transportation of stolen or forged securities/ receiving stolen property) \$20,000 to \$100,000. Very high: Robbery (weapon or threat)				
Breaking and entering (bank or post office-entry or attempted entry to vault).  Drugs:	FILE			
Marihuana, possession with intent to distribute/ sale (large scale (e.g., 2,000 lbs or more)). "Soft drugs", possession with intent to distribute/ sale (over \$5,000).				
"Hard druga", possession with intent to distributely sale (not exceeding \$100,000). Extertion.	20-27	27-34	34-41	41-4
Mann Act (force). Property offenses (theft/forgery/frand/embezzlement/interstate transportation of stolen or forged securities/receiving stolen property) over \$100,000 but not ex-	1 4 4 4			

Offense characteristics—severity of offense behavior

(examples)

Offender characteristics—parole prognosis
(asilent factor score) (in months)

Very good Good Fair Poor
(11 to 9) (8 to 6) (5 to 4) (3 to 0)

Orestest:

Aggravated felony (e.g., robbery, sexual act, aggravated assault)—weapon fired or personal injury.

Aircraft bljacking.

Drugs: "Hard drugs", possession with intent to distributefule distances of \$300,000).

Esphonage.

Esphonage.

Notes.—1. These guidelines are predicated upon good institutional conduct and program performance.

2. If an offense behavior is not listed above, the proper category may be obtained by comparing the severity of the offense behavior with those of similar offense behaviors listed.

3. If an offense behavior involved multiple separate offense, the severity lovel may be increased.

5. If a continuance is to be given, allow 20 d it mo) for release program provision.

6. "Hard drugs" include, but are not limited to, barbiturates, amphetamines. LSD, and hashist.

7. Conspirincy shall be rated for guideline purposes according to the underlying offense behavior if such behavior was consummated. If the offense is unconsummated, the conspiracy will be rated one step below the consummated offense.

Same Register No.

Register No.

Register No.

Register No.

Prove convictions at a convenience of the provenience of the consummated offense.

# Case name Register No. Regis

### § 2.21 Reparole consideration guidelines.

(a) If revocation is based upon administrative violation(s) only [i.e., violations other than new criminal conduct] the following guidelines shall apply.

Customary time to be served be fore rerelease (months)

Positive supervision history (examples):

- a. No serious atcohol/drug abuse and no possession of weapon(s) [and]
- At least 8 months from date of release to date of violation behavior [and]
- c. Positive employment/school record during supervision [and]
- d. Present violation represents first instance of failure to comply with parole regulations of this term....

Negative supervision history (examples):

- a. Serious alcohol/drug abuse (e.g. readdiction to hard drugs) or possession of weapon(s) [or]
- b. Less than 8 months from date of release to date of violation behavior [or]
- c. Negative employment/school record during supervision [or]
   d. Negative attitude toward super-
- d. Negative attitude toward supervision demonstrated by lack of positive efforts to cooperate with parole (aftercare) plan or by repetitious or persistent violations

(b) (1) If a finding is made that the prisoner has engaged in behavior constitting new criminal conduct, the appropriate severity rating for the new criminal behavior shall be calculated. New criminal conduct may be determined either by a new federal, state, or a local conviction or by an independent finding

by the Commission at revocation hearing. As violations may be for state or local offenses, the appropriate severity level may be determined by analogy with listed federal offense behaviors.

(2) The guidelines for parole consideration specified at 28 CFR 2.20 for the poor parole risk category shall then be applied. The original sentence type (i.e. adult, youth), shall determine the applicable guidelines for the parole violator term. Time served on a new state or federal sentence shall be counted as time in custody. This does not affect the computation of the total violator term as provided by §§ 2.47 (b) and (c) and 2.52 (c) and (d).

(c) The above are merely guidelines. A decision outside these guidelines (either above or below) may be made when circumstances warrant. For example, violations of an assaultive nature, or violations by a person with a history of assaultive conduct or by a person with a history of repeated parole failure may warrant a decision above the guidelines. Minor offense(s) (e.g., traffic infractions, disorderly conduct) shall normally be treated under administrative violations.

# § 2.22 Communication with the Commission.

Attorneys, relatives, or interested parties wishing a personal interview to discuss a specific case with a representative of the Commission must submit a written request to the appropriate regional office setting forth the nature of the information to be discussed. Such personal interview may be conducted by Staff Personnel in the regional offices. Personal interviews, however, shall not be held by an examiner or member of the Commission except under the Commission except under the Commission's appeals procedures.

### § 2.23 Delegation to hearing examiners.

(a) There is hereby delegated to hearing examiners the authority necessary to conduct hearings and make recommendations relative to the grant or denial of parole or reparole, revocation or reinstatement of parole or mandatory release, and conditions of parole.

(b) Hearing examiners shall function as two-man panels except as provided by §§ 2.43 and 2.47 and the concurrence of two examiners shall be requested for their recommendation. In the event of a divided recommendation by the panel, the appropriate regional Administrative hearing Examiner shall cast the deciding

vote.

(c) In the event the Administrative Hearing Examiner is serving as a member of a hearing examiner panel or is otherwise unavailable, cases requiring his action under paragraph (b) of this section will be referred to another hearing examiner.

(d) A recommendation of a hearing examiner panel shall become an effective Commission decision upon review at the Regional Office and docketing, unless action is initiated by the regional Commissioner pursuant to § 2.24.

### § 2.24 Review of panel recommendation by the Regional Commissioners.

- (a) A Regional Commissioner may review the recommendation of any examiner panel and refer this recommendation, prior to written notification to the prisoner, with his recommendation and vote to the National Commissioners for consideration and any action deemed appropriate. Written notice of this referral action shall be mailed or transmitted to the prisoner within twentyone days of the date of the hearing. The Regional Commissioner and each National Commissioner shall have one vote and decisions shall be based upon the concurrence of two votes. Action shall be taken by the National Commissioners within thirty days of the date of referral action by the Regional Commissioner, except in emegencles.
- (b) Notwithstanding the provisions of paragraph (a) of this section, a Regional Commissioner may:
- (1) On the motion of the Administrative Hearing Examiner, modify or reverse the recommendation of a hearing examiner panel that is outside the guidelines to bring the decision closer to (or to) the nearer limit of the appropriate guideline range; or
- (2) On his own motion, modify the recommendation of a hearing examiner panel to bring the decision to a date not to exceed six months from the date recommended by the examiner panel.

### § 2.25 Regional appeal.

- (a) A prisoner or parolee may submit to the responsible Regional Commissioner a written appeal of a decision to grant, rescind, deny, or revoke, parole, except that an appeal of a Commission decision pursuant to § 2.17 shall be pursuant to § 2.27. This appeal must be filed on a form provided for that purpose within thirty days from the date of entry of such decision.
- (b) The Regional Commissioner may affirm the decision, order a new institutional hearing on the next docket, order a regional appellate hearing, reverse the decision, or modify a continuance or the effective date of parole. Reversal of a decision or the modification of a decision by more than one hundred eighty days whether based upon the record or following a regional appellate hearing shall require the concurrence of two out of three Regional Commissioners. Decisions requiring a second or additional vote shall be referred to other Regional Commissioners on a rotating basis as established by the Chairman.
- (c) Regional appellate hearings may be held at the regional office before the Regional Commissioner. If a regional appellate hearing is ordered, attorneys, relatives and other interested parties who wish to appear must submit a written request to the Regional Commissioner stating their relationship to the prisoner and the general nature of the information they wish to present. The Regional Commissioner shall determine if the requested appearances will be permitted. The prisoner shall not appear personally.

(d) Within 30 days of receipt of the appeal, except in emergencies, the Regional Commissioner shall inform the applicant in writing of the decision and the reasons therefor.

(e) If no appeal is filed within thirty days of the date of the entry of the original decision, such decision shall stand as the final decision of the Commission.

(f) Appeals under this section may be based on the following grounds;

- (1) That the guidelines were incorrectly applied as to any or all of the following:
  - (i) Severity rating:
  - (ii) Salient factor score;
  - (iii) Time in custody;
- (2) That a decision outside the guidelines was not supported by the reasons or facts as stated;
- (3) That especially mitigating circumstances (for example, facts relating to the severity of the offense or the prisoner's probability of success on parole) justify a different decision;

(4) That a decision was based on erroneous information, and the actual facts justify a different decision;

(5) That the Commission did not follow correct procedure in deciding the case, and a different decision would have resulted if the error had not occurred;

(6) There was significant information in existence but not known at the time

of the hearing;

(7) There are compelling reasons why a more lenient decision should be rendered on grounds of compassion.

### § 2.26 Appeal to National Appeals Board.

- (a) Within 30 days of entry of a Regional Commissioner's decision under § 2.25, a prisoner or parolee may appeal to the National Appeals Board on a form provided for that purpose. However, any matter not raised on a regional level appeal may not be raised on appeal to the National Appeals Board. The National Appeals Board may, upon the concurrence of two members, affirm, modify, or reverse the decision, or order a rehearing at the institutional or regional level.
- (b) The National Appeals Board shall act within 60 days of receipt of the appellant's papers, to affirm, modify, or reverse the decision.
- (c) Decisions of the National Appeals Board shall be final.

# § 2.27 Appeal of original jurisdiction cases.

(a) Cases decided under the procedure specified in § 2.17 may be appealed within thirty days of the entry of the decision on a form provided for this purpose. Attorneys, relatives and other interested parties who wish to submit written information in support of a prisoner's appeal should send such information to the National Appeals Board Analyst, United States Parole Commission, 320 First Street, N.W., Washington, D.C. 20537. Appeals of original jurisdiction cases shall be reviewed by the Commission at its next quarterly meeting. A quorum of five Commissioners shall be required and all decisions shall be by majority vote. This appellate decision shall be final.

(b) Attorneys, relatives, or other interested parties who wish to speak for or against parole at such consideration must submit a written request to the Chairman of the Commission stating their relationship to the prisoner and the general nature of the material they wish to present. The Chairman shall determine if the requested appearances will be permitted.

(c) If no appeal is filed within thirty days of the entry of the decision under § 2.17, that decision shall stand as the final decision of the Commission.

### § 2.28 Reopening of cases.

Notwithstanding the appeal procedure of §§ 2.25 and 2.26, the appropriate Regional Commissioner may, on his own motion, reopen a case at any time upon the receipt of new information of substantial significance and may then take any action authorized under the provisions and procedures of § 2.25. Original jurisdiction cases may be reopened upon the motion of the appropriate Regional Commissioner under the procedures of § 2.17.

### § 2.29 Release on parole.

(a) A grant of parole shall not be deemed to be operative until a certificate of parole has been delivered to the prisoner.

(b) An effective date of parole shall not be set for a date more than six months from the date of the hearing. Residence in a Community Treatment Center as part of a parole release plan generally shall not exceed one hundred

and twenty days.

(c) When an effective date of parole has been set by the Commission, release on that date shall be conditioned upon the completion of a satisfactory plan for parole supervision. The appropriate Regional Commissioner may, on his own motion, reconsider any case prior to release and may reopen and advance or retard an effective parole date. An effective parole grant may be retarded for up to one hundred and twenty days without a hearing for development and approval of release plans.

(d) When an effective date of parole falls on a Saturday, Sunday, or legal holiday, the Warden of the appropriate institution shall be authorized to release the prisoner on the first working day preceding such date.

### § 2.30 False or withheld information.

All paroles are ordered on the assumption that information from the prisoner has not been fraudulently given to or withheld from the Commission. If evidence comes to the attention of the Regional Commissioner that a prisoner willfully concealed or misrepresented information deemed significant, the Regional Commissioner may initiate action pursuant to \$2.34(b) to determine whether such parole should be revoked or rescinded.

§ 2.31 Parole to detainers; statement of policy.

(a) Where a detainer is lodged against a prisoner, the Commission may grant parole if the prisoner in other respects meets the criteria set forth in § 2.18. The presence of a detainer is not in itself a valid reason for the denial of parole.

(b) The Commission will cooperate in working out arrangements for concurrent supervision with other jurisdictions where it is feasible and where release on parole appears to be justified.

### § 2.32 Parole to local or immigration detainers.

(a) When a state or local detainer is outstanding against a prisoner whom the Commission wishes to parole, the Commission may order either of the following:

(1) "Parole to the actual physical custody of the detaining authorities only." In this event, release is not to be effected except to the detainer. When such a detainer is withdrawn, the prisoner is not to be released unless and until the Commission makes a new order of parole.

(2) "Parole to the actual physical custody of the detaining authorities or an approved plan." In this event, release is to be effected even though the detainer might be withdrawn, providing there is an acceptable plan for community supervision.

(b) When the Commission wishes to parole a prisoner subject to a detainer filed by Federal immigration officials, the Commission may order one of the

following:

 "Parole for deportation only." In this event, release is not to be effected unless immigration officials make full arrangements for deportation immediately upon release.

(2) "Parole to the actual physical custody of the immigration authorities only." In this event, release is not to be effected unless immigration officials take the prisoner into custory—regardless of whether or not deportation follows:

- (3) "Parole to the actual physical custody of the immigration authorities or an approved plan." In this event, release is to be effected regardless of whether or not immigration officials take the prisoner into custody, providing there is an acceptable plan for community supervision.
- (c) As used in this section "parole to a detainer" means release to the "physical custody" of the authorities who have lodged the detainer. Temporary detention in a jail in the county where the institution of confinement is located does not constitute release on parole to such detainer. If the authorities who lodged the detainer do not take the prisoner into custody for any reason, he shall be returned to the institution to await further order of the Commission.

### § 2.33 Release of plans.

(a) A grant of parole is conditioned upon the approval of release plans by the Regional Commissioner. In general, the following factors are considered as elements in the prisoner's release plan.

 Availability of legitimate employment and an approved residence for the prospective parolee; and

(2) Availability of necessary aftercare for a parolee who is ill or who requires special care. (b) Generally, parolees will be released only to the place of their legal residence unless the Commission is satisfied that another place of residence will serve the public interest more effectively or will improve the probability of the applicant's readjustment.

(c) Where the circumstances warrant, the Commission on its own motion, or upon recommendation of the probation officer, may require that an adviser who is a responsible, reputable, and lawabiding citizen living in or near the community in which the releasee will reside be available to the releasee. Such advisor shall serve under the direction of and in cooperation with the probation officer to whom the parolee is assigned.

### § 2.34 Rescission of parole.

(a) When an effective date of parole or mandatory parole has been set by the Commission, release on that date shall be conditioned upon continued good conduct by the prisoner. If a prisoner has been granted parole and has subsequently been charged with institutional misconduct sufficient to become a matter of record, the Regional Commissioner shall be advised promptly of such misconduct. The prisoner shall not be released until the institution has been notified that no change has been made in the Commissioner's order to parole.

(1) Upon receipt of information that a prisoner has violated the rules of the institution, the Regional Commissioner may retard the parole grant for up to sixty days without a hearing or may retard the parole grant and schedule the case for a rescission hearing. If the prisoner was confined in a Federal prison at the time of the order retarding parole, the rescission hearing shall be scheduled for the next docket of parole hearings at the institution. If the prisoner was residing in a Federal community treatment center or a state or local halfway house. the rescission hearing shall be scheduled for the first docket of parole hearings after return to a Federal institution. When the prisoner is given written notice of the Commission action retarding parole, he shall be given notice of the charges of misconduct to be considered at the rescission hearing. The purpose of the rescission hearing shall be to determine whether rescission of the parole grant is warranted. At the rescission hearing the prisoner may be represented by a person of his choice and may present documentary evidence.

(2) An institution discipline committee hearing conducted by the institution resulting in a finding that the prisoner has violated the rules of his confinement, may be relied upon by Commission as conclusive evidence of institutional misconduct.

(3) Consideration of disciplinary infractions in cases with presumptive parole dates may be deferred until the commencement of the next in-person hearing or the prerelease record review required by § 2.14(b). While prisoners are encouraged to earn the restoration of forfeited or withheld good time, the Commission will consider the prisoner's overall institutional record in determining

whether the conditions of a presumptive parole date have been satisfied.

(4) If the parole grant is rescinded, the prisoner shall be furnished a written statement of the findings of misconduct and the evidence relied upon.

- (b) (1) Upon receipt of new information adverse to the prisoner regarding matters other than institutional misconduct, the Regional Commissioner may refer the case to the National Commissioners under the procedures of § 2.17(a) with his recommendation and vote, to retard a previously granted parole. If parole is retarded the case shall be scheduled for a hearing on the next docket of parole hearings or at the first docket of parole hearings following return to a federal institution.
- (2) The prisoner shall be given notice of the nature of the new adverse information upon which the rescission consideration is to be based. The hearing shall be conducted in accordance with the procedures set out in §§ 2.12 and 2.13. The purpose of the hearing shall be to determine if the parole grant should be rescinded or if a new parole date should be established.

# § 2.35 Mandatory release in the absence of parole.

A prisoner shall be mandatorily released by operation of law at the end of the sentence imposed by the court less such good time deductions as he may have earned through his behavior and efforts at the institution of confinement. If released pursuant to 18 U.S.C. 4164, such prisoner shall be released, as if on parole, under supervision until the expiration of the maximum term or terms for which he was sentenced less 180 days. If released pursuant to 18 U.S.C. 4205(f), such prisoner shall remain under supervision until the expiration of the maximum term or terms for which he was sentenced. Insofar as possible, release plans shall be completed before the release of any such prisoner.

### § 2.36 Same; youth offenders.

A prisoner committed under the Youth Corrections Act must be initially released conditionally under supervision not later than two years before the expiration of the term imposed by the court.

### § 2.37 Reports to police departments of names of parolees; statement of policy.

Names of parolees under supervision will not be furnished to a police department of a community, except as required by law. All such notifications are to be regarded as confidential.

### § 2.38 Community supervision by United States Probation Officers.

(a) Pursuant to sections 3655 and 4203
(b) (4) of Title 18 of the United States
Code, United States Probation Officers shall provide such parole services as the
Commission may request. In conformity with the foregoing, probation officers function as parole officers and provide supervision to parolees and mandatory releasees under the Commission's jurisdiction.

(b) A parolee or mandatory releasee may be transferred to a new district of supervision with the permission of the probation officers of both the trans-ferring and receiving district, provided such transfer is not contrary to instructions from the Commission.

### § 2.39 Jurisdiction of the Commission.

(a) Jurisdiction of the Commission over a parolee shall terminate no later than the date of expiration of the maximum term or terms for which he was sentenced, except as provided by § 2.35, § 2.43, or § 2.52.

(b) The parole of any parolee shall run concurrently with the period of parole or probation under any other Fed-

eral, State, or local sentence.

(c) The parole of any prisoner sentenced before June 29, 1932, shall be for the remainder of the term or terms specified in his sentence, less good time allowances provided by law.

(d) Upon the termination of jurisdiction, the Commission shall issue a certificate of discharge to such parolee and to such other agencies as it may deter-

mine.

### § 2.40 Conditions of release.

- (a) The conditions of release are printed on the release certificate and are binding regardless of whether the parolee signs the certificate. The following conditions are deemed necessary to provide adequate supervision and to protect the public welfare:
- (1) The parolee shall go directly to the district named in the certificate (unless released to the custody of other authorities). Within three days after his arrival, he shall report to his parole adviser, if he has one, and to the United States Probation Officer whose name appears on the certificate. If in any emergency the parolee is unable to get in touch with his parole adviser or his probation officer or his office, he shall communicate with the United States Parole Commission, Washington, D.C. 20537.
- (2) If the parolee is released to the custody of other authorities, and after release from the physical custody of such authorities, he is unable to report to the United States Probation Officer to whom he is assigned within three days, he shall report instead to the nearest United States Probation Officer.
- (3) The parolee shall not leave the limits fixed by his certificate of parole without written permission from the probation officer.
- (4) The parolee shall notify his probation officer within two days of any change in his place of residence.
- (5) The parolee shall make a complete and truthful written report (on a form provided for that purpose) to his probation officer between the first and third day of each month, and on the final day of parole. He shall also report to his probation officer at other times as the probation officer directs.
- (6) The parolee shall not violate any law, nor shall he associate with persons engaged in criminal activity. The parolee

shall get in touch within two days with his probation officer or office if he is arrested or questioned by a law-enforcement officer.

- (7) The parolee shall not enter into any agreement to act as an informer or special agent for any law-enforcement agency.
- (8) The parolee shall work regularly unless excused by his probation officer, and support his legal dependents, if any, to the best of his ability. He shall report within two days to his probation officer any changes in employment.
- (9) The parolee shall not drink alcoholic beverages to excess. He shall not purchase, possess, use, or administer marihuana or narcotic or other habitforming drugs, unless prescribed or advised by a physician. The parolee shall not frequent places where such drugs are illegally sold, dispensed, used, or given away.
- (10) The parolee shall not associate with persons who have a criminal record unless he has permission of his probation
- (11) The parolee shall not have firearms (or other dangerous weapons) in his possession without the written permission of his probation officer, following prior approval of the United States Parole Commission.
- (b) The Commission or a member thereof may at any time modify or add to the conditions of release pursuant to this section, on its own motion or on the request of the U.S. Probation Officer supervising the parolee. The parolee shall receive notice of the proposed modification and unless waived shall have ten days following receipt of such notice to express his views thereon. Following such ten day period, the Commission shall have 21 days, exclusive of holidays, to order such modification of or addition to the conditions of release.
- (c) The Commission may require a parolee to reside in or participate in the program of a residential treatment center, or both, for all or part of the period of parole.
- (d) The Commission may require a parolee, who is an addict, within the meaning of section 4251(a), or a drug dependent person within the meaning of section 2(8) of the Public Health Service Act, as amended, to participate in the community supervision program authorized by § 4255 for all or part of the period of parole.
- (e) A parolee may petition the Commission on his own behalf for a modification of conditions pursuant to this section.
- (f) The notice provisions of paragraph (b) of this section shall not apply to modification of parole or mandatory release conditions pursuant to a revocation proceeding or pursuant to paragraph (e) of this section.
- (g) A parolee may appeal an order to impose or modify parole conditions under the procedures of \$\$ 2.25 and 2.26 as applicable not later than thirty days after the effective date of such conditions.

- § 2.41 Travel by parolees and mandatory releasees.
- (a) The probation officer may approve travel outside the district without approval of the Regional Commissioner in the following situations:

(1) Vacation trips not to exceed thirty days

(2) Trips, not to exceed thirty days, to investigate reasonably certain employment possibilities.

(3) Recurring travel across a district boundary, not to exceed fifty miles outside the district, for purpose of employ-

ment, shopping, or recreation.

- (b) Specific advance approval by the Regional Commissioner is required for other travel (including travel outside the contiguous forty-eight states, employment more than fifty miles outside the district, and vacations exceeding thirty days). A request for such permission shall be in writing and must demonstrate a substantial need for such travel. In cases falling under the criteria of § 2.17, the concurrence of two out of three Commissioners shall be required to grant such
- (c) A special condition imposed by the Regional Commissioner prohibiting certain travel shall supersede any general rules relating to travel as set forth above.

### § 2.42 Probation Officer's Reports to Commission.

A supervision report shall be submitted by the responsible probation officer to the Commission for each parolee or mandatory release after the completion of 12 months of continuous supervision and annually thereafter. The probation officer shall submit such additional reports as the Commission may direct.

### § 2.43 Early termination of parole.

(a) (1) Upon its own motion or upon request of the parolee, the Commission may terminate supervision, and thus jurisdiction, over a parolee prior to the expiration of his maximum sentence. A committed youth offender may be granted an early termination of jurisdiction (unconditional discharge) at any time after one year of continuous supervision on parole.

(2) Two years after each parolee's release on parole, and at least annually thereafter, the Commission shall review the status of the parole to determine the need for continued supervision. In calculating such two-year prior there shall not be included any period of release on parole prior to the most recent such release, nor any period served in confine-

ment on any other sentence.

(3) Five years after each parolee's release on parole, the Commission shall terminate supervision over such parolee unless it is determined, after a hearing conducted in accordance with the procedures prescribed in 18 U.S.C. 4214(a)(2). that such supervision should not be terminated because there is a likelihood that the parolee will engage in conduct violating any criminal law. Such hearing may be conducted by a hearing examiner or other official designated by the Regional Commissioner.

(4) If supervision is not terminated under paragraph (a) (3) of this section the parolee may request a hearing annually thereafter, and a hearing shall be conducted with respect to such termination of supervision not less frequently than biennially.

(5) In calculating the five-year period referred to in paragraph (a)(3) of this section, there shall not be included any period of release on parole prior to the most recent such release or any period served in confinement on any other sen-

(6) When termination of jurisdiction prior to the expiration of sentence is granted in the case of a youth offender. his conviction shall be automatically set aside. A certificate setting aside his conviction shall be issued in lieu of a certificate of termination.

(b) The Regional Commissioner in the region of supervision may release a parolee from supervision pursuant to this section if warranted by the circumstances of the case and reports of the supervising probation officer. Except that, in the case of a parolee previously considered pursuant to § 2.17, the decision to grant termination of supervision must also be pursuant to the provisions of § 2.17.

(c) A parolee may appeal an adverse decision under paragraphs (a) (3) or (4) of this section pursuant to §§ 2.25, 2.26 or

§ 2.27 as applicable.

### § 2.44 Summons to appear or warrant for retaking of parolee.

(a) If a parolee is alleged to have violated the conditions of his release, and satisfactory evidence thereof is presented, the Commission or a member thereof may:

(1) Issue a summons requiring the offender to appear for a preliminary interview or local revocation hearing.

(2) Issue a warrant for the apprehension and return of the offender to cus-

A summons or warrant may be issued or withdrawn only by the Commission,

or a member thereof.

(b) Any summons or warrant under this section shall be issued as soon as practicable after the alleged violation is reported to the Commission, except when delay is deemed necessary. Issuance of a summons or warrant may be withheld until the frequency or seriousness of violations, in the opinion of the Commission, requires such issuance. In the case of any parolee charged with a criminal offense, issuance of a summons or warrant may be withheld, or a warrant may be issued and held in abeyance pending disposition of the charge.

(c) A summons or warrant may be issued only within the prisoner's maximum term or terms except that in the case of a prisoner released as if on parole pursuant to 18 U.S.C. 4164, such summons or warrant may be issued only within the maximum term or terms, less one-hundred eighty days. A summons or warrant shall be considered issued when signed and placed in the mail at the Commission Headquarters or appropri-

ate regional office.

(d) The issuance of a warrant under this section suspends the running of a sentence until such time as the parolee may be retaken into custody and a final determination of the charges may be

made by the Commission.

(e) A summons or warrant issued pursuant to this section shall be accom-panied by a statement of the charges against the parolee, the applicable procedural rights under the Commission's regulations and the possible actions which may be taken by the Commission. A summons shall specify the time and place the parolee shall appear for a revocation hearing. Failure to appear in response to a summons shall be grounds for issuance of a warrant.

### § 2.45 Same, youth offenders.

(a) In addition to the issuance of a summons or warrant pursuant to § 2.44 above, the Commission or a member thereof, when of the opinion that a youth offender will be benefitted by further treatment in an institution or other facility, may direct his return to custody or issue a warrant for his apprehension and return to custody.

(b) Upon his return to custody, such youth offender shall be scheduled for a

revocation hearing.

### § 2.46 Execution of warrant and service of summons.

- (a) Any officer of any Federal correctional institutional or any Federal officer authorized to serve criminal process within the United States, to whom a warrant is delivered shall execute such warrant by taking the prisoner and returning him to the custody of the Attorney General.
- (b) On arrest of the parolee the officer executing the warrant shall deliver to him a copy of the Warrant Application listing the charges against the parolee, the applicable procedural rights under the Commission's regulations and the possible actions which may be taken by the Commission
- (c) If execution of the warrant is delayed pending disposition of local charges, for further investigation, or for some other purpose, the parolee is to be continued under supervision by the probation officer until the normal expiration of the sentence, or until the warrant is executed, whichever first occurs, Monthly supervision reports are to be submitted, and the parolee must con-tinue to abide by all the conditions of release.
- (d) A summons to appear at a preliminary interview or revocation hearing shall be served upon the parolee in person by delivering to the parolee a copy of the summons. Service shall be made by any federal officer authorized to serve criminal process within the United States, and certification of such service shall be returned to the appropriate regional office of the Commission.

# 7 Warrant placed as a detainer and dispositional Review.

(a) In those instances where a parolee is serving a new sentence in an institution, a parole violation warrant may be placed against him as a detainer. Such warrant shall be reviewed by the regional Commissioner not later than 180 days following notification to the Commission of such placement. The parolee shall receive notice of the pending review, and shall be permitted to submit a written application containing information relative to the disposition of the warrant. He shall also be notified of his right to request counsel under the provisions of § 2.48(b) to assist him in completing his written application.

(b) Following a dispositional review under this section, the Regional Com-

missioner may:

(1) Let the detainer stand and order further review at an appropriate time:

(2) Withdraw the detainer and: (i) Order reinstatement of the parolee to supervision upon release from custody. or (ii) Close the case if the expiration date has passed:

(3) Order a revocation hearing to be conducted by a hearing examiner or an official designated by the regional Commissioner at the institution in which the

parolee is confined.

Following a revocation hearing conducted pursuant to this section, the Commission may take any action specified at \$ 2.52 including the ordering of concurrent or consecutive service of all or part of any violator term imposed. Such revocation hearing shall be conducted under the applicable procedures at § 2.50, and the parolee may be represented by his own or appointed counsel as provided in § 2.48(b),

(c) It shall be the general policy of the Commission that, in the absence of substantial mitigating circumstances the violator term of a parolee convicted of a new offense subsequent to release on parole shall run consecutively to any term imposed for the new offense.

### § 2.48 Revocation by the Commission, preliminary interview.

- (a) Interviewing Officer. A parolee who is retaken on a warrant issued by a Commissioner shall be given a preliminary interview by an official designated by the Regional Commissioner to enable the Commission to determine if there is probable cause to believe that the parolee has violated his parole as charged, and if so, whether a revocation hearing should be conducted. The offi-cial designated to conduct the preliminary interview may be a United States Probation Officer in the district where the prisoner is confined, provided he is not the officer who recommended that the warrant be issued.
- (b) Notice and Opportunity to Pestpone Interview. At the beginning of the preliminary interview, the interviewing officer shall ascertain that the Warrant Application has been given to the prisoner as required by § 2.46(b), and shall advise the prisoner that he may have the preliminary interview postponed in order to obtain representation by an attorney or arrange for the attendance of witnesses. The prisoner shall also be advised that if he cannot afford to retain an attorney he may apply to a United

States District Court for appointment of counsel to represent him at the preliminary interview and the revocation hearing pursuant to 18 U.S.C. 3006A. In addition, the prisoner may request the Commission to obtain the presence of persons who have given information upon which revocation may be based, Such adverse witnesses shall be requested to attend the preliminary interview unless the prisoner admits a violation or has been convicted of a new offense while on supervision or unless the interviewing officer finds good cause for their non-attendance. Pursuant to § 2.49(a) a subpoena may issue for the appearance of adverse witnesses or the production of documents.

(c) Review of the charges. At the preliminary interview, the interviewing officer shall review the violation charges with the prisoner, apprise the prisoner of the evidence which has been presented to the Commission, receive the statements of witnesses and documentary evidence on behalf of the prisoner, and allow cross-examination of those witnesses in attendance. Disclosure of the evidence presented to the Commission shall be made pursuant to § 2.50(e).

(d) At the conclusion of the preliminary interview, the interviewing officer shall inform the parolee of his recommended decision as to whether there is probable cause to believe that the parolee has violated the conditions of his release, and shall submit to the Commission a digest of the interview together with his recommended decision.

(1) If the interviewing officer's recommended decision is that no probable cause may be found to believe that the parolee has violated the conditions of his release, the responsible regional Commissioner shall review such recommended decision and notify the parolee of his final decision concerning probable cause as expeditiously as possible following receipt of the interviewing officer's digest. A decision to release the parolee shall be implemented without

(2) If the interviewing officer's recommended decision is that probable cause may be found to believe that the parolee has violated a condition (or conditions) of his release, the responsible regional Commissioner shall notify the parolee of his final decision concerning probable cause within 21 days of the date of the preliminary interview.

(3) Notice to the parolee of any final decision of a regional Commissioner finding probable cause and ordering a revocation hearing shall state the charges upon which probable cause has been found and the evidence relied upon.

(e) Release notwithstanding probable cause: If the Commission finds probable cause to believe that the parolee has violated the conditions of his release, reinstatement to supervision or release pending further proceeding may nonetheless be ordered if it is determined that:

(1) Continuation of revocation proceedings is not warranted despite the violations found; or

(2) Incarceration pending further revocation proceedings is not warranted by the alleged frequency or seriousness of such violation or violations, and that the parolee is not likely to fail to appear for further proceedings, and that the parolee does not constitute a danger to himself or others.

(f) Conviction as probable cause: Conviction of a Federal, State, or Local crime committed subsequent to release on parole or mandatory release shall constitute probable cause for the purposes of this section and no preliminary interview shall be conducted unless otherwise ordered by the regional Commissioner.

(g) Local revocation hearing: A postponed preliminary interview may be conducted as a local revocation hearing by an examiner panel or other interviewing officer designated by the regional Commissioner provided that the prisoner has been advised that the postponed preliminary interview will constitute his final revocation hearing.

### § 2.49 Place of revocation hearing.

(a) If the prisoner requests a local revocation hearing, he shall be given a revocation hearing reasonably near the place of the alleged violation(s) or arrest, if the following conditions are met:

(1) The prisoner has not been convicted of a crime committed while under supervision; and

(2) The prisoner denies that he has violated any condition of his release.

- (b) If there are two or more alleged violations, the hearing may be conducted near the place of the violation chiefly relied upon as a basis for the issuance of the warrant or summons as determined by the regional Commissioner.
- (c) A prisoner who voluntarily waives his right to a local revocation hearing, or who admits any violation of his release. or who is retaken following conviction of a new crime, shall be given a revocation hearing upon his return to a Federal institution. However, the Regional Commissioner may, on his own motion, designate a case for a local revocation hearing.
- (d) A prisoner retaken on a warrant issued by the Commission shall be retained in custody until final action relative to revocation of his release, unless otherwise ordered by the regional Commissioner under § 2.48(d) (2). A parolee who has been given a revocation hearing pursuant to the issuance of a summons under § 2.44 shall remain on supervision pending the decision of the Commission.
- (e) Local revocation hearings shall be scheduled to be held within sixty days of the probable cause determination. Institutional revocation hearings shall be scheduled to be held within ninety days of the date of the execution of the violator warrant upon which the prisoner was retaken. However, if a prisoner requests and receives any postponement of his preliminary interview or revocation hearing, or consents to a postponed revocation proceeding initiated by the Commission; or if a prisoner by his actions otherwise precludes the prompt

conduct of such proceedings, the above stated time limits may be extended.

### § 2.50 Revocation hearing procedure.

(a) A revocation hearing shall be conducted by a hearing examiner panel or, in a local revocation hearing only, may be conducted by another official designated by the Regional Commissioner. In the case of a revocation hearing conducted by such other official or in the case of a revocation hearing conducted by a single examiner pursuant to § 2.47, a recommendation relative to revocation shall be made by the concurrence of two examiners on the basis of a review of the record. A revocation decision may be appealed under the provisions of § 2.25 and § 2.26, or § 2.27 as applicable.

(b) The purpose of the revocation hearing shall be to determine whether the prisoner has violated the conditions of his release and, if so, whether his parole or mandatory release should be

revoked or reinstated.

(c) The alleged violator may present witnesses and documentary evidence in his behalf. However, the presiding hearing officer or examiner panel may limit or exclude any irrelevant or repetitious statement or documentary evidence.

(d) At a local revocation hearing, the Commission may on the request of the alleged violator or on its own motion, request the attendance of persons who have given statements upon which revocation may be based. Those witnesses who are present shall be made available for questioning and cross-examination in the presence of the alleged violator unless the presiding hearing officer or examiner panel finds good cause for their non-attendance. Adverse witnesses will not be requested to appear at institutional revocation hearings.

(e) All evidence upon which the finding of violation may be based shall be disclosed to the alleged violator at the revocation hearing. The hearing officer or examiner panel may disclose documentary evidence by permitting the alleged violator to examine the document during the hearing, or where appro-priate, by reading or summarizing the document in the presence of the alleged violator.

(f) In lieu of an attorney, an alleged violator may be represented at a revocation hearing by a person of his choice. However, the role of such non-attorney representative shall be limited to offering a statement on the alleged violator's behalf with regard to reparole or reinstatement to supervision.

### § 2.51 Issuance of a subpoena for the appearance of witnesses or production of documents.

(a) (1) Preliminary Interview or Local Revocation Hearing: If any person who has given information upon which revocation may be based refuses, upon request by the Commission to appear, the regional Commissioner may issue a subpoena for the appearance of such witness. Such subpoena may also be issued at the discretion of the regional Commissioner in the event such adverse witness is judged unlikely to appear as re(2) In addition, the regional Commissioner may, upon his own motion or upon a showing by the parolee that a witness whose testimony is necessary to the proper disposition of his case will not appear voluntarily at a local revocation hearing or provide an adequate written statement of his testimony, issue a subpoena for the appearance of such witness at the revocation hearing.

(3) Both such subpoenas may also be issued at the discretion of the regional Commissioner if it is deemed necessary for orderly processing of the case.

(b) A subpoena issued pursuant to paragraph (a) of this section above may require the production of documents as well as, or in lieu of, a personal appearance. The subpoena shall specify the time and the place at which the person named therein is commanded to appear, and shall specify any documents required to be produced.

(c) A subpoena may be served by any Federal officer authorized to serve criminal process. The subpoena may be served at any place within the judicial district in which the place specified in the subpoena is located, or any place where the witness may be found. Service of a subpoena upon a person named therein shall be made by delivering a copy thereof to

such person.

(d) If a person refuses to obey such subpoena, the Commission may petition a court of the United States for the judicial district in which the parole proceeding is being conducted, or in which such person may be found, to require such person to appear, testify, or produce evidence. The court may issue an order requiring such person to appear before the Commission, and failure to obey such an order is punishable by contempt.

### § 2.52 Revocation of parole or mandatory release.

- (a) Whenever a parolee is summoned or retaken by the Commission, and the Commission finds by a preponderance of the evidence, that the parolee has violated a condition of the parole, the Commission may take any of the following actions:
- (1) Restore the parolee to supervision including where appropriate: (1) Reprimand (ii) Modification of the parolee's conditions of release (iii) Referral to a residential community treatment center for all or part of the remainder of his original sentence; or

(2) Revoke parole.

(b) If parole is revoked pursuant to this section, the Commission shall also determine, on the basis of the revocation hearing, whether reparole is warranted or whether the prisoner should be continued for further review.

(c) A parolee whose release is revoked by the Commission will receive credit on service of his sentence for time spent under supervision, except as provided

below:

(1) If the Commission finds that such parolee intentionally refused or failed to respond to any reasonable request, order, summons or warrant of the Commission or any agent thereof, the Commission may order the forfeiture of the time during which the parolee so refused or failed to respond, and such time shall not be tredited to service of the sentence.

- (2) If the parolee has been convicted of a new offense committed subsequent to his release on parole, which is punishable by a term of imprisonment, forfeiture of the time from the date of such release to the date of execution of the warrant shall be ordered, and such time shall not be credited to service of the sentence. An actual term of confinement or imprisonment need not have been imposed for such conviction; it suffices that the statute under which the parolee was convicted permits that trial court to impose any term of confinement or imprisonment in any penal facility. If such conviction occurs subsequent to a revocation hearing (i) which the Commission makes an independent finding of violation of conditions of parole), the Commission may reopen the case and schedule a further hearing relative to time forfeiture and such further disposition as may be appropriate. However, in no event shall the violator term imposed under this subsection, taken together with the time served before release, exceed the total length of the original sentence.
- (d) (1) Notwithstanding the above, prisoners committed under the Narcotic Addict Rehabilitation Act or the Youth Corrections Act shall not be subject to any forfeiture provision, but shall serve uninterrupted sentences from the date of conviction, except as provided in § 2.10 (b) and (c).
- (2) The commitment of a juvenile offender under the Federal Juvenile Delinquency Act may not be extended past the offender's twenty-first birthday unless the juvenile has attained his nineteenth birthday at the time of his commitment, in which case his commitment shall not exceed the lesser of two years or the maximum term which could have been imposed on an adult convicted of the same offense.

### § 2.53 Mandatory parole.

(a) A prisoner (including a prisoner sentenced under the Narcotic Addiction Rehabilitation Act, Federal Juvenile Delinquency Act, or the provisions of 5010 (c) of the Youth Corrections Act) serving a term or terms of five years or longer shall be released on parole after completion of two-thirds of each consecutive term or terms or after completion of thirty years of each term or terms of more than 45 years (including life terms), whichever comes earlier, unless pursuant to a hearing under this section, the Commission determines that there is a reasonable probability that the prisoner will commit any Federal, State, or local crime or that the prisoner has frequently or seriously violated the rules of the institution in which he is confined. If parole is denied pursuant to this section such prisoner shall serve until the expiration of his sentence less good time. The forfeiture of statutory good time shall be deemed in itself to indicate that the prisoner has frequently or seriously

violated the rules of the institution or institutions in which he has been confined.

(b) When feasible, at least sixty days prior to the scheduled two-thirds date, a review of the record shall be conducted by an examiner panel. If a mandatory parole is ordered following this review, no hearing shall be conducted.

- (c) A prisoner released on mandatory parole pursuant to this section shall remain under supervision until the expiration of the full term of his sentence unless the Commission terminates parole supervision pursuant to § 2.43 prior to the full term date of the sentence.
- (d) A prisoner whose parole has been revoked and whose parole violator term is five years or more shall be eligible for mandatory parole under the provisions of this section upon completion of twothirds of the violator term and shall be considered for mandatory parole under the same terms as any other eligible prisoners.

# § 2.54 Reviews pursuant to 18 U.S.C. 4203/4215.

- (a) The Attorney General, within thirty days after entry of a Regional Commissioner's decision, may request in writing that the National Appeals Board review such decision. Within sixty days of the receipt of the request the National Appeals Board shall, upon the concurrence of two members, affirm, modify, or reverse the decision, or order a rehearing at the institution or regional level. The Attorney General and the prisoner affected shall be informed in writing of the decision, and the reasons therefor.
- (b) Notwithstanding the above provision, the Commission, by majority vote, may, upon its own motion, review any decision of a Regional Commissioner relative to grant or denial of parole, imposition of parole conditions or revocation of parole, upon the receipt of new and significant information. Referrals for this purpose may be made by not less than three Commissioners, Such review by the Commission must be made within thirty days following the entry of the decision by the Regional Commissioner. Following the review the Commission shall inform the affected prisoner in writing of its action and, if parole is denied, the reasons therefor.
- (c) Notwithstanding the provisions of §§ 2.23-2.26 and § 2.28, any decision made by a Regional Commissioner or the National Appeals Board shall, upon the petition of not less than three Commissioners, be referred to the full Commission for review and, by majority vote, affirmed, modified, or reversed. Such petition must be submitted to the Chairman of the Commission and be acted upon by the Commission not later than 30 days from the date of entry of the decision to be reviewed. The prisoner shall receive a written, notice of this referral, which shall stay the decision in his case until such review has been completed. Following review by the full Commission, the prisoner shall be informed in writing of the Commission's decision and, if parole is denied, of the reasons therefor.

### § 2.55 Disclosure of Records.

(a) Prior to an initial parole hearing conducted pursuant to § 2.13 or any review hearing thereafter, a prisoner may review reports and other documents in the institution file which will be considered by the Commission at his parole hearing. These documents are generally limited to official reports bearing on the prisoner's offense behavior, personal history, and institutional progress. Review of such reports shall be permitted by the Bureau of Prisons pursuant to its regulations within seven days of a request by the prisoner, except that in the case of reports which must be sent to the originating agency for clearance pursuant to paragraph (c) of this section, a reasonable amount of time shall be permitted to obtain such clearance. Copies of reports and documents may be furnished under applicable Bureau of Prisons regulations.

(b) A report shall not be disclosed to

the extent it contains:

(1) Diagnostic opinions which, if known to the prisoner, could lead to a serious disruption of his institutional program;

(2) Material which would reveal sources of information obtained upon a

promise of confidentiality; or

(3) Any other information which, if disclosed, might result in harm, physical or otherwise, to any person. The term "otherwise" shall be deemed to include the legitimate privacy interests of such person under the Privacy Act of 1974.

- (c) It shall be the duty of the agency which originated any report or document referred to in paragraph (a) of this section to determine whether or not to apply any of the exceptions to disclose set forth in paragraph (b) of this section. If any report or portion thereof is deemed by the originating agency to fall within an exception to disclosure, such agency shall prepare and furnish for inclusion in the institution file a summary of the basic contents of the material to be withheld, bearing in mind the need for confidentiality or impact on the prisoner, or both. In the case of a report prepared by an agency other than the Bureau of Prisons, the Bureau shall refer such report to the originating agency for a determination relative to disclosure, if the report has not been previously cleared or prepared for disclosure.
- (d) Upon request by the prisoner, the Commission shall make available a copy

of any record which it has retained of a parole or parole revocation hearing pursuant to 18 U.S.C. 4208(f).

- (e) Except for deliberative memoranda referred to in paragraph (f) of this section, reports or documents received at regional offices which may be considered by the Commission at any proceeding shall be forwarded for inclusion in the prisoner's institutional file so that he may review them pursuant to paragraph (a) of this section. Such reports will first be referred by the Commission to originating agencies pursuant to paragraph (c) of this section for a determination relative to disclosure if the report has not previously been cleared or prepared for disclosure.
- (f) Duplicate copies of records in a prisoner's institutional file as well as deliberative memoranda among Commission Members or staff which do not contain new factual information relative to the parole release determination are retained in Parole Commission regional office files following initial hearing. Records maintained in these files, shall be made available to prisoners, parolees, mandatory releasees, their authorized representative and members of the public upon written request in accordance with applicable law and Department of Justice regulations at 28 CFR Part 16, Subparts C & D. The Commission reserves the right to invoke statutory exemptions to disclosure of its files in appropriate cases under the Freedom of Information Act or Privacy Act text provisions and Alternate Means of Access.

### § 2.56 Special parole terms.

- (a) The Drug Abuse Prevention and Control Act, 21 U.S.C. 801 to 966, provides that, on conviction of certain offenses, mandatory "special parole terms" must be imposed by the court as part of the sentence. This term is an additional period of supervision which follows the completion of the regular sentence (including competition of any period on parole or mandatory release).
- (b) At the time of release under the regular sentence, whether under full term expiration or under a mandatory release certificate or a parole certificate, a separate Special Parole Term certificate will be issued to the prisoner by the Bureau of Prisons.
- (e) Should a release be found to have violated conditions of release during supervision under his regular sentence,

i.e., before commencement of the Special Parole Term, he will be returned as a violator of his basic supervision period under his regular sentence; the Special Parole Term will follow unaffected, as in paragraph (a) of this section. Should a release violate conditions of release during the Special Parole Term he will be subject to revocation on the Special Parole Term as provided in § 2.52, and subject to reparole or mandatory release under the Special Parole Term.

(d) If the prisoner is reparoled under the revoked Special Parole Term a certificate of parole to Special Parole Term is issued by the Commission. If the inmate is mandatorily released under the revoked "special parole term" a certificate of mandatory release to Special Parole Term will be issued by the Bureau

of Prisons

(e) If the prisoner is terminated from regular parole under § 2.43, the Special Parole Term commences to run at that point in time. Early termination from supervision from a Special Parole Term may occur as in the case of a regular parole term, except that the time periods considered shall commence from the beginning of the Special Parole Term.

### 8 2.57 Prior orders.

Any order of the United States Board of Parole entered prior to May 14, 1976, including, but not limited to, orders granting, denying, rescinding or revoking parole or mandatory release, shall be a valid order of the United States Parole Commission according to the terms stated in the order.

### § 2.58 Absence of hearing examiner.

In the absence of a hearing examiner, a regional commissioner may exercise the authority delegated to hearing examiners in § 2.23.

### § 2.59 Appointment of committees.

The Chairman shall appoint four permanent committees, as follows: (a) Policy, (b) Budget, (c) Personnel and training, (d) Research, and in addition such ad hoc committees as may from time to time be approved by a majority of the Commissioners, to study, review, and recommend to the Commission and Chairman regarding policies and procedures of the Commission. Such Committees shall be appointed from among the Commissioners.

[FR Doc.77-22623 Filed 8-4-77;8:45 am]

FRIDAY, AUGUST 5, 1977
PART IV



DEPARTMENT OF HEALTH, EDUCATION, AND WELFARE

Office of the Secretary

NONDISCRIMINATION
IN FEDERALLY ASSISTED
PROGRAMS

Proposed Annual Operating Plan for FY 1978

### DEPARTMENT OF HEALTH, EDUCATION, AND WELFARE

Office of the Secretary

NONDISCRIMINATION IN FEDERALLY ASSISTED PROGRAMS

Proposed Annual Operating Plan for FY 1978

PUBLICATION FOR COMMENT

The Office for Civil Rights (OCR) is issuing its proposed FY 1978 Annual Operating Plan for public comment. The plan, set forth in detail in the accompanying narrative and charts, establishes the numbers of complaint investigations, compliance reviews, and the other related activities which will be conducted by the investigative, managerial, and non-clerical support staff of OCR's Headquarters and ten Regional offices.

Responses to OCR's solicitation for comments should be received no later than September 6, 1977, and should be addressed to:

Director
Office for Civil Rights
Department of Health, Education, and Welfare
Washington, D.C. 20201

OCR will review its proposed plan in light of the comments received and, within the overall staffing levels included in the plan, will revise the plan to reflect suggestions which have merit. Commentators who recommend OCR undertake activities in addition to those proposed in the plan are urged to indicate the specific activity they would displace, together with the reason for their position, or to identify the specific area to be changed and the impact of that change on the remaining elements of the proposed plan.

Persons seeking clarification of any of the provisions of the proposed plan may contact the Office for Civil Rights, at the above address.

> DAVID S. TATEL, Director.

AUGUST 1, 1977.

OFFICE FOR CIVIL RIGHTS
PROPOSED ANNUAL OPERATING PLAN FOR
FY 1978

I. Purpose

The Office for Civil Rights (OCR) within the Office of the Secretary is responsible for assuring that Federally assisted programs are free from unlawful discrimination. The Annual Operating Plan (AOP) represented in the narrative and charts below describes the compliance and compliance related activities OCR will undertake in FY 1978 to carry out that responsibility.

II. Jurisdiction

OCR is charged with enforcing the following statutes and executive orders as they relate to the expenditures of HEW funds:

Title VI of the Civil Rights Act of 1964, which prohibits discrimination on the basis of race, color, or national origin in federally assisted programs and activities; Title IX of the Education Amendments

Title IX of the Education Amendments of 1972, which prohibits discrimination on the basis of sex in federally assisted education programs;

Section 504 of the Rehabilitation Act of 1973, which prohibits discrimination against qualified handicapped persons on the basis of physical or mental handicap in federally assisted programs;

Sections 799A and 855 of the Public Health Service Act, which require that schools of medicine and nurse training, as a condition of receiving Federal support, must provide assurances that there is no discrimination on the basis of sex in admissions:

Executive Order 11246, as amended, which prohibits employment discrimination on the basis of race, color, sex, religion, and national origin and requires affirmative action on the part of Federal contractors and subcontractors:

Title VII of the Education Amendments of 1972 (referred to as the Emergency School Aid Act or ESAA), which provides aid to desegregating institutions;

Age Discrimination Act of 1975, which prohibits discrimination in all federally funded programs on the basis of an individual's age.

### III. Balanced Compliance Program

OCR has established as its goal for FY 1978 the conduct of a balanced compliance program designed in view of existing resources to be responsive to the needs of all protected groups served by all institutions and programs over which OCR has jurisdiction. There are two major assumptions underlying the development of this annual operating plan. First, it is OCR's best judgment regarding the implementation of a balanced enforcement program that 55 percent of the total investigative time should be assigned to complaint investigations and 45 percent to compliance reviews. Reserving 45 percent of the time for reviews allows OCR the flexibility to conduct OCR selected compliance reviews as well as pre-grant and pre-award reviews required by law.

Second, OCR recognizes that limiting the resources assigned to complaint processing will result in an increase in the backlog of complaints. However, OCR intends to insure that this resulting increase in the backlog is only a shortterm phenomenon. Through increased resources and efficiencies, OCR plans to attempt to reduce the backlog and become current in complaint processing by the end of fiscal year 1980 or 1981. Last year's Annual Operating Plan which indicated that the backlog could be reduced by the end of this fiscal year, was totally unrealistic and disregarded OCR's other lawful responsibilities. The proposed fiscal year 1978 operating plan. unlike last year's, represents an effort to utilize existing resources to fulfill OCR obligations to complainants and at the same time to fulfill other statutory responsibilities. The principle of this plan. together with similar plans for future fiscal years, which will reflect increased resources and efficiencies, will lay the ground work for the elimination of the complaint backlog and the permanent installation of a balanced and responsible enforcement program. The annual operating plan for FY 1978 represents an effort to use resources to meet OCR's obligations to complainants in the fairest and most expeditious manner possible and, at the same time, allow OCR to carry out its other statutory responsibilities.

The essential elements of a balanced compliance program are:

Compliance reviews; Fre-award and pre-grant reviews; Complaint investigations; and Expansion of civil rights responsibilities into all HEW programs.

Balance will be achieved by distributing staff resources and determining numbers of activities to be conducted in a manner that insures coverage of all of OCR's legislative and executive order authorities and all types of recipients/ contractors (for example, public school districts, colleges and universities, vocational schools and programs, hospitals, nursing homes, medical laboratories, and various state and local government agencies receiving Federal financial assistance). Since complaint investigations and pre-funding reviews are limited in scope, compliance reviews will be planned and scheduled to include those issues, jurisdictions, and protected groups which would not otherwise be addressed or which would receive inadequate attention were OCR's activities limited to complaint activities.

The actual numbers of activities to be conducted are computed on a total expected OCR staff of 1102 positions in FY 1978. Subtracted from this total is a 3 percent vacancy rate considered acceptable by the Office of Management and Budget. The staffing figures in the plan indicate professional staff only.

### IV. Compliance Program

A. Complaint Investigations.-A complaint is an allegation that an HEW contractor or a recipient of HEW funds has violated one or several of the legal authorities OCR is responsible for enforcing by discriminating against an individual or group of individuals, OCR will begin FY 1978 with a backlog of approximately 3025 complaints. During the fiscal year, it expects to receive an additional 2455 complaints. By allocating 232 person years or 55 percent of its total investigative staff, OCR will be able to resolve 42 percent of the backlog and will investigate 10 percent of the new re-ceipts, or a total of 1501 complaints. Within the total staff allocation for complaints, the distribution of resources to the regions to conduct complaint investigations will be based on each region's proportion of the total complaints on hand as of October 1, 1977, and the total number of complaints expected to be received during the fiscal year.

During FY 1978, each regional office

<sup>\*</sup>This is the staffing level expected to be authorized by Congress. However, the Office of Management and Budget job ceilings may prevent OCR from attaining the authorized level. Accordingly, the activities projected in this AOP may have to be reduced to reflect actual staffing levels.

NOTICES

will first bring to completion all complaint investigations active on the first of October. All remaining complaints (backlogged and new receipts) will be divided by enforcement area (race, national origin, sex, and handicap), and investigative resources allocated in proportion to the number of complaints in each enforcement area. For example, if 25 percent of a regional office's complaints allege national origin discrimination, then 25 percent of that office's complaint investigation staff, remaining after the assignment of staff to active complaints, will be allocated to the investigation of national origin complaints. Within the proportions established for each regional office, OCR will investigate backlogged and newly received complaints on a one-to-one ratio and within these categories, in chronological order by date of receipt. The proportions for each regional office will be revised on a semi-annual basis and the allocation of resources altered to conform to changes in the proportions. Certain complaint investigations will be consolidated as fol-

 Before any complaint investigation is initiated, the Regional Office will determine if there are other similar allegations on hand against the same recipient/contractor. If so, all such complaints will be consolidated into one investigation.

2. If a compliance review of a recipient/contractor is scheduled, investigation of all outstanding complaints against the recipient/contractor will be incorporated into the review.

During FY 1978, OCR will concentrate heavily on developing new techniques for increasing complaint handling efficiency. Revised Complaint Handling Procedures, improved work measurements, better policy articulation, improved training, and new approaches to resolving complaints will help OCR reduce its backlog and handle new complaints more effi-

ciently and expeditiously.

B. Pre-Funding Activity.--Under Federal law and executive order OCR, is responsible for conducting a variety of pre-funding reviews. Pre-grant reviews are required to ensure compliance with the civil rights requirements of Section 706 of the Emergency School Aid Act (ESAA). During FY 1978, OCR will allocate 61.44 investigative person years to the processing of ESAA applications, including Title VI action on ineligible or waiver-denied applicant school districts. Routine health and human development pre-grant clearances relevant to Title VI compliance will require a total of 15.5 person years during FY 1978. Also since under Executive Order 11246, OCR has been designated a compliance agency by the Department of Labor's Office of Federal Contract Compliance Programs (OFCCP), 16.1 investigative person years will be devoted to the conduct of preaward reviews to determine the equal employment compliance status of prospective recipients of HEW contracts in excess of \$1 million. In addition to these three specific areas of pre-funding activity, OCR will be working with the HEW principal operating components (POC's) through OCR's new Office of

Program Review and Assistance to develop uniform, effective pre-grant review procedures for the Department's funding programs.

C. Compliance Reviews.-A compliance review is an OCR initiated investigation of the policies and practices of a reciptent of Federal funds to determine whether that recipient is in compliance with the legal authorities OCR enforces. Unlike a complaint investigation or a required pre-funding review, a compliance review allows OCR to select the site of an investigation, the jurisdiction(s) under which the investigation will be conducted, and the issue(s) the investigation will include. Thus, compliance reviews provide OCR with a tool essential to addressing those areas of discrimination not targeted by complaints or pre-grant reviews.

During FY 1978, OCR will conduct 185 compliance reviews requiring a staff allocation of approximately 113 investigative person years. Investigative staff from the regional offices as well as staff from the headquarters Compliance and Enforcement Office's Compliance Program Branch will participate in the investigations, assisted by staff from the Office of Policy, Planning and Research where policy and procedures are not firmly established.

The reviews will be geographically disbursed and, on a national level, will cover a comprehensive range of issues in all program areas and under all jurisdictional authorities. The selection of the actual issues and jurisdictions to be covered will be made following an assessment of the issues and jurisdictions covered in the complaint investigations and pre-funding reviews to be conducted. Types of recipients and issues to be addressed in compliance reviews include:

Program area	Recipient	Issue
Elementary and secondary educa- tion:	State education agencies	cies affecting the treatment of students and the employment of faculty by SEA's.
	Vocational/technical schools.	Employment within agency.  Admission to schools.  Admission to programs.
	Nonvocational/technical State administered schools.	Admission to schools.
	Local education agencies.	Admission to programs. Discipline. Special education.
Higher education	State education agencies which administer formerly dual systems.	Migrant education. Failure to eliminate vestiges of formal racial duality. Potential discriminatory impact of all policies affecting the treatment of students and employment of staff and faculty by public
-	Other State education agencies	higher education institutions.  Agency employment.  Potential discriminatory impact of policies.  Agency employment.
	Professional schools.	Admissions. Retention. Supportive services.
	Graduate schools (nonprofessional)	Admissions, Retention.
	Undergraduate schools	Retention.
Nonprofit organiza-	Community colleges.	Supportive services. Employment.
tions. Health and human development.	Residential health facilities, (e.g., hospitals, nursing homes).	Site selection. Access to facilities. Room assignment.
	Nonresidential health and human develop- ment facilities (e.g., family health centers, vocational rehabilitation centers and work shops). State and local health and human develop-	Comparability of treatment. Site selection. Access to facilities. Access to services. Delivery of services. Policies affecting the location and provision of services.
	ment agencies.	Adequacy of monitoring vendors.  Adequacy of complaint handling procedures.

During the fiscal year, OCR will also complete 44 on-going elementary and secondary education Title VI reviews specifically aimed at ensuring equal educational opportunity for non-native English speakers (commonly referred to as Lau reviews) and will continue, subject to careful reevaluation and possible limitation, the comprehensive reviews of the public school systems in New York City, Philadelphia, Chicago, and Los Angeles now underway. In addition, OCR will conduct Title VI student assignment reviews where evidence indicates that unconstitutional segregation or its effects exists and where constitutionally accept-

able remedies are achievable. These reviews will cover Title IX and Section 504 issues as well.

### V. Support

Conduct of the compliance activities described in IV above requires a variety of regional and headquarters support staff. While, with few exceptions, these staff will not participate directly in the conduct of reviews and investigations, they will provide the management and staff support necessary to the full development and implementation of the balanced compliance program. The descriptions below are not meant to be exhaustive, nor do they delineate all of the

functions to be performed by the staff. They are meant only to explain the activities which are most critical to the implementation of the compliance program.

A. Regional Management and Professional Support Staff. This staff of 111 provides overall regional office management and management support, including reporting, assessments of regional progress, evaluations of compliance activities, provision of data collection services, provision of administrative support services, and coordination of the correspondence of the regional office, which is largely prepared by the regional investigative staffs.

B. Headquarters Staff.

Professi	
Unit sta	
Office of the Director	4
Executive Secretariat	3
Office of Inter-Governmental Af-	
fairs	8
Office of Public Affairs	7
Office of the Assistant Director, Ad-	
ministration and Management	27
Office of the Deputy Director, Com-	
pliance and Enforcement	3
Division of Operational Planning	-
and Support	77
Division of Technical Review and	-
	65
Assistance	10
Division of Training	10
Office of the Associate Director, Policy,	-
Planning and Research	2
Division of Policy and Procedures	32
Division of Planning and Research.	12
Office of the Deputy Director, Pro-	
gram Review and Assistance	2
District of Personal Physician and	
Division of Program Planning and	-
Appraisal	8
Division of Intra-Departmental	
Technical Assistance	13
THE RESERVE TO SERVE THE PARTY OF THE PARTY	700
Total	203

1. Director.-In addition to the Director, the staff will include the Executive Secretariat responsible for document control: the Office of Public Affairs responsible for planning and directing a comprehensive civil rights public information program; and the Office of Inter-Governmental Affairs responsible for lialson with the Congress, Federal Departments and agencies, and State and local governments and their representative organizations. All three of these offices are responsible for coordinating the preparation of responses to the large volume of congressional, governmental, and other correspondence received by OCR; much of the actual letterwriting is done by the staffs of the operational and policy divisions.

2. Administration and Management.— This staff will provide financial management and personnel and supply services, as well as support of information systems (the automated case tracking system and the other elements of the overall information management system). In addition to providing management analysis, the staff will coordinate the development of the short-term and annual operating plans.

3. Compliance and Enforcement.— This office will provide a wide range of

support services to the compliance program and will be the direct link between headquarters and the regions. Staff will oversee the conduct of the compliance activities to ensure that all policies, procedures, and standards are uniformly and effectively implemented. They will also provide direct technical assistance to the regions in the development of cases. Approximately one-third of the staff will participate in the compliance reviews conducted by the regions.

The staff assigned to training will develop, maintain, and administer an OCR staff training program. FY 1978 will be the first year of implementation of an intensive training program designed to provide staff with the skills necessary to implement an effective compliance program. This first year will focus on investigation and negotiation skills development.

4. Policy, Planning and Research.—Roughly two thirds of this staff will be devoted to the development of policies and procedures. Organized by type of discrimination (age, sex, race, handicap) and by function (legislative analysis, dissemination), this staff will concentrate its efforts on developing policies and procedures in all areas, including but not limited to:

Complaint investigation and compliance review procedures.

Guidelines for state health and human development agencies.

Anti-discrimination provisions for the location of health and human development facilities and services.

Policy on vocational education programs.
Policy on discipline in schools.
Title IX policies.

Guidelines, standards, and remedies for language barrier problems in all facilities. Procedures for monitoring corrective action agreements.

Policy on recipient/contractor record keeping. Procedures for full implementation of the 504 regulation.

The remaining staff will be responsible for long range planning, research and evaluation, and the design and conduct of surveys and other bulk data collection activities. Projects stemming from this last activity may, subject to the availability of funds, include: the Adams Higher Education state plan evaluation instrument, the survey of college and university enrollments and degrees conferred, the comprehensive urban review data collection efforts, vocational education survey, special school survey, planning and development of a comprehensive health and human development survey, and the elementary and secondary education survey.

In addition, staff will provide data collection and analysis support in the development of individual cases.

5. Program Review and Assistance.— This office, created in FY 1977, is to carry out the new efforts of the Department to infuse civil rights responsibilities into the activities of the HEW program agencies (principal operating components or POC's). It is these agencies that have direct contact with the recipients and beneficiaries of Federal funds. The staff will concentrate their FY 1978 efforts on developing and promoting civil rights responsibilities and providing technical assistance. The focus will be on those operating agencies which have the most critical civil rights responsibilities.

Specific objectives to be accomplished in FY 1978 include:

- Develop and implement a Self-Assessment Guide with POC's.
- Develop short and long range goals for OCR/FOC civil rights initiatives.
   Develop a civil rights orientation program
- for POC program and contracting offices.

  4. Develop pre-award grant and contract clearance procedures to be implemented
- jointly by POC's and OCR.

  5. Participate in the revision of the procurement manual used by POC's.
- Develop consolidated language for all assurances required in grant applications and in contracts.
- Establish a process for OCR clearance of Vocational Education State Plans.
- Coordinate, with the POC's, the 504 technical assistance effort of the Department.
- Develop and implement the initial phase of an OCR/POC technical assistance, education and advocacy program.

### C. General Counsel.

nit:	Professional staff
Headquarters	33
Total	60

The staff of the Civil Rights Division of the Office of the General Counsel (OGC) work in direct support of OCR's compliance program. In addition to OGC involvement in preparing a case for and following it through the administrative enforcement process, emphasis will be placed on OGC involvement in all aspects of OCR's program, including participation in the development and conduct of reviews and investigations, reviews of pertinent legislation and regulations, provision of formal opinions, and participation in the development of policies and procedures.

### VI. Work Measurement

To determine investigative staff requirements for complaint and review activity, standard times for each type of activity were computed and then applied against the number of activities to be performed during FY 1978.

The standard time for complaints is 44 person days per investigation. This is an average. The actual work time required to process any one given complaint can vary from as little as a day to as much as the equivalent of several months, depending on the nature of the allegation. OCR has set a goal to increase its efficiency in the investigation of complaints by 30 percent by the end of the fiscal year. Since this goal will be attained throughout the year, the numbers of new investigations planned for FY 1978 reflect an overall increase of 15 percent.

The standard times for reviews are as follows:

NOTICES 39827

ype: Person days per re	view
ESAA applications	14
Executive order 11246 preaward revisions.	23
Lau reviews	20
Brown reviews	406
State elementary and secondary education agencies.	118
Vocational/technical schools	84
Health pre-grant reviews	1
Monitoring of Adams State higher education systems.	84
Nonvocational/technical State ad-	84
ministered elementary and sec- ondary education schools.	
Elementary and secondary school district discipline practices.	62
Elementary and secondary school district special education pro- grams.	46
Elementary and secondary school district migrant education practices.	76
Formerly dual State higher educa- tion systems.	169
Other State higher education sys- tems.	157
Professional schools	105
Graduate schools	105
Undergraduate schools	111
Community colleges	105
Nonprofit organizations	81
Residential health facilities Nonresidential health and human	118
development facilities.  State and local health and human	35
development agencies.	149

The time required to conduct any given review depends on the number of issues investigated and the number of jurisdictional authorities covered. The standard times above assume that each type of review listed will cover the issues delineated in Section III and will be conducted under all applicable authorities.

The final standard times for both complaints and reviews shown on the activ-

ity charts are stated in person years. A person year contains 169 person days, that figure representing the number of days available to each investigator for complaint and review activity. The 169day figure was derived by subtracting from the total working days available in a year, Federal holidays, average annual and sick leave, and an average of days an investigator can be expected to spend performing other activities not directly related to the conduct of specific complaint investigations and reviews, such as providing technical assistance, answering non-case related correspondence, attending meetings and conferences, and participating in training. During FY 1978 efforts will be made to reduce the amount of time spent by investigators on non-case related activities. In addition, for all activities on the charts on which some investigative work will have been completed prior to October 1, the standard times have been reduced by 50 percent.

### VII. AOP Monitoring and Adjustment

OCR will regularly monitor progress under the AOP. At the end of the second quarter of the fiscal year it will revise the second six months of the plan as required by assessments of the first six months. The revising will include both the numbers and kinds of activities and the time measures used. For example, if incoming complaints are being received at a rate different than expected at the beginning of the fiscal year, the proportions will be recomputed; or if the standard times previously estimated for completing compliance activities were inaccurate, they will likewise be recomputed and built into the plan.

PLANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

STRAMARY

(Staff time shown in investigative person years)

					11 11 11		
TOTAL		PERSON YEARS	73.20	95.14	23.142	₹T*#E	231.90
		NAMESS	203	999	git	183	1501
INVESTIGATIONS		PERSONS YEARS	17.04	17.72	5.09	12.63	22.48
	NEW ENCENTES	MINERR	#	80	ສ	27	237
	90	PERSON YEARS	47.81	18.17	18.4	12,41	53.59
	ENCKLOG	NUMBER	8	.88	83	35	SHS
CARRIOVER*	No.	PERSON YEARS	ao***	59.25	13,46	от.е	125.83
		MUNGER	346	504	104	2	1022
BASIS			RACE 2	SEX	MATIONAL	BANDICAP	TOTALS

\*Carryover means all complaints in which an investigation was initated prior to October 1, 1977 but which have not been resolved as of that date.

PLANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

PECTON I

(Staif time shown in investigative person years)

13		,			N DES		
		PERSON YEARS	3.48	9.6	15.	8.	11.64
TOTAL		NUMBER	₹	ŧ	m	10	19
	59	PERSONS YEARS	#:	1.33	87	22.	12.2
CORS	NEW RECEIPTS	MUMBER	a	9	н	1	70
NEW INVESTIGATIONS	500	PERSON YEARS	4	1,33	22.	8.	2.2
	EACKLOG	MAGER	3	v	τ	1	og .
WER*		PERSON YEARS	2.60	3.97	£,	8.	7.22
CARRIOVER*		MINESE	50	22	1	4	47
HASIS			BACE	SEX	NATIONAL ORIGIN	RANDICAP	TOTALS

FIANNEL COMPLAINT RESOLUTION ACTIVITY-FY 1978 (Staff time shown in investigative person years)

REGION II

						1	
		PERSON YEARS	64.6	9.74	2.89	5.72	27.84
TOTAL		NUMBER	表	8	15	æ	151
	10	PERSONS YEARS	2.88	2.66	21.12	2.21	8.87
SHO	MEW RECEIPTS	MARKE	13	12		10	94
NEW INVESTIGATIONS	500	PERSON YEARS	3.10	2.66	777	2,21	60.6
	BACKLOG	MUNERER	Ħ.	टा	5	OI	4
VER*		PERSON YEARS	3.51	24°4	59*	1,30	9.88
CABRYOVER*		MUMBER	Z	ā	5	OT.	2
MASTS	araus	The same	HACE	XEX	MATICHAL	HAMDICAP	TOTALE

PLANNED COMPLAINT RESOLUTION ACTIVILIY-FY 1978

REGION III

(Staff time shown in investigative person years)

_							
		PERSON YEARS	64.9	0.70	1.78	3.57	18.46
TOTAL	THE PERSON	MINEER	88	23	п	19	121
	23	PERSONS YEAR	1.77	1,55	4	1.33	5.09
SHOOT	NEW RECEIPTS	MUNGER	8	7	a	vo	23
NEW INVESTIGATIONS	507	PERSON YEARS	1.99	1.77	8:	1,33	5.31
	BACKLOG	MOMBER	6,	89	7	10	ħ2
YER*		PERSON YEARS	2.73	3.38	1.04	16.	8.06
CABRYOVER*	The state of the s	MONERA	E E	38	9	7	74
BISWE	The San		FACE	SEX	RATIONAL ORIGIN	HANDICAP	TOTALE

\*Carryover means all complaints in which an investigation was initated prior to October 1, 1977 but which have not been resolved as of that date.

PIANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

WINDING IV

	-		,	F III			
		PERSON YEARS	13.61	8,45	1,21	6.80	30.13
TOTAL		MUNEER	28	83	7	₹€	176
1000	(0)	PERSONS YEARS	3.5	2,21	ā,	2.88	70.6
ONES	NEW RECEIPTS	MUNESE	16	07	8	13	1/2
NEW INVESTIGATIONS	507	PERSON YEARS	3.76	2,21	4	2,88	9.29
100	BACKLOG	MANABER	17	OT .	2	13.	24
WER*		PERSON YEARS	6.37	4.03	.33	1.04	7.11.
CARRYOVER.*		MUNEES	64	R	£	. 8	88
BASIS			SACE	TES	NATIONAL	HANDICAP	TOTALS

\*Carryover means all complaints in which an investigation was initated prior to October 1, 1977 but which have not been resolved as of that date.

PLANDED COMPLAINT RESOLUTION ACTIVITY-FY 1978

(Staff time shown in investigative person years)

EASIS	CAERTOVER*	YER*		NEW INVESTIGATIONS			TOTAL	13
			BACKLOG	907	NEW RECEIPTS	93		
	MUMBER	PERSON YEARS	MUNESER	PERSON YEARS	MUMBER	PERSONS YEARS	NUMBER	PERSON YEARS
RACE	бц	5.98	7	1.55	7	1.55	8	9.68
SEX	148	17,24	41	3.76	17	3.76	182	24.76
NATIONAL	10	1,30	1.	22.	1	.22	12	1.74
EANDICAP	17	2.2	6	1.99	6.	1.99	35	6.19
TOTALS	22 <sup>4</sup>	26.73	34	7.52	• **	7.2	262	h1.77

PLANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

REGION VI

(Staff time shown in investigative person years)

		PERSON YEARS	24,36	13.44	7.08	3.83	148.71
TOTAL		MUNERS	155	85	9	13	307
	62	PERSONS YEARS	5,09	2,88	1,33	1.33	10,63
Sigo	NEW RECEIPTS	MUMBER	83	13	9	9	84
NEW INVESTIGATIONS	00	PERSON YEARS	5.09	2,88	1.33	1.33	10,63
	BACKLOG	MUNGER	23	13	9	9	84
42R*		PERSON YEARS	14.18	89.7	24.4	101	27.45
CARRYOVER*		MUMBER	109	26	表	6.	an
BASIS		The state of the s	RACE	XXX	NATIONAL CRIGIN	HANDICAP	TOTALS

PLANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

REGION VII

(Staff time shown in investigative person years)

_									
		PERSON YEARS	1.58		STATE OF THE PERSON NAMED IN	3.83	1.35	2,43	62*6
TOTAL		MUMBER	80			×	0.	ET.	. 8
	SZ	PERSONS YEARS	99.			1.12	22.	86.	2.89
SHOO	NEW RECEIPTS	MINERE	т			us	1	*	13
NEW INVESTIGATIONS	50	PERSON YEARS	99*			1.12	82.	68,	2.89
	BACKLOG	MUNESS	8	The same of		2	н		នា
WER*	The state of the s	PERSON YEARS	98.			1.69	16.	.65	3.51
CARRYOVER*	THE PERSON NAMED IN	MUNCERR	2.	Charles of the Control of the Contro	100	83	7	i.	<b>%</b>
BASIS		Total Control of the last	BACE	The second second		SEX	MATIONAL	HANDICAP	TOTAIS

PLANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

REGION VIII

				THE RE			
		PERSON TEARS	F. 78	5.32	1,01	1,8	9.26
TOTAL		MAGER	п	E)	5	9	65
	S	PERSONS YEARS	22.	38.	ą	78.	1.98
CONS	NEW RECEIPTS	MARKE	1	6	a	3	6
NEW INVESTIGATIONS	907	PERSON YEARS	4	8.	#	4	2.21
	BACKLOG	NUMBER	2	4	2	2	10
******		PERSON YEARS	1.04	3.77	.13	.13	5.07
CARRITOVER*		MUNERS	æ	36	1	1	94
BASIS			BACE	SEX	NATIONAL ORIGIN	EANDICAP	TOTALS

\*Carryover means all complaints in which an investigation was initated prior to October 1, 1977 but which have not been resolved as of that date.

PLANNEL COMPLAINT RESOLUTION ACTIVITY-PY 1978

NECTOR IX

SASIS	CARRY	CARRYOVER*		NEW INVESTIGATIONS	OWS		TOTAL	
		THE REAL PROPERTY.	BACKLOG	100	NEW RECEIPTS	199		
	MUNCHER	PRESON YEARS	MUMBER	PERSON YEARS	NUMBER	PERSONS YEARS	NUMBER	PERSON YEARS
FACE	3	5.14	4	8.	7	8.	R	26.9
Mass	f)	6.37 3	0	1,33	10	1.33	đ	9.03
MATIONAL ORIGIN	eq.	2.99	6	78.	m	*8	58	4,31
HAIDICAP	9	87.	5	21.12	3	21.12	16	3.02
TOTALS	150	15.28	18	4,00	92	4.00	156	23.28
				The second secon				

\*Carryover means all complaints in which an investigation was initated prior to October 1, 1977 but which have not been resolved as of that date.

PLANNED COMPLAINT RESOLUTION ACTIVITY-PY 1978

REGION X

							LE WHILL
		PERSON TEARS	2.43	47.7	1,56	£.	11.52
TOTAL		MUMBER	18	59	टा	m	8.
	59	PERSONS YEARS	0	25.	0	0	22.
CORES	NEW RECEIPTS	MARKE	0	1	0	0	1
NEW INVESTIGATIONS	507	PERSON YEARS	22.	27.	0	0	#.
	BACKLOG	MARKE	1	1	0	0	2
VSR*		PERSON YEARS	2.21	6.70	1.56	.39	10.86
CARRYOVER*		NUMBER	17	63	12	ണ	32
BASIS			BACE	SEX	NATIONAL ORIGIN	HANDICAE	TOTALS

\*Carryover means all complaints in which an investigation was initated prior to October 1, 1977 but which have not been resolved as of that date.

	-	-													
	Married 1/	1	61,44	16.12	15,55 2/	50.7	20,00	83	16.80	7.60	3,33	2,16	100	2.70	3
	-	1	13	124	2339	00	-	3	-	2	0	10	3	3	**
	2000		11 .86 763	25	55.				2.40						
	CT Lewis L			-	88				7						
	200	1000	3.76	1.35	2.21		6.00	177	4.80						
(3)	Che Pare			n	.59 341		1		2						
1978 Cn ven	-	Service Constru	4,40	C.	.59	1000									
T - TI	Parent and	1		"	18								1	150	
CTIVITY	S core	STORY	1.45	11	1.24										
TIES A	goests crr	ACTE A	199	(h	192								-		
CLAIF LIKE Shows in investigative person vents)	35	STORES	20,40	1 22	1.62	1.02		5,16							
PELIN	PATTAS	-	255	2	250	"		53		ME.		1		7	9
16 th	200	State	3,68	3,12	3,12		9,00		9.	96					
650	CHICAGO	Serving.	55	77	480	T	-		4				-		1000
	TAN T	Maria Maria	16,00	2,47	2,11	1.53									
	ATTANTA	STATE ACTV. STATE	200 1	19	326	m		-			-		-	-	
		TATE	4.72	1.82	1.40	1.53	6.00								
	PRICA.		65	14		-	-				1	775		-	
	15	STATE ACTU	4.24	1.56	1.32 216		2.00								7
	NEW YORK		53	12 1	707	1100	- 1		24112						
-		1000	1.92	1,43	1,37 2										
	POSTON	ACTA STREET STREET	24 1	11 1				100			Nijen		-		7,650
	1	2		7	MS 2111	-	40	1				7	-	-	9
	四四 0 201		1. Stargency School Aid Act Fre-Stant Seviews	2. Exerutive Order 11246 Pre-Award Forlaws of Contractors	3. Sealth Agency and Program Pre-Grant Reviews	4, Peritoring of Adars State Righer Elecation Systems	5. Equal Educational Services Reviews of Major City School Systems	6. Reviews of Elementary and Secondary School District Larguage Programs (Lau)	<ol> <li>Conjectenative Raviews of Elementary and Secondary School Districts in the North and Watt(Brown)</li> </ol>	8. Feriews of State Elementary and Secondary Education Agencies	9, Reviews of Elementary and Secondary Vocational Technical Schools/Programs	0, Periews of Elementary and Secondary Special Purpose State Administered Schools	II. Reviews of Elementary and Secondary School District Discipline Practices	12. Reviews of Elementary and Secondary School District Special Education Programs	13. Reviews of Elementary and Secondary School District Migrant Ducation, Cractices

The second secon	The second name of the second	Conclusion with the last	- Company	The state of the s	-	-
Seriews of Formerly Duel State Migher Edocation Systems	2 2,00	1 1	1.00		14	4.00
15, seriews of State Pigher Education Systems	100					5.22
16. Regiews of Professional Schools					2	1,255
17. Egystews of Criscoste Schools						1,55
16. Agrices of undergraduate Schools	24				01	9.60
Berievs of Community Colleges/Community College Systems					10	6.20
20. Prejews of Non Profit Organizations					10	2,40
11. Arrievs of Sesidential Scalth Parilities					0	6.307
Faviews of Non-Presidential Essith Facilities					n	2,33
Azriers of State Bealth Agencies					4	27
arriews of Mon-Tamidontial Human Development.					1	15.2
15. Tevievs of State Human Development Agencies						3.5
STANCE.					1011	¥10,081 7701

[FR Doc.77-22486 Filed 8-4-77:8:45 am]